Studies on Airway Inflammation and Remodeling in Chronic Lung Disease

The EGFR-ADAM17 axis controls inflammatory responses in Cystic Fibrosis and Chronic Obstructive Pulmonary Disease



Marta Stolarczyk

Studies on Airway Inflammation and Remodeling in Chronic Lung Disease

The EGFR-ADAM17 axis controls inflammatory responses in Cystic Fibrosis and Chronic Obstructive Pulmonary Disease

> To obtain the degree of Doctor from the Erasmus University Rotterdam by command of the Rector Magnificus

> > Prof.dr.H.A.P.Pols

and in the accordance with the decision of the Doctorate Board

The public defense shall be held on Tuesday 21 March 2017 at 13.30

By

Marta Stolarczyk Born in Iława, Poland

zafing

Erasmus University Rotterdam

Doctoral Committee:

Promotor:	Prof.dr. FG Grosveld
Other members:	Prof.dr. PS Hiemstra Dr. R Rottier Prof.dr. RW Hendriks
Co-Promotor:	Dr. BJ Scholte

Dla mamy Basi i dla taty "Once we accept our limits, we go beyond them" Albert Einstein

Table of contents:

List of abbreviations	8
Scope of the thesis	10
Chapter 1 General introduction:	13
The EGFR-ADAM17 axis in pro-inflammatory airway responses and remodeling in CFTR related lung diseases: CF and COPD	
Chapter 2 ADAM17 and EGFR regulate IL-6 receptor and amphiregulin mRNA expression and release in cigarette smoke-exposed primary bronchial epithelial cells from patients	61
Chapter 3 Enhanced ADAM17-mediated AREG release from CFTR deficient polarized bronchial	77
epithelial cells is caused by enhanced extracellular redox potential, affecting ADAM17 and EGFR activity	
Chapter 4 Correction of lung inflammation in F508del CFTR murine Cystic Fibrosis model by the sphingosine-1-phosphate lyase inhibitor	111
Chapter 5 General discussion: Is abnormal activation of EGFR-ADAM17 axis in COPD and CF a relevant target for therapeutic intervention?	129
Appendix Role of IL-1β in experimental CF upon <i>P.aeruginosa</i> infection	157
Summary	176
Samenvatting	183
Curriculum vitae and PhD portfolio	189
Acknowledgements	195

Abbreviations (in alphabetical order)

ADAM17	A disintegrin and metalloprotease 17; f.k.a TACE
ALI-CFBE41o-	CFBE410- a CF Bronchial Epithelial Cell Line cultured at Air-Liquid Interface
ALI-HBEC	Primary Human Bronchial Epithelial Cells cultured at Air-Liquid Interface
AREG	Amphiregulin
ASL	Air Surface Liquid
ASM	Airway Smooth Muscle cells
ATP	Adenosine Triphosphate
BALF	Bronchoalveolar Lavage Fluid
cAMP	Cyclic Adenosine Monophosphate
CF	Cystic Fibrosis
CFTR	Cystic Fibrosis Transmembrane Conductance Regulator
COPD	Chronic Obstructive Pulmonary Disease
CS	Cigarette Smoke
Duox1	Dual oxidase 1, NADPH oxidase (NOX) family; transmembrane, produces extracellular peroxide (ROS)
Duox2	Dual oxidase 2, NADPH oxidase (NOX) family; transmembrane, produces extracellular peroxide (ROS)
ECM	Extracellular Matrix
EGFR	Epidermal Growth Factor Receptor
ERK1/2	Extracellular Signal-Regulated Kinase 1/2
F508del CFTR	Most prevalent pathogenic mutation in CF patients, deletion of a phenylalanine in the NBD1 domain of the CFTR gene causes abnormal folding and trafficking of CFTR protein
G551D CFTR	A missense mutation in an ATP binding pocket of CFTR, causing a channel gating defect
Gp130	Signal Transducing Glycoprotein 130, IL-6ST, CD130, cytokine receptor.
Grx-1-roGFP	Glutaredoxin-1 The reduction-oxidation sensitive Green Fluorescent Protein
Grx-1-roGFP-GPI	Glutaredoxin-1-The reduction-oxidation sensitive Green Fluorescent Protein fused to a Glycosylphosphatidylinositol anchor
GSH	Glutathione; reduced form
GSSG	Glutathione dimer; oxidized form of glutathione
IL1R	Interleukin-1 Receptor
IL-1 β	Interleukin-1 beta
IL-6	Interleukin-6
IL-6R	Interleukine-6 Receptor
IL-8	Interleukin-8
МАРК	Mitogen-activated protein kinases
MPD	Membrane proximal domain (of ADAM17)
MUC5AC	Mucin 5AC
MUC5B	Mucin 5B
NADPH	Nicotinamide Adenine Dinucleotide Phosphate (reduced)
РЗ8 МАРК	p38 Mitogen-Activated Protein Kinases
PDI	Protein Disulfide Isomerase
ROS	Reactive Oxygen Species
S1P	Sphingosine-1-Phosphate
STAT3	Signal Transducer and Activator of Transcription 3

TACE	TNF- $lpha$ -Converting Enzyme; former name of ADAM17
TGF-α	Transforming Growth Factor alpha
TIMP-3	Metalloproteinase Inhibitor 3, extracellular, regulates ADAM activity
TMI-1	4-[[4-(2-butynyloxy)phenyl]sulfonyl]-N-hydroxy-2,2-dimethyl- (3S)thiomorpholinecarboxamide; inhibitor of MMPs, ADAM10 and ADAM17
TMI-2	(2R, 3S)-2-([[4-(2-butynyloxy)phenyl]sulfonyl]amino)-N,3-dihydroxybutanamide; selective ADAM17 inhibitor
TNF- α	Tumor Necrosis Factor alpha
VX-770	Ivacaftor (trade name KALYDECO [®]), CFTR potentiator
VX-809	Lumacaftor, CFTR corrector

SCOPE OF THE THESIS

Cystic fibrosis (CF) and Chronic obstructive pulmonary disease (COPD) are progressive and eventually fatal lung diseases characterized by airway inflammation, bacterial infection, viscous mucus secretion and myofibroblast hyperplasia. An effective cure is not available. Though the etiology of the two diseases is obviously different, the molecular mechanisms involved are likely related, but not completely elucidated. By studying pathological processes in the CF and COPD airway model systems, we aim to establish novel therapeutic targets.

In this thesis we focus on the role of ADAM17 metalloprotease activity, which sheds a range of bioactive protein ligands, including most epithelial growth factor receptor (EGFR) ligands. We hypothesized that the ADAM17-EGFR axis is a link between CF and COPD lung pathology. When triggered by stress signals, ADAM17 releases a number of protein substrates from airway epithelial cells, including IL-6 receptor (IL-6R) and several growth factors such as amphiregulin (AREG) that activate gp130 and EGFR respectively on the epithelial cells (autocrine) and the underlying tissues (paracrine, trans-activation). Both the IL-6/IL-6R/gp130 and AREG/EGFR pathways converge in STAT3 activation, a transcription factor and genetic modulator of CF lung disease, and are involved in the lung tissue repair, inflammation and fibrosis. Thus the focal point of this thesis was to establish the role of ADAM17 and EGFR in pathology of CFTR-related lung diseases: CF and COPD.

Chapter 1 provides a general introduction to CF and COPD pathology and describes the importance of anti-inflammatory therapy in CF. In the second part, ADAM17 and EGFR are introduced and their role in lung inflammation and remodeling is underlined.

Chapter 2 shows the responses of ADAM17 and EGFR to whole cigarette smoke (CS) exposure in differentiated COPD and non-COPD primary human bronchial epithelial cells cultured at the air-liquid interface (ALI-HBEC). Further, it investigates the effect of CS exposure on localization and quantity of phospho-ADAM17/substrate interactions, and shows the involvement of the EGFR-ADAM17 axis in regulation of mRNA and protein levels of ADAM17 substrates, providing new insight into the spatiotemporal ADAM17-mediated cleavage.

Chapter 3 demonstrates that CFTR deficiency hyperactivates epithelial ADAM17 and EGFR as a cell-autonomous defect in a unique model of immortalized CFTR deficient CFBE410-airway epithelial cells with inducible CFTR expression. Druggable targets upstream and downstream of these regulators were defined in order to understand the cascade of events leading to CFTR-mediated activation of ADAM17-EGFR axis and potentially apply them as an early intervention in CF lung therapy.

Chapter 4 provides evidence of chronic lung inflammation in a mouse model of the most common CF mutation (F508del CFTR). In unchallenged mutant mice we observe enhanced infiltration with activated granulocytes and an abnormal dendritic cell balance, consistent with basal pro-inflammatory signalling due to CFTR deficiency. Upon challenge with bacterial toxins, resolution of induced inflammation is delayed in mutant mice. This correlates with abnormal sphingosine metabolism, in particular reduced S1P in lung tissue, which is known to be associated with inflammatory processes through S1P recpetors, Oral administration of an S1P-lyase inhibitor (LX2931), corrected the S1P deficiency and reduced basal and induced inflammation in mutant mice.

Chapter 5 Summarizes and critically discusses the studies presented in this thesis and provides possible future directions in investigation of EGFR-ADAM17 axis in the context of CF and COPD lung disease.

APPENDIX

Appendix presents further evidence that F508del CFTR mutant mice show excessive inflammation, increased IL-1 β production and reduced bacterial clearance. CFTR deficient mice with the IL-1R1 knock-out (double mutant dd x IL-1R1) have attenuated inflammation, providing a rationale for targeting IL-1 β /IL-1R1 pathway in CF patients. Importantly, in airway epithelial cell lines activation of IL-1R activates the ADAM17/EGFR/IL8 signaling (Kim et al, 2013), thus intervention in IL-1 β /IL-R1 β pathway may be also beneficial in attenuation of EGFR/ADAM17 signaling in the CF or COPD disease state, which can be further evaluated in differentiated human airway epithelial cells.

human disease CF lung COPD airway IL6R inflammation AREG remodeling human disease CF lung COPD airway ILBK inflation ALCOHAPTER 1 ADAM17 fibrosis PBEC human disease CF lung COPD airway IL6R infla General introduction EGFR shedding ADAMI EGFR shedding ADAM17 fibrosis PBEC hus human lung COPD airway IL6R inflammation AREG remodeling EGFR shedding ADAM17 The EGFR-ADAM17 axis in pro-inflammatory airway shedding ADA responses and remodeling in CFTR related lung diseases: AREG EGFR sheddin CF and COPD sis PBEC human disease CF lung COPD airway IL6R inflamma EGFR shedding ADAM17 fibrosis PBEC human disease CF lung i (to be submitted) modeling EGFR shedding ADAM17 fibrosis PBEC human disease CF lung COPD airway ILGR inflammation AREG human disease CF lung COPD airway ILGR inflammation Stolarczyk M and Scholte BJ

INTRODUCTION

1.1 CYSTIC FIBROSIS, A CONGENITAL LUNG DISEASE WITH AN EARLY ONSET

Cystic Fibrosis (CF) is an autosomal recessive lung disease caused by more than 2000 different mutations in the cystic fibrosis transmembrane conductance regulator (CFTR) gene [1] with 70-80.000 patients worldwide. The most common CF mutation is the deletion of a phenylalanine at amino acid position 508 in the NBD1 domain of CFTR protein (F508del) [2]. The CFTR gene encodes the Cystic Fibrosis Transmembrane conductance Regulator protein, a chloride channel mainly expressed in the apical membrane of secretory epithelial cells [3]. Later studies showed that CFTR is also involved in transport of other molecules, for instance bicarbonate [4] and glutathione [5] [6]. Bicarbonate is crucial to normal expression of mucins that in CF remains aggregated and poorly solubilized [7], whereas glutathione, as a natural antioxidant, reduces oxidative cellular stress [8].

CFTR has a key role in maintaining ion and water homeostasis in secretory epithelia [3], thus mutations in CFTR cause a multi-organ CF disease affecting lungs, intestine, pancreas, liver, sinuses, and reproductive organs [9]. However, under present treatment the main morbidity and mortality among CF patients is due to lung malfunction, thus most available and experimental treatments aim to prevent the progression of CF lung disease [10].

CF lung disease, an early disorder of distal airways

Advances in imaging and monitoring of the respiratory system in infant patients [11] [12] [13] have revealed that severe CF lung pathology starts early in childhood [14] [15] and progresses irreversibly over time [16] [17]. Characteristic lung functional and structural changes are observed already in pre-school patients [15] [17] [18] [19]. This early onset of lung abnormalities includes bronchiectasis, diagnosed thickening and dilation of the bronchial walls [14] [20] [21], air trapping and atelectasis (partial collapse of the lung) [14] [22]. These symptoms occur simultaneously with reduced mucociliary clearance and mucus plugging [20]. Quantitative and standardized tracking of early lung disease progression in infants with CT scans is pursued to advance the comparative analysis and provide the evaluation of the treatment [18] [23].

In a recent micro CT and histological study of end-stage CF lungs the dilatation and obstruction of distal airways and a severe reduction in the number of functional terminal bronchioles was clearly documented for the first time, confirming that obstruction and remodeling of peripheral airways is a prominent feature in CF lung disease, and therefore are prime targets of experimental therapy [24]. CF mouse models [25] (Chapter 4, Appendix) and large CF animal models, like pig [26] [27] and ferret [28] are helpful in the investigation of the mechanism involved in this early onset of CF lung disease and intervention therapy, but all have considerable limitations. In addition to state of the art clinical and biomarker studies, there is an urgent need for the development of organotypic cell culture models in which the complex molecular and cellular interactions involved in CF lung pathology can be studied (Chapter 2 and 3).

Mucus, Mucociliary transport and ciliary beat frequency

Healthy airways are protected by mucociliary transport that removes pathogens and toxic particles captured in mucus by ciliary beating. In healthy subjects this requires CFTR dependent balanced fluid and proper mucus secretion from surface cells and sub-epithelial glands. However, CF lung epithelial cells produce mucus (MUC5AC, MUC5B) with altered properties, characterized mainly by high viscosity [29] [30]. High mucus viscosity in CF is caused by reduced CFTR-dependent bicarbonate secretion, required for proper expansion of secreted mucus molecules [30] [31].

Furthermore, CF patients have intrinsically impaired ciliary beat frequency (CBF), which additionally reduces muco-ciliary transport of inhaled chemical particles and pathogens. CBF is not only dependent on CFTR-mediated bicarbonate transport, but also regulated by soluble adenyl cyclase (sAC) [32] [33].

Taken together, high mucus viscosity [29] and impaired CBF disable effective clearance of bacteria and inhaled particles [26] [34], which has been recently confirmed by less effective removal of bacteria introduced into the CF pig lung in comparison to healthy animals [35].

Air Surface Liquid

Airway epithelium is covered with a thin layer of air-surface liquid (ASL). Airway epithelial cell cultures from CF patients have reduced ASL height, presumably due to defective CFTR dependent fluid secretion [36] [37] [38]. Additionally, as a consequence of reduced bicarbonate secretion, CFTR deficiency abnormally acidifies ASL [32] [33]. Reduced pH of ASL impairs bacterial killing [39] [40], inhibits the activity of ASL antimicrobials [41], and increases ASL viscosity of newborn CF piglets, which can be improved after reducing ASL Ca²⁺ levels [26].

Taken together, the careful regulation of the balance between fluid and ion secretion and resorption at the airway surface, maintenance of the optimal liquid layer (ASL) height, pH and mucus viscosity, is the primary role of CFTR in the lungs. Impaired CFTR function leads to dehydration and acidification of ASL, which interferes with pathogen and mucus clearance [34] [36] [39] [42].

Bacterial infection and inflammation

Impaired bacterial killing and mucociliary clearance facilitate bacterial colonization in CF lungs. *Pseudomonas aeruginosa, Haemophilus influenza, Staphylococus aureus,* generally harmless in normal individuals, are pathogens frequently found in CF patient lungs [43]. Microbes that colonize airways frequently form biofilms, stimulating persistent inflammatory responses [44]. Despite activation of inflammatory responses mediated by the innate and cellular immune system, eradication of bacterial infection is impaired in CF lungs. It is plausible that persistent inflammation is caused by bacteria captured in sticky mucus, which produce a mucoid biofilm during adaptation to the CF lung environment.

Recurrent infections (mainly with *P. aeruginosa*) and concomitant chronic inflammation, recognized as main hallmarks of CF lung disease, worsen the lung pathology. In airways, bacterial infections induce a massive recruitment of neutrophils [45]. Both bacterial infections and neutrophil infiltration have been correlated with progression of lung disease [14]. This ultimately leads to irreversible airway remodeling, observed as air trapping, bronchiolar obstruction and bronchiectasis [24].

However, it still remains unknown whether elevated levels of inflammatory cytokines, growth factors and mucins always result from bacterial infection, or are an intrinsic property of CFTR deficient mucosa (Chapter 4, Appendix). Some reports argue that bacterial infections are indispensable to inflammatory responses, providing evidence that CF lungs do not have inflammation at birth in humans [46] or CF pig [35]. However, others report that CFTR malfunction leads to overexpression of pro-inflammatory molecules (such as IL-6, IL-8, IL-17, CXCL1, CXCL2) as a cell-autonomous defect occurring before *P.aeruginosa* infection [47] [48] [49] [50] [51] [52]. Thus, it is likely that mucosal inflammation is an autonomous defect of CF epithelial cells, which may occur before bacterial infection. Furthermore, it is

also unclear whether the elevated levels of inflammatory mediators in the bronchioalveolar lavage fluid (BALF) of young and adult CF patients [53] are caused by functional abnormalities in CF myeloid cells, in particular macrophages [54], dendritic cells (Appendix: CF mouse studies), and neutrophils [55] or are primarily related to abnormal cytokine signaling by CFTR deficient airway epithelial cells. This thesis aims to address the question whether inflammation is an inherited property of CF airway epithelial cells, which is covered by Chapter 2.

In summary, due to the early onset of the lung disease and its irreversible nature, it is clear that CF patients require early intervention therapy [17]. Excessive lung inflammation and tissue remodeling observed in CF may be an inherent property of CFTR deficient lung mucosa. Therefore, it is important to establish whether alleviation of inflammatory responses is beneficial in management of CF lung disease [56].

1.2 STRATEGIES OF CF THERAPY

The most prominent cause of death of CF patients is lung malfunction caused by frequent airway infections, persistent inflammatory responses, mucus plugging and airway remodeling (bronchiectasis, scarring) [24]. Indirect pharmacological management targets these different pathophysiological aspects of the disease and mainly focuses on antiinflammatory agents [57] [58], antibiotics [59], and mucolytic agents [60] [61] [62]. Direct pharmacological management of CF disease, which is being developed at this time, intends to restore the functional expression of mutated CFTR at the plasma membrane by correcting its folding and gating defect [63] and thus reverse the abnormalities in chloride transport. Many attempts have been made to define correctors that aim to correct the folding defect and enable mutant CFTR to reach the apical membrane, and potentiators that increase the open-probability of the mutant CFTR channel at the plasma membrane, thereby increasing CFTR conductance [64] [65] [66] [67] [68]. On the basis that patients with residual CFTR activity have less severe disease [69] [70], it is claimed that for therapeutic strategies aimed at correcting mutant CFTR function or gene therapy around 10 % of normal CFTR activity is required to restore chloride transport [71].

So far, a potentiator lvacaftor (VX-770, trade name KALYDECO®) was tested with excellent results in G551D CFTR patients [72] [73] and has been approved by the U.S Food and Drug Administration (FDA) for patients with gating mutations (G551D, G1244E, G1349D, G178R, G551S, S1251N, S1255P, S549N, or S549R), which together account for 4-5% of all CF alleles [3]. In clinical trial, the corrector VX-809 (lumacaftor) resulted in significant improvement of lung disease, albeit only in a subpopulation of homozygous F508del CFTR mutants [68]. In 2015 the FDA approved ORKAMBI® for homozygous F508del CFTR patients, the combination of CFTR corrector Lumacaftor (VX-809) and potentiator Ivacaftor (VX-770) in one pill. This therapy improves lung function in patients homozygous for the F508del mutation, although modestly and not in all patients [10] [74], so further investigations with different combinations of new correctors and potentiators are in progress to find more effective therapy [75] [76]. Long-term effects on lung function after treatment with correctors and potentiators are still not well investigated. Current developments use stem cell based intestinal organoids as a new test platform for high throughput screening and personalized evaluation of available compounds and combination therapies [77] [78] [79]. Gene therapies (viral and non-viral gene transfer, gene editing, mRNA repair) as well as strategies targeting protein repair other than potentiators and correctors (like premature termination codon (PTC) read-through drugs or proteostasis regulators) are also under investigation, but so far with limited success in clinical trials [65] [66] [67] [80] [81]. Editing the CFTR gene using the CRISPR-Cas9 system in stem cells is a potential next step to improve CF therapy

[82]. However, this approach has to overcome similar issues of efficiency, delivery and safety as conventional gene therapy.

Since the 1990s excessive airway inflammation, which positively correlates with the progression of CF lung malfunction, is recognized as the major factor in the pathogenesis of CF lung disease [83]. However, so far the effects of correctors and potentiators on infections and the release of inflammatory mediators have not been broadly investigated in clinical studies. *In vitro* studies by Pohl et al. show that Ivacaftor improves extracellular *P.aerugnosa* killing by neutrophils isolated from F508/G551D or F508del homozygous patients and corrects neutrophilic degranulation and Rab27a activation [84]. Rowe et al. showed that Ivacaftor also reduces *P. aeruginosa* isolated from CF patients after 6 months treatment. However, the free neutrophil elastase and other inflammation markers like IL-1 β , IL-6 and IL-8 in sputum samples remained unchanged [73]. Therefore anti-inflammatory, anti-bacterial and other additional therapies are still important targets of investigation [85].

In summary, recent breakthroughs in the development of small-molecule compounds targeting the mutant CFTR protein have raised hope to find a cure for CF. However, the presently available CFTR correctors and potentiators are not sufficiently effective in a majority of CF patients, and the search for new compounds and additional therapies is still highly relevant. Targeting the CFTR gene by gene editing in stem cells is a next development. At this point, anti-inflammatory and anti-bacterial therapies remain important targets of investigation.

1.3 COPD, ACQUIRED CF?

Chronic obstructive pulmonary disease (COPD), the 5th ranking cause of death worldwide, is a complex inflammatory lung condition that interferes with normal breathing due to obstruction of the lung airflow, excessive sputum production and a chronic cough. Usually COPD is characterized by chronic bronchitis (inflammation of the bronchi and bronchioles) and emphysema [86]. Similar to CF, bronchiectasis and peripheral airway thickening are also observed in the lungs of COPD patient [87]. Phenotypic expression of COPD is highly variable. Some individuals develop lung disease dominated by emphysema, while others exhibit chronic bronchitis. This heterogeneous phenotype likely reflects the contribution of multiple pathogenic mechanisms. Once COPD starts to develop, it tends to worsen over time, and so far its progress cannot be controlled effectively in most patients.

The most prominent etiological factor leading to COPD is cigarette smoke, but also exposure to fumes, chemicals and dust [88]. Although COPD and CF differ in primary cause, the spectra of the pathological events overlap considerably (Figure 1). Both diseases are characterized by excessive mucus production and insufficient clearance, leading to lower airways obstruction with chronic neutrophilic infiltration. In CF and COPD airway surface liquid (ASL) dehydratation and viscous mucus secretion impair mucociliary and bacterial clearance, causing chronic inflammation and facilitating recurrent infections [89]. Furthermore, during chronic CF lung disease, *P.aeruginosa* clones adapt, resulting in a highly diverse bacterial community that is extremely difficult to eradicate therapeutically. This is also observed in severe COPD, but not in mild disease [90] [91]. Recent publications show the involvement of a wide variety of inflammatory mediators in COPD (like IL8, IL-6, CCL18), that overlap with CF related mediators [85] [92]. There are also parallels on the cellular level that include goblet cell metaplasia, hyperplasia of myoblasts, extensive extracellular matrix production [93].



Figure 1. Common features of CF and COPD. CF and COPD despite the distinct nature of the disease development share many common clinical manifestations. Mutations in CFTR gene contribute to the CF outcome. Recently CS has been shown to affect CFTR channel activity and since is recognized as a factor that may contribute to a CF-like phenotype in COPD. Thus COPD has been recently described as acquired CF.

The broad spectrum of common features and events observed in CF and COPD encouraged researchers to seek for the common factors for both diseases. Studies showed that cigarette smoke decreases CFTR mRNA expression, and reduced CFTR protein level through accelerated degradation and altered channel activity [94] [95] [96] [97]. CFTR dysfunction and a clinical phenotype similar to mild CF developed by COPD patients has made researchers to describe COPD as an "acquired CF" [89] [93] [98] [99]. Nasal potential differences (NPD) in patients with no detectable CFTR gene mutation reflected a pattern typical of CFTR deficiency [100]. As a result of CS exposure, CFTR rapidly internalizes leading to its decreased activity, diminished CFTR-mediated anion transport, impaired mucociliary clearance and depleted ASL *in vitro* and *in vivo* [94] [95]. However, the effect presented in these papers is transient, so whether CS-induced inhibition of CFTR contributes to the chronic COPD in ex-smokers remains to be investigated.

1.4 STRATEGIES OF THERAPIES IN COPD

The CF-like symptoms of COPD, impaired CFTR function observed in COPD patients, and the direct evidence that CS exposure inhibits CFTR activity have led to a trial in which smokers were treated with a CFTR potentiator. Sloane et al. showed that Ivacaftor (VX-770) activated CFTR-dependent chloride transport in non-CF patients, increased ASL height and mucociliary transport [95]. These findings contrasted with the results of Cholon et al showing that chronic treatment with VX-770 reduced non-CF CFTR function and the amount of mature CFTR [101]. These results thus discredit the possibility to use VX-770 as an additional therapy to enhance CFTR function in COPD.

Inflammation is recognized as the major pathophysiological mechanism of COPD progression, with molecular targets overlapping those of CF (IL-6, IL8, CCL18) [85]. It is also chronic and persists even after cessation of smoking, suggesting that apart from the role of CFTR also epigenetic changes may play a role [102] [103]. Thus airway inflammation remains an important therapeutic target in COPD management. Presently, several compounds targeting inflammatory responses in COPD are under investigation [104] [105] [106] [107], though none of these have been shown to be beneficial in COPD patients as yet.

COPD is a complex multifactorial disease, with large variation in the patient population due to undefined genetic and environmental factors. Therefore, a personalised approach using multiple treatments will likely be required, but a robust trial strategy is elusive. This also applies to CF anti-inflammatory treatment, though this disease is considered monogenetic, the downstream responses to CFTR deficiency, and responses to therapeutic intervention are highly variable in the population. The perfect corrector and potentiator combination tailored to the individual patient, supported by additional anti-inflammatory medication, would likely be the best solution.

1.5 EPITHELIAL EGFR/ADAM17 AXIS, A POTENTIAL THERAPEUTIC TARGET IN CF AND COPD LUNG DISEASE

For decades the design of anti-inflammatory agents aimed mainly to decrease the neutrophil influx into the lung and concomitant inflammatory responses mediated by eosinophils, macrophages, basophils, mast cells, NK, dendritic cells, B cells and T cells [108]. Recently the importance of airway epithelial cells in inflammatory responses has been also recognised [109]. Airway epithelium serves as a first barrier and acts as defence against daily inhaled air pollutants and microbes by mucociliary clearance and secretion of a range of cytokines, cytokines receptors, growth factors, growth factor receptors and antimicrobial peptides [110] [109]. Airway epithelial cells not only release inflammatory mediators to ASL, but they also signal to the underlying tissues (myocytes or fibroblasts).

A disintegrin and metalloproteinase 17 (ADAM17) is involved in the immune defence mechanisms mediated by epithelial cells. ADAM17, also known as a Tumor Necrosis Factor- α Converting Enzyme (TACE), is an enzyme with proteolytic activity that releases extracellular domains of transmembrane proteins to produce soluble bioactive signaling proteins.

Mature ADAM17 consists of metalloprotease domain (catalytic), a disintegrin domain, membrane proximal domain (MPD) rich in cysteine residues comprising 5 disulfide bonds (this cysteine-rich segment is shorter in comparison to other ADAMs), "Conserved ADAM-seventeeN Dynamic Interaction Sequence" (CANDIS), unlike other ADAM family members it lacks EGF-like domain [111] [112] [113] [114] [115]. These extracellular domains are followed by transmembrane region and cytoplasmic tail with phosphorylation sites that potentially are involved in ADAM17 activation (Figure 2).

Ectodomain shedding mediated by the metalloprotease domain of ADAM17 provides a mechanism for both membrane protein downregulation and subsequent initiation or inhibition of autocrine/paracrine signaling. So far 76 proteins have been identified as substrates of ADAM17 [116]. They encompass membrane bound cytokines (TNF- α), cytokine receptors (IL-6R, TNF-R), growth factors (AREG), ligands of EGFR (TGF- α , AREG, EREG, HB-EGF, Epigen), adhesion proteins (L-selectin, ICAM-1) and transmembrane mucins (MUC-1). The shed soluble forms of these proteins are bioactive transducers of cell signaling via activation of cellular receptors on underlying cells (transactivation/paracrine activation), but they also are involved in activation of the shedding cells and neighbouring cells

(autocrine activation) (Figure 3).



Figure 2. ADAM17 domain structure.

ADAM17 (and its close relative ADAM10) is an atypical member of the ADAM family. It has additional disulfide bonds in the metalloprotease domain, it lacks two calcium binding sites in disintegrin domain. The Membrane Proximal Domain (MPD), replacing the cysteine-rich and EGF-like domains, with a novel alpha/beta fold has a shorter cysteine-rich segment [111] [113] [114]. The MPD has cysteine residues determining the ADAM17 conformation (open/closed) and ADAM17 protease activity (active/inactive switch). The MPD is in close proximity to the active site likely due to a C-shaped conformation of the extracellular part of mature ADAM17 [111] [112]. ADAM17 lacks an EGF-like domain, so the MPD is followed by the juxtamembrane region "Conserved ADAM17 Dynamic Interaction Sequence" (CANDIS) involved in some substrate recognition (IL-6R, but not TNF α) [117], transmembrane region and a cytoplasmic tail [111] [112] [113] [114] [115] with phosphorylation sites, which are likely important for ADAM17 activity.

Because most of the epidermal growth factor receptor (EGFR) ligands are cleaved by ADAM17, this sheddase has emerged as an important transducer of the airway epithelial autocrine and paracrine signaling (Figure 3). EGFR and ADAM17 are both involved in the broad spectrum of events that is characteristic of both CF and COPD lung disease, like excessive mucus expression [118] [119] [120], cytokine secretion [121], airway epithelial cell wound healing [122] [123], abnormal airway proliferation [124], maintenance of barrier integrity and progressive lung tissue scarring [125]. They both are activated upon bacterial or viral infection and during inflammation [126] [127] [128]. While this is an effective and necessary response, it is suggested that exaggerated airway epithelial signaling in the chronic state may enhance inflammation and may lead to damage of the lung structure.

EGFR functions as a sensor of airway epithelial integrity [129]. When cells have intact tight junctions EGFR is not activated. But disruption of the epithelial cell integrity, either by mechanical injury or cytokine treatment (TNF- α /IFN- γ), leads to EGFR phosphorylation and concomitant inhibition of protein phosphatase 2A activity [130]. Cigarette smoke exposure of differentiated HBEC also leads to damage of the lung tissue observed as destruction of epithelial cell integrity, loss of E-cadherin/ β -catenin complex and disappearance of cilia [42]. This coincides with phosphorylation and peri-nuclear trafficking of EGFR [42] suggesting the importance of EGFR in maintenance of epithelial cell barrier integrity. The response of ADAM17 to loss of pulmonary epithelial cell integrity has been also shown by neuregulin-1 (NRG-1) shedding and concomitant activation of human epidermal growth factor receptor-2

(HER2) [131]. This raises the question whether and how EGFR and ADAM17 co-operate in sensing responses to airway injury.

Taken together, EGFR and ADAM17, expressed by airway epithelial cells, have emerged as potential therapeutic targets in COPD and CF lung disease [125] [132]. Therefore, we aimed to understand the mechanism that provides a link between the loss of functional CFTR expression and the EGFR/ADAM17 axis. In order to do this, we studied the involvement of EGFR and ADAM17 in models of COPD and CF disease together, since it seems likely that they share molecular mechanisms and therapeutic options.



Figure 3. ADAM17 dependent paracrine and autocrine signaling.

A Disintegrin and Metalloproteinase 17 (ADAM17), also known as a Tumor Necrosis Factor- α converting enzyme (TACE), is involved in the immune defense mechanisms mediated by the epithelial cells. ADAM17 releases extracellular domains of transmembrane proteins to produce soluble bioactive signaling proteins taking part in autocrine (activation of receptors within the same epithelial cell layer) and paracrine signaling (activation of cellular receptors on underlying neighbouring cells, also termed trans-activation). Among ADAM17 substrates there are (1) adhesion proteins (L-selectin, ICAM) (2) transmembrane mucins (MUC-1), (3) membrane bound cytokines (TNF- α), (4) growth factors (AREG) and other ligands of EGFR (TGF- α , EREG, HB-EGF, Epigen), (5) cytokine receptors (IL-6R, TNF-R); in this picture the examples of these proteins are presented. Ectodomain shedding provides the mechanism for membrane protein downregulation and subsequent initiation or inhibition of autocrine/paracrine signaling. For instance, shed IL-6R from epithelial cells autocrinally activate EGFR on the neighboring epithelial cells or paracrinally on the underlying fibroblasts. ADAM17 natural inhibitor TIMP-3 inhibits ADAM17 proteolytic activity.

EGFR and ADAM17 are important in developmental processes

ADAM17 knockout mice survive between 17.5 embryonic days up to a few days after birth, underlining the need of the ectodomain shedding in development. The few survivors have defects in epithelium and lung and also in vascular system, eye, hair, heart, and skin [133]. Tissue specific deletions of ADAM17 or hypomorphic ADAM17 knock-in demonstrate an *in vivo* role of ADAM17 in controlling inflammation and regeneration [134] [135]. Similar to ADAM17-KO mice, EGFR-KO mice survive for up to 8 days after birth and suffer from impaired epithelial development in several organs, including lung, indicating the need of a functional ADAM17/EGFR axis for proper functioning and development of lung epithelium [136].

ADAM17 and EGFR crosstalk

ADAM17 works in association with a tyrosine kinase receptor EGFR by shedding most of its ligands (e.g. AREG, HB-EGF, TNF- α , EPGN (Epigen), Epiregulin) [116]. Only two EGFR ligands, EGF and betacellulin are shed by ADAM10 [137] [138], a close relative of ADAM17 [116]. Crosstalk of ADAM17 and EGFR in inflammatory signaling transduction is also defined by the establishment of a positive ADAM17/EGFR feedback loop likely involving activation of ADAM17 via the EGFR/MAPK pathway [126] [139] (Figure 4). The exact molecular mechanism of this feedback signaling has not been firmly established.



Figure 4. A positive ADAM17/EGFR feedback loop. ADAM17/EGFR positive feedback loop likely involves activation of ADAM17 via the EGFR/MAPK pathway and direct interaction of ERK1/2 with ADAM17. Additionally, by cleavage of Notch1 from a non-small lung carcinoma cell line ADAM17 regulates transcription of EGFR mRNA and increases EGFR expression on the cell surface, providing other mechanism contributing to positive feedback regulation of the ADAM17/EGFR axis.

ADAM17 regulates transcription of EGFR mRNA by cleavage of Notch1 and increases EGFR expression on the cell surface, as shown in a non-small lung carcinoma cell line [140], providing another positive feedback mechanism of the ADAM17/EGFR axis. In our studies we aimed to understand the mechanism of cigarette smoke induced activation of the EGFR/ADAM17 cascade (Chapter 2), and the role of CFTR in this process (Chapter 3).

The role of ADAM17 and EGFR in balanced regulation of lung inflammation and regeneration

ADAM17, expressed in human bronchial epithelial cells from the large and small bronchi, in lung smooth muscle cells, lung muscular vessels, alveolar macrophages, perivascular leukocytes and lung endothelial cells, is an important regulator of lung tissue homeostasis [132] [141] [142] and development [133], pro- and anti-inflammatory responses [116] and is involved in tissue regeneration [139] [143] [144].

The anti- and pro-inflammatory properties of the EGFR/ADAM17 signaling pathway are very much context dependent [116]. For instance, *in vitro* in airway epithelial cells ADAM17 together with EGFR induces mRNA expression and protein release of IL8, a neutrophil chemotactic factor that promotes inflammation [121] [145]. Whereas, by shedding TNF receptor type 2 (TNFR2) [146], which antagonises TNF- α , ADAM17 exhibits also anti-inflammatory properties [147]. Thus it is likely that the type of stimulus and substrate selection determine the nature of the EGFR and ADAM17 activity [148] [149] [150].

The involvement of EGFR/ADAM17 signaling in tissue regeneration encompasses wound healing [139], proliferation [143] [144], differentiation [151] [152] and cell migration [153]. However, when pathogenic stimuli or cell stress induces ADAM17 and EGFR activity, this contributes to cancer development [116] [154] [155] [156] and cardiac tissue remodeling after an infarct [157].

Due to the involvement of ADAM17/EGFR paracrine and autocrine signaling in several lung disorders [125] [132], modulation of ADAM17 and EGFR activation in COPD and CF is important, to keep the balance between anti-inflammatory processes and promotion of inflammation, and also between regeneration and excessive tissue remodeling. However, the mechanism of this modulation is still not fully understood.

Does CFTR deficiency affect the ADAM17/EGFR axis?

EGFR activation has been observed in NCI-H292 cells after inhibition of CFTR (inh-172) [49]. This CFTR-dependent induction of EGFR activity was likely ADAM17 dependent [49]. In CFTR deficient IB3 cells EGFR phosphorylation was markedly diminished by inhibition of ADAM17 (siRNA knockdown and broad spectrum MMPs inhibitor TAPI-1), plausibly due to reduced shedding of EGFR substrates. Interestingly, the inhibitory effect was much lower in C38 cells with corrected CFTR expression [49], suggesting that aberrant ADAM17 activity may contribute to induced activation of EGFR in CFTR deficiency. However, the relationship between ADAM17/EGFR signaling and CFTR deficiency is still poorly understood [158] and may involve a variety of actions. As a consequence of ADAM17-mediated release of EGFR ligands, the activation of downstream signaling occurs (i.e. MEK/ERK, PI3K/AKT, STAT3 or mTOR), which suggest that EGFR/ADAM17 pathway may be also abnormal in CF and COPD lung diseases [125] [132].

In our laboratory it was observed that F508del CFTR mutant mice compared to normal mice develop spontaneous and progressive chronic lung inflammation [25] (Chapter 4, Appendix). Additionally, naphthalene provoked injury induced abnormal responses in mutant mice compared to normal [159] with stronger expression of AREG, IL-6, elastin and collagen mRNA levels seven days after injury. Exaggerated responses mediated by the EGFR/ADAM17/ERK1/2/MAPK pathway have also been reported in CFTR deficient cells. Martel et al. showed that CFTR deficient cell line (CuFi-1) in response to heat-inactivated *P. aeruginosa* produce more IL-8 than non-CFTR deficient cell line (NuLi-1) and this involves

EGFR phosphorylation and ERK1/2 activation [160]. However, CuFi and NuLi are independent subclones and genetically different, and therefore not comparable models. Further studies by Kim et al. showed that inhibition of CFTR with CFTR-inh172 induces IL-8 production in ADAM17 and EGFR dependent manner and involves binding of IL-1 α to IL-1R [49]. These observations led to the hypothesis that ADAM17 dependent IL-6/IL-6R/gp130 and AREG/EGFR pathways contribute to abnormal resolution of injury and inflammation in CF lung disease.

These data together suggest altered responses of the ADAM17-EGFR signaling pathway in CFTR deficiency. However, most of the studies cited here use undifferentiated, submerged immortalized cell lines [161], compare different cell lines (like CuFi and NuLi or IB3 and C38 cells) [49] [160] [162] [163] or use a CFTR inhibitor [49] with reported off-target effects [164]. Therefore, we addressed the mechanism by which CFTR deficiency activates the EGFR/ADAM17 axis in differentiated CFBE cells with inducible CFTR expression (Chapter 3), thereby overcoming both the high donor variation observed in primary airway cell culture, and the genetic and epigenetic variability of distinct immortalized cell lines.

Release of cytokines, growth factors and mucins depends on ADAM17/EGFR signaling

External stress factors like oxidative stress, viral and bacterial toxins, and CS exposure activate the EGFR/ADAM17 signaling pathway. Several studies showed that pathogens inhaled into the airways, or exposure of airway epithelial cells to other external stimuli activate toll-like receptors (TLR) [165] [166] [167] and G-coupled receptors (GPCR) [168] that cross talk with downstream EGFR/ADAM17 signaling. As a result of cigarette smoke extract exposure, secretion of downstream proinflammatory cytokines (including IL8) [121] [169] [145], growth factors (TGF α , AREG, HB-EGF) [121], mucins (MUC5AC) and phosphorylated MUC1 is induced in an ADAM17/EGFR dependent manner [119] [170] [171] [42]. Rhinovirus infection of airway epithelial cells showed that this cascade of events involves ERK1/2, MAPK and p38 MAP kinase [127] [172]. Particular matter exposure also induces IL-8 secretion in an EGFR/ERK1/2/MAPK dependent manner, but does not involve p38 MAP kinase [173]. Upon activation of the ERK or p38 MAPK pathway, ADAM17 dissociates from an endogenous extracellular tissue inhibitor of metalloproteinase-3 (TIMP-3) [174] [175] [176] [177], accumulates on the cell surface [157] [178] and induces release of TGF- α [157] [178]. Acrolein, an active component of cigarette smoke, which induces MUC5AC mRNA in an ADAM17 and EGFR dependent manner, also decreases TIMP-3 transcript levels, suggesting a role of TIMP-3 in ADAM17 activation. However in TIMP3-KO cells PMA-stimulated shedding of ADAM17 is not affected, which contradicts the model in which pre-bound TIMP3 needs to be removed for ADAM17 activation [138] and supporting that ADAM17 can be rapidly activated by conformational changes [138] [179] [180]. Thus, the role of TIMP-3 in HBEC signaling is still unclear.

Is Redox potential a link between CFTR deficiency and ADAM17/EGFR signaling?

The intra and extracellular redox potential changes in response to physiological processes and in pathophysiological conditions [181]. Reactive oxygen species (ROS), produced during cellular stress [85], is an important regulator of redox state and is also involved in activation of the EGFR/ADAM17 signaling pathway, affecting TGF- α and AREG release and mucin expression [119] [120] [165] [173]. Some studies point towards the role of NADPH oxidases (NOX), in particular dual oxidase 1 (DUOX1) [120] [165] [182] or dual oxidase 2 (DUOX2) [166], which produce ROS at an extracellular or possibly intra-vesicular domain. ATPmediated DUOX1 activation has been shown to involve a TGF- α /ADAM17/ERK1/2/EGFR signaling pathway [182]. Other studies point towards intracellular NOX/ROS dependent SRC activity in EGFR/ADAM17 activation [139]. Recent studies indicated the involvement of DUOX1 in allergen dependent SRC/EGFR activation in airway cells [183]. However, the mechanism by which ROS affects the EGFR/ADAM17 signaling pathway in intact airway cells is likely highly complex and still remains not well understood.

ROS not only affects signaling molecules in the EGFR/ADAM17 pathway, but it has been shown that ADAM17 [180] and EGFR [184] are redox sensitive proteins. EGFR has intracellular cysteine residues in the active site that are direct targets of ROS and determine EGFR kinase activity, likely through association of EGFR with NADPH oxidase, NOX2 [184]. ADAM17 activity, however, is regulated by thiol-disulfide isomerisation in the extracellular MPD domain mediated by protein disulfide isomerase (PDI), an oxidoreductase sensitive to redox changes [179]. PDI, by direct interaction with the membrane proximal domain (MPD) [180], changes the disulfide bridge pattern and thus the conformation of the extracellular protease domain from open, active to closed, inactive state, leading to the inhibition of ADAM17 proteolytic activity (Figure 5) [111] [112] [113] [114] [115] [117] [179] [180]. Redox dependent conformational changes likely make ADAM17 sensitive to the extracellular redox potential [180]. These findings are in line with previous observations showing that ADAM17 activity is redox sensitive [185] and blocking PDI induces ADAM17-mediated L- selectin shedding [186].



closed, inactive ADAM17 conformation

open, active ADAM17 conformation

Figure 5. ADAM17 is a redox sensitive protein.

The ADAM17 membrane proximal domain (MPD), which is in close proximity to the active site, is sensitive to extracellular redox changes. This redox-sensitive ADAM17 activity is regulated by thioldisulfide isomerisation mediated by protein disulphide isomerase (PDI), an oxidoreductase sensitive to redox changes. PDI by direct interaction with membrane proximal domain (MPD) changes the disulfide bridge pattern and thus the conformation of the extracellular protease domain from open, active to closed, inactive state, leading to the inhibition of ADAM17 proteolytic activity. Redox sensitive conformational changes likely make ADAM17 sensitive to the extracellular redox potential.

Since CFTR deficiency is thought to increase ROS levels [187], which in turn may activate ADAM17 and EGFR [139] and inactivate protein phosphatases [188], we propose that it is a link between EGFR/ADAM17 signaling and CFTR deficiency. CFTR deficiency may also affect the redox potential of ASL, due to CFTR mediated trans-epithelial glutathione transport at the apical membrane, which serves as a natural antioxidant [8]. Thus, by using the ratiometric fluorescence measurements [181], we questioned whether extracellular or intracellular redox potential is affected by CFTR and takes part in CFTR-mediated activation of EGFR/ADAM17 pathway (Chapter 2).

All in all, these studies show the involvement of different regulatory molecules leading to activation of EGFR/ADAM17 signaling, however, the exact cascade of events is still not well defined. In order to understand the regulation of EGFR/ADAM17 signaling, we aimed to define the molecules regulating ADAM17 shedding activity in CF and COPD. SRC [139], ROS [139], P38MAP [143] and ERK1/2 [189] appear to be the most important modulators of the ADAM17/EGFR pathway. Therefore, this thesis focuses on the role of these molecules in EGFR/ADAM17 signaling cascade in advanced COPD and CF cellular models (Chapter 2 and Chapter 3).

1.6 ROLE OF AREG IN EGFR/ADAM17 DEPENDENT MUCUS AND CYTOKINE SECRETION, AND REMODELING

Amphiregulin (AREG) is an ADAM17 substrate and EGFR agonist, expressed ubiquitously in the human lung [190] by human airway epithelial cells [121], smooth muscle cells [191] and fibroblasts [192]. AREG is also expressed by infiltrating and resident lung myeloid cells, including activated macrophages [193], eosinophils [194], dendritic cells [195], neutrophils [196] and mast cells [197]. Human AREG is synthetized as an N-glycosylated transmembrane precursor (50 kDa) [198] with a basolateral sorting motif [199] and shed mainly by ADAM17 [126]. Proteolytic cleavage of the AREG extracellular domain releases several AREG soluble forms, predominantly an α -N-glycosylated 43 kDa form [198], which is one of the EGFR ligands, with lower affinity to EGFR than EGF and TGF- α .

AREG is induced in lung disease

AREG is involved in inflammation and repair responses through autocrine and paracrine activation of EGFR, and generally induced in lung disease [200]. In CF sputum samples, elevated levels of AREG have been shown in airway blood neutrophils [196]. Until now, the airway epithelial secretion of AREG has not been investigated in CF and COPD condition in comparison to controls (Chapters 2 and 3). However, in lung biopsies from asthma patients more AREG is expressed than in healthy controls [190], suggesting that this is also a case in CF and COPD lung disease. Other studies showed that in sputum of asthma patients AREG is upregulated only during an acute attack, which contrasts with the presence of EGF up to 7 days after an acute asthma attack [201], suggesting its role in a quick cellular responses to the triggers. Increase of AREG in sputum samples from children with asthma negatively correlates with lung function [202] and positively correlates with the number of eosinophils [203].

Regulation of AREG transcription and shedding

In vitro AREG mRNA expression and protein release are induced upon exposure to different stress factors like histamine [190], diesel exhaust particles [204], cigarette smoke extract exposure [121] [124] [205] and rhinoviruses [206] [127]. Human airway trypsin-like protease, a serine protease found in the sputum of patients with chronic airway diseases, induces AREG mRNA by the protease-activated receptor-2 (PAR-2)/ERK pathway in the human airway cell line (NCI-H292) [207]. Du et al showed that cigarette smoke extract triggered transcription of AREG is cAMP and EGFR dependent in the epithelial cell line MSK-Leuk1, and involves binding of the cAMP-responsive element binding protein to the AREG gene promoter in an EGFR dependent manner [208]. Also, AREG protein secretion is dependent on the EGFR/ERK1/2/MAPK pathway in an airway epithelial cell line treated with particulate matter [173].

These data together suggest the role of EGFR/ADAM17 signaling in regulation of AREG and IL-6R mRNA production and shedding in COPD human airway epithelial cells and in CFBE410- cells with inducible CFTR expression, which is covered by Chapter 2 and 3,

respectively.

AREG affects mucus and cytokine secretion

In asthmatic patients AREG produced by mast cells enhances mucus production [209]. *In vitro* the AREG-dependent MUC5AC mRNA level induction has been shown upon exposure to particulate matter in NCI-H292 cells [210]. Importantly, prolonged exposure to AREG induces MUC5AC expression in NHBE-ALI [201] and promotes also TNF- α -induced IL-8 secretion from NCI-H292 cells [211]. Interestingly, *in vitro* eosinophil-airway epithelial cell line interaction induces AREG dependent secretion of MUC5AC, TGF- β 1, and IL-8 in culture supernatants [203]. All these findings show an important role of AREG in mucus secretion and cytokine release in airway epithelial cells.

AREG in paracrine signaling and tissue remodeling

The role of airway epithelial AREG in activation of underlying tissues has also been studied. Conditioned media from AREG-stimulated airway epithelial cells induced expression of IL8, VEGF, COX-2 and AREG in human airway smooth muscle cells (HASMC), providing proof for crosstalk between epithelial cells and HASMC in an AREG-dependent manner [191], and underlined the important role of AREG in paracrine signaling. Inhibition of AREG or EGFR in TGF- β 1-stimulated lung fibroblasts diminished AREG-dependent fibroblast proliferation, expression of α -smooth muscle actin and collagen [212], strongly suggesting a role of AREG/ADAM17/EGFR signaling in pulmonary fibrosis *in vivo*. AREG stimulation also induces airway smooth muscle cell (ASM) proliferation, which leads to airway remodeling *in vivo* [213]. The AREG/EGFR/ERK1/2/MAPK signaling pathway is also involved in cell migration through extracellular matrix (ECM), due to plasmin-dependent degradation of matrix proteins in MSK-Leuk1 cells treated with a saline extract of cigarette smoke (CSE) [214]. CSE and exogenous AREG treatment also lead to induction of DNA synthesis in an EGFR dependent manner, providing further proof of AREG and EGFR involvement in cell proliferation [208].

These data together suggest that exaggerated and chronic AREG release may contribute to mucus plugging and excessive inflammation and tissue remodeling in CF and COPD. Therefore, the AREG/EGFR/ADAM17 signaling pathway is a potential therapeutic target to regulate both inflammation and lung cell proliferation and migration, which may prevent lung tissue inflammation, remodeling and fibrosis in COPD and CF. However, AREG is only one of the many players involved in lung tissue autocrine and paracrine signaling leading to lung pathology. There is no available literature presenting intervention in AREG function in clinical trials.

The role of AREG and EGFR trafficking in cellular signaling

The majority of EGFR signaling is believed to occur at the plasma membrane. However, many studies show that upon ligand-dependent activation, EGFR is rapidly internalized into endosomes, where hypothetically it may continue to signal [215] [216] [217] [218] [219]. Up till now, the endocytic transport of EGFR has been interpreted as a signal attenuation, due to lysosomal degradation [220]. However, recent reviews elaborate on the role of EGFR endocytic trafficking and recycling in active signal transduction [220] [221] [222] [223] [224].

The full-length forms of AREG and other EGFR substrates have also been observed in intracellular vesicular compartments, i.e. lysosomes [225] and endosomes [199]. Mono-ubiquitination of pro-AREG at Lys240 (pro-AREGmUb) leads to quick endocytosis of AREG from the cell surface and its localization into early endosomes [226]. AREG has been also observed in exosomes negative for the ER marker calreticulin [224]. Higginbotham et al.

showed the importance of paracrine exosomal AREG-mediated signaling in breast cancer cells [224]. Recipient LM2-4175 cells rapidly take up AREG containing exosomes in an EGFR-dependent manner. In comparison to other EGFR substrates, namely TGF- α and HB-EGF, exosomal AREG uptake enhanced invasion of LM2-4175 cells through the Matrigel to the recipient cells, which were placed under the Transwell filter [224]. These exosomes also improve wound healing.

Taken together, presented data indicate the existence of a non-canonical trafficking pathway of EGFR-mediated signaling after EGFR endocytosis. So far, these studies largely focus on deregulated EGFR trafficking in the context of carcinogenesis. However, the EGFR trafficking in CFTR deficiency and its effect on the signaling cascade has not been investigated yet. As a first step to study the role of ADAM17 substrate and receptor trafficking, we investigated the intracellular localization of EGFR, ADAM17 and its substrates in CS treated ALI-HBEC and in CFBE410- cells with induced CFTR activity (Chapter 2 and 3).

1.7 ROLE OF IL-6-R IN HUMAN LUNG INFLAMMATION AND FIBROSIS

Pro and anti-inflammatory responses of IL-6R

The IL-6 co-receptor (IL-6R) is a transmembrane receptor type I that in classic signaling binds to IL-6, which evokes an association with the homodimer of signal transducing glycoprotein, gp130. This trimer dimerises to form a hexameric complex composed of IL-6, IL-6R and gp130 [227] (Figure 6). IL-6R can also evoke a trans-mediated signaling cascade, when it is shed close to the transmembrane region by ADAM17 [228] or ADAM10 [229] [228] [230]. Subsequently, γ-secretase cleaves the IL-6R C-terminal fragment, which is rapidly degraded, so intracellular part does not take part in further signaling [231]. sIL-6R can be also generated by alternative splicing [232] [233] [234], however it is thought that shedding rather than the alternative splicing takes part in regulated generation of the sIL6-R [146]. Both the alternatively spliced form and the shed form of sIL-6R can create functional complexes with IL-6 (IL-6/sIL-6R) to dimerize and trans-activate gp130 finally creating a hexameric complex on the underlying myofibroblasts [227] [232] [235] [236] (Figure 6). The soluble form of gp130 (sgp130) selectively blocks trans-signaling and is described as a natural inhibitor of IL-6R trans-signaling *in vivo* [237] [238] [238].

It is thought that IL-6 activated membrane bound IL-6R has anti-inflammatory, antiapoptotic and regenerative properties [146] [235] [240] [241]. In contrast to classic IL-6R signaling in which IL-6 binds to membrane bound IL-6R associating with gp130, transsignaling responses mediated by IL-6/sIL-6R complexes binding to gp130 are thought to maintain inflammation and promote inflammation-associated cancer [146] [235] [242] [243] (Figure 6). IL-6R-mediated signaling leads to activation of STAT3 [239] [244], ERK [245] and PI3K [246], linking this signaling pathway with the EGFR/ADAM17 axis, which also activates STAT3 through a different pathway.

IL-6R expression is restricted to specific cell types

IL-6R expression is restricted to several cell types, such as lymphoid cells (megacaryocytes, neutrophils, monocytes/macrophages, leukocytes, certain lymphocytes), hepatocytes and airway epithelial cells [146] [246] [247] [248]. Cells that do not express IL-6R, but do express gp130 can still transduce IL-6 signals via binding of IL-6/sIL-6R complexes. Many cells including smooth muscle cells [249] and endothelial cells [250] are only responsive to IL-6 through IL-6/sIL-6R trans-signaling [249] [250] (Figure 6).

The role of IL-6R in inflammation and remodeling

Bacterial toxins, bacterial metalloproteinases and apoptosis are known to stimulate release of sIL-6R. It has been shown that in polarized airway epithelial cell lines bacterial treatment

induces shedding of sIL-6R [248]. Expression of IL-6 and sIL-6R is increased significantly in serum and lung biopsies from adenocarcinoma patients [243]. Further, in an adenocarcinoma mouse model blocking of the sIL-6R mediated trans-signaling with an anti-IL-6R antibody or the inhibitor sgp130Fc alleviates lung cancer pathogenesis showing the importance of IL-6R signaling in tumour signaling [243], proliferation and migration [251].



Figure 6 Classic and trans-signal activation of IL-6R signaling. In classic signaling membrane bound IL-6R stimulated with autocrinally or paracrinally secreted IL-6 associates with the homodimer of signal transducing glycoprotein, gp130 to activate downstream molecules, like STAT-3. In trans-signaling soluble form of IL-6R (sIL-6R) generated either by ADAM17 or by alternative splicing binds to IL-6 and this IL-6/IL-6R complex activates gp130 on the airway epithelial cells (autocrine IL-6R trans-signaling) or underlying fibroblasts/myoblasts/smooth muscle cells that do not express IL-6R (paracrine IL-6R trans-signaling), but express gp130 to evoke activation of downstream molecules, for instance STAT-3 signaling.

However, the role of IL-6R in CF and COPD has not been studied in detail. In COPD patients, elevated levels of IL-6R have been observed in peripheral blood leukocytes [252] and sputum samples [189]. Recently, genetic variants of IL-6R have been linked with lung function [253] and COPD severity [254]. Elevated levels of IL-6R were also detected in protein lysates from lungs of the patients with idiopathic pulmonary fibrosis (IPF), which is a lethal lung disease with progressive fibrosis [255]. In contrast to elevated IL-6R levels in sputum of COPD patients [189] and in bronchoalveolar lavage fluids (BALF) from patients with allergic asthma [256], the levels of sIL-6R in BALF form CF patients were not different in comparison to control [257]. This was explained by enhanced degradation of sIL-6R by serine proteases. Furthermore, it is not evident what the impact of luminal IL6-R on the activity of the sub-epithelial tissue is, which is separated from the lumen by tight junctions.

Shedding of IL6-R from epithelial cells occurs mainly towards the basal side (Chapter 2) [248]. Importantly, in bleomycin challenged murine lungs inhibition of sIL-6R by intraperitoneal injection of a recombinant decoy receptor (gp130Fc) attenuates pulmonary fibrosis observed as reduction in myofibroblasts, fibronectin, and collagen, whereas activation of IL-6 trans-signaling in cell lines enhances fibroblast proliferation and extracellular matrix protein production [255], which are the hallmarks of the pulmonary fibrosis progression.

Together these data show the importance of IL-6R trans-signaling in inflammation and lung remodeling [24]. We focused our studies on the mechanism regulating the IL-6R mRNA and protein shedding from COPD airway epithelial cells (Chapter 2).

Species specificity of IL-6R signaling

IL-6R is a substrate for both ADAM17 and its close relative ADAM10 [229]. IL-6R needs to bind IL-6 to evoke downstream signaling, neither IL-6R alone nor IL-6 has affinity for gp130 [258]. Species differences in IL-6R mediated signaling have been observed. Human IL-6 stimulates human and murine cells, whereas murine IL-6 only stimulates murine IL-6R signaling [259]. Because of this species specificity transgenic mice expressing human sIL-6R from a liver promotor did not bind the endogenous murine IL-6, and as a consequence the transgenic animals do not have a transgene specific phenotype [259] [260]. Garberts et al for the first time questioned the species specificity of IL-6R shedding by showing that human IL-6R is a substrate of human ADAM17, but murine IL-6R is a substrate of murine ADAM10 [261]. Subsequently, Schumacher et al. revealed that trigger-induced shedding of both human and mouse IL-6R is mediated by murine ADAM17, but constitutive release of IL-6R is largely mediated by ADAM10 in mice [230]. Additionally, in humans, but not in mice, the sIL-6R can be generated by translation from an alternatively spliced mRNA [262] [230]. Of note, the soluble gp130 form, which circulates in human plasma, blocks IL-6R trans signaling responses and does not show species specificity, meaning that it interacts with human and mouse sIL-6R complexes [263]. All of this has implications in the extrapolation of IL-6R data obtained in mouse models to human. Thus, to provide clinically relevant data, we focused our studies on primary and immortalized human cells from COPD and CF patients.

1.8 IS ADAM17 PHOSPHORYLATION IMPORTANT FOR CS-INDUCED AREG AND IL-6R SHEDDING?

The serine- and threonine-rich ADAM17 cytoplasmic tail [130] has three phosphorylation sites that have been proposed to activate ADAM17: Thr735, Ser791 and Ser819 (Figure 2). Phosphorylated ADAM17 is detected under basal conditions [264]. Stimulation of ADAM17 further increases its phosphorylation, without changing the total protein level [130] [264], suggesting that phosphorylation plays a role in the regulation of the proteolytic activity of ADAM17. However, mutation of the phosphorylation sites individually or in combination and even removal of the whole cytoplasmic tail has no significant effect on stimulated shedding suggesting that phosphorylation may be an additional but not a crucial regulatory mechanism of ADAM17 proteolytic activity, which does not exclude that it plays a role in transport, processing, and maturation [130] [138] [185] [265] [266].

Different stimuli affect distinct phosphorylation sites on the ADAM17 molecule. Cigarette smoke extract increases phosphorylation of Ser and Thr, but not Tyr residues on ADAM17 molecule in a PKC- ε dependent manner. ADAM17 **Ser819** residue, but not on Thr or Tyr, is phosphorylated when ERK/MAP kinase signaling pathway is induced by growth factor (GF) stimulation [130]. However, an ADAM17-Ser819 mutant does not lose its proteolytic activity when simulated with fibroblast growth factor (FGF) [130], suggesting that this phosphorylation is not required for ADAM17 proteolytic activity. To date, the kinase

responsible for phosphorylation of this site has not been identified. Another phosphorylation residue on ADAM17 molecule, a *Ser791* is phosphorylated in resting cells, whereas in response to GF stimulation *Ser791* undergoes dephosphorylation, which is independent of the ERK/MAPK pathway [130]. Pro-ADAM17 and mature ADAM17 are also phosphorylated at *Thr735* under resting condition. Phorbol esther (PMA) treatment further increases phosphorylation at this site [264]. ERK 1/2 co-immunoprecipitates with ADAM17 and directly phosphorylates it at Thr735 [143] [264] [267], providing evidence that it regulates ADAM17 maturation, trafficking to the cell surface and activation. Also p38 has been shown to phosphorylate Thr735 on ADAM17 [264] [267]. Stimulation of ERK and PKC with PMA also activates ADAM17, leading to shedding of IL-6R [143]. P38 α MAP kinase interacts with the cytoplasmic domain of ADAM17 and phosphorylates it on Thr735 resulting in release of TGF- α and enhanced cell proliferation [143], which is likely dependent on the association of ERK and ADAM17 [264].

Taken together, some studies show that GF stimulation induces ADAM17 phosphorylation at Thr735 [264], while others find contradicting data [130], which may be cell or context dependent. In Chapter 2 we addressed the association of Thr735 phosphorylated ADAM17 with its substrates (IL-6R and AREG) in CS treated airway epithelial cells, and correlated this with ADAM17 shedding activity.

1.9 Murine models to study activity of ADAM17 in airway epithelia

Inactivation of ADAM17 by deletion of the zinc binding domain through homologous recombination in mice led to severely hypoplastic lungs at birth, reduced branching morphogenesis and alveolar development, impaired epithelial cell proliferation and differentiation and delay in vasculogenesis [268]. Due to the early mortality of ADAM-KO mice, several alternative ADAM17 mouse models have been developed, with reduced activity (hypomorphic) or conditional mutants [132]. Conditional knockout ADAM17 mice can serve as a model to investigate the role of ADAM17 in organogenesis, in specific cell types, and in inflammatory responses and tissue remodeling.

Induced dysfunction of ADAM17 (ADAM17flox/flox SPC-rtTA TetO-Cre) in developing lung epithelial cells reduced saccular formation, decreased cell proliferation and reduced lung epithelial cell differentiation, but the mice were born without severe respiratory distress [152]. Phenotypic differences were reduced after alveolarisation of the immature lung (P14). Knocking out the gene in mesenchymal cells using a *Dermo1-Cre* transgene did not produce a detectable phenotype [152]. This suggests that epithelial, but not mesenchymal ADAM17 plays a prominent, but not decisive role in lung development.

Conditional knockout mice in the hematopoietic system (Adam17flox/ Δ Zn driven by Vav-Cre) treated with LPS have significantly reduced LPS-induced alveolar TNF- α and L-selectin levels, but IL-6R levels are reduced by only 25% [269], consistent with reports that in mice IL-6R is a preferred substrate of ADAM10, but not ADAM17 as in human [261] [230]. In the mutant mice alveolar neutrophil counts are reduced after LPS challenge, but initial infiltration of neutrophils occurs quicker, in comparison to control animals [269]. Other studies show that conditional knockout mice (ADAM17flox/flox with R26Cre ER) treated with human neutrophil elastase have attenuated goblet cell metaplasia in comparison to the wild-type mice [270], suggesting that ADAM17 is involved in the airway metaplasia.

In our laboratory we took an effort to generate CF ADAM17 conditional knockout mice in lung epithelial cells (SPC-rtta/Tet-O-Cre/Adam17flox/flox), which can serve as a model of inflammatory and repair responses in normal and CF airway epithelial cells, by crossing *Cftr^{tm1EUR}* F508del CFTR mutant mice with ADAM17flox/flox [271] and SPC-rtTA/tetO-cre [272].

HYPOTHESIS:

ADAM17 is hyperactive in CF and COPD, promoting inflammation and tissue remodeling by shedding of EGFR binding growth factor AREG and the cytokine receptor IL-6R from airway epithelial cells

We hypothesized that both CF and COPD lung disease are associated with hyperactivity of the EGFR/ADAM17 signaling cascade in airway epithelial cells, causing enhanced or sustained release of growth factors, pro-inflammatory cytokines and cytokine receptors. This may contribute to epithelial metaplasia, fibroblast and smooth muscle cell activation, and net deposition of extracellular matrix. Interventions in this pathway may therefore reduce CF and COPD lung pathology.

We focused our studies on two ADAM17 substrates: growth factor AREG and proinflammatory IL-6 receptor (IL-6R). AREG, a type I trans-membrane protein [273], proteolytically cleaved by ADAM17 from airway epithelial cells, activates EGFR not only on the airway epithelial cells, but also on the underlying fibroblasts. IL-6R trans-signaling requires ADAM17-mediated shedding of the extracellular domain [228] [230] or alternative mRNA splicing [232] [233] [234]. The shed form and alternatively spliced form bind to IL-6 creating IL-6/IL-6R complexes that dimerize and trans-activate gp130 on the underlying myofibroblasts [232] [235] [236]. gp130 is ubiquitously expressed, meaning that cells that do not express IL-6R, but do express gp130 can still transduce IL-6R signal by binding IL-6/IL-6R complexes. Many cells including smooth muscle cells [249] and endothelial cells [250] are only responsive to IL-6 through IL-6/IL-6R trans-signaling.

Taken together, we hypothesised that pathology driven trans-signaling at least in part depends on airway epithelial AREG and IL-6R shedding. Importantly, signals transduced by shed AREG and IL-6R from airway epithelial cells may converge in activation of the transcription factor STAT3 in lung fibroblasts, myofibroblasts and smooth muscle cells (Figure 7), which is involved in fibrotic responses and inflammatory lung disease [274]. This aspect has not been investigated during the course of this thesis; however, it is the subject of ongoing investigations in our institute.

Considering the similarity of the pathology of COPD and CF, we investigated the influence of COPD disease on the basal and CS-induced EGFR/ADAM17 signaling in differentiated primary bronchial epithelial cells. Furthermore, by using CFBE410- cells with inducible expression of CFTR, we aimed to establish whether CFTR deficiency hyper-activates EGFR/ADAM17 signaling as a cell-autonomous defect. Finally, we aimed to define the druggable targets in the EGFR/ADAM17 signaling pathway, which could be applied in management of COPD and as an early intervention CF lung therapy.



Figure 7 Is ADAM17/EGFR signaling an important player in CF and COPD lung disease?

We hypothesized that both CFTR deficiency and cigarette smoke exposure are associated with hyperactivity of ADAM17/EGFR signaling cascade in airway epithelial cells, causing enhanced release of pro-inflammatory cytokines and growth factors. We focused our studies on two ADAM17 substrates: pro-inflammatory IL-6 receptor (IL-6R) and growth factor AREG. AREG proteolytically cleaved by ADAM17 from airway epithelial cells activates EGFR autocrinally in the airway epithelial cells and paracrinally on the underlying fibroblasts. IL-6R trans-signaling requires ADAM17-mediated shedding of the extracellular domain or alternative mRNA splicing. The alternatively spliced and shed form binds to IL-6 creating IL-6/IL-6R complexes to dimerize and trans-activate gp130 on the underlying myofibroblasts. Signals transduced by shed AREG and IL-6R from airway epithelial cells may converge in activation of a transcription factor STAT3 in lung fibroblasts, myofibroblasts and smooth muscle cells, which is involved in fibrotic responses and inflammatory lung disease. Enhanced ADAM17/EGFR signaling may contribute to hyper-inflammation and epithelial metaplasia, fibroblast and smooth muscle cell activation, and net deposition of extracellular matrix (tissue remodeling) leading to lung pathology observed in COPD and CF.

LITERATURE

- 1. http://www.genet.sickkids.on.ca/StatisticsPage.html.
- 2. Cheng SH, Gregory RJ, Marshall J, Paul S, Souza DW, White GA, ORiordan CR, Smith AE. Defective intracellular transport and processing of CFTR is the molecular basis of most cystic fibrosis. *Cell* 1990; 63: 827–834.
- Kreda SM, Mall M, Mengos A, Rochelle L, Yankaskas J, Riordan JR, Boucher RC. Characterization of Wild-Type and ΔF508 Cystic Fibrosis Transmembrane Regulator in Human Respiratory Epithelia. *Mol Biol Cell* 2005; 5: 2154–2167.
- 4. Poulsen JH, Fischer H, Illek B, Machen TE. Bicarbonate conductance and pH regulatory capability of cystic fibrosis transmembrane conductance regulator. *Proc Natl Acad Sci U S A* 1994; 5340–5344.
- 5. Linsdell P, Hanrahan JW. Glutathione permeability of CFTR. *Am J Physiol* 1998; 275: C323–C326.
- 6. Gao L, Kim KJ, Yankaskas JR, Forman HJ. Abnormal glutathione transport in cystic fibrosis airway epithelia. *Am J Physiol Lung Cell Mol Physiol* 1999; 277: L113–L118.
- 7. Quinton PM. Cystic fibrosis: impaired bicarbonate secretion and mucoviscidosis. *The Lancet* 2008; 372: 415–417.
- 8. Ballatori N, Krance SM, Notenboom S, Shi S, Tieu K, Hammond CL. Glutathione dysregulation and the etiology and progression of human diseases. *Biol Chem* 2009; 390: 191–214.
- 9. Edelman A, Sallenave J-M. Cystic fibrosis, a multi-systemic mucosal disease: 25 years after the discovery of CFTR. *Int J Biochem Cell Biol* 2014; 52: 2–4.
- 10. Elborn SJ. Cystic fibrosis. *The Lancet* 2016; S0140-6736: 00576–00576.
- 11. Owens CM, Aurora P, Stanojevic S, Bush A, Wade A, Oliver C, Calder A, Price J, Carr SB, Shankar A, Stocks J, on behalf of the London Cystic Fibrosis Collaboration. Lung Clearance Index and HRCT are complementary markers of lung abnormalities in young children with CF. *Thorax* 2011; 66: 481–488.
- 12. Ramsey BW. Use of Lung Imaging Studies as Outcome Measures for Development of New Therapies in Cystic Fibrosis. *Proc Am Thorac Soc* 2007; 4: 359–363.
- 13. Castile R. Novel Techniques for Assessing Infant and Pediatric Lung Function and Structure. *Pediatr Infect Dis J* 2004; 23: S246–S253.
- 14. Mott LS, Park J, Murray CP, Gangell CL, de Klerk NH, Robinson PJ, Robertson CF, Ranganathan SC, Sly PD, Stick SM, on behalf of AREST CF. Progression of early structural lung disease in young children with cystic fibrosis assessed using CT. *Thorax* 2012; 67: 509–516.
- 15. Ramsey KA, Ranganathan S. Interpretation of lung function in infants and
young children with cystic fibrosis. *Respirology* 2014; 19: 792–799.

- 16. Fuchs SI, Gappa M, Eder J, Unsinn KM, Steinkamp G, Ellemunter H. Tracking Lung Clearance Index and chest CT in mild cystic fibrosis lung disease over a period of three years. *Respir Med* 2014; 108: 865–874.
- 17. Stick S, Tiddens H, Aurora P, Gustafsson P, Ranganathan S, Robinson P, Rosenfeld M, Sly P, Ratjen F. Early intervention studies in infants and preschool children with cystic fibrosis: are we ready? *Eur Respir J* 2013; 42: 527–538.
- 18. Mott LS, Graniel KG, Park J, de Klerk NH, Sly PD, Murray CP, Tiddens HAWM, Stick SM. Assessment of Early Bronchiectasis in Young Children With Cystic Fibrosis Is Dependent on Lung Volume. *Chest* 2013; 144: 1193–1198.
- 19. Stick SM, Brennan S, Murray C, Douglas T, Ungern-Sternberg von BS, Garratt LW, Gangell CL, De Klerk N, Linnane B, Ranganathan S, Robinson P, Robertson C, Sly PD, AREST CF ARESTFCFACARESTFCF. Bronchiectasis in infants and preschool children diagnosed with cystic fibrosis after newborn screening. J Pediatr 2009; 155: 623–628.e1.
- 20. Rossi UG. The radiology of chronic lung disease in children. *Arch Dis Child* 2005; 90: 601–607.
- 21. Long FR, Williams RS, Castile RG. Structural airway abnormalities in infants and young children with cystic fibrosis. *J Pediatr* 2004; 144: 154–161.
- 22. Long FR, Williams RS, Adler BH, Castile RG. Comparison of quiet breathing and controlled ventilation in the high-resolution CT assessment of airway disease in infants with cystic fibrosis. *Pediatr Radiol* 2005; 35: 1075–1080.
- 23. Rosenow T, Oudraad MCJ, Murray CP, Turkovic L, Kuo W, de Bruijne M, Ranganathan SC, Tiddens HAWM, Stick SM. PRAGMA-CF. A Quantitative Structural Lung Disease Computed Tomography Outcome in Young Children with Cystic Fibrosis. *Am J Respir Crit Care Med* 2015; 191: 1158–1165.
- 24. Boon M, Verleden SE, Bosch B, Lammertyn EJ, McDonough JE, Mai C, Verschakelen J, de Corput MK-V, Harm A W Tiddens, Proesmans M, Vermeulen FL, Verbeken EK, Cooper J, Van Raemdonck DE, Decramer M, Verleden GM, Hogg JC, Dupont LJ, Vanaudenaerde BM, De Boeck K. Morphometric analysis of explant lungs in cystic fibrosis. *Am J Respir Crit Care Med* 2016; 193: 516–526.
- 25. Wilke M, Buijs-Offerman RM, Aarbiou J, Colledge WH, Sheppard DN, Touqui L, Bot A, Jorna H, de Jonge HR, Scholte BJ. Mouse models of cystic fibrosis: Phenotypic analysis and research applications. *J Cyst Fibros* 2011; 10: S152– S171.
- 26. Tang XX, Ostedgaard LS, Hoegger MJ, Moninger TO, Karp PH, McMenimen JD, Choudhury B, Varki A, Stoltz DA, Welsh MJ. Acidic pH increases airway surface liquid viscosity in cystic fibrosis. *J Clin Invest* 2016; 126: 879–891.
- 27. Meyerholz DK. Lessons learned from the cystic fibrosis pig. *Theriogenology*

2016; 86: 427-432.

- 28. Jeong JH, Joo NS, Hwang PH, Wine JJ. Mucociliary clearance and submucosal gland secretion in the ex vivo ferret trachea. *Am J Physiol Lung Cell Mol Physiol* 2014; 307: L83–L93.
- 29. Burgel P-R, Montani D, Danel C, Dusser DJ, Nadel JA. A morphometric study of mucins and small airway plugging in cystic fibrosis. *Thorax* 2007; 62: 153–161.
- 30. Quinton PM. Role of epithelial HCOFormula transport in mucin secretion: lessons from cystic fibrosis. *Am J Physiol Cell Physiol* 2010; 299: C1222–C1233.
- 31. Shamsuddin AKM, Quinton PM. Native Small Airways Secrete Bicarbonate. *Am J Respir Cell Mol Biol* 2014; 50: 796–804.
- 32. Schmid A, Sutto Z, Schmid N, Novak L, Ivonnet P, Horvath G, Conner G, Fregien N, Salathe M. Decreased Soluble Adenylyl Cyclase Activity in Cystic Fibrosis Is Related to Defective Apical Bicarbonate Exchange and Affects Ciliary Beat Frequency Regulation. *J Biol Chem* 2010; 285: 29998–30007.
- 33. Schmid A, Meili D, Salathe M. Soluble adenylyl cyclase in health and disease. *Biochim Biophys Acta* 2014; 1842: 2584–2592.
- 34. Derichs N, Jin BJ, Song Y, Finkbeiner WE, Verkman AS. Hyperviscous airway periciliary and mucous liquid layers in cystic fibrosis measured by confocal fluorescence photobleaching. *FASEB J* 2011; 25: 2325–2332.
- 35. Stoltz DA, Meyerholz DK, Pezzulo AA, Ramachandran S, Rogan MP, Davis GJ, Hanfland RA, Wohlford-Lenane C, Dohrn CL, Bartlett JA, Nelson GA, Chang EH, Taft PJ, Ludwig PS, Estin M, Hornick EE, Launspach JL, Samuel M, Rokhlina T, Karp PH, Ostedgaard LS, Uc A, Starner TD, Horswill AR, Brogden KA, Prather RS, Richter SS, Shilyansky J, McCray PBJ, Zabner J, et al. Cystic fibrosis pigs develop lung disease and exhibit defective bacterial eradication at birth. *Sci Transl Med* 2010; 2: 29ra31.
- 36. Tarran R, Button B, Picher M, Paradiso AM, Ribeiro CM, Lazarowski ER, Zhang L, Collins PL, Pickles RJ, Fredberg JJ, Boucher RC. Normal and cystic fibrosis airway surface liquid homeostasis The effects of phasic shear stress and viral infections. *J Biol Chem* 2005; 280: 35751–35759.
- 37. Harvey PR, Tarran R, Garoff S, Myerburg MM. Measurement of the Airway Surface Liquid Volume with Simple Light Refraction Microscopy. *Am J Respir Cell Mol Biol* 2011; 45: 592–599.
- 38. Thiagarajah JR, Song Y, Derichs N, Verkman AS. Airway surface liquid depth imaged by surface laser reflectance microscopy. *J Gen Physiol* 2010; 136: 353–362.
- Pezzulo AA, Tang XX, Hoegger MJ, Alaiwa MHA, Ramachandran S, Moninger TO, Karp PH, Wohlford-Lenane CL, Haagsman HP, van Eijk M, Bánfi B, Horswill AR, Stoltz DA, McCray PB, Welsh MJ, Zabner J. Reduced airway surface pH impairs bacterial killing in the porcine cystic fibrosis lung. *Nature* 2012; 487: 109–113.

- 40. Shah VS, Meyerholz DK, Tang XX, Reznikov L, Abou Alaiwa M, Ernst SE, Karp PH, Wohlford-Lenane CL, Heilmann KP, Leidinger MR, Allen PD, Zabner J, McCray PB, Ostedgaard LS, Stoltz DA, Randak CO, Welsh MJ. Airway acidification initiates host defense abnormalities in cystic fibrosis mice. *Science* 2016; 351: 503–507.
- Abou Alaiwa MH, Reznikov LR, Gansemer ND, Sheets KA, Horswill AR, Stoltz DA, Zabner J, Welsh MJ. pH modulates the activity and synergism of the airway surface liquid antimicrobials β-defensin-3 and LL-37. *Proc Natl Acad Sci U S A* 2014; 111: 18703–18708.
- 42. Chen Y-T, Gallup M, Nikulina K, Lazarev S, Zlock L, Finkbeiner W, McNamara N. Cigarette smoke induces epidermal growth factor receptor-dependent redistribution of apical MUC1 and junctional beta-catenin in polarized human airway epithelial cells. *Am J Pathol* 2010; 177: 1255–1264.
- 43. Linnane B, Vaish S, Clarke D, O'Sullivan N, McNally P. The findings of a clinical surveillance bronchoalveolar lavage programme in pre-school patients with cystic fibrosis. *Pediatr Pulmonol* 2014; 50: 327–332.
- 44. Moreau-Marquis S, Stanton BA, O'Toole GA. Pseudomonas aeruginosa biofilm formation in the cystic fibrosis airway. *Pulm Pharmacol Ther* 2008; 21: 595–599.
- 45. Birrer P, McElvaney NG, Rüdeberg A, Sommer CW, Liechti-Gallati S, Kraemer R, Hubbard R, Crystal RG. Protease-antiprotease imbalance in the lungs of children with cystic fibrosis. *Am J Respir Crit Care Med* 1994; 150: 207–213.
- 46. Armstrong DS, Grimwood K, Carlin JB, Carzino R, Gutièrrez JP, Hull J, Olinsky A, Phelan EM, Robertson CF, Phelan PD. Lower airway inflammation in infants and young children with cystic fibrosis. *Am J Respir Crit Care Med* 1997; 156: 1197–1204.
- 47. Hartl D, Gaggar A, Bruscia E, Hector A, Marcos V, Jung A, Greene C, McElvaney G, Mall M, Doring G. Innate immunity in cystic fibrosis lung disease. *J Cyst Fibros* 2012; 11: 363–382.
- 48. Balough K, McCubbin M, Weinberger M, Smits W, Ahrens R, Fick R. The relationship between infection and inflammation in the early stages of lung disease from cystic fibrosis. *Pediatr Pulmonol* 1995; 20: 63–70.
- 49. Kim S, Beyer BA, Lewis C, Nadel JA. Normal CFTR Inhibits Epidermal Growth Factor Receptor-Dependent Pro-Inflammatory Chemokine Production in Human Airway Epithelial Cells. *PLoS ONE* 2013; 8: e72981.
- 50. Osika E, Cavaillon JM, Chadelat K, Boule M, Fitting C, Tournier G, Clement A. Distinct sputum cytokine profiles in cystic fibrosis and other chronic inflammatory airway disease. *Eur Respir J* 1999; 14: 339–346.
- 51. Tsuchiya M, Kumar P, Bhattacharyya S, Chattoraj S, Srivastava M, Pollard HB, Biswas R. Differential regulation of inflammation by inflammatory mediators in cystic fibrosis lung epithelial cells. *J Interferon Cytokine Res* 2013; 33: 121–129.

- 52. Veit G, Bossard F, Goepp J, Verkman AS, Galietta LJV, Hanrahan JW, Lukacs GL. Proinflammatory cytokine secretion is suppressed by TMEM16A or CFTR channel activity in human cystic fibrosis bronchial epithelia. *Mol Biol Cell* 2012; 23: 4188–4202.
- 53. Kirchner KK, Wagener JS, Khan TZ, Copenhaver SC, Accurso FJ. Increased DNA levels in bronchoalveolar lavage fluid obtained from infants with cystic fibrosis. *Am J Respir Crit Care Med* 1996; 154: 1426–1429.
- 54. Huaux F, Noel S, Dhooghe B, Panin N, Re Lo S, Lison D, Wallemacq P, Marbaix E, Scholte BJ, Lebecque P, Leal T. Dysregulated Proinflammatory and Fibrogenic Phenotype of Fibroblasts in Cystic Fibrosis. *PLoS ONE* 2013; 8: e64341.
- 55. Tirouvanziam R, Gernez Y, Conrad CK, Moss RB, Schrijver I, Dunn CE, Davies ZA, Herzenberg LA. Profound functional and signaling changes in viable inflammatory neutrophils homing to cystic fibrosis airways. *Proc Natl Acad Sci U S A*. 2008; 4335–4339.
- 56. Torphy TJ, Allen J, Cantin AM, Konstan MW, Accurso FJ, Joseloff E, Ratjen FA, Chmiel JF. Considerations for the Conduct of Clinical Trials with Antiinflammatory Agents in Cystic Fibrosis. *Annals ATS* 2015; 12: 1398–1406.
- 57. Cheng K, Ashby D, Smyth RL. Oral steroids for long-term use in cystic fibrosis. Cochrane Database Syst Rev 2015.
- 58. Lands LC, Stanojevic S. Oral non-steroidal anti-inflammatory drug therapy for lung disease in cystic fibrosis. Cochrane Database Syst Rev 2016.
- 59. Littlewood KJ, Higashi K, Jansen JP, Capkun-Niggli G, Balp M-M, Doering G, Tiddens HAWM, Angyalosi G. A network meta-analysis of the efficacy of inhaled antibiotics for chronic Pseudomonas infections in cystic fibrosis. *J Cyst Fibros* 2012; 11: 419–426.
- 60. Yang C, Chilvers M, Montgomery M, Nolan SJ. Dornase alfa for cystic fibrosis. *Cochrane Database Syst Rev* 2016
- 61. Aitken ML, Bellon G, De Boeck K, Flume PA, Fox HG, Geller DE, Haarman EG, Hebestreit HU, Lapey A, Schou IM, Zuckerman JB, Charlton B. Long-Term Inhaled Dry Powder Mannitol in Cystic Fibrosis. *Am J Respir Crit Care Med* 2012; 185: 645–652.
- 62. Donaldson SH, Bennett WD, Zeman KL, Knowles MR, Tarran R, Boucher RC. Mucus clearance and lung function in cystic fibrosis with hypertonic saline. *N Engl J Med* 2006; 354: 241–250.
- 63. Cai Z-W, Liu J, Li H-Y, Sheppard DN. Targeting F508del-CFTR to develop rational new therapies for cystic fibrosis. *Acta Pharmacol Sin* 2011; 32: 693–701.
- 64. Rogan MP, Stoltz DA, Hornick DB. Cystic Fibrosis Transmembrane Conductance Regulator Intracellular Processing, Trafficking, and Opportunities for Mutation-Specific Treatment. *Chest* 2011; 139: 1480–1490.

- 65. Lubamba B, Dhooghe B, Noel S, Leal T. Cystic fibrosis: insight into CFTR pathophysiology and pharmacotherapy. *Clin Biochem* 2012; 45: 1132–1144.
- 66. Dhooghe B, Haaf JB, Noel S, Leal T. Strategies in early clinical development for the treatment of basic defects of cystic fibrosis. *Expert Opin Investig Drugs* 2016; 25: 423–436.
- 67. Lukacs GL, Verkman AS. CFTR: folding, misfolding and correcting the ΔF508 conformational defect. *Trends Mol Med* 2012; 18: 81–91.
- 68. Clancy JP, Rowe SM, Accurso FJ, Aitken ML, Amin RS, Ashlock MA, Ballmann M, Boyle MP, Bronsveld I, Campbell PW, De Boeck K, Donaldson SH, Dorkin HL, Dunitz JM, Durie PR, Jain M, Leonard A, McCoy KS, Moss RB, Pilewski JM, Rosenbluth DB, Rubenstein RC, Schechter MS, Botfield M, Ordonez CL, Spencer-Green GT, Vernillet L, Wisseh S, Yen K, Konstan MW. Results of a phase IIa study of VX-809, an investigational CFTR corrector compound, in subjects with cystic fibrosis homozygous for the F508del-CFTR mutation. *Thorax* 2011; 67: 12–18.
- Bronsveld I, Mekus F, Bijman J, Ballmann M, de Jonge HR, Laabs U, Halley DJ, Ellemunter H, Mastella G, Thomas S, Veeze HJ, Tümmler B. Chloride conductance and genetic background modulate the cystic fibrosis phenotype of ΔF508 homozygous twins and siblings. J Clin Invest 2001; 108: 1705–1715.
- 70. Hirtz S, Gonska T, Seydewitz HH, Thomas J, Greiner P, Kuehr J, Brandis M, Eichler I, Rocha H, Lopes AI, Barreto C, Ramalho A, Amaral MD, Kunzelmann K, Mall M. CFTR Cl– channel function in native human colon correlates with the genotype and phenotype in cystic fibrosis. *Gastroenterology* 2004; 127: 1085–1095.
- 71. Patel S, Sinha IP, Dwan K, Echevarria C, Schechter M, Southern KW. Potentiators (specific therapies for class III and IV mutations) for cystic fibrosis. Cochrane Database Syst Rev 2015.
- 72. Accurso FJ, Rowe SM, Clancy JP, Boyle MP, Dunitz JM, Durie PR, Sagel SD, Hornick DB, Konstan MW, Donaldson SH, Moss RB, Pilewski JM, Rubenstein RC, Uluer AZ, Aitken ML, Freedman SD, Rose LM, Mayer-Hamblett N, Dong Q, Zha J, Stone AJ, Olson ER, Ordoñez CL, Campbell PW, Ashlock MA, Ramsey BW. Effect of VX-770 in Persons with Cystic Fibrosis and the G551D- CFTRMutation. N Engl J Med 2010; 363: 1991–2003.
- 73. Rowe SM, Heltshe SL, Gonska T, Donaldson SH, Borowitz D, Gelfond D, Sagel SD, Khan U, Mayer-Hamblett N, Van Dalfsen JM, Joseloff E, Ramsey BW. Clinical Mechanism of the Cystic Fibrosis Transmembrane Conductance Regulator Potentiator Ivacaftor in G551D-mediated Cystic Fibrosis. Am J Respir Crit Care Med 2014; 190: 175–184.
- 74. Cholon DM, Esther CR, Gentzsch M. Efficacy of lumacaftor-ivacaftor for the treatment of cystic fibrosis patients homozygous for the F508del-CFTR mutation. *Expert Rev Precis Med Drug Dev* 2016; 1: 235–243.
- 75. Boyle MP, Bell SC, Konstan MW, McColley SA, Rowe SM, Rietschel E, Huang X, Waltz D, Patel NR, Rodman D, group V-8-1S. A CFTR corrector (lumacaftor) and

a CFTR potentiator (ivacaftor) for treatment of patients with cystic fibrosis who have a phe508del CFTR mutation: a phase 2 randomised controlled trial. *Lancet Respir Med* 2014; 2: 527–538.

- 76. Wainwright CE, Elborn JS, Ramsey BW, Marigowda G, Huang X, Cipolli M, Colombo C, Davies JC, De Boeck K, Flume PA, Konstan MW, McColley SA, McCoy K, McKone EF, Munck A, Ratjen F, Rowe SM, Waltz D, Boyle MP. Lumacaftor–Ivacaftor in Patients with Cystic Fibrosis Homozygous for Phe508del CFTR. N Engl J Med 2015; 373: 220–231.
- 77. Ikpa PT, Bijvelds MJC, de Jonge HR. Cystic fibrosis: Toward personalized therapies. *Int J Biochem Cell Biol* 2014; 52: 192–200.
- 78. Dekkers JF, Wiegerinck CL, de Jonge HR, Bronsveld I, Janssens HM, de Winterde Groot KM, Brandsma AM, de Jong NW, Bijvelds MJ, Scholte BJ, Nieuwenhuis EE, van den Brink S, Clevers H, van der Ent CK, Middendorp S, Beekman JM. A functional CFTR assay using primary cystic fibrosis intestinal organoids. Nat Med 2013; 19: 939–945.
- 79. Dekkers JF, Gogorza Gondra RA, Kruisselbrink E, Vonk AM, Janssens HM, de Winter-de Groot KM, van der Ent CK, Beekman JM. Optimal correction of distinct CFTR folding mutants in rectal cystic fibrosis organoids. *Eur Respir J* 2016; 48: 451–458.
- 80. Alton E, Armstrong DK, Ashby D, Bayfield KJ, Bilton D, Bloomfield EV, Boyd AC, Brand J, Buchan R, Calcedo R, Carvelli P, Chan M, Cheng SH, Collie DS, Cunningham S, Davidson HE, Davies G, Davies JC, Davies LA, Dewar MH, Doherty A, Donovan J, Dwyer NS, Elgmati HI, Featherstone RF, Gavino J, Gea-Sorli S, Geddes DM, Gibson J, Gill DR, et al. A randomised, double-blind, placebo-controlled trial of repeated nebulisation of non-viral cystic fibrosis transmembrane conductance regulator (CFTR) gene. *Efficacy and Mechanism Evaluation* 2016
- 81. Tosco A, De Gregorio F, Esposito S, De Stefano D, Sana I, Ferrari E, Sepe A, Salvadori L, Buonpensiero P, Di Pasqua A, Grassia R, Leone CA, Guido S, De Rosa G, Lusa S, Bona G, Stoll G, Maiuri MC, Mehta A, Kroemer G, Maiuri L, Raia V. A novel treatment of cystic fibrosis acting on-target: cysteamine plus epigallocatechin gallate for the autophagy-dependent rescue of class II-mutated CFTR. *Cell Death Differ* 2016; 23: 1380–1393.
- 82. Merkert S, Martin U. Targeted genome engineering using designer nucleases: State of the art and practical guidance for application in human pluripotent stem cells. *Stem Cell Res* 2016; 16: 377–386.
- 83. Cantin AM, Hartl D, Konstan MW, Chmiel JF. Inflammation in cystic fibrosis lung disease: Pathogenesis and therapy. *J Cyst Fibros* 2015; 14: 419–430.
- 84. Pohl K, Hayes E, Keenan J, Henry M, Meleady P, Molloy K, Jundi B, Bergin DA, McCarthy C, McElvaney OJ, White MM, Clynes M, Reeves EP, McElvaney NG. A neutrophil intrinsic impairment affecting Rab27a and degranulation in cystic fibrosis is corrected by CFTR potentiator therapy. *Blood* 2014; 124: 999–1009.
- 85. Schnúr A, Hegyi P, Rousseau S, Lukacs GL, Veit G. Epithelial Anion Transport as

Modulator of Chemokine Signaling. *Mediators Inflamm* 2016; 2016: 7596531.

- 86. Hogg JC, Timens W. The Pathology of Chronic Obstructive Pulmonary Disease. *Annu Rev Pathol Mech Dis* 2009; 4: 435–459.
- 87. Silva S, Paschoal I, De Capitani E, Moreira M, Palhares L, Pereira M. COPD phenotypes on computed tomography and its correlation with selected lung function variables in severe patients. *Int J Chron Obstruct Pulmon Dis* 2016; 11: 503–513.
- 88. Mehta AJ, Miedinger D, Keidel D, Bettschart R, Bircher A, Bridevaux P-O, Curjuric I, Kromhout H, Rochat T, Rothe T, Russi EW, Schikowski T, Schindler C, Schwartz J, Turk A, Vermeulen R, Probst-Hensch N, Künzli N. Occupational Exposure to Dusts, Gases, and Fumes and Incidence of Chronic Obstructive Pulmonary Disease in the Swiss Cohort Study on Air Pollution and Lung and Heart Diseases in Adults. *Am J Respir Crit Care Med* 2012; 185: 1292–1300.
- 89. Raju SV, Solomon GM, Dransfield MT, Rowe SM. Acquired Cystic Fibrosis Transmembrane Conductance Regulator Dysfunction in Chronic Bronchitis and Other Diseases of Mucus Clearance. *Clin Chest Med* 2016; 37: 147–158.
- 90. Döring G, Parameswaran IG, Murphy TF. Differential adaptation of microbial pathogens to airways of patients with cystic fibrosis and chronic obstructive pulmonary disease. *FEMS Microbiol Rev* 2011; 35: 124–146.
- 91. Beck JM, Young VB, Huffnagle GB. The microbiome of the lung. *Transl Res.* 2012; 160: 258–266.
- 92. Cantin AM. Cystic Fibrosis Transmembrane Conductance Regulator. Implications in Cystic Fibrosis and Chronic Obstructive Pulmonary Disease. Ann Am Thorac Soc. 2016; S150–S155.
- 93. Mall MA, Hartl D. CFTR: cystic fibrosis and beyond. *Eur Respir J* 2014; 44: 1042–1054.
- 94. Clunes LA, Davies CM, Coakley RD, Aleksandrov AA, Henderson AG, Zeman KL, Worthington EN, Gentzsch M, Kreda SM, Cholon D, Bennett WD, Riordan JR, Boucher RC, Tarran R. Cigarette smoke exposure induces CFTR internalization and insolubility, leading to airway surface liquid dehydration. *FASEB J* 2012; 26: 533–545.
- 95. Sloane PA, Shastry S, Wilhelm A, Courville C, Tang LP, Backer K, Levin E, Raju SV, Li Y, Mazur M, Byan-Parker S, Grizzle W, Sorscher EJ, Dransfield MT, Rowe SM. A Pharmacologic Approach to Acquired Cystic Fibrosis Transmembrane Conductance Regulator Dysfunction in Smoking Related Lung Disease. *PLoS ONE* 2012; 7: e39809.
- 96. Bodas M, Min T, Vij N. Critical role of CFTR-dependent lipid rafts in cigarette smoke-induced lung epithelial injury. *Am J Physiol Lung Cell Mol Physiol* 2011; 300: L811–L820.
- 97. Bebok Z, Varga K, Hicks JK, Venglarik CJ, Kovacs T, Chen L, Hardiman KM, Collawn JF, Sorscher EJ, Matalon S. Reactive Oxygen Nitrogen Species

Decrease Cystic Fibrosis Transmembrane Conductance Regulator Expression and cAMP-mediated Cl- Secretion in Airway Epithelia. *J Biol Chem* 2002; 277: 43041–43049.

- 98. Andrew J Thorley TDT. Pulmonary epithelium, cigarette smoke, and chronic obstructive pulmonary disease. *Int J Chron Obstruct Pulmon Dis.* 2007; 2: 409.
- 99. Rab A, Rowe SM, Raju SV, Bebok Z, Matalon S, Collawn JF. Cigarette smoke and CFTR: implications in the pathogenesis of COPD. *Am J Physiol Lung Cell Mol Physiol* 2013; 305: L530–L541.
- Cantin AM, Hanrahan JW, Bilodeau G, Ellis L, Dupuis A, Liao J, Zielenski J, Durie
 P. Cystic Fibrosis Transmembrane Conductance Regulator Function Is
 Suppressed in Cigarette Smokers. *Am J Respir Crit Care Med* 2006; 173: 1139–1144.
- 101. Cholon DM, Quinney NL, Fulcher ML, Esther CR, Das J, Dokholyan NV, Randell SH, Boucher RC, Gentzsch M. Potentiator ivacaftor abrogates pharmacological correction of F508 CFTR in cystic fibrosis. *Sci Transl Med* 2014; 6: 246ra96.
- 102. Schamberger AC, Mise N, Meiners S. Epigenetic mechanisms in COPD: implications for pathogenesis and drug discovery. *Expert Opin Drug Discov* 2014; 9: 609–628.
- 103. Comer BS, Ba M, Singer CA, Gerthoffer WT. Epigenetic targets for novel therapies of lung diseases. *Pulm Pharmacol Ther.* 2015; 147: 91–110.
- 104. Gross NJ. Novel Antiinflammatory Therapies for COPD. *Chest* 2012; 142: 1300–1307.
- 105. Cazzola M, Page CP, Calzetta L, Matera MG. Emerging anti-inflammatory strategies for COPD. *Eur Respir J* 2012; 40: 724–741.
- 106. Lambert JA, Raju SV, Tang LP, McNicholas CM, Li Y, Courville CA, Farris RF, Coricor GE, Smoot LH, Mazur MM, Dransfield MT, Bolger GB, Rowe SM. Cystic Fibrosis Transmembrane Conductance Regulator Activation by Roflumilast Contributes to Therapeutic Benefit in Chronic Bronchitis. *Am J Respir Cell Mol Biol* 2014; 50: 549–558.
- 107. Durham AL, Caramori G, Chung KF, Adcock IM. Targeted anti-inflammatory therapeutics in asthma and chronic obstructive lung disease. *Transl Res* 2016; 167: 192–203.
- 108. Robb CT, Regan KH, Dorward DA, Rossi AG. Key mechanisms governing resolution of lung inflammation. *Semin Immunopathol* 2016; 38: 425–448.
- 109. Hiemstra PS, McCray PB, Bals R. The innate immune function of airway epithelial cells in inflammatorylung disease. *Eur Respir J* 2015; 45: 1150–1162.
- 110. Thorley AJ, Tetleyy TD. Pulmonary epithelium, cigarette smoke, and chronic obstructive pulmonary disease. *Int J Chron Obstruct Pulmon Dis* 2007; 2: 409–428

- 111. Janes PW, Saha N, Barton WA, Kolev MV, Wimmer-Kleikamp SH, Nievergall E, Blobel CP, Himanen J-P, Lackmann M, Nikolov DB. Adam Meets Eph: An ADAM Substrate Recognition Module Acts as a Molecular Switch for Ephrin Cleavage In trans. *Cell* 2005; 123: 291–304.
- 112. Takeda S, Igarashi T, Mori H, Araki S. Crystal structures of VAP1 reveal ADAMs' MDC domain architecture and its unique C-shaped scaffold. *EMBO J* 2006; 25: 2388–2396.
- 113. Takeda S. Three-dimensional domain architecture of the ADAM family proteinases. *Semin Cell Dev Biol* 2009; 20: 146–152.
- 114. Lorenzen I, Lokau J, Düsterhöft S, Trad A, Garbers C, Scheller J, Rose-John S, Grötzinger J. The membraneproximal domain of A Disintegrin and Metalloprotease 17 (ADAM17) is responsible for recognition of the interleukin6 receptor and interleukin1 receptor II. *FEBS Lett* 2012; 586: 1093–1100.
- 115. Düsterhöft S, Michalek M, Kordowski F, Oldefest M, Sommer A, Röseler J, Reiß K, Grötzinger J, Lorenzen I. Extracellular Juxtamembrane Segment of ADAM17 Interacts with Membranes and Is Essential for Its Shedding Activity. *Biochemistry* 2015; 54: 5791–5801.
- 116. Scheller J, Chalaris A, Garbers C, Rose-John S. ADAM17: a molecular switch to control inflammation and tissue regeneration. *Trends Immunol* 2011; 32: 380–387.
- 117. Düsterhöft S, Höbel K, Oldefest M, Lokau J, Waetzig GH, Chalaris A, Garbers C, Scheller J, Rose-John S, Lorenzen I, Grötzinger J. A Disintegrin and Metalloprotease 17 Dynamic Interaction Sequence, the Sweet Tooth for the Human Interleukin 6 Receptor. *J Biol Chem* 2014; 289: 16336–16348.
- 118. Deshmukh HS, Case LM, Wesselkamper SC, Borchers MT, Martin LD, Shertzer HG, Nadel JA, Leikauf GD. Metalloproteinases Mediate Mucin 5AC Expression by Epidermal Growth Factor Receptor Activation. *Am J Respir Crit Care Med* 2005; 171: 305–314.
- 119. Shao MXG, Nakanaga T, Nadel JA. Cigarette smoke induces MUC5AC mucin overproduction via tumor necrosis factor-α-converting enzyme in human airway epithelial (NCI-H292) cells. Am J Physiol Lung Cell Mol Physiol 2004; 287: L420–L427.
- 120. Shao MXG, Nadel JA. Dual oxidase 1-dependent MUC5AC mucin expression in cultured human airway epithelial cells. *Proc Natl Acad Sci U S A* 2005; 102: 767–772.
- 121. Richter A, O'Donnell RA, Powell RM. Autocrine ligands for the epidermal growth factor receptor mediate interleukin-8 release from bronchial epithelial cells in response to cigarette smoke. *Am J Respir Crit Care Med* 2002; 27: 85–90.
- 122. Pasqualon T, Pruessmeyer J, Weidenfeld S, Babendreyer A, Groth E, Schumacher J, Schwarz N, Denecke B, Jahr H, Zimmermann P, Dreymueller D,

Ludwig A. A transmembrane C-terminal fragment of syndecan-1 is generated by the metalloproteinase ADAM17 and promotes lung epithelial tumor cell migration and lung metastasis formation. *Cell. Mol. Life Sci* 2015; 72: 3783–3801.

- 123. Broekman W, Amatngalim GD, de Mooij-Eijk Y, Oostendorp J, Roelofs H, Taube C, Stolk J, Hiemstra PS. TNF-α and IL-1β-activated humanmesenchymal stromal cells increase airwayepithelial wound healing in vitro viaactivation of the epidermal growth factorreceptor. *Respir Res* 2016; 17: 1–12.
- 124. Lemjabbar H, Li D, Gallup M, Sidhu S, Drori E, Basbaum C. Tobacco Smokeinduced Lung Cell Proliferation Mediated by Tumor Necrosis Factor αconverting Enzyme and Amphiregulin. J Biol Chem 2003; 278: 26202–26207.
- 125. Vallath S, Hynds RE, Succony L, Janes SM, Giangreco A. Targeting EGFR signaling in chronic lung disease: therapeutic challenges and opportunities. *Eur Respir J* 2014; 44: 513–522.
- 126. Breshears LM, Schlievert PM, Peterson ML. A Disintegrin and Metalloproteinase 17 (ADAM17) and Epidermal Growth Factor Receptor (EGFR) Signaling Drive the Epithelial Response to Staphylococcus aureus Toxic Shock Syndrome Toxin-1 (TSST-1). J Biol Chem 2012; 287: 32578–32587.
- 127. Zhu L, Lee P, Lee W, Zhao Y, Yu D, Chen Y. Rhinovirus-induced major airway mucin production involves a novel TLR3-EGFR–dependent pathway. *Am J Respir Cell Mol Biol* 2009; 40: 610–619.
- 128. Hudy MH, Proud D. Cigarette smoke enhances human rhinovirus-induced CXCL8 production via HuR-mediated mRNA stabilization in human airway epithelial cells. *Respir Res* 2013; 14: 88.
- 129. Vermeer PD, Panko L, Welsh MJ, Zabner J. erbB1 functions as a sensor of airway epithelial integrity by regulation of protein phosphatase 2A activity. *J Biol Chem* 2006; 281: 1725–1730.
- 130. Fan H, Turck CW, Derynck R. Characterization of Growth Factor-induced Serine Phosphorylation of Tumor Necrosis Factor-alpha Converting Enzyme and of an Alternatively Translated Polypeptide. *J Biol Chem* 2003; 278: 18617–18627.
- 131. Finigan JH, Faress JA, Wilkinson E, Mishra RS, Nethery DE, Wyler D, Shatat M, Ware LB, Matthay MA, Mason R, Silver RF, Kern JA. Neuregulin-1-Human Epidermal Receptor-2 Signaling Is a Central Regulator of Pulmonary Epithelial Permeability and Acute Lung Injury. *J Biol Chem* 2011; 286: 10660–10670.
- 132. Dreymueller D, Uhlig S, Ludwig A. ADAM-family metalloproteinases in lung inflammation: potential therapeutic targets. *Am J Physiol Lung Cell Mol Physiol* 2015; 308: L325–L343.
- 133. Peschon JJ, Slack JL, Reddy P, Stocking KL, Sunnarborg SW, Lee DC, Russell WE, Castner BJ, Johnson RS, Fitzner JN, Boyce RW, Nelson N, Kozlosky CJ, Wolfson MF, Rauch CT, Cerretti DP, Paxton RJ, March CJ, Black RA. An essential role for ectodomain shedding in mammalian development. *Science* 1998; 282: 1281– 1284.

- 134. Horiuchi K, Kimura T, Miyamoto T, Takaishi H, Okada Y, Toyama Y, Blobel CP. Cutting Edge: TNF-alpha-Converting Enzyme (TACE/ADAM17) Inactivation in Mouse Myeloid Cells Prevents Lethality from Endotoxin Shock. *J Immunol* 2007; 179: 2686–2689.
- 135. Chalaris A, Adam N, Sina C, Rosenstiel P, Lehmann-Koch J, Schirmacher P, Hartmann D, Cichy J, Gavrilova O, Schreiber S, Jostock T, Matthews V, Häsler R, Becker C, Neurath MF, Reiß K, Saftig P, Scheller J, Rose-John S. Critical role of the disintegrin metalloprotease ADAM17 for intestinal inflammation and regeneration in mice. *J Exp Med* 2010; 207: 1617–1624.
- 136. Miettinen PJ, Berger JE, Meneses J, Phung Y, Pedersen RA, Werb Z, Derynck R. Epithelial immaturity and multiorgan failure in mice lacking epidermal growth factor receptor. *Nature* 1995; 376: 337–341.
- 137. Le Gall SM, Bobé P, Reiss K, Horiuchi K, Niu XD, Lundell D, Gibb DR, Conrad D, Saftig P, Blobel CP. ADAMs 10 and 17 represent differentially regulated components of a general shedding machinery for membrane proteins such as transforming growth factor α. *Mol Biol Cell* 2009; 20: 1785–1794.
- 138. Le Gall SM, Maretzky T, Issuree PDA, Niu XD, Reiss K, Saftig P, Khokha R, Lundell D, Blobel CP. ADAM17 is regulated by a rapid and reversible mechanism that controls access to its catalytic site. *J Cell Sci* 2010; 123: 3913– 3922.
- 139. Sham D, Wesley UV, Hristova M, van der Vliet A. ATP-Mediated Transactivation of the Epidermal Growth Factor Receptor in Airway Epithelial Cells Involves DUOX1-Dependent Oxidation of Src and ADAM17. *PLoS ONE* 2013; 8: e54391.
- 140. Baumgart A, Seidl S, Vlachou P, Michel L, Mitova N, Schatz N, Specht K, Koch I, Schuster T, Grundler R, Kremer M, Fend F, Siveke JT, Peschel C, Duyster J, Dechow T. ADAM17 Regulates Epidermal Growth Factor Receptor Expression through the Activation of Notch1 in Non-Small Cell Lung Cancer. *Cancer Res* 2010; 70: 5368–5378.
- 141. Ermert M, Pantazis C, Duncker HR, Grimminger F, Seeger W, Ermert L. In situ localization of $TNF\alpha/\beta$, TACE AND TNF receptors TNF-R1 and TNF-R2 in control and LPS-treated lung tissue. *Cytokine* 2003; 22: 89–100.
- 142. Dijkstra A, Postma DS, Noordhoek JA, Lodewijk ME, Kauffman HF, Hacken ten NHT, Timens W. Expression of ADAMs ("a disintegrin and metalloprotease") in the human lung. *Virchows Arch* 2009; 454: 441–449.
- 143. Xu P, Derynck R. Direct Activation of TACE-Mediated Ectodomain Shedding by p38 MAP Kinase Regulates EGF Receptor-Dependent Cell Proliferation. *Mol Cell* 2010; 37: 551–566.
- 144. Booth BW, Sandifer T, Martin EL, Martin LD. IL-13-induced proliferation of airway epithelial cells: mediation by intracellular growth factor mobilization and ADAM17. *Respir Res* 2007; 8: 51.
- 145. Amatngalim GD, van Wijck Y, de Mooij-Eijk Y, Verhoosel RM, Harder J,

Lekkerker AN, Janssen RAJ, Hiemstra PS. Basal Cells Contribute to Innate Immunity of the Airway Epithelium through Production of the Antimicrobial Protein RNase 7. *J Immunol* 2015; 194: 3340–3350.

- 146. Rose-John S. IL-6 Trans-Signaling via the Soluble IL-6 Receptor: Importance for the Pro-Inflammatory Activities of IL-6. *Int J Biol Sci* 2012; 8: 1237–1247.
- 147. Nam H-S, Lee SY, Kim SJ, Kim JS, Kwon SS, Kim YK, Kim KH, Moon HS, Song JS, Park SH, Kim SC. The soluble tumor necrosis factor-alpha receptor suppresses airway inflammation in a murine model of acute asthma. *Yonsei Med J* 2009; 50: 569–575.
- 148. Dang M, Dubbin K, D'Aiello A, Hartmann M, Lodish H, Herrlich A. Epidermal growth factor (EGF) ligand release by substrate-specific a disintegrin and metalloproteases (ADAMs) involves different protein kinase C (PKC) isoenzymes depending on the stimulus. *J Biol Chem* 2011; 286: 17704–17713.
- 149. Killock DJ, Ivetić A. The cytoplasmic domains of TNFα-converting enzyme (TACE/ADAM17) and L-selectin are regulated differently by p38 MAPK and PKC to promote ectodomain shedding. *Biochem. J.* 2010; 428: 293–304.
- 150. Horiuchi K, Le Gall S, Schulte M, Yamaguchi T, Reiß K, Murphy G, Toyama Y, Hartmann D, Saftig P, Blobel CP. Substrate Selectivity of Epidermal Growth Factor-Receptor Ligand Sheddases and their Regulation by Phorbol Esters and Calcium Influx. *Mol Biol Cell* 2007; 18: 176–188.
- 151. Wang X-J, Feng C-W, Li M. ADAM17 mediates hypoxia-induced drug resistance in hepatocellular carcinoma cells through activation of EGFR/PI3K/Akt pathway. *Mol Cell Biochem* 2013; 380: 57–66.
- 152. Xu W, Liu C, Kaartinen V, Chen H, Lu CH, Zhang W, Luo Y, Shi W. TACE in perinatal mouse lung epithelial cells promotes lung saccular formation. *Am J Physiol Lung Cell Mol Physiol* 2013; 305: L953–L963.
- 153. Maretzky T, Evers A, Zhou W, Swendeman SL, Wong P-M, Rafii S, Reiß K, Blobel CP. Migration of growth factor-stimulated epithelialand endothelial cells depends on EGFRtransactivation by ADAM17. *Nat Commun* 2011; 2: 229.
- 154. Ni S-S, Zhang J, Zhao W-L, Dong X-C, Wang J-L. ADAM17 is overexpressed in non-small cell lung cancer and its expression correlates with poor patient survival. *Tumor Biol.* 2013; 34: 1813–1818.
- 155. Huang Y, Benaich N, Tape C, Kwok HF, Murphy G. Targeting the Sheddase Activity of ADAM17 by an Anti-ADAM17 Antibody D1(A12) Inhibits Head and Neck Squamous Cell Carcinoma Cell Proliferation and Motility via Blockage of Bradykinin Induced HERs Transactivation. *Int J Biol Sci* 2014; 10: 702–714.
- 156. Caiazza F, McGowan PM, Mullooly M, Murray A, Synnott N, O'Donovan N, Flanagan L, Tape CJ, Murphy G, Crown J, Duffy MJ. Targeting ADAM-17 with an inhibitory monoclonal antibody has antitumour effects in triple-negative breast cancer cells. *Br J Cancer* 2015; 112: 1895–1903.
- 157. Zheng D-Y, Zhao J, Yang J-M, Wang M, Zhang X-T. Enhanced ADAM17

expression is associated with cardiac remodeling in rats with acute myocardial infarction. *Life Sci* 2016; 151: 61–69.

- 158. Nadel JA. The CFTR and EGFR relationship in airway vascular growth, and its importance in cystic fibrosis. *Eur Respir J* 2013; 42: 1438–1440.
- 159. Carvalho-Oliveira IM, Charro N, Aarbiou J, Buijs-Offerman RM, Wilke M, Schettgen T, Kraus T, Titulaer MK, Burgers P, Luider TM, Penque D, Scholte BJ. Proteomic Analysis of Naphthalene-Induced Airway Epithelial Injury and Repair in a Cystic Fibrosis Mouse Model. *J Proteome Res* 2009; 8: 3606–3616.
- 160. Martel G, Roussel L, Rousseau S. The protein kinases TPL2 and EGFR contribute to ERK1/ERK2 hyper- activation in CFTRDF508-expressing airway epithelial cells exposed to Pseudomonas aeruginosa. *Biochem Biophys Res Commun* 2013; 441: 689–692.
- 161. Martin C, Coolen N, Wu Y, Thevenot G, Touqui L, Pruliere-Escabasse V, Papon JF, Coste A, Escudier E, Dusser DJ, Fajac I, Burgel P-R. CFTR dysfunction induces vascular endothelial growth factor synthesis in airway epithelium. *Eur Respir J* 2013; 42: 1553–1562.
- 162. Berube J, Roussel L, Nattagh L, Rousseau S. Loss of Cystic Fibrosis Transmembrane Conductance Regulator Function Enhances Activation of p38 and ERK MAPKs, Increasing Interleukin-6 Synthesis in Airway Epithelial Cells Exposed to Pseudomonas aeruginosa. *J Biol Chem* 2010; 285: 22299–22307.
- 163. Trinh NTN, Prive A, Maille E, Noel J, Brochiero E. EGF and K+ channel activity control normal and cystic fibrosis bronchial epithelia repair. *Am J Physiol Lung Cell Mol Physiol* 2008; 295: L866–L880.
- 164. Kelly M, Trudel S, Brouillard F, Bouillaud F, Colas J, Nguyen-Khoa T, Ollero M, Edelman A, Fritsch J. Cystic Fibrosis Transmembrane Regulator Inhibitors CFTRinh-172 and GlyH-101 Target Mitochondrial Functions, Independently of Chloride Channel Inhibition. *J Pharmacol Exp Ther.* 2010; 333: 60–69.
- 165. Koff JL, Shao MXG, Ueki IF, Nadel JA. Multiple TLRs activate EGFR via a signaling cascade to produce innate immune responses in airway epithelium. *Am J Physiol Lung Cell Mol Physiol* 2008; 294: L1068–L1075.
- 166. Yu H, Zhou X, Wen S, Xiao Q. Flagellin/TLR5 responses induce mucus hypersecretion by activating EGFR via an epithelial cell signaling cascades. *Exp Cell Res* 2012; 318: 723–731.
- 167. Wang R, Ahmed J, Wang G, Hassan I, Strulovici-Barel Y, Salit J, Mezey JG, Crystal RG. Airway Epithelial Expression of TLR5 Is Downregulated in Healthy Smokers and Smokers with Chronic Obstructive Pulmonary Disease. *J Immunol* 2012; 189: 2217–2225.
- 168. Gschwind A, Hart S, Fischer OM, Ullrich A. TACE cleavage of proamphiregulin regulates GPCR-induced proliferation and motility of cancer cells. *EMBO J* 2003; 22: 2411–2421.
- 169. Heijink IH, Brandenburg SM, Noordhoek JA, Slebos D-J, Postma DS, van

Oosterhout AJ. Role of aberrant metalloproteinase activity in theproinflammatory phenotype of bronchialepithelium in COPD. *Respir Res* 2011; 12: 110.

- 170. Baginski TK, Dabbagh K, Satjawatcharaphong C, Swinney DC. Cigarette Smoke Synergistically Enhances Respiratory Mucin Induction by Proinflammatory Stimuli. *Am J Respir Cell Mol Biol* 2006; 35: 165–174.
- 171. Kim S, Lewis C, Nadel JA. CCL20/CCR6 Feedback Exaggerates Epidermal Growth Factor Receptor-Dependent MUC5AC Mucin Production in Human Airway Epithelial (NCI-H292) Cells. *J Immunol* 2011; 186: 3392–3400.
- 172. Liu K, Gualano RC, Hibbs ML, Anderson GP, Bozinovski S. Epidermal growth factor receptor signaling to Erk1/2 and STATs control the intensity of the epithelial inflammatory responses to rhinovirus infection. *J Biol Chem* 2008; 283: 9977–9985.
- 173. Blanchet S, Ramgolam K, Baulig A, Marano F, Baeza-Squiban A. Fine Particulate Matter Induces Amphiregulin Secretion by Bronchial Epithelial Cells. *Am J Respir Cell Mol Biol* 2004; 30: 421–427.
- 174. Amour A, Slocombe PM, Webster A, Butler M, Knight CG, Smith BJ, Stephens PE, Shelley C, M H, Knäuper V, AJ D, G M. TNF-α converting enzyme (TACE) is inhibited by TIMP-3. *FEBS Lett* 1998; 435: 39–44.
- 175. Mahmoodi M, Sahebjam S, Smookler D, Khokha R, Mort JS. Lack of tissue inhibitor of metalloproteinases-3 results in an enhanced inflammatory response in antigen-induced arthritis. *Am J Pathol.* 2005; 166: 1733–1740.
- 176. Smookler DS, Mohammed FF, Kassiri Z, Duncan GS, Mak TW, Khokha R. Cutting Edge: Tissue Inhibitor of Metalloproteinase 3 Regulates TNF-Dependent Systemic Inflammation. *J Immunol* 2006; 176: 721–725.
- 177. Wisniewska M, Goettig P, Maskos K, Belouski E, Winters D, Hecht R, Black R, Bode W. Structural Determinants of the ADAM Inhibition by TIMP-3: Crystal Structure of the TACE-N-TIMP-3 Complex. *J Mol Biol* 2008; 381: 1307–1319.
- 178. Xu P, Liu J, Sakaki-Yumoto M, Derynck R. TACE Activation by MAPK-Mediated Regulation of Cell Surface Dimerization and TIMP3 Association. *Sci Signal* 2012; 5: ra34.
- 179. Willems SH, Tape CJ, Stanley PL, Taylor NA, Mills IG, Neal DE, McCafferty J, Murphy G. Thiol isomerases negatively regulate the cellular shedding activity of ADAM17. *Biochem. J.* 2010; 428: 439–450.
- 180. Düsterhöft S, Jung S, Hung C-W, Tholey A, Sönnichsen FD, Grötzinger J, Lorenzen I. Membrane-Proximal Domain of a Disintegrin and Metalloprotease-17 Represents the Putative Molecular Switch of Its Shedding Activity Operated by Protein-disulfide Isomerase. *J Am Chem Soc* 2013; 135: 5776–5781.
- Gutscher M, Pauleau A-L, Marty L, Brach T, Wabnitz GH, Samstag Y, Meyer AJ, Dick TP. Real-time imaging of the intracellular glutathione redox potential. *Nat Meth* 2008; 5: 553–559.

- 182. Boots AW, Hristova M, Kasahara DI, Haenen GRMM, Bast A, van der Vliet A. ATP-mediated Activation of the NADPH Oxidase DUOX1 Mediates Airway Epithelial Responses to Bacterial Stimuli. *J Biol Chem* 2009; 284: 17858–17867.
- Hristova M, Habibovic A, Veith C, Janssen-Heininger YM, Dixon AE, Geiszt M, van der Vliet A. Airway epithelial dual oxidase 1 mediates allergen-induced IL-33 secretion and activation of type 2 immune responses. J Allergy Clin Immunol. 2016; 137: 1545–1556.
- 184. Paulsen CE, Truong TH, Garcia FJ, Homann A, Gupta V, Leonard SE, Carroll KS. Peroxide-dependent sulfenylation of the EGFR catalytic site enhances kinase activity. *Nat Chem Biol* 2011; 8: 57–64.
- 185. Wang Y, Herrera AH, Li Y, Belani KK, Walcheck B. Regulation of Mature ADAM17 by Redox Agents for L-Selectin Shedding. *J Immunol* 2009; 182: 2449–2457.
- 186. Bennett TA, Edwards BS, Sklar LA, Rogelj S. Sulfhydryl Regulation of L-Selectin Shedding: Phenylarsine Oxide Promotes Activation-Independent L-Selectin Shedding from Leukocytes. *J Immunol* 2000; 164: 4120–4129.
- 187. Ziady AG, Hansen J. Redox balance in cystic fibrosis. *Int J Biochem Cell Biol* 2014; 52: 113–123.
- 188. Juarez JC, Manuia M, Burnett ME, Betancourt O, Boivin B, Shaw DE, Tonks NK, Mazar AP, Donate F. Superoxide dismutase 1 (SOD1) is essential for H2O2mediated oxidation and inactivation of phosphatases in growth factor signaling. *Proc Natl Acad Sci U S A*. 2008; 105: 7147–7152.
- 189. Ravi AK, Khurana S, Lemon J, Plumb J, Booth G, Healy L, Catley M, Vestbo J, Singh D. Increased levels of soluble interleukin-6 receptor and CCL3 in COPD sputum. *Respir Res* 2014; 15: 103.
- 190. Hirota N, Risse PA, Novali M, McGovern T, Al-Alwan L, McCuaig S, Proud D, Hayden P, Hamid Q, Martin JG. Histamine may induce airway remodeling through release of epidermal growth factor receptor ligands from bronchial epithelial cells. *FASEB J* 2012; 26: 1704–1716.
- 191. Deacon K, Knox AJ. Human airway smooth muscle cells secrete amphiregulin via bradykinin/COX-2/PGE 2, inducing COX-2, CXCL8, and VEGF expression in airway epithelial cells. *Am J Physiol Lung Cell Mol Physiol* 2015; 309: L237–L249.
- 192.Vermeer PD. Differentiation of human airway epithelia is dependent on erbB2.Am J Physiol Lung Cell Mol Physiol 2006; 291: L175–L180.
- 193. Meng C, Liu G, Mu H, Zhou M, Zhang S, Xu Y. Amphiregulin may be a new biomarker of classically activated macrophages. *Biochem Biophys Res Commun* 2015; 466: 393–399.
- 194. Matsumoto K, Fukuda S, Nakamura Y, Saito H. Amphiregulin Production by Human Eosinophils. *Int Arch Allergy Immunol* 2009; 149: 39–44.

- 195. Bles N, Di Pietrantonio L, Boeynaems JM, Communi D. ATP confers tumorigenic properties to dendritic cells by inducing amphiregulin secretion. *Blood* 2010; 116: 3219–3226.
- 196. Adib-Conquy M, Thierry Pedron, Petit-Bertron A-F, Tabary O, Harriet Corvol, Jacquot J, Annick Clément, Cavaillon J-M. Neutrophils in cystic fibrosis display a distinct gene expression pattern. *Mol Med* 2008; 14: 36–44.
- 197. Wang S-W, Oh CK, Cho SH, Hu G, Martin R, Demissie-Sanders S, Li K, Moyle M, Yao Z. Amphiregulin expression in human mast cells and its effect on the primary human lung fibroblasts. *J Allergy Clin Immunol.* 2005; 115: 287–294.
- 198. Brown CL, Meise KS, Plowman GD, Coffey RJ, PJ D. Cell surface ectodomain cleavage of human amphiregulin precursor is sensitive to a metalloprotease inhibitor. Release of a predominant N-glycosylated 43-kDa soluble form. *J Biol Chem* 1998; 273: 17258–17268.
- 199. Gephart JD, Singh B, Higginbotham JN, Franklin JL, Gonzalez A, Fölsch H, Coffey RJ. Identification of a Novel Mono-Leucine Basolateral Sorting Motif Within the Cytoplasmic Domain of Amphiregulin. *Traffic* 2011; 12: 1793–1804.
- 200. Zaiss DMW, Gause WC, Osborne LC, Artis D. Emerging Functions of Amphiregulin in Orchestrating Immunity, Inflammation, and Tissue Repair. *Immunity* 2015; 42: 216–226.
- 201. Enomoto Y, Orihara K, Takamasu T, Matsuda A, Gon Y, Saito H, Ra C, Okayama Y. Tissue remodeling induced by hypersecreted epidermal growth factor and amphiregulin in the airway after an acute asthma attack. *J Allergy Clin Immunol* 2009; 124: 913–920.e1–7.
- 202. Kim KW, Jee HM, Park YH, Choi BS, Sohn MH, Kim K-E. Relationship Between Amphiregulin and Airway Inflammation in Children With Asthma and Eosinophilic Bronchitis. *Chest* 2009; 136: 805–810.
- 203. Shimizu S, Kouzaki H, Ogawa T, Takezawa K, Tojima I, Shimizu T. Eosinophilepithelial cell interactions stimulate the production of MUC5AC mucin and profibrotic cytokines involved in airway tissue remodeling. *Am J Rhinol Allergy* 2014; 28: 103–109.
- 204. Parnia S, Hamilton LM, Puddicombe SM, Holgate ST, Frew AJ, Davies DE. Autocrine ligands of the epithelial growth factor receptor mediate inflammatory responsesto diesel exhaust particles. *Respir Res* 2014; 15: 1–10.
- 205. Lemjabbar-Alaoui H, Sidhu SS, Mengistab A, Gallup M, Basbaum C. TACE/ADAM-17 Phosphorylation by PKC-Epsilon Mediates Premalignant Changes in Tobacco Smoke-Exposed Lung Cells. *PLoS ONE* 2011; 6: e17489.
- 206. Leigh R, Oyelusi W, Wiehler S, Koetzler R, Zaheer RS, Newton R, Proud D. Human rhinovirus infection enhances airway epithelial cell production of growth factors involved in airway remodeling. *J Allergy Clin Immunol.* 2008; 121: 1238–1245.e4.
- 207. Chokki M, Eguchi H, Hamamura I, Mitsuhashi H, Kamimura T. Human airway

trypsin-like protease induces amphiregulin release through a mechanism involving protease-activated receptor-2-mediated ERK activation and TNF alpha-converting enzyme activity in airway epithelial cells. *FEBS J* 2005; 272: 6387–6399.

- 208. Du B, Altorki NK, Kopelovich L, Subbaramaiah K, Dannenberg AJ. Tobacco Smoke Stimulates the Transcription of Amphiregulin in Human Oral Epithelial Cells: Evidence of a Cyclic AMP-Responsive Element Binding Protein-Dependent Mechanism. *Cancer Res* 2005; 65: 5982–5988.
- 209. Okumura S, Sagara H, Fukuda T, Saito H, Okayama Y. FccRI-mediated amphiregulin production by human mast cells increases mucin gene expression in epithelial cells. *J Allergy Clin Immunol.* 2005; 115: 272–279.
- 210. Val S, Belade E, George I, Boczkowski J, Baeza-Squiban A. Fine PM induce airway MUC5AC expression through the autocrine effect of amphiregulin. *Arch Toxicol* 2012; 86: 1851–1859.
- 211. Chokki M, Mitsuhashi H, Kamimura T. Metalloprotease-dependent amphiregulin release mediates tumor necrosis factor-α-induced IL-8 secretion in the human airway epithelial cell line NCI-H292. *Life Sci* 2006; 78: 3051–3057.
- Zhou Y, Lee JY, Lee CM, Cho WK, Kang MJ, Koff JL, Yoon PO, Chae J, Park HO, Elias JA, Lee CG. Amphiregulin, an epidermal growth factor receptor ligand, plays an essential role in the pathogenesis of transforming growth factor-β-induced pulmonary fibrosis. *J Biol Chem* 2012; 287: 41991–42000.
- 213. Shim JY, Park SW, Kim DS, Shim JW, Jung HL, Park MS. The Effect of Interleukin-4 and Amphiregulin on the Proliferation of Human Airway Smooth Muscle Cells and Cytokine Release. *J Korean Med Sci* 2008; 23: 857.
- 214. Du B, Leung H, Khan KMF, Miller CG, Subbaramaiah K, Falcone DJ, Dannenberg AJ. Tobacco Smoke Induces Urokinase-Type Plasminogen Activator and Cell Invasiveness: Evidence for an Epidermal Growth Factor Receptor Dependent Mechanism. *Cancer Res* 2007; 67: 8966–8972.
- 215. Huang F, Khvorova A, Marshall W, Sorkin A. Analysis of Clathrin-mediated Endocytosis of Epidermal Growth Factor Receptor by RNA Interference. *J Biol Chem* 2004; 279: 16657–16661.
- 216. Sigismund S, Argenzio E, Tosoni D, Cavallaro E, Polo S, Di Fiore PP. Clathrin-Mediated Internalization Is Essential for Sustained EGFR Signaling but Dispensable for Degradation. *Dev Cell* 2008; 15: 209–219.
- 217. Rappoport JZ, Simon SM. Endocytic trafficking of activated EGFR is AP-2 dependent and occurs through preformed clathrin spots. *J Cell Sci* 2009; 122: 1301–1305.
- 218. Goh LK, Huang F, Kim W, Gygi S, Sorkin A. Multiple mechanisms collectively regulate clathrin-mediated endocytosis of the epidermal growth factor receptor. *J Cell Biol* 2010; 189: 871–883.
- 219. Henriksen L, Grandal MV, Knudsen SLJ, van Deurs B, Grøvdal LM.

Internalization Mechanisms of the Epidermal Growth Factor Receptor after Activation with Different Ligands. *PLoS ONE* 2013; 8: e58148.

- 220. Tan X, Lambert PF, Rapraeger AC, Anderson RA. Stress-Induced EGFR Trafficking: Mechanisms, Functions, and Therapeutic Implications. *Trends Cell Biol* 2016; 28: 352–366.
- 221. Tomas A, Futter CE, Eden ER. EGF receptor trafficking: consequences for signaling and cancer. *Trends Cell Biol* 2014; 24: 26–34.
- 222. Singh B, Coffey RJ. Trafficking of Epidermal Growth Factor Receptor Ligands in Polarized Epithelial Cells. *Annu Rev Physiol* 2014; 76: 275–300.
- 223. Jones S, Cunningham DL, Rappoport JZ, Heath JK. The non-receptor tyrosine kinase Ack1 regulates the fate of activated EGFR by inducing trafficking to the p62/NBR1 pre-autophagosome. *J Cell Sci* 2014; 127: 994–1006.
- Higginbotham JN, Beckler MD, Gephart JD, Franklin JL, Bogatcheva G, Kremers
 G-J, Piston DW, Ayers GD, McConnell RE, Tyska MJ, Coffey RJ. Amphiregulin
 Exosomes Increase Cancer Cell Invasion. *Curr Biol* 2011; 21: 779–786.
- 225. Ebsen H, Lettau M, Kabelitz D, Janssen O. Subcellular localization and activation of ADAM proteases in the context of FasL shedding in T lymphocytes. *Mol Immunol* 2015; 65: 416–428.
- 226. Fukuda S, Nishida-Fukuda H, Nakayama H, Inoue H, Higashiyama S. Monoubiquitination of pro-amphiregulin regulates its endocytosis and ectodomain shedding. *Biochem Biophys Res Commun* 2012; 420: 315–320.
- 227. Boulanger MJ, Chow DC, Brevnova EE, Garcia KC. Hexameric structure and assembly of the interleukin-6/IL-6 alpha-receptor/gp130 complex. *Science* 2003; 300: 2101–2104.
- 228. Müllberg J, Oberthür W, Lottspeich F, Mehl E, Dittrich E, Graeve L, Heinrich PC, Rose-John S. The soluble human IL-6 receptor. Mutational characterization of the proteolytic cleavage site. *J Immunol* 1994; 152: 4958–4968.
- 229. Baran P, Dunitz JM, Grotzinger J, Scheller J, Garbers C. Minimal interleukin 6 (IL-6) receptor stalk composition for IL-6 receptor shedding and IL-6 classic signaling. *J Biol Chem* 2013; 288: 14756–14768.
- 230. Schumacher N, Meyer D, Mauermann A, Heyde von der J, Wolf J, Schwarz J, Knittler K, Murphy G, Michalek M, Garbers C, Bartsch JW, Guo S, Schacher B, Eickholz P, Chalaris A, Rose-John S, Rabe B. Shedding of Endogenous Interleukin-6 Receptor (IL-6R) Is Governed by A Disintegrin and Metalloproteinase (ADAM) Proteases while a Full-length IL-6R Isoform Localizes to Circulating Microvesicles. *J Biol Chem* 2015; 290: 26059–26071.
- 231. Chalaris A, Gewiese J, Paliga K, Fleig L, Schneede A, Krieger K, Rose-John S, Scheller J. ADAM17-mediated shedding of the IL6R induces cleavage of the membrane stub by gamma-secretase. *Biochim Biophys Acta* 2010; 1803: 234–245.

- 232. Lust JA, Donovan KA, Kline MP, Greipp PR, Kyle RA, NJ M. Isolation of an mRNA encoding a soluble form of the human interleukin-6 receptor. *Cytokine* 1992; 4: 96–100.
- 233. Rath KS, Funk HM, Bowling MC, Richards WE, Drew AF. Expression of soluble interleukin-6 receptorin malignant ovarian tissue. *YMOB* 2010; 203: 230.e1–230.e8.
- Lamas JR, Rodríguez-Rodríguez L, Tornero-Esteban P, Villafuertes E, Hoyas J, Abasolo L, Varadé J, Álvarez-Lafuente R, Urcelay E, Fernández-Gutiérrez B. Alternative splicing and proteolytic rupture contribute to the generation of soluble IL-6 receptors (sIL-6R) in rheumatoid arthritis. *Cytokine* 2013; 61: 720– 723.
- 235. Rose-John S, Waetzig GH, Scheller J, Grötzinger J, Seegert D. The IL-6/sIL-6R complex as a novel target for therapeutic approaches. *Expert Opin Ther Targets* 2007; 11: 613–624.
- 236. Lee SY, Buhimschi IA, Dulay AT, Ali UA, Zhao G, Abdel-Razeq SS, Bahtiyar MO, Thung SF, Funai EF, Buhimschi CS. IL-6 Trans-Signaling System in Intra-Amniotic Inflammation, Preterm Birth, and Preterm Premature Rupture of the Membranes. *J Immunol* 2011; 186: 3226–3236.
- 237. Müllberg J, Dittrich E, Graeve L, Gerhartz C, Yasukawa K, Taga T, Kishimoto T, Heinrich PC, S R-J. Differential shedding of the two subunits of the interleukin-6 receptor. *FEBS Lett* 1993; 332: 174–178.
- 238. Diamant M, Rieneck K, Mechti N, Zhang X-G, Svenson M, Bendtzen K, Klein B. Cloning and expression of an alternatively spliced mRNA encoding a soluble form of the human interleukin-6 signal transducer gpl30. *FEBS Lett* 1997; 412: 379–384.
- 239. Heinrich PC, Behrmann I, Haan S, Hermanns HM, Müller-Newen G, Schaper F. Principles of interleukin (IL)-6-type cytokine signaling and its regulation. *Biochem J* 2003; 374: 1–20.
- 240. Grivennikov S, Karin E, Terzic J, Mucida D, Yu G-Y, Vallabhapurapu S, Scheller J, Rose-John S, Cheroutre H, Eckmann L, Karin M. IL-6 and Stat3 Are Requiredfor Survival of Intestinal Epithelial Cellsand Development of Colitis-Associated Cancer. *Cancer Cell* 2009; 15: 103–113.
- 241. Barkhausen T, Tschernig T, Rosenstiel P, van Griensven M, Vonberg R-P, Dorsch M, Mueller-Heine A, Chalaris A, Scheller J, Rose-John S, Seegert D, Krettek C, Waetzig GH. Selective blockade of interleukin-6 trans-signaling improves survival in a murine polymicrobial sepsis model. *Crit Care Med* 2011; 39: 1407–1413.
- 242. Nowell MA, Williams AS, Carty SA, Scheller J, Hayes AJ, Jones GW, Richards PJ, Slinn S, Ernst M, Jenkins BJ, Topley N, Rose-John S, Jones SA. Therapeutic Targeting of IL-6 Trans Signaling Counteracts STAT3 Control of Experimental Inflammatory Arthritis. *J Immunol* 2008; 182: 613–622.
- 243. Brooks GD, McLeod L, Alhayyani S, Miller A, Russell PA, Ferlin W, Rose-John S,

Ruwanpura S, Jenkins BJ. IL6 Trans-signaling Promotes KRAS-Driven Lung Carcinogenesis. *Cancer Res* 2016; 76: 866–876.

- 244. Oberg HH, Wesch D, Grüssel S, Rose-John S, Kabelitz D. Differential expression of CD126 and CD130 mediates different STAT-3 phosphorylation in CD4+CD25and CD25high regulatory T cells. *Int Immunol* 2006; 18: 555–563.
- 245. Mihara M, Hashizume M, Yoshida H, Suzuki M, Shiina M. IL-6/IL-6 receptor system and its role in physiological and pathological conditions. *Clin Sci* 2012; 122: 143–159.
- 246. Chalaris A, Garbers C, Rabe B, Rose-John S, Scheller J. The soluble Interleukin 6 receptor: Generation and role in inflammation and cancer. *Eur J Cell Biol* 2011; 90: 484–494.
- 247. Takizawa H, Ohtoshi T, Yamashita N, Oka T, Ito K. Interleukin 6-receptor expression on human bronchial epithelial cells: regulation by IL-1 and IL-6. *Am J Physiol* 1996; 270: L346–L352.
- 248. Gomez MI, Sokol SH, Muir AB, Soong G, Bastien J, Prince AS. Bacterial Induction of TNF-α Converting Enzyme Expression and IL-6 Receptor Shedding Regulates Airway Inflammatory Signaling. J Immunol 2005; 175: 1930–1936.
- 249. Klouche M, Bhakdi S, Hemmes M, Rose-John S. Novel path to activation of vascular smooth muscle cells: up-regulation of gp130 creates an autocrine activation loop by IL-6 and its soluble receptor. *J Immunol* 1999; 163: 4583–4589.
- 250. Romano M, Sironi M, Toniatti C, Polentarutti N, Fruscella P, Ghezzi P, Faggioni R, Luini W, van Hinsbergh V, Sozzani S, Bussolino F, Poli V, Ciliberto G, Mantovani A. Role of IL-6 and its soluble receptor in induction of chemokines and leukocyte recruitment. *Immunity* 1997; 6: 315–325.
- 251. Sun W, Liu DB, Li WW, Zhang LL, Long GX, Wang JF, Mei Q, Hu GQ. Interleukin-6 promotes the migration and invasion of nasopharyngeal carcinoma cell lines and upregulates the expression of MMP-2 and MMP-9. *Int J Oncol* 2014; 44: 1551–1560.
- 252. Edmiston JS, Archer KJ, Scian MJ, Joyce AR, Zedler BK, Murrelle EL. Gene expression profiling of peripheral blood leukocytes identifies potential novel biomarkers of chronic obstructive pulmonary disease in current and former smokers. *Biomarkers* 2010; 15: 715–730.
- 253. Wang Y, Hu H, Wu J, Zhao X, Zhen Y, Wang S, Li W, Liang M, Bin Wu, Ma G. The IL6R gene polymorphisms are associated with sIL-6R, IgE and lung function in Chinese patients with asthma. *Gene* 2016; 585: 51–57.
- 254. Pérez-Rubio G, Silva-Zolezzi I, Fernández-López JC, Camarena Á, Velázquez-Uncal M, Morales-Mandujano F, Hernández-Zenteno RDJ, Flores-Trujillo F, Sánchez-Romero C, Velázquez-Montero A, Espinosa de los Monteros C, Sansores RH, Ramírez-Venegas A, Falfán-Valencia R. Genetic Variants in IL6Rand ADAM19are Associated with COPD Severity in a Mexican Mestizo

Population. COPD 2016; 13: 610–615.

- 255. Le TT, Karmouty-Quintana H, Melicoff E, Weng T, Chen NY, Pedroza M, Zhou Y, Davies J, Philip K, Molina J, Luo F, George AT, Garcia-Morales LJ, Bunge RR, Bruckner BA, Loebe M, Seethamraju H, Agarwal SK, Blackburn MR. Blockade of IL-6 Trans Signaling Attenuates Pulmonary Fibrosis. J Immunol 2014; 193: 3755–3768.
- 256. Doganci A, Eigenbrod T, Krug N, De Sanctis GT, Hausding M, Erpenbeck VJ, Haddad E-B, Schmitt E, Bopp T, Kallen K-J, Herz U, Schmitt S, Luft C, Hecht O, Hohlfeld JM, Ito H, Nishimoto N, Yoshizaki K, Kishimoto T, Rose-John S, Renz H, Neurath MF, Galle PR, Finotto S. The IL-6R α chain controls lung CD4+CD25+ Treg development and function during allergic airway inflammation in vivo. J Clin Invest 2005; 115: 313–325.
- 257. McGreal EP, Davies PL, Powell W, Rose-John S, Spiller OB, Doull I, Jones SA, Kotecha S. Inactivation of IL-6 and soluble IL-6 receptor by neutrophil derived serine proteases in cystic fibrosis. *Biochim Biophys Acta* 2010; 1802: 649–658.
- 258. Taga T, Hibi M, Hirata Y, Yamasaki K, Yasukawa K, Matsuda T, Hirano T, Kishimoto T. Interleukin-6 triggers the association of its receptor with a possible signal transducer, gp130. *Cell* 1989; 58: 573–581.
- 259. Peters M, Jacobs S, Ehlers M, Vollmer P, Müllberg J, Wolf E, Brem G, Meyer zum Büschenfelde KH, Rose-John S. The Function of The Soluble Interleukin 6 (IL-6) Receptor in vivo: Sensitization of human soluble IL-6 Receptor Transgenic Mice Towards IL-6 and prolongation of the Plasma Half-life of IL-6. J Exp Med. 1996; 183: 1399–1406.
- 260. van Dam M, Müllberg J, Schooltink H, Stoyan T, Brakenhoff JP, Graeve L, Heinrich PC, Rose-John S. Structure-function analysis of interleukin-6 utilizing human/murine chimeric molecules. Involvement of two separate domains in receptor binding. *J Biol Chem* 1993; 268: 15285–15290.
- 261. Garbers C, Jänner N, Chalaris A, Moss ML, Floss DM, Meyer D, Koch-Nolte F, Rose-John S, Scheller J. Species Specificity of ADAM10 and ADAM17 Proteins in Interleukin-6 (IL-6) Trans-signaling and Novel Role of ADAM10 in Inducible IL-6 Receptor Shedding. *J Biol Chem* 2011; 286: 14804–14811.
- 262. Scheller J, Chalaris A, Schmidt-Arras D, Rose-John S. The pro- and antiinflammatory properties of the cytokine interleukin-6. *Biochim Biophys Acta* 2011; 1813: 878–888.
- 263. Jostock T, Müllberg J, Özbek S, Atreya R, Blinn G, Voltz N, Fischer M, Neurath MF, Rose-John S. Soluble gp130 is the natural inhibitor of soluble interleukin-6 receptor transsignaling responses. *Eur J Biochem* 2001; 268: 160–167.
- 264. Díaz-Rodríguez E, Montero JC, Esparís-Ogando A, Yuste L, Pandiella A. Extracellular Signal-regulated Kinase Phosphorylates Tumor Necrosis Factor alpha -converting Enzyme at Threonine 735: A Potential Role in Regulated Shedding. *Mol Biol Cell* American Society for Cell Biology; 2002; 13: 2031– 2044.

- 265.Reddy P, Slack JL, Davis R, Cerretti DC, Kozlosky CJ, Blanton RA, Shows D,
Peschon JJ, RA B. Functional analysis of the domain structure of tumor necrosis
factor-α converting enzyme. J Biol Chem 2000; 275: 14608–14614.
- 266. Li C, Hao M, Cao Z, Ding W, Graves-Deal R, Hu J, Piston DW, Coffey RJ. Naked2 Acts as a Cargo Recognition and Targeting Protein to Ensure Proper Delivery and Fusion of TGF. *Mol Biol Cell* 2007; 18: 3081–3093.
- 267. Soond SM, Everson B, Riches DWH, Murphy G. ERK-mediated phosphorylation of Thr735 in TNFα-converting enzyme and its potential role in TACE protein trafficking. *J Cell Sci* 2005; 118: 2371–2380.
- Zhao J, Chen H, Peschon JJ, Shi W, Zhang Y, Frank SJ, Warburton D. Pulmonary Hypoplasia in Mice Lacking Tumor Necrosis Factor-α Converting Enzyme Indicates an Indispensable Role for Cell Surface Protein Shedding during Embryonic Lung Branching Morphogenesis. *Dev Biol* 2001; 232: 204–218.
- 269. Arndt PG, Strahan B, Wang Y, Long C, Horiuchi K, Walcheck B. Leukocyte ADAM17 Regulates Acute Pulmonary Inflammation. *PLoS ONE* 2011; 6: e19938.
- 270. Park JA, Sharif AS, Shiomi T, Kobzik L, Kasahara DI, Tschumperlin DJ, Voynow J, Drazen JM. Human neutrophil elastase-mediated goblet cell metaplasia is attenuated in TACE-deficient mice. *Am J Physiol Lung Cell Mol Physiol* 2013; 304: L701–L707.
- 271. Weskamp G, Mendelson K, Swendeman S, Le Gall S, Ma Y, Lyman S, Hinoki A, Eguchi S, Guaiquil V, Horiuchi K, Blobel CP. Pathological neovascularization is reduced by inactivation of ADAM17 in endothelial cells but not in pericytes. *Circ Res* 2010; 106: 932–940.
- 272. Perl A, Tichelaar JW, Whitsett JA. Conditional gene expression in the respiratory epithelium of the mouse. *Transgenic Res* 2002; 11: 21–29.
- 273. Isokane M, Hieda M, Hirakawa S, Shudou M, Nakashiro K, Hashimoto K, Hamakawa H, Higashiyama S. Plasma-membrane-anchored growth factor proamphiregulin binds A-type lamin and regulates global transcription. *J Cell Sci* 2008; 121: 3608–3618.
- 274. Knight D, Mutsaers SE, Prêle CM. STAT3 in tissue fibrosis: is there a role in the lung? *Pulm Pharmacol Ther* 2011; 24: 193–198.

uman disease CF lung COPD airway IL6R inflammation AREC prosis PBEC human disease CF lung COPD airway ILGR infla CHAPTER 2 EGER she S So I FR G remodeling a sinflammation hui a iuman ing COPD airway ILER inflammation AREG remodeling EGFR shedding ADAM17 Torosis brosis PBEC hu ADAM17 and EGFR regulate IL-6 receptor and amphiregulin and release in cigarette smoke-exposed of primary bronchial epithelial cells from patients with chronic obstructive pulmonary disease (COPD). (published - Physiol Rep) remodeling EGFR shedding ADAM17 fibrosis PBEC sease CF lung COPD airway IL6R inflammation AREG remodeling EGFR shedding ADAM17 fibrosis prosis PBEC human disease CF lung COPD airway Stolarczyk M, Amatngalim GD, EGER sh DAM17 fibrosis PBEC human disease Yu X, Veltman M, Hiemstra PS, Scholte BJ odeling

Physiological Reports

ORIGINAL RESEARCH

ADAM17 and EGFR regulate IL-6 receptor and amphiregulin mRNA expression and release in cigarette smoke-exposed primary bronchial epithelial cells from patients with chronic obstructive pulmonary disease (COPD)

Marta Stolarczyk^{1,*}, Gimano D. Amatngalim^{2,*}, Xiao Yu¹, Mieke Veltman¹, Pieter S. Hiemstra² & Bob J. Scholte¹

1 Cell Biology, Erasmus MC, Rotterdam, The Netherlands

2 Pulmonology, Leiden University Medical Center (LUMC), Leiden, The Netherlands

Keywords

A disintegrin and metalloprotease 17 (ADAM17), amphiregulin (AREG), Chronic Obstructive Pulmonary Disease (COPD), epidermal growth factor receptor (EGFR), IL6 receptor (IL6R), TACE.

Correspondence

Bob J Scholte, ErasmusMC, Cell Biology Ee1034, PO box 2040 3000 CA Rotterdam, The Netherlands. Tel: +31 10 7043205 E-mail: b.scholte@erasmusmc.nl

Funding Information

This study (MS) was supported by grants from the Lung Foundation Netherlands (#3.3.10.027). the Dutch Cystic fibrosis Foundation NCFS (MV HIT-CF7) and Galapagos NV, The Netherlands.

Received: 13 June 2016; Revised: 8 July 2016; Accepted: 9 July 2016

doi: 10.14814/phy2.12878

Physiol Rep, 4 (16), 2016, e12878, doi: 10.14814/phy2.12878

*Equal contribution.

Abstract

Aberrant activity of a disintegrin and metalloprotease 17 (ADAM17), also known as TACE, and epidermal growth factor receptor (EGFR) has been suggested to contribute to chronic obstructive pulmonary disease (COPD) development and progression. The aim of this study was to investigate the role of these proteins in activation of primary bronchial epithelial cells differentiated at the air-liquid interface (ALI-PBEC) by whole cigarette smoke (CS), comparing cells from COPD patients with non-COPD. CS exposure of ALI-PBEC enhanced ADAM17-mediated shedding of the IL-6 receptor (IL6R) and the EGFR agonist amphiregulin (AREG) toward the basolateral compartment, which was more pronounced in cells from COPD patients than in non-COPD controls. CS transiently increased IL6R and AREG mRNA in ALI-PBEC to a similar extent in cultures from both groups, suggesting that posttranslational events determine differential shedding between COPD and non-COPD cultures. We show for the first time by in situ proximity ligation (PLA) that CS strongly enhances interactions of phosphorylated ADAM17 with AREG and IL-6R in an intracellular compartment, suggesting that CS-induced intracellular trafficking events precede shedding to the extracellular compartment. Both EGFR and ADAM17 activity contribute to CS-induced IL-6R and AREG protein shedding and to mRNA expression, as demonstrated using selective inhibitors (AG1478 and TMI-2). Our data are consistent with an autocrinepositive feedback mechanism in which CS triggers shedding of EGFR agonists evoking EGFR activation, in ADAM17-dependent manner, and subsequently transduce paracrine signaling toward myeloid cells and connective tissue. Reducing ADAM17 and EGFR activity could therefore be a therapeutic approach for the tissue remodeling and inflammation observed in COPD.

Introduction

Chronic obstructive pulmonary disease (COPD) is a progressive lung disorder characterized by irreversible airflow obstruction due to airway inflammation, infection, and tissue remodeling (Vestbo et al. 2013). Airway epithelial cells play a central role in the pathogenesis of COPD through a variety of mechanisms, including production of inflammatory mediators, antimicrobial peptides, and growth factors (Hiemstra et al. 2015). Exposure of the lung tissue to triggers like cytotoxic particles and gasses, including cigarette smoke, microbes, and innate immune

Page 1

2016 | Vol. 4 | Iss. 16 | e12878

^{© 2016} The Authors. *Physiological Reports* published by Wiley Periodicals, Inc. on behalf of the American Physiological Society and The Physiological Society.

This is an open access article under the terms of the Creative Commons Attribution License.

which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

mediators (Koff et al. 2008; Kim et al. 2011) (Lemjabbar-Alaoui et al. 2011) (Shao 2004) (Zhou et al. 2012) activate various matrix metalloproteinases (MMPs) and a disintegrin and metalloproteinases (ADAMs) expressed by airway epithelial cells (Dijkstra et al. 2009; Paulissen et al. 2009). The activity of MMPs and ADAMs contributes not only to proteolytic degradation of lung tissue, but also to regulation and processing of numerous receptor activating proteins (Richter et al. 2002; Gomez et al. 2005; Bell et al. 2007; Baumgart et al. 2010). Through these various activities, MMPs and ADAMs are implicated in a broad spectrum of processes ranging from inflammatory responses to airway epithelial repair. It has been proposed that their aberrant activity might lead to chronic inflammation and abnormal tissue remodeling in the lungs of COPD patients (Paulissen et al. 2009).

One of the ADAMs, a ubiquitously expressed Zn²⁺dependent disintegrin and metalloprotease 17 (ADAM17), formerly known as TNFa converting enzyme (TACE), is recognized as an important regulator of pulmonary inflammation, cell proliferation, and epithelial barrier function (Gooz 2010; Lemjabbar-Alaoui et al. 2011). In bronchial epithelial cells, ADAM17 modulates these processes by cleaving membrane-bound cytokines (TNFa), several EGF receptor (EGFR) agonists (TGF-a, amphiregulin, epiregulin, HB-EGF), cytokine receptors (IL6R, TNF-R), growth factor receptors (NOTCH receptors), and adhesion proteins (L-selectin, ICAM-1, E-cadherin) (Gomez et al. 2005; Bell et al. 2007; Baumgart et al. 2010; Gooz et al. 2012). Moreover, ADAM17 phosphorylation and activity is enhanced in airway epithelial cell lines and in undifferentiated primary cells upon exposure to cigarette smoke extract (Lemjabbar et al. 2003; Shao 2004; Lemjabbar-Alaoui et al. 2011).

Our studies focus on two ADAM17 substrates implicated in COPD pathogenesis: the IL-6 cytokine receptor (IL6R) and the growth factor amphiregulin (AREG), one of the EGFR agonists produced by bronchial epithelial cells (Richter et al. 2002). Elevated levels of IL6R have been observed in peripheral blood leukocytes of COPD patients (Edmiston et al. 2010), and recently genetic variants of IL6R have been linked with COPD severity (Pérez-Rubio et al. 2016). However, the regulation of shedding of IL6R and AREG from COPD airway epithelium has not been studied. Upon shedding from epithelial cells, IL6R and AREG activate the shared interleukin receptor gp130 and EGFR, respectively, on epithelial cells (autocrine), as well as on underlying myofibroblasts and myeloid cells (paracrine) (Burgel and Nadel 2008; Nechemia-Arbely et al. 2008; Kasina et al. 2009; Rose-John 2012). Both IL6/IL6R/ gp130 and AREG/EGFR/ERK pathways are involved in the resolution of lung inflammation and repair of injury, but also in progression of subepithelial fibrosis and collagen deposition (Zhou et al. 2012). These signaling pathways involve the JAK kinase and/or MAP kinase pathway, which are druggable targets in COPD pathology (Barnes 2013). Excessive ligand-mediated EGFR activation results in epithelial hyperproliferation and increased production of the inducible mucin MUC5AC, processes observed in smokers with or without COPD (Lemjabbar et al. 2003; Shao 2004; Deshmukh et al. 2005; Kasina et al. 2009; Lemjabbar-Alaoui et al. 2011; Li et al. 2011; Zhang et al. 2014). Moreover, EGFR activation results in subsequent transcriptional regulation of inflammatory mediators such as IL-8 (Richter et al. 2002), a chemokine that has been implicated in COPD development.

So far, studies on the effect of cigarette smoke on epithelial ADAMs activity has largely relied on the use of airway epithelial cell lines or undifferentiated primary cell cultures, stimulated with an aqueous extract of cigarette smoke. However, whole cigarette smoke exposure and primary differentiated airway cell cultures represent more relevant physiological conditions. Firstly, fresh whole cigarette smoke (CS) contains unstable active components and particulate matter that are largely absent from extracts. Furthermore, immortalized epithelial cells are poor models of bronchial epithelium in situ, since they are frequently karyotypically unstable and heterogeneous, do not show characteristic features of differentiation and inherently carry multiple mutations in pathways essential for growth, differentiation, cell-cell interaction, and polarization. Furthermore, submerged cultures of primary airway epithelial cells fail to differentiate. Finally, using cell lines does not allow a comparison of patient populations. Therefore, we examined the effect of whole CS exposure on shedding of the soluble interleukin-6 receptor (sIL6R) and the EGFR-ligand amphiregulin (AREG) by well-differentiated, air-liquid interface cultured human primary bronchial epithelial cells (ALI-PBEC).

This allowed us to compare CS-induced ADAM17mediated protein shedding and mRNA expression of sIL6R and AREG in well-differentiated ALI-PBEC from COPD and non-COPD (ex)smokers. Moreover, we established in this model the involvement of both EGFR and ADAM17 not only in shedding of ADAM17 substrates, but also in the regulation of mRNA levels of ADAM17 substrates and IL-8. Finally, for the first time, we observed intracellular CS-induced phosphorylated ADAM17-substrate interaction via an in situ proximity ligation assay. Overall, our results provide novel insights into the activation of airway epithelial cells by cigarette smoke in COPD, and highlight a possible role of ADAMs and EGFR in COPD pathology.

2016 | Vol. 4 | Iss. 16 | e12878 Page 2

Materials and Methods

Air-liquid interface cell culture of human primary bronchial epithelial cells

Human airway epithelial cells were obtained from macroscopically normal, anonymous bronchial tissue obtained from lung cancer patients undergoing resection surgery for lung cancer at LUMC. This material was be used for research according to the "Code of Conduct for Responsible Use" (FEDERA code) based on the condition that the patient has no objection against such use. Primary bronchial epithelial cells (PBEC) were isolated from tumorfree lung resection material (Wetering et al. 2000), and passage 2 expanded cells were cultured at the air-liquid interface (ALI) to achieve mucociliary differentiation as previously described (Amatngalim et al. 2015). Briefly, 40,000 cells were seeded on 0.65 cm Transwell inserts (Corning Costar, Cambridge, MA) with a 0.4 µm pore size, which were coated with 30 µg/mL PureCol (Advanced BioMatrix, San Diego, CA), 10 µg/mL Bovine serum albumin (Sigma-Aldrich, St. Louis, MO), and 10 µg/mL Fibronectin (isolated from plasma). Cells were cultured in Bronchial epithelial growth medium (BEGM) (Lonza, Verviers, Belgium) and Dulbecco's modified Eagle's medium (DMEM) (Gibco, Bleiswijk, The Netherlands) (1:1 mixture) containing 1 mmol/L Hepes (Lonza) and supplemented with SingleQuot supplements and growth factors according to the manufacturer's instructions (bovine pituitary extract, hydrocortisone, human epidermal growth factor, epinephrine, transferrin, insulin, T3 and retinoic acid; all from Lonza), additional 15 ng/ mL retinoic acid (Sigma-Aldrich), 1 mg/mL BSA (Sigma-Aldrich), 100 U/mL penicillin, and 100 µg/mL streptomycin (Lonza). PBEC were initially cultured on inserts in submerged conditions until cell layers were confluent. Next, apical medium was removed and cells were cultured at air-exposed conditions for at least 2 weeks to allow mucociliary differentiation. Clinical history and lung function data were obtained from anonymized patients (Table 1), and COPD disease status was based on lung function data according to the Global Initiative for Chronic Obstructive Lung Disease (GOLD) classification (Vestbo et al. 2013). Donor cells were randomly allocated to experimental groups.

Cigarette smoke exposure

Air-liquid interface cultured human primary bronchial epithelial cells were exposed to whole cigarette smoke (CS) in an exposure model, adapted from (Beisswenger et al. 2004) and previously described in more detail (Amatngalim et al. 2015). In this model, ALI-PBEC cultures were

Table 1. COPD and non-COPD	patient characteristics.
---------------------------------	--------------------------

	COPD	non-COPD	P-value
Number of donors	15	11	
Gender (females/males)	4/11	2/9	
Age, years	70 ± 8	66 ± 6	0.1667
FEV ₁ , % predicted	65 ± 16	81 ± 16	< 0.01
FEV ₁ /FVC %	55 ± 9	79 ± 9	< 0.0001

Characteristics of PBEC donors. Age in years, and lung function as FEV_1 (% predicted) and FEV_1/FVC are shown as means \pm SD. The mean differences were compared using the nonparametric Mann–Whitney test. COPD, chronic obstructive pulmonary disease, FEV1, Forced expiratory volume in one-second, FVC, forced vital capacity.

placed in either a CS- or air (negative control) exposure chamber located in a tissue incubator at 37° C and 5%CO₂. Smoke derived from one cigarette (3R4F reference cigarettes [University of Kentucky, Lexington, KY]) was infused into the exposure chamber by a mechanical pump with a constant flow of 1 L/min, and equally distributed by a ventilator inside the chamber. After infusion (approximately 4–5 min), residual smoke was removed by infusion of air from the tissue incubator for 10 min. Directly after CS exposure, the basal medium of the cell cultures was refreshed and cells were incubated for the indicated periods of time. Untreated cells used as controls were subjected to the same procedure omitting the smoke (AIR).

Inhibitors

TMI-2 (1 μ mol/L; PF-5480090), a highly selective inhibitor of ADAM17 activity (Zhang et al. 2004), was obtained from Wyeth inc. (Philadelphia, Pennsylvania) and the selective EGFR inhibitor AG1478 (1 μ mol/L) was from Sigma Aldrich. Cells were preincubated for 1 h with inhibitors before CS exposure, and directly after CS exposure media were replaced and inhibitors were freshly added.

ELISA

ALI-PBEC conditioned culture media were collected from the basolateral side of the inserts (1 mL), and apical washes were obtained by washing the apical surface with 100 μ L PBS at different time points, dependent on the experiment. Collected samples were diluted 1:1 with BEGM media and analyzed for IL6R and AREG by human IL6R or AREG ELISA kit, R&D. Further steps were performed according to the manufacturer's protocol. Data were corrected for the dilution factor and insert size, and the amount of the shed IL6R and AREG was expressed as pg/mL per cm².

[@] 2016 The Authors. *Physiological Reports* published by Wiley Periodicals, Inc. on behalf of the American Physiological Society and The Physiological Society.

RNA isolation and quantitative real-time PCR

RNA was isolated using the miRNeasy Mini Kit (Qiagen) according to the manufacturer's instructions, and cDNA was synthesized by reverse-transcription PCR using oligo (dT) primers (Qiagen) and Moloney murine leukemia virus (M-MLV) polymerase (Promega, Leiden, The Netherlands). mRNA expression was determined by quantitative real-time PCR as described previously (Amatngalim et al. 2015) with primer pairs presented in Table 2. mRNA expression was quantified using the standard curve method (Larionov et al. 2005), in which arbitrary expression levels were normalized to the housekeeping genes RPL13A and ATP5B. The housekeeping genes were selected based on stable expression using the "Genorm method" (Vandesompele et al. 2002).

Proximity ligation assay

Chronic obstructive pulmonary disease ALI-PBEC werefixed with 4% paraformaldehyde and permeabilized with 0.5% Triton-X100 in PBS twice for 15 min, blocked with 1% BSA and 0.15% glycine. Next, they were incubated overnight in 4°C with two different first antibodies simultaneously: against ADAM17 (rabbit polyclonal, C-terminal ADAM17, 25 μg/mL, ab78162, Abcam) or ADAM17P^{T735} (rabbit polyclonal, phospho-ADAM17 in position T735, 25 µg/mL, ab60996, Abcam) and IL6R (goat anti-human IL6R recognizing extracellular domain, 20 µg/mL, AF-227-NA, R&D) or AREG (polyclonal goat anti-human Areg, 25 µg/mL, AF-262, R&D). All washing steps were repeated three times with 0.5% Triton-X100 in PBS (Sigma-Aldrich). Further steps were performed according to the DuoLink manufacturer's protocol. Briefly, Proximity ligation assay (PLA) probes for anti-goat PLUS and anti-rabbit MINUS were incubated at 37°C for 1 h. Ligation and amplification steps were performed with Detection Red Reagent. Finally, inserts with ALI-PBEC were cut out and mounted on the slides with DAPI (Vectashield mounting medium for fluorescence with DAPI, H-1200). *z*-stacks were acquired using confocal microscopy (Leica604).

PLA image analysis

The number of dots was counted in the whole *z*-stack with Image J software. The threshold values were adjusted with the *Intermodes* algorithm (the filter size set between 10 and 437 microns to exclude the small and large dots, which were in the range of 10% of the total dot count). The objects on the edges of the culture inserts were excluded from the analysis. The number of nuclei was counted in each *z*-stack by hand to express the number of dots per nucleus.

Statistical analysis

Data were analyzed with GraphPad Prism Software, using the appropriate statistical test, as indicated underneath each figure. Cells from various donors (number indicated by n in the legends of figures) were used for each experiment, samples were collected from two or three wells from a single donor, averaged and represented as a single dot in the figure. Statistical analysis was performed on the averaged data. Values are presented as mean with SEM values. Differences at *P*-values <0.05 were considered to be statistically significant. ns > 0.05, *P < 0.05, **P < 0.01, **P < 0.001, ****P < 0.0001.

Results

CS induces shedding of IL6R and AREG by ALI-PBEC into basolateral medium, but not apical

We first examined the effect of cigarette smoke (CS) exposure on the release of sIL6R and AREG by ALI-PBEC

Table 2. Quantitative-PCR primers.				
Primer name		[ref]		
IL8	FW 5'-CAGCCTTCCTGATTTCTG-3'	Amatngalim et al. (2015)		
	RV 5'-CACTTCTCCACAACCCTCTGC-3'			
AREG	FW 5'GTGGTGCTGTCGCTCTTGATA 3'	Clarke et al. (2013)		
	REV 5'ACTCACAGGGGAAATCTCACT3'			
IL6R full-length	FW 5'GCTGTGCTCTTGGTGAGGAAGTTT3'	Rath et al. (2010)		
form (full-IL6R)	REV 5'CTGAGCTCAAACCGTAGTCTGTAGAAA3'			
IL6R alternatively spliced	FW 5'GCGACAAGCCTCCCAGGTT3'	Rath et al. (2010)		
variant (spliced IL6R)	REV 5'CCGCAGCTTCCACGTCTTCTT3'			

FW, Forward primer; REV, Reversed primer.

2016 | Vol. 4 | Iss. 16 | e12878 Page 4

at the apical surface and in the basal medium, which contains a maintenance level of EGF, associated with a basal level of EGFR activity. This was done, using a previously described whole CS exposure model (Amatngalim et al. 2015), in which CS caused a transient disruption in the airway epithelial barrier integrity, accompanied by minor cytotoxic effects measured at the apical surface. Both sIL6R and AREG were barely detectible in the apical washes collected from ALI-PBEC of 17 COPD donors at different stages of disease, following exposure to either CS or air (Fig. 1A and B). In contrast, sIL6R and AREG were markedly released into the basal medium in both conditions. CS significantly increased release of sIL6R into the basal medium at 12 h postexposure, while AREG levels were increased at 12 and 24 h after CS exposure. These results demonstrate that shedding of sIL6R and AREG by ALI-PBEC occurs mainly to the basolateral compartment, and is enhanced by CS exposure.

CS significantly induces shedding of IL6R and AREG in COPD ALI-PBEC but not in non-COPD ALI-PBEC

Next, we explored whether shedding of sIL6R and AREG differs between ALI-PBEC isolated from COPD patients and non-COPD (ex)-smokers upon CS and air exposure. Based on the previous result, the release was only determined in the basal medium 24 h after exposure. Shedding of sIL6R and AREG did not differ between COPD and non-COPD ALI-PBEC exposed to air (Fig. 2A and B), indicating no differences at baseline conditions. In

contrast, shedding of sIL6R (Fig. 2A) and AREG (Fig. 2B) was significantly higher after CS exposure only in COPD ALI-PBEC, and not in non-COPD ALI-PBEC. These data show that CS-induced release of sIL6R and AREG was more pronounced in airway epithelial cells from COPD in comparison to non-COPD donors.

CS-induced IL6R and AREG mRNA expression is lower in COPD ALI-PBEC compared to non-COPD cultures

We further determined mRNA expression of IL6R and AREG in CS and air exposed ALI-PBEC cultures from COPD and non-COPD patients. The soluble form of IL6R can be generated either by shedding of the membrane anchored form or by de novo synthesis of the alternatively spliced isoform that differs at the C-terminus (Rose-John 2012). Therefore, we determined mRNA expression levels of both IL6R variants: the membrane-anchored (full-IL6R mRNA) and the alternatively spliced (spliced-IL6R mRNA) variant.

Time-course analysis revealed that CS increased full-IL6R mRNA 3 h after exposure, but not at later time points (Fig. 3A). In contrast, baseline expression of spliced-IL6R mRNA did not differ from the expression after CS treatment (Fig. 3B), suggesting that the increase in sIL6R protein levels in culture supernatants did not result from alternative splicing. Similar to full-IL6R mRNA, CS significantly induced AREG mRNA expression 3 h after exposure, but not at later time points (Fig. 3C). These findings suggest that the CS-induced increase in



Figure 1. Cigarette smoke induces shedding of ILGR and AREG by ALI-PBEC into basolateral medium. ILGR (A) and AREG (B) were mainly shed to the basolateral compartment in ALI-PBEC (COPD donors at different stages, n = 12 for ILGR and n = 5 for AREG). Basolateral media (basolateral) were collected and apical PBS washes (apical) were performed 12 h and 24 h after CS or air exposure. Both ILGR and AREG were readily detectable in the basolateral compartment, and barely present in the apical washes. The response of cells from each donor was analyzed within one experiment using duplicate or triplicate inserts. Statistical analysis was performed by two-way ANOVA (Bonferroni) on the averaged data from each donor, comparing apical versus basolateral shedding at air and CS exposure, and basolateral shedding at air versus CS exposure.

© 2016 The Authors. *Physiological Reports* published by Wiley Periodicals, Inc. on behalf of the American Physiological Society and The Physiological Society.



Figure 2. Cigarette smoke significantly induces shedding of IL6R and AREG into basolateral medium by COPD ALI-PBEC. Soluble forms of IL6R (A) and AREG (B) shed into the basolateral compartment were detected 24 h after CS or air exposure in ALI-PBEC derived from non-COPD and COPD donors (Table 1). (A) IL6R levels were significantly increased 24 h after CS treatment in COPD-ALI-PBEC (n = 15), but this increase was not significant in the non-COPD group (n = 11 donors). (B) Similarly, CS exposure significantly increased AREG levels in ALI-PBEC cells derived from COPD donors (n = 10), but not in non-COPD ALI-PBEC (n = 8 donors). Statistical analysis: paired *t*-test. *n* refers to the number of donors, duplicate or triplicate data were averaged per donor.

IL6R and AREG shedding is mediated at least in part *via* regulation of their mRNA expression levels.

Baseline expression of full-IL6R and AREG mRNA did not differ between COPD and non-COPD ALI-PBEC (Fig. 3D and F). After CS exposure, full-IL6R and AREG mRNA were expressed at higher levels in both non-COPD and COPD ALI-PBEC. Interestingly, after CS induction, COPD cells expressed full-ILR and AREG at lower levels on average but this did not reach statistical significance (Fig. 3D and F). Spliced-IL6R mRNA expression did not differ between investigated groups either after CS or air exposure (Fig. 3E). These findings suggest that COPD patients may have impaired transcriptional or posttranscriptional responses to inflammatory and tissue regenerative triggers. The apparent contrast with the more pronounced shedding from COPD cells after CS challenge (Fig. 2) suggests that posttranslational mechanisms determine shedding rate, rather than substrate mRNA levels.

ADAM17 is required for CS-induced release of IL6R and AREG in ALI-PBEC

To confirm the previously established involvement of ADAM17 in the shedding process of IL6R and AREG in our model, we used the selective ADAM17 inhibitor TMI-2 (Wyeth) (Zhang et al. 2004). TMI-2 only partially decreased baseline IL6R release at all investigated time points (Fig. 4A), plausibly because release of the product of spliced-IL6R mRNA, which cannot be distinguished from shed IL-6R with the available antibodies, is not sensitive to inhibitors of ADAMs (Vermes et al. 2012). In contrast, TMI-2 significantly decreased baseline AREG

shedding at all time points (Fig. 4B). Importantly, CSinduced shedding of IL6R and AREG was significantly inhibited by TMI-2 at all time points after CS exposure, indicating that ADAM17 activity is involved in CSinduced ADAM17 substrate release (Fig. 4).

ADAM17- and ADAM17P-substrate interactions are increased after CS exposure in an intracellular compartment of ALI-PBEC

Next, we explored the interactions of IL6R or AREG with ADAM17 3 h after CS treatment in ALI-PBEC with an in situ proximity ligation assay (PLA) (Fredriksson et al. 2002), using antibodies against ADAM17 phosphorylated at Thr735 (ADAM17-P^{T735}) or total ADAM17. Protein IL6R/AREG-ADAM17 and IL6R/AREG-ADAM17-PT735 interactions were visualized as fluorescent red dots in x-y confocal sections (representative confocal pictures shown in Fig. 5A and B). In air-exposed cells, PLA signals were largely confined to the basal region, as in the control incubations, and not significantly higher than background (data not shown), as indicated by red lines in Figure 5C-F (relevant control data are shown in Figure S1). Interestingly, CS exposure significantly increased the total number of PLA signals for interactions of IL6R or AREG with ADAM17 (Fig. 5C and E). We observed that CS strongly enhanced interactions of IL6R or AREG with ADAM17-P^{T735} (Fig. 5D and F), which further extends previous findings showing that ADAM17 is phosphorylated after smoke extract (CSE) exposure in submerged immortalized NCI-H292 cells (Lemjabbar-Alaoui et al. 2011). CSinduced PLA signals of substrate-ADAM17 and substrate-ADAM17-P^{T735} were primarily detected in the apical

2016 | Vol. 4 | Iss. 16 | e12878 Page 6



Figure 3. CS exposure transiently enhances IL6R and AREG mRNA expression in COPD and non-COPD ALI-PBEC. mRNA levels of the IL6R fulllength variant (full-IL6R) (A), the IL6R splice variant (spliced-IL6R) (B) and AREG (C) were determined by qPCR 3, 12, and 24 h after CS (black bars) or air exposure (open bars) (n = 14 unspecified donors). A transient induction of full-IL6R (A) and AREG (C), but not spliced-IL6R (B) was observed at 3 h after CS exposure. In COPD (n = 7) and non-COPD (n = 8) ALI-PBEC, mRNA of full-IL6R (D), spliced-IL6R (E), and AREG (F) were determined 3 h after CS exposure. mRNA expression of full-IL6R and AREG, was lower on average but not statistically significant in COPD compared to non-COPD donors. Data were normalized for expression against two reference genes (ATP5B and RPL13A). n refers to the number of donors. The response of cells from each donor was analyzed within one experiment using duplicate inserts and data were averaged per donor. Statistical analysis was performed on the averaged data from each donor. Statistical analysis: Two-way ANOVA With Tukey's multiple comparison test.



Figure 4. ADAM17 is involved in the release of soluble IL6R and AREG in ALI-PBEC. The selective ADAM17 inhibitor, TMI-2 (Zhang et al. 2004) decreases basal and CS-induced IL6R (A,B) and AREG (C,D) shedding in ALI-PBEC cells (n = 3 COPD donors) at 3, 6 (A,C), 12, and 24 h time points (B,D). n refers to the number of donors. The response of cells from each donor was analyzed within one experiment using duplicate or triplicate inserts and data were averaged per donor. Statistical analysis was performed on the averaged data from each donor, by two-way ANOVA (Bonferroni), confirming first the effect of CS on IL6R and AREG shedding at different time points, and second the effect of TMI-2 on shedding during air and CS exposure.

region of the cells and were not confined to a lateral membrane pattern suggesting an intracellular vesicular localization of protein complexes in ALI-PBEC. These data for the first time demonstrate that CS exposure strongly increases the interaction of ADAM17 and ADAM17-P^{T735} with IL6R or AREG in an intracellular vesicular compartment of ALI-PBEC, suggesting a CS induced effect on protein trafficking.

© 2016 The Authors. *Physiological Reports* published by Wiley Periodicals, Inc. on behalf of the American Physiological Society and The Physiological Society.



Figure 5. CS increases substrate-ADAM17 and substrate-ADAM17-P^{T735} interactions in pseudostratified COPD ALI-PBEC cells. The proximity ligation assay (PLA) signal in COPD- ALI-PBEC exposed to CS or air was performed for the following interactions: IL6R-ADAM17, IL6R-ADAM17P^{T735}, AREG-ADAM17, and AREG-ADAM17-PT⁷³⁵. Relevant control data are shown in Figure S1. Here, we show a representative figure of IL6R-ADAM17P^{T735} 3 h after air (A) and CS exposure (B). Left panels show merged signals of nuclei (blue) and PLA (red) in the *x*-*y* sections of the confocal *z*-stack and right panel presents PLA signal in the apical region (red dots). The number of PLA interactions was counted for all interactions as described in the methods section and expressed per nucleus 3 h after CS or air exposure in the whole *z*-stack of the ALI-PBEC (C–F). The red lines on the graphs indicate the maximal dot count in the PLA assay controls, in which one of the antibodies for the interaction was omitted (background staining not shown). For each interaction, cells from one donor were analyzed. Different filters (*n* = 4) were used to show distinct interactions. Statistical analysis: unpaired *t*-test.

EGFR is required for basal and induced AREG shedding in ALI-PBEC

ADAM17-dependent shedding of EGFR ligands such as AREG results in activation of EGFR through an autocrine feedback loop, which modulates basal EGFR activity (DeWitt et al. 2001). This mechanism was shown to be activated by CS extract in submerged cultured PBEC and in cell lines (Lemjabbar et al. 2003). In our experimental set-up, we have previously shown that CS enhances basal EGFR activity by increasing its phosphorylation (Amatngalim et al. 2015). To illustrate the involvement of EGFR in CS-induced ADAM17-related shedding in ALI-PBEC, we assessed sIL6R and AREG shedding after starvation for

2016 | Vol. 4 | Iss. 16 | e12878 Page 8

growth factors, using medium devoid of EGF and bovine pituitary extract (BPE). Removing these factors from the medium substantially reduced baseline shedding of IL6R and AREG (Fig. 6), when compared to standard culture conditions including EGF and BPE (Fig. 4A and C). Both sIL6R and AREG release were significantly increased at 3 h after CS exposure. The selective ADAM17 inhibitor TMI-2 and the EGFR tyrosine kinase inhibitor (AG1478) added prior to CS exposure, partially inhibited sIL6R shedding, consistent with a substantial contribution of the ADAMinsensitive splice variant sIL-6R levels in the basal medium (Fig. 6A). AG1478 strongly impaired AREG shedding, to a similar extent as TMI-2 (Fig. 6B). These findings together demonstrate a critical role of EGFR activation in ADAM17mediated basal and CS-induced shedding activity.

EGFR and ADAM17 are required for CSinduced IL6R and AREG mRNA expression

We previously observed that EGFR activation is involved in CS-induced expression of several genes in ALI-PBEC (Amatngalim et al. 2015). The molecular mechanism by



Figure 6. IL6R and AREG shedding depends on ADAM17 and EGFR activity in COPD-ALI-PBEC. COPD ALI-PBEC (n = 3 donors) were starved for growth factors for 48 h prior to CS or air exposure. Three hours after CS exposure, IL6R (A) and AREG (B) shedding were significantly increased compared to air. The ADAM17 inhibitor (1 µmol/L TMI-2) and the EGFR inhibitor (1 µmol/L AG1478) significantly reduced AREG, but IL6R shedding to a lesser extent. The response of cells from each donor was analyzed within one experiment using triplicate inserts and data were averaged per donor. Statistical analysis was performed on the averaged data from each donor by one way ANOVA (Tukey multiple comparison test), only relevant comparisons are shown, air versus CS-treated cells and the effect of inhibitors in CS-treated cells.

which CS activates EGFR are not known. Here, we explored the effect of ADAM17 and EGFR inhibition on CS-induced IL6R and AREG mRNA levels in ALI-PBEC. At 3 hours after CS exposure in the absence of EGF in the medium, both TMI-2 and AG1478 significantly impaired CS-induced expression of full-L6R mRNA (Fig. 7A), but not the splice variant (Fig. 7B). Both inhibitors strongly diminished CSinduced AREG mRNA levels (Fig. 7C) as well as IL-8



Figure 7. ILGR and AREG mRNA expression are regulated by ADAM17 and EGFR activity in ALI-PBEC. ALI-PBEC (n = 3 donors) were starved for growth factors for 48 h prior to CS exposure. At 3 h, CS-induced mRNA levels of full-length ILGR (A), AREG (C), and IL8 (D) were diminished upon ADAM17 (1 µmol/L TMI-2) and EGFR (1 µmol/L AG1478) inhibition, whereas that of the alternatively spliced form of ILGR was not affected (B). n refers to the number of donors. The response of cells from each donor was analyzed within one experiment using triplicate inserts and data were averaged per donor. Statistical analysis was performed on the averaged data from each donor, by one-way ANOVA (Tukey multiple comparison test), only relevant comparisons are shown, air versus CS-treated cells and the effect of inhibitors in CS-treated cells.

© 2016 The Authors. *Physiological Reports* published by Wiley Periodicals, Inc. on behalf of the American Physiological Society and The Physiological Society.

mRNA expression (Fig. 7D). Overall, these findings for the first time demonstrate that ADAM17, next to EGFR, is essential in the CS-induced mechanism regulating not only the mRNA of ADAM17 substrates (IL6R and AREG), but also IL-8 in ALI-PBEC.

Discussion

Many studies have demonstrated that airway epithelial cells are activated by exposure to environmental triggers like cigarette smoke, which contributes to COPD pathology (Rusznak et al. 2000; Schulz et al. 2004; Heijink et al. 2012; Amatngalim et al. 2015). In contrast to most studies, we used fresh whole cigarette smoke instead of (aged) cigarette smoke extract, and ALI-differentiated PBEC from COPD and non-COPD donors instead of submerged cultures of nondifferentiated primary cells or cell lines.

While the obvious advantage of this approach is that we can study well-differentiated primary cells from different patient populations, a limitation is that confirmation of data obtained with experimental pharmaceutics by, for example, gene editing or RNAi technology is not feasible in this context. Aside from efficiency issues and off-target effects in primary cells, knocking down EGFR or ADAM17 likely affects the growth and differentiation of primary bronchial epithelial cells, which essentially defeats our purpose. However, the two inhibitors that we apply here to inhibit EGFR (AG1478) and ADAM17 (TMI-2), respectively are widely used and are known to be highly selective.

Importantly, our data demonstrate for the first time that CS triggered increase of basal shedding of IL6R and AREG into the basal medium, in the presence of EGF in the growth medium providing basal EGFR activity, was more pronounced in ALI-PBEC derived from COPD patients compared to non-COPD controls. We further report the ability of CS to increase mRNA expression of these genes in an EGFR- and ADAM17-dependent way in ALI-PBEC cells, under these conditions, with a lower tendency to induction in the COPD group. These results extend previous studies showing dysregulated responses of COPD airway epithelial cells to cellular stress, and provide novel evidence for the mechanism of CS-induced and COPD-related proinflammatory and profibrotic responses (Fig. 8).

The differential effect of CS on sIL6R and AREG release between COPD and non-COPD ALI-PBEC might be related to differences in epithelial barrier function as previously described (Heijink et al. 2014). Using the current CS exposure system, we have previously shown that CS causes a transient decrease in epithelial barrier function (Amatngalim et al. 2015). However, in contrast to Heijink et al., we did not observe differences between COPD and non-COPD cultures at baseline conditions and upon CS exposure (G.D. Amatngalim et al., unpubl.

data), which may be explained by the fact that Heijink et al. focused on severe (GOLD stage IV) COPD. Another explanation might be differences in epithelial cell differentiation, as it has been shown that COPD epithelial cells display a more mesenchymal phenotype due to enhanced autocrine expression of TGF- β 1 (Gohy et al. 2015).

As previously shown, the EGFR-ADAM17 pathway is essential for IL-8 release from a bronchial epithelial cell line exposed to particulate air pollution (Ovrevik et al. 2011) and implicated in CS extract-induced expression of the mucin MUC5AC (Shao 2004). Further, autocrine production of EGFR ligands is involved in CS-induced IL-8 release from airway epithelial cells (Richter et al. 2002). Our studies extend these observations by showing the involvement of the ADAM17-EGFR pathway in the release of IL6R and AREG upon CS exposure of differentiated PBEC, both in the presence (Fig. 4), and absence (Fig. 6) of EGFR ligand (EGF) in the basal medium respectively. Notably, the basal shedding rates are considerably lower in cells preincubated in medium lacking EGF, resulting in a much larger ADAM17- and EGFRdependent stimulation effect of CS (compare Figs 4 and 6). Which of these extreme conditions of basal EGFR activation apply in normal and COPD lungs in situ, and to what extent autocrine feedback signaling through ADAM-dependent EGFR ligand shedding determines EGFR activity (Fig. 8) remains to be established.

Additionally, EGFR and ADAM17 were both essential for CS-induced IL6R and AREG mRNA expression (Fig. 7). These results provide novel insights into the mechanisms of airway epithelial cell activation by cigarette smoke in COPD, and highlight a role of ADAMs and EGFR in this process (Fig. 8).

We further found that CS increases shedding of IL6R and AREG to the basal medium, but not to the apical side (Fig. 1). This is in line with report in polarized Madin-Darby canine kidney cells (MDCK cells) showing that newly synthetized AREG is directly delivered to the basolateral surface with >95% efficiency (Brown et al. 2001). However, this is in contrast to the secretion of the innate immune mediators IL-8 and ribonuclease 7, which were also detected at the apical surface (Amatngalim et al. 2015). A polarized ADAM17-mediated secretion toward underlying tissue may be relevant for lung tissue remodeling through autocrine, paracrine, extracrine (exosomal targeted receptor activation) pathways in COPD (Booth et al. 2007; Zhou et al. 2012). Further examination of this phenomenon in epithelial-mesenchymal co-culture systems is in progress.

Amphiregulin release and phosphorylation of ADAM17 after CS extract treatment in ALI-PBEC has been previously detected by ELISA or western Blotting (Lemjabbar-Alaoui et al. 2011). Our proximity ligation assay (PLA)

2016 | Vol. 4 | Iss. 16 | e12878 Page 10



Figure 8. Cigarette smoke exposure activates EGFR-ADAM17 axis in airway epithelial cells. Under basal conditions, there is ADAM17 related AREG and IL6R shedding activity, depending on the level of EGFR activity (compare Figs 4 and 6). Cigarette smoke exposure (CS) initiates an interaction of the phosphorylated form of ADAM17 (ADAM17-P) with the full-length transmembrane forms of IL6R and AREG in an intracellular compartment of the airway epithelial cell (Fig. 5), resulting in proteolysis and subsequent secretion (shedding) of the soluble active domains of IL6R and AREG toward the basolateral compartment. This involves both ADAM17 and EGFR activity (Figs 4 and 6). CS exposure also affects IL6R and AREG gene expression or mRNA stabilization in airway epithelial cells through ADAM17 and EGFR activation. sAREG and sIL6R secreted towards the basolateral compartment may change the level of activity of EGFR and the interleukin receptor IL6st/gp130 on the airway epithelial cells (autocrine). This may contribute to the activity of the EGFR/ADAM17 axis (positive feedback), which is likely kept in check by inactivation of internalized EGFR. Paracrine activity of sAREG and sIL6R may further transactivate EGFR and the interleukin receptor IL6st/gp130 on the underlying myofibroblasts, and myeloid cells, activating downstream pathways, including STAT3, involved in inflammation, collagen deposition, and myofibroblast proliferation.

data show for the first time that CS-induced shedding involves an intracellular interaction between phosphorylated ADAM17 and its substrates (Figs 5 and 8), whereas the majority of the literature suggests that shedding occurs mainly at the plasma membrane surface. This interaction likely takes place in intracellular

@ 2016 The Authors. *Physiological Reports* published by Wiley Periodicals, Inc. on behalf of the American Physiological Society and The Physiological Society.
phosphorylated membranes that sequester active ADAM17 and its transmembrane substrates upon activation. This process may relate to the transient change in barrier function upon CS treatment in our system and the subsequent activation of EGFR (Amatngalim et al. 2015). Our observation is supported by other reports showing the presence ADAM17 or its substrates in a vesicular compartment in lysosomes (Ebsen et al. 2015), endosomes (Gephart et al. 2011; Dombernowsky et al. 2015), and exosomes negative for the ER marker calreticulin (Higginbotham et al. 2011). Moreover, Gutwein et al. demonstrated that ADAM10-mediated L1 migration factor cleavage occurs in Golgi-derived vesicles in tumor cells (Gutwein et al. 2002). This was further supported by a recent paper suggesting that also ADAM10/ADAM17mediated release of soluble FasL occurs from an intracellular vesicular pool of secretory lysosomes in stimulated T lymphocytes (Ebsen et al. 2015).

Moreover, after ligand binding, EGFR traffics in endosomes from the plasma membrane to an intracellular compartment to continue its signaling (Vieira et al. 1996; Teis et al. 2006). EGF-dependent MAPK signaling occurs from late endosomes and lysosomes (de Araujo et al. 2013). Interestingly, the MAPK/ERK pathway regulates trafficking of ADAM17 phosphorylated at Thr735 from the endoplasmic reticulum toward the plasma membrane (Soond 2005; Hilliard et al. 2011), which can be also activated through ligand binding to EGFR. Higginbotham et al. showed that AREG containing exosomes are rapidly internalized by recipient cells in an EGFR-dependent manner (Higginbotham et al. 2011), enhancing invasion of LM2-4175 cells through Matrigel and wound healing. In our ALI-PBEC system, we observed a predominantly lateral localization of EGFR under basal culture conditions. After exposure to CS, we observed a more cytoplasmic localization, consistent with EGFR activation (Figure S2). Therefore, in line with these and published observations, our findings suggest that in HBEC-ALI, CS triggers EGFR-mediated trafficking of ADAM17 and its substrates to a common subcellular compartment to allow proteolysis and subsequent secretion of soluble products (Fig. 8). At this time, we cannot establish to what extent autocrine signaling through shed ADAM substrates determine this response, or whether alternative mechanisms such as transactivation by intracellular kinases or oxidation or the extracellular receptor domain plays a role. Additional studies of triggered trafficking of EGFR, ADAM17-P, and its substrates in polarized airway cells are required to further establish this mechanism.

In addition to CS-enhanced release through ADAM17 enzymatic activity, we observed transiently enhanced mRNA expression of both AREG and full-IL6R in ALI-PBEC upon CS exposure (Figure 3). CS did not affect the level of the alternatively spliced form of IL6R, so we conclude that alternative splicing unlikely contributes to CS-enhanced release of IL6R. Previously, we observed upregulated IL-8 mRNA expression in ALI-PBEC exposed to CS as a result of enhanced EGFR phosphorylation and activation of the downstream MAPK/ERK1/2 signaling pathway (Amatngalim et al. 2015). Here, we report that CS-induced IL6R and AREG mRNA expression was also reduced upon EGFR inhibition. In addition, we show here for the first time that inhibition of ADAM17 has the same effect on these mRNA levels (Fig. 7). Therefore, our data suggest that CS enhances factors common for activation of IL6R, AREG, and IL-8 mRNA expression likely via an autocrine ADAM17-EGFR axis (Fig. 8).

The transcriptional and posttranscriptional regulation of these genes upon inhaled toxic substances has not been fully elucidated. Induced EGFR signaling is able to activate transcription of target genes. In addition, it has been shown that CS extract enhances HuR-mediated IL-8 mRNA stability in airway epithelial cells (Hudy and Proud 2013). Moreover, UV-exposure of keratinocytes enhances mRNA HuR-mediated stability of AREG, in an EGFR-dependent manner (Nakayama et al. 2013). These observations suggest that CS-induced activation of EGFR enhances sIL6R, AREG, and IL8 mRNA stability in ALI-PBEC. Furthermore, mRNA regulation may be altered in cultured airway epithelial cells from COPD patients (Steiling et al. 2013). CS-treated COPD ALI-PBEC expressed lower AREG and IL6R mRNA levels on average compared to non-COPD controls, but this difference was not statistically different (Fig. 3). Further studies on mRNA stability in this system are required to establish this. Nevertheless, this observation contrasts with the shedding data (Fig. 2) and suggests that ADAM17dependent AREG and sIL6R output is not primarily regulated on the mRNA level, but involves posttranslational regulation.

Our data support the relevance of the ADAM17/EGFR pathway in COPD development and progression. Selective inhibitors of ADAM17, EGFR, and other components of this signaling pathway such as JAK and MAPK potentially expand therapeutic possibilities. The development of ADAM inhibitors for clinical use has been studied intensively (Moss et al. 2008; Duffy et al. 2011; Dreymueller et al. 2015). In cellular and animal tumor models, positive results were recorded (Witters et al. 2008). An ADAM17 inhibitor, TAPI-0, reduced bleomycin-induced lung inflammation (Lee et al. 2012). The selective inhibitor TMI-2 used in this study, reduced LPS-induced inflammation in vivo (Zhang et al. 2004). However, due to a lack of target specificity of available compounds, and side effects associated with the various other biological functions of

© 2016 The Authors. *Physiological Reports* published by Wiley Periodicals, Inc. on behalf of the American Physiological Society and The Physiological Society.

ADAMs, chronic and systemic application of these compounds in humans is so far prohibited (Arribas and Esselens 2009). Clearly, more advanced intervention tools are required. Our data offer new insights in the regulation of mRNA expression, secretion, and release of ADAM17 substrates in airway epithelial cells upon triggering, which in combination with state of the art molecular design and advanced organotypic cellular modeling of airways could allow development of more selective inhibitors, targeted to specific cells and subcellular domains.

In summary, this study provides evidence that ADAM17-mediated release and shedding of IL-6R and AREG is highly enhanced in airway epithelial cells in response to CS-induced injury. Next to ADAM17, we highlight the importance of EGFR in the regulation of IL6R and AREG release and mRNA expression. Moreover, CS-induced ADAM17-mediated shedding of IL6R and AREG is especially high in COPD ALI-PBEC, suggesting that reducing ADAM17 activity in COPD might be a potential therapeutic approach.

Acknowledgements

The ADAM inhibitors TMI-1 and TMI-2 were kindly provided by Wyeth Research, Cambridge, MA 02140, USA. The authors thank the Department of Thoracic Surgery at LUMC for collection of lung tissue, and Winifred Broekman and Jorn Nützinger (LUMC) for their help in collecting patient data.

Conflict of Interest

Dr. Hiemstra reports receiving research grants from Galapagos NV for the submitted work, and research grants from Boehringer Ingelheim and Grifols outside the submitted work. Dr Scholte reports a research grant from Lexicon inc not related to the submitted work.

References

- Amatngalim, G. D., Y. van Wijck, Y. de Mooij-Eijk, R. M. Verhoosel, J. Harder, A. N. Lekkerkerker, et al. 2015. Basal cells contribute to innate immunity of the airway epithelium through production of the antimicrobial protein RNase 7. J. Immunol. 194:3340–3350.
- de Araujo, M. E. G., T. Stasyk, N. Taub, H. L. Ebner, B. Furst, P. Filipek, et al. 2013. Stability of the endosomal scaffold protein LAMTOR3 depends on heterodimer assembly and proteasomal degradation. J. Biol. Chem. 288:18228–18242.
- Arribas, J., and C. Esselens. 2009. ADAM17 as a therapeutic target in multiple diseases. Curr. Pharm. Des. 15:2319–2335.
- Barnes, P. J. 2013. New anti-inflammatory targets for chronic obstructive pulmonary disease. Nat. Rev. Drug Discov. 12:543–559.

- Baumgart, A., S. Seidl, P. Vlachou, L. Michel, N. Mitova, N. Schatz, et al. 2010. ADAM17 regulates epidermal growth factor receptor expression through the activation of Notch1 in non-small cell lung cancer. Cancer Res. 70:5368–5378.
- Beisswenger, C., J. Platz, C. Seifart, C. Vogelmeier, and R. Bals. 2004. Exposure of differentiated airway epithelial cells to volatile smoke in vitro. Cell. Physiol. Biochem. 71:402–409.
- Bell, J. H., A. H. Herrera, Y. Li, and B. Walcheck. 2007. Role of ADAM17 in the ectodomain shedding of TNF- α and its receptors by neutrophils and macrophages. J. Leukoc. Biol. 82:173–176.
- Booth, B. W., T. Sandifer, E. L. Martin, and L. D. Martin. 2007. IL-13-induced proliferation of airway epithelial cells: mediation by intracellular growth factor mobilization and ADAM17. Respir. Res. 8:51.
- Brown, C. L., R. J. Coffey, and P. J. Dempsey. 2001. The proamphiregulin cytoplasmic domain is required for basolateral sorting, but is not essential for constitutive or stimulus-induced processing in polarized Madin-Darby canine kidney cells. J. Biol. Chem. 276:29538–29549.
- Burgel, P.-R., and J. A. Nadel. 2008. Epidermal growth factor receptor-mediated innate immune responses and their roles in airway diseases. Eur. Respir. J. 32:1068–1081.
- Clarke, L. A., L. Sousa, C. Barreto, and M. D. Amaral. 2013. Changes in transcriptome of native nasal epithelium expressing F508del-CFTR and intersecting data from comparable studies. Respir. Res. 14:38.
- Deshmukh, H. S., L. M. Case, S. C. Wesselkamper, M. T. Borchers, L. D. Martin, H. G. Shertzer, et al. 2005. Metalloproteinases mediate mucin 5AC expression by epidermal growth factor receptor activation. Am. J. Respir. Crit. Care Med. 171:305–314.
- DeWitt, A. E., J. Y. Dong, H. S. Wiley, A. Douglas, and D. A. Lauffenburger. 2001. Quantitative analysis of the EGF receptor autocrine system reveals cryptic regulation of cell response by ligand capture. J. Cell Sci. 114:2301–2313.
- Dijkstra, A., D. S. Postma, J. A. Noordhoek, M. E. Lodewijk, H. F. Kauffman, N. H. T. ten Hacken, et al. 2009. Expression of ADAMs ("a disintegrin and metalloprotease") in the human lung. Virchows Arch. 454:441–449.
- Dombernowsky, S. L., J. Samsoe-Petersen, C. H. Petersen, R. Instrell, A. M. Hedegaard, L. Thomas, et al. 2015. The sorting protein PACS-2 promotes ErbB signalling by regulating recycling of the metalloproteinase ADAM17. Nat. Commun. 6:7518.
- Dreymueller, D., S. Uhlig, and A. Ludwig. 2015. ADAM-family metalloproteinases in lung inflammation: potential therapeutic targets. Am. J. Physiol. Lung Cell. Mol. Physiol. 308:L325–L343.
- Duffy, M. J., M. Mullooly, N. O'Donovan, S. Sukor, J. Crown, A. Pierce, et al. 2011. The ADAMs family of proteases: new biomarkers and therapeutic targets for cancer? Clin. Proteomics 8:9.

@ 2016 The Authors. *Physiological Reports* published by Wiley Periodicals, Inc. on behalf of the American Physiological Society and The Physiological Society.

2016 | Vol. 4 | Iss. 16 | e12878 Page 13 fibrosis PBEC human disease CF lung COPD airway ILGR Inflammation APCHAPTER 3 PREC huss 💒 iumani disea lung COPD airway IL6R Inflammation AREG remodeling EGFR shedding ADAM17 Torosis PBEC Lack of CFTR function in polarized bronchial epithelial cells results shedd in increased extracellular redox potential leading to augmented ADAM17 and EGFR activity and enhanced AREG release (submitted) for AREG remodeling EGFR shedding ADAM17 fibrosis PBEC human disea Stolarczyk M¹*, Veit G²*, human disease CF lung COPD airway IL6R inflammation Schnur A², Bidaud P-A², Veltman M¹, Lukacs GL², Scholte BJ¹ ¹Erasmus MC, Cell Biology, Rotterdam, The Netherlands ² McGill University, Department of Physiology, Montréal, Québec Inflamm*) Equal contributionEGFR shedding ADAM17 fibrosis PBEC human disease CF lung

Abstract

A disintegrin and metalloprotease 17 (ADAM17) activity is involved in chronic inflammation and tissue remodeling. Since these are hallmarks of CF lung disease we aimed to establish whether CFTR deficiency affects ADAM17 in polarized CFBE41o- cells lacking functional CFTR (iCFTR-) compared to cells with induced CFTR expression (iCFTR+). We found enhanced ADAM17-mediated shedding of an EGFR ligand, amphiregulin (AREG) in iCFTR- compared to iCFTR+ cells. This difference was observed at air-liquid interface, but not in submerged culture. Expression of the G551D mutant form of CFTR does not have this effect. As previously reported, ADAM17-mediated AREG shedding is tightly regulated by tyrosine kinases and phosphatases, particularly by EGFR/MAPK and SRC. iCFTR- cells displayed enhanced apical presentation of the basolateral marker EGFR compared to iCFTR+, which suggests its abnormal trafficking in CFTR deficient cells. Furthermore, AREG shedding is controlled by extracellular redox potentials, since the non-permeant natural antioxidant glutathione (GSH), but not its permeant precursor GSH ester, significantly inhibits AREG release. The apical extracellular redox state measured with a fluorescent ROS probe (Grx1roGFP-GPI) indicated enhanced oxidation of the apical surface fluid (ASL) in iCFTRcompared to iCFTR+ cells. In contrast, an intracellular fluorescent ROS probe (Grx1-roGFP) did not reveal a difference between iCFTR- and iCFTR+ cells. In summary our data suggest that in iCFTR- cells a hyper-oxidized state of the extracellular membrane, likely caused by defective GSH secretion, results in enhanced activity of the EGFR/ADAM17 signaling axis. This may contribute to the development of CF lung pathology, which should be investigated in future studies.

INTRODUCTION

More than 2000 different mutations in the cystic fibrosis transmembrane conductance regulator gene (*CFTR*) have been identified. Most of these mutations impair the function of the CFTR protein (68) (76), an anion channel involved in fluid homeostasis in multiple secretory epithelia (19) (46), leading to a recessive inherited disease Cystic Fibrosis (CF). The morbidity and mortality of CF patients is mainly due to chronic lung disease characterized by mucus plugging, chronic inflammation, recurrent bacterial infections, and tissue remodeling (11).

So far, the results of gene therapy and small molecule therapies aimed at correcting the activity of mutant CFTR are promising, but not sufficiently effective to correct CFTR function to the WT level in CF patients (16). Therefore, additional therapeutic strategies targeting downstream effects of CFTR deficiency are required to prevent or reduce loss of lung function, preferably as early intervention (44). This requires in depth knowledge of the relationship between the primary functional CFTR defect and disease pathology. In the present study we focused on the effect of CFTR dysfunction on airway epithelial pro-inflammatory and pro-fibrotic signaling through the EGFR/ADAM17 axis.

The transmembrane protease A Disintegrin And Metalloprotease 17 (ADAM17; formerly known as TACE) is likely involved in CF lung disease (17). ADAM17 function is closely linked to Epithelial Growth Factor Receptor (EGFR) activity. An activated ADAM17 releases a soluble EGFR binding domain of transmembrane growth factors and cytokines into the extracellular space in a process called shedding. This allows both autocrine signaling towards the epithelial cells and paracrine epithelial-mesenchymal trans-signaling (6) (8) (61). Inhibition of either ADAM17 or EGFR affect ADAM17-mediated release of EGFR ligands, suggesting that the EGFR/ADAM17 axis is established by a positive feedback loop, likely involving the MAPK pathway (5) (35) (61) (66) (64) (79).

In the lungs, the ADAM17/EGFR pathway is activated during inflammation and bacterial infections (9). EGFR and ADAM17 activities are involved in cytokine (e.g. IL-8) and growth factor (e.g. TGF α) release and in regulation of their mRNA expression levels (2) (51) (69). Among other substrates, ADAM17 sheds most of the epidermal growth factor receptor (EGFR) ligands (64), including amphiregulin (AREG). AREG autocrine signaling affects mucus expression (75) (21) and cytokine secretion (12), whereas its paracrine signaling has been linked to airway smooth muscle cell proliferation (15) and TGF β induced fibrosis characterized by fibroblast proliferation, enhanced α -smooth muscle actin and collagen expression (88).

ADAM17/EGFR signaling is also involved in regulation of mucus production. Its hyperactivity leads to mucus plugging of distal airways, as frequently observed in CFTR deficient subjects (71) (85) (87). Importantly, the hyperactivity may also contribute not only to excessive inflammation and mucus secretion, but due to paracrine signaling also to lung tissue remodeling in CF lung disease. EGFR/ADAM17 activity regulates tissue repair by increasing epithelial proliferation and differentiation (8) (10) (40) (53). Conversely, inhibition of either AREG or EGFR diminishes proliferation of TGF- β -stimulated lung fibroblasts, expression of α -

smooth muscle actin and collagen (88) suggesting that hyperactivation of ADAM17/AREG/EGFR signaling may contribute to pulmonary fibrosis.

Altered epithelial EGFR expression and signaling have been observed in CF lung disease (55) (34), but its cross-talk with ADAM17 has not been studied in detail. We recently found that cigarette smoke (CS) leads to a more pronounced increase in EGFR dependent ADAM17 activity in differentiated primary epithelial cells from COPD patients in comparison to non-COPD controls, suggesting that chronic inflammation changes the EGFR/ADAM signaling axis (69). An inhibitory effect of CS on CFTR activity has recently been reported (13) (67), which raises the possibility that lack of functional CFTR is a contributing factor to the increased EGFR/ADAM17 signaling.

The EGFR/ADAM17 signaling pathway is sensitive to oxidative stress (ROS), which activates the PKC and MAPK pathways (52). ROS is also a direct regulator of both ADAM17 and EGFR activity by oxidizing or rearranging their cysteine residues (18) (81) (31). In CF bronchi proper oxidant/antioxidant homeostasis (25) is disrupted and increased concentration of reactive oxygen species (ROS), lower levels of glutathione (GSH) and reduced nitric oxide (NO) have been reported (25) (89). This abnormal redox signaling is possibly linked to impaired proteostasis and autophagy in CF epithelial cells (45) (74), and may enhance EGFR/ADAM17 responses in CFTR deficient airway epithelial cells.

To test this hypothesis we have investigated the ADAM17 activity in a CF cell model of airway epithelium that allows inducible expression of CFTR (77). We observed that CFTR deficiency in airway epithelial cells enhances ADAM17 dependent growth factor shedding. This activity is regulated by EGFR, MAPK and SRC. Ratiometric measurements of redox potentials show that the enhanced ADAM17/EGFR responses in CFTR deficient cells are not related to a substantial change in intracellular redox potential, but can instead be attributed to an increased oxidation state of the extracellular fluid. Oxidation of the apical lining fluid in CFTR deficient condition may be caused by defective glutathione transport (25) (38) (89). Overall, our study provides novel insights in the involvement of the EGFR/ADAM17 axis in CF pathology and suggests that intervention in this pathway may be beneficial for CF patients.

RESULTS

Absence of CFTR expression enhances AREG release in human CF bronchial epithelial cells. We used the immortalized CF bronchial epithelial cell line CFBE41o- (CFBE) to study the effect of functional CFTR expression on the release of amphiregulin (AREG), an autocrine and paracrine factor for EGFR activation that is involved in lung morphogenesis, tissue repair following infection or injury, and the resolution of inflammation (83). CFBE cells have a CFTR F508del/F508del genotype with no detectable endogenous F508del CFTR protein expression (20). These cells were engineered to express WT-CFTR under control of a doxycycline responsive transactivator to a level that is comparable with the endogenous expression in Calu-3 (77). Apical expression and function of CFTR was confirmed by immunostaining and short-circuit current (I_{sc}) measurement, respectively (77). CFBE cells were seeded onto permeable filter supports, allowed to differentiate for 4 days post-confluence under liquid-liquid culture (LLC) conditions in presence (iCFTR+) or absence

(iCFTR-) of induced CFTR expression, followed by a shift to air-liquid interface (ALI) to mimic the native environment of respiratory epithelia. iCFTR+ and iCFTR- ALI filters do not differ with respect to protein content or total metabolic activity (Scholte, Veit unpublished data). To examine the release of AREG at the apical and basolateral plasma membrane, polarized CFBE cells were cultured at ALI for 24 h followed by medium change and re-establishment of LLC conditions for 3 h, medium collection and ELISA. The AREG release at the basolateral membrane was four times higher in comparison to the apical release, which, considering three times larger area of the basolateral in comparison to the apical membrane, suggests little polarity of AREG secretion in iCFTR cells (Fig. 1A).



Figure 1 AREG shedding is enhanced in CFTR deficient CFBE. (A) AREG release is mainly towards the basolateral compartment. The AREG release to basolateral and apical compartment was examined (ELISA) in 24h ALI polarized iCFBE+ and iCFBE- cells followed by medium change and re-establishment of LLC conditions for 3 h. The AREG release at the basolateral plasma membrane (PM) was 3-4 times higher in comparison to the apical release. However, considering the larger area of the basolateral cellular membrane in comparison to the apical (approximately 3 times), it suggests modest basolateral polarity of AREG secretion in iCFTR cells. Statistical analysis: paired t-test. Values represent means of n independent experiments performed in triplicate (three filters in parallel within one experiment). (B) Functional CFTR expression decreases AREG release in CFBE cells. Constitutive basolateral AREG release from under LLC conditions and from 4-8h and 8-24h ALI differentiated CFBE41o- cells with (iCFTR+) and without (iCFTR-) induced CFTR expression. The low basolateral AREG release under LCC was increased upon transfer to ALI conditions. After ALI exposure, during the time span 4-8h induction of CFTR expression partially attenuated AREG release, which was maintained in subsequent 8-24h medium collections. (C) Expression of the mutant G551D-CFTR in CFBE41o- cells (iG551D+) did not significantly reduce AREG release in comparison to CFBE41o- cells not expressing G551D-CFTR (iG551D-).

AREG levels were determined by ELISA and expressed as pg AREG per filter produced per hour. Data were collected in parallel for iCFTR- and CFTR+ in triplicate (three different filters per experiment), averaged and presented as a single value in a figure. Horizontal bars indicate averages of averaged triplicates of n separate experiments with SEM values. Statistical analysis: unpaired t-test.



Figure 2 CFTR expression does not alter the mRNA levels of AREG, ADAM17 or EGFR. The mRNA levels of AREG (A), ADAM17 (B) and EGFR (C) were determined by qPCR in CFBE differentiated for 5 days on filter supports under LLC and shifted to ALI for the indicated times. Data are means ± SEM of three independent experiments.

The low constitutive AREG release under LLC conditions (~2 pg/filter*hour) was increased by four to five fold upon transfer to ALI conditions (Fig. 1B). This was associated with a transient increase in AREG mRNA expression (Fig. 2A). The increase in AREG release was attenuated by induction of CFTR expression, which reduced the AREG secretion rate by ~35% from ~9 to 6 pg/filter*hour in the basolateral medium (Fig. 1B). The CFTR suppressive effect was maintained in subsequent medium collections (8-24h) under ALI, however it was absent in epithelia under LLC (Fig. 1B). Expression of the mutant G551D-CFTR, which has a comparable biosynthetic processing and stability to the WT CFTR, but functionally is severely impaired (77) (3), did not significantly reduce AREG release (Fig. 1C). Furthermore, this result excludes a direct influence of doxycycline on AREG shedding activity in this system. Similarly, removing doxycycline from the medium for 24h before the experiment did not affect shedding activity (n=1 in triplicate, data not shown). Also, doxycycline did not affect the relative expression levels of AREG in these cells (Fig. 2A). These data indicate that constitutive WT CFTR activity is necessary and sufficient to limit the excess secretion of AREG in CF epithelia (Fig. 1B).

The increased AREG release of CFBE epithelia is mediated by ADAM17

The increased release of AREG in iCFTRcompared to iCFTR+ CFBE cells (Fig. 1B) was not reflected by a difference in AREG mRNA levels (Fig. 2A), suggesting that the CFTR

may influence the shedding, rather than the expression of AREG. AREG is expressed as a type-I transmembrane protein (pro-AREG) that is predominantly directed to the basolateral membrane by a cytosolic sorting motif, where it undergoes constitutive internalization and recycling (28). The ectodomain shedding of pro-AREG is mediated by the metalloproteinase



Figure 3 Increased CFTR-dependent basolateral AREG release is mediated by ADAM17 (A) Differentiated CFBE were shifted to ALI for 24 h followed by addition of the fluorogenic ADAM17 substrate (10 μ M, 1h) to the basolateral medium in presence or absence of TMI1 (2 μ M, n = 3), TMI-2(1 μ M, n = 5) or an equivalent amount of carrier (DMSO, n =5). Data are means ± SEM of averaged independent experiments normalized to iCFTR- (100%), statistical analysis was performed by one way ANOVA. (B) 8h treatment with the selective ADAM17 inhibitor, TMI-2 (1 μ M, Wyeth) decreased AREG release to the same extend as treatment with an inhibitor targetting a broad spectrum of metalloproteases, TMI-1 (2 μ M, Wyeth), providing the involvement of ADAM17 in AREG shedding in 24h ALI differentiated iCFTR+ and iCFTR- CFBE410- cells.

ELISA data were normalized to iCFTR- condition. Values represent means of n independent experiments in triplicates (three different filters per experiment). Statistical analysis: unpaired t-test. Statistic of differences between iCFTR+ and iCFTR- indicated at the top of the graph, differences between control condition and treatment are indicated just above the bars.

ADAM17 (60) and occurs mainly intracellularly in differentiated airway epithelial cells, likely in endosomal compartments (69).

To measure the ADAM17 activity in polarized CFBE monolayers under ALI conditions, a fluorogenic ADAM17 substrate was added to the basolateral medium for 2 h. In the absence of CFTR expression significantly higher amounts of substrate were cleaved (Fig. 3A). The highly specific ADAM17 inhibitor TMI-2 (1μ M) (86) reduced production of the fluorescent product to a similar extent as a broad specificity ADAM/MMP inhibitor (TMI-1) (2µM) confirming the involvement of ADAM17 in this activity (Fig 3A). Since CFTR expression or culture conditions did not alter the ADAM17 mRNA levels (Fig. 2B), these results suggest that under ALI conditions active CFTR decreases the ADAM17 enzyme activity. Consistent with this observation and similar to the fluorogenic ADAM17 substrate release, inhibition of the ADAM17 activity with the highly specific inhibitor TMI-2 (1 μ M) (86) also reduced AREG release to a similar extent as a broad specificity ADAM/MMP inhibitor (TMI-1) (2μ M), confirming the involvement of ADAM17 in this activity (Fig 3B) (85). These data suggest that, as in other epithelial cell lines (9) and primary human bronchial epithelia (69), ADAM17 is the main shedding protease involved in AREG release in CFBE cells. Moreover, inhibition of ADAM17 reduced the CFTR-dependent differences in proteolysis of a fluorescent substrate (Fig 3A) and in AREG release (Fig 3B), suggesting that CFTR deficiency increases ADAM17 dependent shedding activity.



Figure 4 Basolateral AREG shedding is tightly regulated by protein phosphatases and kinases, in particular EGFR/MAPK

(A) The broad range tyrosine kinase inhibitor genistein (100 μ M) strongly reduces the AREG shedding in iCFBE410- cells. (B) Herbimycin A (1 μ M), as a consequence of Hsp90 inhibition destabilizes multiple kinases and results in decreased AREG release, particularly in iCFTR- CFBE410- cells. (C) Conversely, the tyrosine phosphatase inhibitor sodium orthovanadate (25 μ M) induces AREG shedding, also in particular in iCFTR- CFBE410- cells. (D) SRC kinase family inhibitor, PP2 (10 μ M) decreased AREG shedding, while the inactive analog PP3 had no inhibitory effect (10 μ M) (E) The canonical inhibitor of EGFR AG1478 (1 μ M) strongly inhibits AREG release, to the similar extent as selective ADAM17 inhibitor TMI-2 presented in Figure 2 (F) Inhibition of downstream MAP kinase MEK1/2 with U0126 inhibitor (10 μ M) also reduces AREG shedding. Reduction of AREG shedding upon inhibition of EGFR/MAPK pathway proofs the existence of positive feedback loop in EGFR/ADAM17 axis.

ELISA data were normalized to iCFTR- condition. Values represent means of n independent experiments in triplicates (three different filters per experiment). Statistical analysis: unpaired t-test. Statistic of differences between iCFTR+ and iCFTR- indicated at the top of the graph, differences between control condition and treatment are indicated just above the bars.



Figure 5

CFTR deficiency increases apical localization and phosphorylation of EGFR in CFBE41o- cells cultured at ALI

(A-D) Western blot analysis. In four separate independent experiments iCFTR- and CFTR+ cells were grown in quadruplicate for 8h at ALI and analyzed (see Material and Method section for details) (A) EGFR (green fluorescence) and (B) EGFR-P (luminescence) show representative results of blots from a single experiment. Quantitative analysis of EGFR (C) and EGFR-P (D) of all data, normalized to the average percentage for iCFTR- cells per experiment show no significant effect of CFTR expression on total EGFR expression, but a significant reduction of EGFR phosphorylation. (E,F) Laser confocal fluorescence microscopy pictures of EGFR immunostaining in 24h ALI exposed CFTR- and CFTR+ iCFBE41o- cells showed that EGFR is predominantly, but not exclusively localized to lateral membrane. Particularly in iCFTR- CFBE41o- cells apical EGFR staining is localized to the apical membrane. (G) EGFR apical plasma membrane density (PMD, cell surface labeling ELISA, see Material and Methods) is higher in iCFTR- 24h ALI CFBE410- cells than in iCFTR+, and this difference is not present in LCC cells (H) EGFR basolateral PMD in CFBE41o- cells does not show differences in EGFR expression between iCFTR- and iCFTR+ CFBE41o- cells neither under ALI nor LCC condition. (I) EGFR apical and (J) basolateral PMD did not show the differences in apical and basolateral EGFR expression in iG551D- and iG551D+ CFBE41o- cells (n=number of independent experiments, in triplicate each). Statistical analysis: unpaired t-test.



Figure 6

EGF stimulation causes EGFR internalization and AREG shedding

(A) EGFR aggregates 3h after EGF (10ng/ml) stimulation in 24h ALI exposed CFBE41o- cells (untreated control: Fig. 5A), (B) This is accompanied by strong induced AREG shedding at the 8h and 24h time points (nearly 20 and 40-fold respectively).

ELISA data were normalized to the iCFTR- condition. Values represent means of n independent experiments in triplicates (three different filters per experiment). Statistical analysis: unpaired t-test. Statistic of differences between iCFTR+ and iCFTR- indicated at the top of the graph, differences between control condition and treatment are indicated just above the bars.

Inhibition of EGFR signaling abrogates the CFTR-dependent differences in AREG release

Since it is not obvious how the loss of an apical anion transporter could directly lead to an increase in ADAM17 activity, which is active in intracellular compartments (69) and at the plasma membrane (8) (63), we measured the involvement of signaling pathways upstream of ADAM17. Among other factors, ADAM17 activity is regulated by receptor tyrosine kinases (RTKs) (23). The broad-range tyrosine kinase inhibitor genistein (100 μ M) strongly reduced the AREG shedding and diminished the CFTR-dependent differences in cells under ALI conditions (Fig. 4A). Similarly, treatment with herbimycin A (1 μ M), which inhibits Hsp90 (80) leading to the destabilization of multiple kinases (70), resulted in decreased AREG release in non-CFTR expressing cells (Fig. 4B). Consistent with the notion that tyrosine kinase signaling is involved in the augmented AREG shedding in the absence of CFTR, the tyrosine phosphatase inhibitor sodium ortho-vanadate (25 μ M) increased AREG release (Fig. 4C).

Next, we inhibited specific candidate tyrosine kinases that are involved in ADAM17 activation. The SRC-kinase in part regulates ADAM17 activity (63). The SRC family inhibitor PP2 decreased AREG shedding (10 μ M, Fig. 4D; lower concentrations 1 μ M and 3 μ M; were equally effective, data not shown), while the inactive analog PP3 (10 μ M) had no effect. The SRC-kinase may also directly phosphorylate and activate the EGFR, bypassing EGFR ligand binding, which results in downstream MAP kinase activation and in turn ADAM17 activation (27) (43). The canonical inhibitor of EGFR signaling AG1478 (1 μ M) strongly inhibited AREG

release (Fig. 4E). Similarly, inhibition of the downstream MAP kinase MEK1/2 with the inhibitor U0126 (10 μ M) reduced AREG release (Fig. 4F). These data are consistent with the existence of a positive feedback loop in the EGFR/ADAM17 axis that involves the MAPK pathway (9) (63) (69). Most importantly, inhibition of SRC, EGFR or MEK1/2 abrogated the CFTR-dependent differences in AREG shedding, indicating that non-functional CFTR activates the EGFR-ADAM17 axis.



Figure 7 CFTR mediates oxidation of the air surface liquid (ASL), but not cytosol in CFBE41o- cells

(A) Laser confocal fluorescence microscopy pictures of iCFBE41o- cells lentivirally transduced with glutaredoxin-1-redox sensitive GFP (Grx-1-GFP) (B) The ratiometric fluorescence measurements (the ratio of 405 nm emission and 488 nm excitation) of Grx-1-GFP probe expressed in iCFBE41o- cells revealed no difference in cytosolic glutathione redox potential between iCFTR+ and iCFTR- cells. Cell permeable reducing agent DTT (0,5 mM) was used as a control for ratiometric fluorescence measurements. (C) Laser confocal fluorescence microscopy pictures of iCFBE41o- cells lentivirally transduced with glutaredoxin-1-redox sensitive GFP fused to C-terminal GPI anchor and N-terminal signal sequence (Grx-1-GFP-GPI) (D) The ratiometric fluorescence measurements of apically expressed Grx-1-GFP-GPI probe in iCFBE41o- cells measured as the ratio of 405 nm emission and 488 nm excitation of the probe revealed oxidation of ASL in CFTR deficient cells. The cell impermeable reducing agent MESNA (50 μM) was used as a control.

Statistical analysis: unpaired t-test; n indicates the number of independent experiments in duplicate or triplicate, meaning that two or three different filters were used per experiment.

CFTR deficiency increases apical localization and phosphorylation of EGFR in CFBE cells cultured at ALI

To test whether the EGFR inhibitor-sensitive and CFTR modulated AREG release is the result of increased EGFR expression, the mRNA and protein levels of the receptor were determined by qPCR (Fig. 2C) and immunoblot (Fig. 5A-D), respectively. The levels of EGFR mRNA at different times before and after shift to ALI conditions were not significantly different between CFTR and non-CFTR expressing cells (Fig. 2C). Consistent with this, the total EGFR antigen, as determined by western blotting, did not differ significantly between iCFTR- and iCFTR+ (Fig. 5A, C).

Indirect immunostaining and laser confocal fluorescence microscopy showed EGFR was predominantly confined to the basolateral membrane in both iCFTR+ and iCFTRexpressing cells (33), but a small proportion of the EGFR staining was localized to the apical membrane (Fig. 5E,F). The specificity of the EGFR immunofluorescence signal was confirmed by the intracellular localization of EGFR antigen after 3 h stimulation with EGF (Fig. 6A), a known trigger for internalization and lysosomal degradation of the receptor (57). To establish an effect of CFTR deficiency on EGFR localization, the plasma membrane density of EGFR was determined at the apical and basolateral cell surface (Fig. 5G,H). Here an anti-EGFR antibody recognizing the extracellular domain of EGFR, followed by an HRP-coupled secondary antibody was used to detect the cell surface localized receptor by fluorescence spectroscopy with the cell impermeable Amplex Red as HRP substrate (see Methods). The apical EGFR PM density was increased by ~30% in non-CFTR expressing cells cultured for 24h under ALI conditions compared to CFTR-expressing cells (Fig. 5G). In contrast, at the basolateral membrane in ALI cultured cells, and in cells grown under LLC conditions no CFTR-dependent differences in the EGFR PM expression were observed (Fig. 5H). Furthermore, expression of G551D-CFTR in cells cultured under ALI did not suppress apical (Fig. 5I) or basolateral EGFR PM density (Fig. 5J).

Taken together, these results indicate that functional CFTR deficiency alters the net polarity of EGFR under ALI conditions. However, the majority of EGFR is still localized at or near the basolateral membrane in iCFTR- cells (Fig 5A).

The constitutive phosphorylation of EGFR at Y1173, a measure of EGFR activity, was decreased in CFTR expressing cells compared to CFTR deficient cells cultured for 8 h under ALI as determined by phospho-immunoblotting (Fig. 5 B,D,H), whereas the total amount of EGFR was not significantly affected by CFTR induction (Figure 5C), consistent with the EGFR mRNA results (Figure 2A). Activation of EGFR by exogenous stimulation with 10 ng/ml EGF at both the apical and basolateral PM resulted in a massive CFTR-independent increase in AREG release, and gradually abolished the CFTR-dependent differences (Fig. 6B). Taken together, these data indicate that CFTR expression modulates EGFR trafficking and the level of EGFR constitutive activity, but not the expression or maximal signaling capacity of the EGFR pathway.

CFTR mediated shift of the extracellular redox potential attenuates AREG release.

Since the expression of functional CFTR is sufficient to limit the augmented AREG release in CF cells, it is plausible to assume that CFTR-mediated ion transport at the apical membrane attenuates the constitutive signaling that leads to AREG shedding. CFTR transports chloride, bicarbonate and reduced glutathione (GSH) that regulate the height, pH and redox potential

of the airway surface liquid (ASL), respectively (62). It was previously established that oxidation activates ADAM17 directly, by changing its conformation (18) (81) and indirectly, by initiating upstream signaling that increases ADAM17 activity through activating SRC (63) or EGFR (54). To determine if CFTR expression leads to changes in the cytosolic and ASL redox state, which may lead to induced ADAM17 activity, we used the glutaredoxin1–redox-sensitive GFP (Grx1-roGFP) fusion protein that allows for measurement of the glutathione redox potential by ratiometric fluorescence measurement (31). The unmodified (Grx1-roGFP) or the N-terminal signal sequence and C-terminal GPI-anchor signal fused (SS-Grx1-roGFP-GPI) redox probes were delivered by lentiviral transduction into the CFBE-iCFTR cells for expression in the cytosol or tethered to the outer leaflet of the plasma membrane, respectively (Fig 7A).



Figure 8 ROS scavengers decrease AREG release from CFBE41o- cell

(A) Glutathione (10 mM GSH), but not the cell permeable GSH ester (1 mM), strongly inhibits AREG release in iCFTR CFBE410- cells abrogating differences between CFTR- and CFTR+ condition (B) A ROS scavenger, N-acetyl cysteine (5 mM NAC) and NADPH oxidase inhibitor diphenylene-iodonium (5 μ M DPI) significantly decreases AREG release in CFTR+ and CFTR- iCFTR CFBE cells, showing the role of ROS in regulation of ADAM17 activity.

ELISA data were normalized to iCFTR- condition. Values represent means of n independent experiments in triplicates (three different filters per experiment). Statistical analysis: unpaired t-test. Statistic of differences between iCFTR+ and iCFTR- indicated at the top of the graph, differences between control condition and treatment are indicated just above the bars.

In submerged non-polarized CFBE the cytosolically confined Grx1-roGFP redox probe was in a relatively reduced state, as measured with a fluorescence plate reader (data not shown). The redox probe was slightly further reduced by the addition of the cellpermeant reducing agent dithiothreitol (DTT), but not the cell impermeable 2mercaptoethanesulfonic acid sodium salt (MESNA). Grx1-roGFP was strongly oxidized by addition of hydrogen peroxide (H₂O₂) (data not shown). In contrast, the GPI-anchored Grx1roGFP was in oxidized state under similar conditions and was only slightly oxidized further by H₂O₂, but strongly reduced by both the cell-permeant DTT and the non-permeant MESNA addition, confirming the extracellular accessibility and redox sensitivity of the probe (data not shown).

On polarized filter grown CFBE at ALI, the apical Grx1-roGFP fluorescence was measured by ratiometric laser confocal fluorescence microscopy using an upright confocal microscope after shifting the cells for 24 h to ALI (Fig 7A). Under these conditions, CFTR expression had no significant effect on the redox state of the cytosolic probe in iCFBE cells (Fig. 7B), which already was almost at the maximally reduced state, as revealed by DTT addition (Fig. 7B). In contrast, induction of CFTR expression resulted in a more reduced state of the ASL probe Grx1-roGFP-GPI (Fig. 7C), approaching the minimal fluorescence ratio detected in the presence of MESNA (Fig. 7D).

Supplementation of the basolateral and apical medium with GSH shortly before establishing ALI strongly inhibited the AREG release in non-CFTR and CFTR expressing cells, and abrogated the CFTR-dependent differences in AREG shedding (Fig 8A). Addition of extracellular GSH (3 mM as in Fig 8) inhibited AREG shedding to the same extent, but did not significantly reduce EGFR Y1173 phosphorylation, suggesting that the effect was not due to a change in the activation of EGFR (n=3 independent experiments in fourfold, data not shown). In contrast to GSH, the cell permeable GSH ester, which is cleaved by phosphodiesterases and causes an increase in the intracellular GSH concentration did not diminish AREG shedding significantly (Fig. 8A). N-acetyl cysteine, a cell permeant precursor of GSH synthesis that by itself has an antioxidant effect (59), reduced AREG release (Fig. 8B). Complementary to these results, reducing the cellular reactive oxygen species (ROS) production using diphenyleneiodonium (DPI), an inhibitor of intracellular NADPH oxidases (NOX) and dual oxidases (DUOX), also reduced AREG release, although to a lesser extend (Fig. 8B).

Taken together, our data suggest that high extracellular GSH does not exert its inhibitory effect on AREG shedding through inactivation of EGFR, but rather switches ADAM7 to its closed, inactive conformation (18). Reduced extracellular GSH in CFTR deficient epithelia would have the opposite effect.

DISCUSSION

ADAM17 shedding of AREG is enhanced by CFTR-deficiency.

Here we report that dysfunctional CFTR in differentiated CFBE airway epithelial cells cultured at air-liquid interface induces activity of ADAM17 (Fig 1B). Importantly, this effect was not observed in a normally expressed but inactive form of CFTR (G551D) (Fig 1C), which shows that CFTR activity is required for ADAM17 regulation. The increased ADAM17-mediated AREG shedding caused by CFTR deficiency observed in our model likely contributes to mucus hypersecretion, inflammation and tissue remodeling in CF lung disease. Further studies should test whether ADAM17 activation is also observed in the CF airways *in situ*, and whether attenuation of ADAM activity reduces CF pathology.

Polarity of shedding.

In differentiated ALI-CFBE410- cells, we found that total AREG shedding towards the basal compartment is fourfold higher than towards the apical side (Fig 1A). This is consistent with the relative membrane surface area of the two compartments, suggesting that there is no

strong preference for shedding polarity. Importantly, induction of CFTR does not have a significant effect on the polarity of secretion (Fig 1A). In ALI differentiated primary bronchial epithelial cells the basolateral to apical ratio is substantially higher (69), which may be explained by the fact that these cells have a higher columnar morphology compared to the low cubical CFBE cells. Since the volume of apical lining fluid is small, the actual apical concentration of ligand, which determines the rate of receptor binding, is potentially high. Therefore, we have to consider that also ADAM ligands shed apically from epithelial cells contribute to luminal signaling in the lung, where myeloid cells likely play a major role in determining the downstream effects of AREG signals, in particular activated macrophages during acute injury response (82). Furthermore, enhanced expression of AREG mRNA was reported in alveolar neutrophils from CF patients, consistent with a heightened state of inflammation (1).

Regulation of ADAM17 mediated shedding activity by EGFR.

To establish the mechanism by which CFTR deficiency changes the level of ADAM17 dependent agonist shedding in iCFTR CFBE cells, we analyzed the involvement of several known effectors. First we showed that in this model ADAM17-mediated shedding is tightly regulated by protein phosphatases and tyrosine kinases, particularly by SRC, EGFR and MAPK (Fig 4). Under basal conditions EGFR emerged as a potent ADAM17 regulator. When inhibited, it reduced ADAM17 mediated shedding to the same extent as the selective ADAM17 inhibitor TMI-2 (Fig 3B), whereas addition of the EGFR agonist EGF to the medium strongly enhances AREG release (Fig 6B). This is consistent with results with other epithelial models (9) (56) and our data in primary ALI-HBEC (69). In the latter, inhibition of EGFR reduced ADAM17 dependent AREG and IL-6R shedding, as well as AREG, IL6R and IL8 mRNA levels to the same extent as an ADAM17 selective inhibitor (69). Therefore, there is a close link between EGFR and ADAM17 activity in airway cells. It is likely that the EGFR/ADAM17 axis is established by a feedback loop in which EGFR agonists such as AREG are shed by ADAM17, activating EGFR and downstream MAPK pathway, which in turn activates ADAM17 (9), or ADAM17 regulates EGFR expression by activation of Notch1 (5). Hyperactivity of this EGFR/ADAM17 axis in CFTR deficient epithelia may affect CF lung pathology (24).

EGFR phosphorylation is enhanced by CFTR-deficiency.

Exaggerated epithelial EGFR expression and abnormal signaling are observed in chronic lung disease such as COPD and asthma (2) (34) (55), which emphasizes its potential relevance in CF. Here we show that induction of functional CFTR expression in CFTR deficient airway cells does not affect total EGFR protein (Fig. 5) or mRNA expression (Fig 2), however it significantly reduces the level of EGFR Y1173 phosphorylation, a measure of EGFR activation (Fig. 5D). Our data are consistent with increased phosphorylated EGFR (EGFR-P) levels reported in airway epithelium and inflammatory cells from CF patients in comparison to non-CF controls (47). Furthermore, our data show increased EGFR-dependent AREG shedding in iCFTR- cells (Fig 4E and Fig. 6B) suggesting that CFTR deficiency enhances cellular signaling through the EGFR/ADAM17 axis. In accordance with our results, several reports suggest that active CFTR suppresses other EGFR related inflammatory responses, like IL8 secretion and mucin production (7) (56) (37) (48).

CFTR-deficiency affects EGFR trafficking.

As summarized in the introduction, one of the factors regulating EGFR activity is its localization relative to its target molecules. Upon ligand binding EGFR dimerizes and is internalized to intracellular compartments where it is either inactivated or continues signaling (36) (73) (90) (72). We showed that in CFBE-ALI EGFR is primarily, but not exclusively localized at the basolateral membrane under basal ALI conditions (Fig 5 E,F). Surface ELISA showed an increase of EGFR trafficking to the apical membrane in CFTR deficient ALI-CFBE cells (Fig 5G). Addition of an inhibitory antibody against EGFR (cetixumab) to polarized iCFTR did not reduce AREG shedding, whereas basal addition was effective (Scholte et al, unpublished data). This does not support a major role of apical EGFR in this process, however, abnormal trafficking of EGFR would likely be an important factor in the regulation of the EGFR/ADAM17 axis.

EGF binding leads to EGFR internalization and lysosomal degradation, whereas AREG stimulation causes EGFR internalization and recycling to the membrane (24) (57). Thus, CFTR dependent enhanced shedding of AREG (Fig. 1A) may lead to activation of EGFR (Fig. 5B,D) and more frequent recycling to the membrane (Fig. 5 G), sustaining enhanced EGFR dependent ADAM17 activity through the feedback loop.

As expected, EGF induced EGFR internalization towards a perinuclear compartment in iCFTR (Fig. 6A,B) and caused a massive increase in ADAM17 shedding activity (Fig 6B), gradually reducing the difference between iCFTR+ and iCFTR-. However, further investigations are needed to establish whether abnormal polarisation and trafficking of EGFR plays a decisive role in CFTR dependent EGFR/ADAM17 activity (39). Furthermore, we need to establish whether this is specific for EGFR or reflects a general imbalance of proteostasis and deficient autophagy in reported in CF epithelial cells (22) (45). Our data support the notion that CFTR deficiency changes the kinetics of EGFR/ADAM17/AREG trafficking and activity (24). Further studies in primary bronchial epithelial cells with well-defined tight junctions and preferably in CF lung biopsies need to be performed to confirm our observations.

Regulation of shedding activity by SRC.

The ubiquitous tyrosine kinase c-SRC reportedly causes ROS dependent activation of EGFR and ADAM17 in immortalized epithelial cells (41),(63) suggesting it has a regulatory role in the EGFR/ADAM17 signaling cascade. Indeed, in CFTR-deficient CFBE cells inhibition of SRC (PP2) reduced AREG release to the level observed in cells with induced CFTR activity (Fig. 4D).

A role of other SRC family members in the enhanced AREG shedding in CFTR deficient CFBE cells cannot be ruled out, considering the limited specificity of PP2 inhibition. Transactivation of EGFR by SRC has been reported (27) (43), but the mechanism and relationship with ligand or stress induced trafficking remains to be established. Thus, further studies are warranted to define an effect of CFTR deficiency on SRC trafficking and activity towards its substrates.

Extracellular REDOX potential regulates EGFR/ADAM17-dependent AREG shedding from CFBE cells.

ADAM17 activity is dependent on the interaction between a protein (thiol) disulfide isomerase (PDI) with the extracellular membrane proximal domain (MPD) of ADAM17,

which maintains ADAM17 in an inactive, "closed" conformation. This thiol isomerase is active in the endoplasmic reticulum, presumably involved in protein maturation, but has also been observed in the extracellular space (18). It is inactivated when the extracellular environment is oxidized, which results in an active "open" conformation of ADAM17 (81) (18). In addition to this direct effect of oxidation on ADAM17 activity, reactive oxygen species (ROS) may affect ADAM17 indirectly through the activation of ROS sensitive molecules in the ADAM17 signaling pathway, like SRC (63) (84) and EGFR (54). Therefore, we speculated that a change in redox homeostasis is involved in enhanced EGFR/ADAM17 dependent AREG shedding in CFTR deficient cells.

Indeed, the impermeant natural thiol antioxidant Glutathione (GSH) (29) strongly inhibited EGFR/ADAM17-mediated shedding of AREG in CFBE cells (Fig. 8), but not the cell permeant GSH ester, which is cleaved by intracellular phosphodiesterases increasing the intracellular GSH concentration (Fig. 8). N-acetyl cysteine (NAC), a ROS scavenger and reducing agent, and the NOX inhibitor diphenyleneiodonium (DPI) reduced AREG release to a lesser extent. These data suggest that EGFR/ADAM17-mediated AREG release is primarily sensitive to the extracellular redox potential through the extracellular PDI-MPD domain interaction (18), whereas the EGFR and SRC redox sensitive cysteine residues that are localized intracellularly (54) play a limited role. This would explain why high extracellular GSH effectively blocked AREG shedding, likely by inhibiting ADAM activity at the PDM domain, but did not reduce EGFR phosphorylation (Scholte et al, data not shown).

Abnormal extracellular redox potential in iCFTR- ALI contributes to enhanced EGFR/ADAM activity.

The redox sensitivity of ADAM17 could explain its enhanced activity in CFTR deficient cells. CFTR is involved in the maintenance of intra and extracellular redox potential in airways through its role in the transport of GSH (30) (89). In CFTR-KO mice reduced levels of GSH were reported in the luminal fluid in comparison to wild type mice, whereas total GSH levels in the lung tissue did not differ (78). Furthermore, in wild type mice, *P aeruginosa* infection increases GSH in the epithelial lining fluid, however, this infection induced-increase is absent in CFTR-/- mice (14). Similarly, GSH levels were reduced in lung lavage fluids of CF patients compared to healthy subjects (58). This is attributed to defective GSH transport in CFTR deficient epithelial cells (26), likely contributing to enhanced downstream inflammatory signaling in CF lung disease (32) (89). Importantly, treatment of CF patients with GSH reduced inflammatory markers in BALF (32). The decreased GSH in the epithelial lining fluid may contribute to the poor response to infections and aberrant activation of inflammatory molecules sensitive to oxidation (25).

Consistent with the role of CFTR in GSH transport we established enhanced oxidation of ASL in iCFTR- cells compared to iCFTR+, monitored by Grx1-roGFP2 anchored to the apical membrane (Grx1-roGFP2-GPI) (Fig. 7). In contrast, the ratiometric measurement of cytosolic roGFP-Grx-1 fluorescence did not reveal significant differences in the intracellular redox potential between iCFTR- cells and iCFTR+ cells (31) (49) (Fig. 7D,E). We propose that this oxidized extracellular redox potential, which is likely due to reduced GSH transport in iCFTR- cells, activates ADAM17 shedding activity through the redox sensitive extracellular MPD. How this GSH redox signal is transferred to the active ADAM17 pool remains to be established in future studies. ADAM17 and its substrates actively recycle through the

plasma membrane, exposing the MPD and protease sensitive domains to the extracellular environment. Though ADAM17 reportedly has a preferentially basolateral sorting route, we cannot exclude that it passes the apical membrane, in particular in CFTR deficient cells that have improper trafficking of membrane proteins, as shown by our (Fig 5) and published data (45). Furthermore, we cannot exclude that the apical GSH pool affects the lateral and basal domains of the epithelial plasma membrane, in particular in CFTR deficient cells that have instable tight junctions (42). Further studies with modified targeted redox probes in polarized are required to establish this. Finally, we cannot exclude that extracellular GSH affects the intravesicular compartments, in which ADAM17 is active. We have shown previously (69) that CS induced interactions between ADAM17 and its substrates occur intracellularly in polarized airway cells. GSH can enter the cell through endocytosis/pinocytosis, trans-cytosis, and in the shape of glutathione-protein conjugates (65), which could allow an interaction with the ADAM17 ectodomain even when that is not apically localized. Further studies in advanced cell culture models are required to establish relative ADAM17 and GSH localization in CF airway epithelia.

Taken together, We propose that supplementation with extracellular GSH severely inhibits ADAM17 related AREG shedding through reduction and inactivation of its extracellular domain (18). Conversely, an oxidized extracellular redox potential (Fig 6), due to insufficient GSH secretion in CFTR- cells (4) and consistent with a lower ASL GSH concentration in the ASL of CF patients and CF cell models in comparison to normal controls (26) (58) may activate ADAM17, resulting in enhanced AREG release (Fig 1). This may involve the feedback loop action of the EGFR/ADAM17 axis and enhanced EGFR activity (Fig 5).

Summary and conclusions

Our data provide more insight in the role of CFTR deficiency in regulating the EGFR/ADAM17 axis, which is likely involved in the triggering and resolution of CF lung inflammation. Activity of the EGFR/ADAM17 shedding system is higher in differentiated CFTR deficient CFBE cells at air-liquid interface compared to CFTR expressing cells. This is related to stronger EGFR activation and abnormal trafficking in CFTR deficient cells, likely involving a paracrine feedback loop through enhanced ADAM17 dependent AREG shedding. The extracellular REDOX potential, which as we show strongly affects ADAM17 shedding activity, was changed to a more oxidized state in CFTR deficient cells, likely due to reduced GSH secretion. We propose that abnormal extracellular redox potential in CFTR deficient epithelial cells enhances the basal activity of the EGFR/ADAM17 axis. Further studies are required to establish the link between extracellular redox potential and EGFR/ADAM activity in primary cells and airway in situ. Based on our results we propose that Intervention in this pathway may be beneficial to CF patients.

MATERIALS AND METHODS

Compounds and antibodies

TMI-2 (PF-5480090) and TMI-1 (85) were obtained from Wyeth inc. (Philadelphia, Pennsylvania, USA). AG1478 (1 μ M), diphenyleneiodonium chloride (DPI), genistein, reduced glutathione (GSH), herbimycin A, hydroxyl peroxide (H2O2), N-acetylcystein (NAC), orthovanadate, tert-butyl-hydroxyperoxide (t-BHP) were purchased from Sigma-Aldrich. EGF, PP2, PP3, UO126 were purchased from R&D. Rat monoclonal anti-EGFR antibodies specific for extracellular domain used in dilution 1:200 for IF and 1:1000 for PM, were purchased from Abcam.

CFBE cell culture.

The generation of the human immortalized CFBE41o- cell line (abbreviated as CFBE) harboring the inducible expression of WT- or G551D CFTR has been described before (77). The CFBE cells were propagated on plastic dishes coated with 3 mg/ml collagen (Pure Col, Advanced Biomatrix), 1 mg/ml human fibronection (BD Biosciences) with and 100 µg/ml bovine serum albumin (Sigma-Aldrich) in LHC-basal medium (Biosource). Cells were cultured in MEM (Gibco) supplemented with 10% fetal bovine serum (FBS, Gibco), 10 mM 4-(2hydroxyethyl)-1-piperazineethanesulfonic acid (HEPES, Gibco), 200mM L-glutamine (Gibco). Cells containing the transactivator alone (TetON) were maintained in medium containing 200 µg/ml G418 (Invitrogen), and cells harboring inducible CFTR expression were maintained in presence of 200 µg/ml G418 (Invitrogen) and 3 µg/ml Puromycin (Invitrogen). When cells reached 80% confluence, they were transferred on the 1.12 cm² Transwell filters with $0.4\mu m$ pore size (Corning) at a density of 1×10^5 cells per filter and cultured with 0.5 ml and 1.5 ml medium in the apical and basolateral chamber, respectively. 24 h after seeding (day 1) CFTR expression was induced with 500ng/ml doxycycline. At day 4 the medium was changed to OptiMEM (+/- doxycycline) supplemented with 2 mM L-glutamine (Gibco). At day 5 compound treatments were performed in OptiMEM and samples were collected at different time points after ALI exposure: 4h (0-4h), 8h (4-8h), 24h (8-24h). At each investigated time point after sample collection, whole medium was changed.

Generation of Redox Sensor Cells

The ratiometric redox sensor Grx1-roGFP (31) was shuttled into the lentiviral vectors pLVX-IRES-Hyg (clontech) for stable expression in different cellular compartments. The basic variant is expressed in the cytosol as described by Gutsher et al. Another variant was equipped with a N-terminal signal-sequence and a C-terminal glycosylphosphatidylinositol (GPI) anchor consensus sequence (SS-Grx1-roGFP-GPI) to determine the redox potential of the ASL. CFBE cells with inducible expression of WT-CFTR were transduced with these lentiviral particles followed by selection with 200 µg/ml hygromycin B.

Measurement of redox state in CFBE redox sensor cells

The initial characterization of the redox senor cells was performed in a fluorescence plate reader. For this purpose CFBE cells expressing the redox sensor variants were seeded into black 96 wells and kept for \geq 4 days post confluent. Fluorescent intensities were measured

at 520 nm (\pm 20 nm) using TECAN plate reader after excitation with 405 nm and 488 nm, respectively. Higher 405/488-ratio value represents a higher oxidative state (49).

To determine the redox state in differentiated CFBE under ALI CFBE redox sensor cells, grown on Snapwell inserts (Corning) were used for ratiometric confocal imaging. Generally a Z-stack of 20-30 optical xy-sections was acquired (0.7-1- μ m apart) with LSM700 microscope (Carl Zeiss, Jena, Germany) equipped with a water immersion objective lens (40x, numerical aperture, 0.8; achroplan; Zeiss). For immersion of the objective lens, cells were covered with non-toxic the hydrophobic, FluorinertTM FC-70 fluid (boiling point, 215°C; refractive index, 1.303; Hampton Research) that allows normal gas exchange. During the experiment cells were maintained at 37°C, with 5% CO₂ in a humidified top-stage incubator mounted on the microscope. The biosensor was excited sequentially with 405 nm and 488 nm lasers and emitted intensities were collected from 520 nm. CFBE cells without Grx1-roGFP2 were used as a background control. Ratios of intensities were calculated from 5 cells on a single apical (GPI) or intracellular z-slice that was selected manually from the Z-stack images for each cell. Three microscopic areas were recorded on each filter. The redox probe maximal and minimal oxidation emission ratios were determined in some experiments by basal addition of H₂O₂ (100 μ m) and 500 μ M DTT or 50 mM MESNA, respectively.

AREG ELISA

Samples for AREG ELISA were collected form the basolateral and apical medium of differentiated CFBE410⁻ conditioned for the indicated times. AREG levels were determined by ELISA (Duoset human AREG, DY262, R&D Systems) following the manufactures instructions. The final detection was performed with a SuperSignal ELISA Femto luminescence substrate (Pierce) on a VICTOR Light plate reader (PerkinElmer). Data were corrected for the dilution factor and insert size. The amount of the shed AREG was expressed as pg/filter per hour. Measurement of the metabolic activity of the cells with alamar blue (Invitrogen) showed that the induced CFTR expression had no effect on the cell number or viability of the CFBE cells (data not shown).

EGFR plasma membrane density measurements (PMD)

The PM density of EGFR was determined by cell-surface ELISA (50). CFBE cells for PM density measurements were grown on 1.12 cm² Transwell filters with 0.4µm pore size (Corning), as described above. At the indicated time points, the cells were incubated in reagent diluent (1% BSA in PBS containing 0.1 mM CaCl₂ and 1mM MgCl₂(PBSCM) for 30 min at RT. EGFR localized at the apical or basolateral membrane was detected by incubation with rat monoclonal anti-EGFR antibody specific for the extracellular domain (1:500, ab231, Abcam) followed by extensive washing and subsequent incubation with anti-rat-HRP antibody (1:1000, 112-035-167, Jackson ImmunoResearch), both for 60 min at RT. Excess antibody was removed by extensive washing with PBSCM, and specific binding was determined with Amplex Red (Invitrogen) HRP substrate. The fluorescence intensity was measured at 544 nm excitation and 590 nm emission wavelengths using an Infinite M1000 (Tecan) fluorescence plate reader, and values determined from cells without primary antibody labeling were subtracted as background.

mRNA isolation and qPCR

The mRNA isolation and q-PCR was performed as described previously (77). The primers are listed in Table 1.

Gene symbol	Direction	Sequence
AREG	forward	GTGGTGCTGTCGCTCTTGATA
AREG	reverse	ACTCACAGGGGAAATCTCACT
ADAM17	forward	TCTCCTATTCCTGACCAGCG
ADAM17	reverse	GTCTGAGAGCAAAGAATCAAGC
EGFR	forward	GGGCTCTGGAGGAAAAGAAA
EGFR	reverse	AAATTCCCAAGGACCACCTC
GAPDH	forward	CATGAGAAGTATGACAACAGCCT
GAPDH	reverse	AGTCCTTCCACGATACCAAAGT

1

Table 1: qPCR primers used in this study

1

Immunofluorescence (IF)

Cells for PM measurements were grown under ALI on the 1.12cm² Transwell filters with 0.4µm pore size (Corning), as described above. At the investigated time points, cells were fixed with 4% paraformaldehyde (Sigma Aldrich) for 10 min at RT, followed by permeabilization with 4% paraformaldehyde and 0,5% Triton-X (Sigma Aldrich) in PBS for 5 min at RT. This was followed by incubation with 100 mM glycine pH 7.5 for 10 min and blocking with 1% BSA (in PBSCM) for 45 min, incubation for 1h with rat monoclonal anti-EGFR antibodies specific for the extracellular domain (1:200, ab231, Abcam), and subsequent incubation with the secondary fluorescent antibody Alexa Fluor 488 (1:1000, A11006, Invitrogen) for 1h at RT in darkness. Antibodies were diluted in 0.5% BSA in PBSCM. Nuclei were stained with 300 ng/ml DAPI for 10 min in darkness. Washing steps were performed with PBSCM. Cut out filters were mounted with mounting medium (100 mM Tris/HCl pH 8.5, 10% mowiol, 25% glycine, 150 mM 1,4-Diazabicyclo[2.2.2]octane) on glass cover slides. 20-30 z-optical sections were acquired using a confocal LSM-780 microscope equipped with a Plan-Apochromat 63X/1.40 oil differential interference contrast objective (Carl Zeiss), reconstituted using the Zen 2012 software package and representative xzsections are shown.

Western blotting

105 iCFTR CFBE cells were seeded per insert ($1.12cm^2$ PET filters, pore size 0.4μ m,Corning) with (iCFTR+) or without (iCFTR-) doxycycline induction. The cells were cultured as

described above in CFBE cell culture section. Four independent experiments were performed with four (3) or three (1) inserts per condition each. At 8h ALI, medium was collected for AREG ELISA (R&D Systems Europe, Abingdon, UK), whereas cell samples were transferred to 100µl RIPA buffer (50 mM Tris HCl, 150 mM NaCl, 1% Triton-X, 0,5% Sodium deoxycholate, 0,1% SDS) containing phosphatase (PhosStop, Sigma-Aldrich) and protease inhibitors (Complete Mini, Roche). Protein concentration was determined with BCA assay (Pierce Rockford, IL). The lysates were centrifuged at 15,000 rpm for 15 min at 4 °C. The clear supernatants were immediately transferred to SDS loading buffer (5µl 4x sample buffer, 1µl 0.1M DTT, 15µl protein sample) and stored at -20°C until use. 7% gel electrophoresis was performed with iCFTR+ and iCFTR- samples loaded in parallel on the same gel and proteins were transferred to PVDF membranes (0.45 micron filter, Merck Millipore). The membranes were blocked with 3% BSA and 0,05% Tween20 in PBS, and then stained for EGFR EP38Y (1:5000, Abcam), and a fluorescent second antibody (1:10.000, 926-32213, Li-Cor). EGFR-P was detected using antibody Tyr1173 (1:200, Santa Cruz) and a HRP labeled second antibody (1:10.000, P0448, Dako). The proteins were detected using Amersham ECL prime Western blotting detection reagent (GE Healthcare Europe GmbH) and scanned with Alliance (luminescence) or Odyssey (fluorescence) scanners. In separate experiments equal amounts of protein from iCFTR- and iCFTR+ cells cultured in parallel were loaded on the same blot. The density of each band was quantitated with ImageJ software. Acquired data for EGFR-P and EGFR were normalized to the average iCFTR- data for each individual blot and presented as percentage of the average of iCFTR- values in Figure 5.

ADAM17 substrate assay

The ADAM17 enzyme activity in polarized CFBE grown on Transwell filters and exposed to ALI for 24 h, was measured by supplementing the basolateral medium with 10 μ M fluorogenic TACE substrate IV (Millipore). Following 1 h incubation the medium was collected and the fluorescence was determined at 320 nm excitation and 420 nm emission wavelengths using an Infinite M1000 (Tecan) fluorescence plate reader. Values determined from filters without cells were subtracted as background control.

Statistical analysis

Statistical analyses were performed with GraphPad Prism Software. The statistical test used for the specific experiment is indicated underneath each figure. Each experiment was performed in triplicate, meaning that samples were collected from three different wells, grown simultaneously, averaged and considered as one experiment in analysis (indicated by n). Graphs present the averages of n experiments with SEM values. Differences at P-value >0.05 were considered to be not statistically significant (ns>0.05), P-values <0.05 were considered to be statistically significant and indicated as *P<0.05, **P<0.01, ***P<0.001, ****P<0.001.

Acknowledgements

Wyeth Research, Cambridge MA 02140 USA kindly provided the ADAM inhibitors TMI-1 and TMI-2. We are deeply indebted to the late Dr Dieter Gruenert, who generated, characterized and kindly provided the original CFBE410–cell line.

Marta Stolarczyk is currently affiliated with the Department of Medicine V, University of Heidelberg, Heidelberg, Germany; Department of Translational Oncology, National Center for Tumor Diseases and German Cancer Research Center, Heidelberg, Germany, and Department of Intelligent Imagining, BioQiant, Heidelberg.

Funding

We thank the Lung Foundation Netherlands (#3.3.10.027) and the Dutch Cystic Fibrosis Foundation NCFS (MV HIT-CF1-C7) for supporting this study. We also acknowledge EMBO (ASTF525–2014) for supporting M.S short-term fellowship at McGill University in Gergely L. Lukacs laboratory with Guido Veit's supervision.

Competing interest

BS obtained a research grant from Lexicon inc (CA, MA USA) for work not related to this study. All other authors declare no conflict of interest.

LITERATURE:

- Adib-Conquy M, Pedron T, Petit-Bertron AF, Tabary O, Corvol H, Jacquot J, Clement A, Cavaillon JM. Neutrophils in cystic fibrosis display a distinct gene expression pattern. *Mol Med* 14: 36–44, 2008.
- Amatngalim GD, van Wijck Y, de Mooij-Eijk Y, Verhoosel RM, Harder J, Lekkerkerker AN, Janssen RAJ, Hiemstra PS. Basal Cells Contribute to Innate Immunity of the Airway Epithelium through Production of the Antimicrobial Protein RNase 7. J Immunol 194: 3340–3350, 2015.
- 3. Anderson MP, Berger HA, Rich DP, Gregory RJ. Nucleoside triphosphates are required to open the CFTR chloride channel. *Cell* 67: 775–784, 1991.
- Ballatori N, Krance SM, Notenboom S, Shi S, Tieu K, Hammond CL. Glutathione dysregulation and the etiology and progression of human diseases. *Biol Chem* 390: 191–214, 2009.
- Baumgart A, Seidl S, Vlachou P, Michel L, Mitova N, Schatz N, Specht K, Koch I, Schuster T, Grundler R, Kremer M, Fend F, Siveke JT, Peschel C, Duyster J, Dechow T. ADAM17 Regulates Epidermal Growth Factor Receptor Expression through the Activation of Notch1 in Non-Small Cell Lung Cancer. *Cancer Res* 70: 5368–5378, 2010.
- 6. Berasain C, Avila MA. Amphiregulin. Semin Cell Dev Biol 28: 31–41, 2014.
- 7. Berube J, Roussel L, Nattagh L, Rousseau S. Loss of Cystic Fibrosis Transmembrane Conductance Regulator Function Enhances Activation of p38 and ERK MAPKs, Increasing Interleukin-6 Synthesis in Airway Epithelial Cells Exposed to Pseudomonas aeruginosa. J Biol Chem 285: 22299–22307, 2010.
- 8. **Booth BW, Sandifer T, Martin EL, Martin LD**. IL-13-induced proliferation of airway epithelial cells: mediation by intracellular growth factor mobilization and ADAM17. *Respir Res* 8: 51, 2007.
- 9. Breshears LM, Schlievert PM, Peterson ML. A Disintegrin and Metalloproteinase 17 (ADAM17) and Epidermal Growth Factor Receptor (EGFR) Signaling Drive the Epithelial Response to Staphylococcus aureus Toxic Shock Syndrome Toxin-1 (TSST-1). J Biol Chem 287: 32578–32587, 2012.
- Broekman W, Amatngalim GD, de Mooij-Eijk Y, Oostendorp J, Roelofs H, Taube C, Stolk J, Hiemstra PS. TNF-α and IL-1β-activated humanmesenchymal stromal cells increase airwayepithelial wound healing in vitro viaactivation of the epidermal growth factorreceptor. *Respir Res* 17: 1–12, 2016.
- 11. **Cantin AM, Hartl D, Konstan MW, Chmiel JF**. Inflammation in cystic fibrosis lung disease: Pathogenesis and therapy. *J Cyst Fibros* 14: 419–430, 2015.

- Chokki M, Mitsuhashi H, Kamimura T. Metalloprotease-dependent amphiregulin release mediates tumor necrosis factor-α-induced IL-8 secretion in the human airway epithelial cell line NCI-H292. *Life Sci* 78: 3051–3057, 2006.
- Clunes LA, Davies CM, Coakley RD, Aleksandrov AA, Henderson AG, Zeman KL, Worthington EN, Gentzsch M, Kreda SM, Cholon D, Bennett WD, Riordan JR, Boucher RC, Tarran R. Cigarette smoke exposure induces CFTR internalization and insolubility, leading to airway surface liquid dehydration. *FASEB J* 26: 533–545, 2012.
- 14. **Day BJ, van Heeckeren AM, Min E, Velsor LW**. Role for Cystic Fibrosis Transmembrane Conductance Regulator Protein in a Glutathione Response to Bronchopulmonary Pseudomonas Infection. *Infect Immun* 72: 2045–2051, 2004.
- 15. **Deacon K, Knox AJ**. Human airway smooth muscle cells secrete amphiregulin via bradykinin/COX-2/PGE 2, inducing COX-2, CXCL8, and VEGF expression in airway epithelial cells. *Am J Physiol Lung Cell Mol Physiol* 309: L237–L249, 2015.
- Dhooghe B, Haaf JB, Noel S, Leal T. Strategies in early clinical development for the treatment of basic defects of cystic fibrosis. *Expert Opin Investig Drugs* 25: 423–436, 2016.
- 17. **Dreymueller D, Uhlig S, Ludwig A**. ADAM-family metalloproteinases in lung inflammation: potential therapeutic targets. *Am J Physiol Lung Cell Mol Physiol* 308: L325–L343, 2015.
- Düsterhöft S, Jung S, Hung C-W, Tholey A, Sönnichsen FD, Grötzinger J, Lorenzen I. Membrane-Proximal Domain of a Disintegrin and Metalloprotease-17 Represents the Putative Molecular Switch of Its Shedding Activity Operated by Protein-disulfide Isomerase. J Am Chem Soc 135: 5776–5781, 2013.
- 19. Ehre C, Ridley C, Thornton DJ. Cystic fibrosis: An inherited disease affecting mucin-producing organs. *Int J Biochem Cell Biol* : 136–145, 2014.
- 20. Ehrhardt C, Collnot E-M, Baldes C, Becker U, Laue M, Kim KJ, Lehr C-M. Towards an in vitro model of cystic fibrosis small airway epithelium: characterisation of the human bronchial epithelial cell line CFBE410. *Cell Tissue Res* 323: 405–415, 2005.
- 21. Enomoto Y, Orihara K, Takamasu T, Matsuda A, Gon Y, Saito H, Ra C, Okayama Y. Tissue remodeling induced by hypersecreted epidermal growth factor and amphiregulin in the airway after an acute asthma attack. *J Allergy Clin Immunol* 124: 913–920.e1–7, 2009.
- 22. Esposito S, Tosco A, Villella VR, Raia V, Kroemer G, Maiuri L. Manipulating

proteostasis to repair the F508del-CFTR defect in cystic fibrosis. *Mol Cell Pediatr* 3: 1–5, 2016.

- 23. **Fan H, Derynck R**. Ectodomain shedding of TGF-α and other transmembrane proteins is induced by receptor tyrosine kinase activation and MAP kinase signaling cascades. *EMBO J* 18: 6962–6972, 1999.
- Fukuda S, Nishida-Fukuda H, Nanba D, Nakashiro K-I, Nakayama H, Kubota H, Higashiyama S. Reversible interconversionand maintenance of mammaryepithelial cell characteristics by theligand-regulated EGFR system. *Sci. Rep.* (2016). doi: 10.1038/srep20209.
- 25. Galli F, Battistoni A, Gambari R, Pompella A, Bragonzi A, Pilolli F, Iuliano L, Piroddi M, Dechecchi MC, Cabrini G. Biochimica et Biophysica Acta. *BBA-MOL BASIS DIS* 1822: 690–713, 2012.
- 26. **Gao L, Kim KJ, Yankaskas JR, Forman HJ**. Abnormal glutathione transport in cystic fibrosis airway epithelia. *Am J Physiol Lung Cell Mol Physiol* 277: L113–L118, 1999.
- 27. Garcia-Recio S, Pastor-Arroyo EM, Marín-Aguilera M, Almendro V, Gascón P. The Transmodulation of HER2 and EGFR by Substance P in Breast Cancer Cells Requires c-Src and Metalloproteinase Activation. *PLoS ONE* 10: e0129661, 2015.
- Gephart JD, Singh B, Higginbotham JN, Franklin JL, Gonzalez A, Fölsch H,
 Coffey RJ. Identification of a Novel Mono-Leucine Basolateral Sorting Motif
 Within the Cytoplasmic Domain of Amphiregulin. *Traffic* 12: 1793–1804, 2011.
- 29. Ghezzi P. Role of glutathione in immunity and inflammation in the lung. *IJGM*4: 105–113, 2011.
- 30. **Gould NS, Min E, Martin RJ, Day BJ**. CFTR is the primary known apical glutathione transporter involved in cigarette smoke-induced adaptive responses in the lung. *Free Radic Biol Med* 52: 1201–1206, 2012.
- Gutscher M, Pauleau A-L, Marty L, Brach T, Wabnitz GH, Samstag Y, Meyer
 AJ, Dick TP. Real-time imaging of the intracellular glutathione redox potential.
 Nat Meth 5: 553–559, 2008.
- 32. Hartl D, Starosta V, Maier K, Beck-Speier I, Rebhan C, Becker BF, Latzin P, Fischer R, Ratjen F, Huber RM, Rietschel E, Krauss-Etschmann S, Griese M. Inhaled glutathione decreases PGE 2 and increases lymphocytes in cystic fibrosis lungs. Free Radic Biol Med 39: 463–472, 2005.
- 33. **He C**. The Epidermal Growth Factor Receptor Juxtamembrane Domain Has Multiple Basolateral Plasma Membrane Localization Determinants, Including a

Dominant Signal with a Polyproline Core. J Biol Chem 277: 38284–38293, 2002.

- 34. **Hirsch FR, Varella-Garcia M, Cappuzzo F**. Predictive value of EGFR and HER2 overexpression in advanced non-small-cell lung cancer. *Oncogene* 28: S32– S37, 2009.
- 35. **Kasina S, Scherle PA, Hall CL, Macoska JA**. ADAM-mediated amphiregulin shedding and EGFR transactivation. *Cell Proliferat* 42: 799–812, 2009.
- 36. **Khan EM, Heidinger JM, Levy M, Lisanti MP, Ravid T, Goldkorn T**. Epidermal Growth Factor Receptor Exposed to Oxidative Stress Undergoes Src- and Caveolin-1-dependent Perinuclear Trafficking. *J Biol Chem* 281: 14486–14493, 2006.
- 37. **Kim S, Beyer BA, Lewis C, Nadel JA**. Normal CFTR Inhibits Epidermal Growth Factor Receptor-Dependent Pro-Inflammatory Chemokine Production in Human Airway Epithelial Cells. *PLoS ONE* 8: e72981, 2013.
- 38. **Kleme M-L, Levy E**. Cystic Fibrosis-Related Oxidative Stress and Intestinal Lipid Disorders. *Antioxid Redox Signal* 22: 614–631, 2015.
- 39. Kuwada SK, Lund KA, Li XF, Cliften P, Amsler K, Opresko LK, Wiley HS. Differential signaling and regulation of apical vs. basolateral EGFR in polarized epithelial cells. Am J Physiol Cell Physiol 275: C1419–C1428, 1998.
- 40. Lemjabbar H, Li D, Gallup M, Sidhu S, Drori E, Basbaum C. Tobacco Smokeinduced Lung Cell Proliferation Mediated by Tumor Necrosis Factor αconverting Enzyme and Amphiregulin. J Biol Chem 278: 26202–26207, 2003.
- 41. Lemjabbar-Alaoui H, Sidhu SS, Mengistab A, Gallup M, Basbaum C. TACE/ADAM-17 Phosphorylation by PKC-Epsilon Mediates Premalignant Changes in Tobacco Smoke-Exposed Lung Cells. *PLoS ONE* 6: e17489, 2011.
- 42. **LeSimple P, Liao J, Robert R, Gruenert DC, Hanrahan JW**. Cystic fibrosis transmembrane conductance regulator trafficking modulates the barrier function of airway epithelial cell monolayers. *J Physiol* 588: 1195–1209, 2010.
- 43. Lin CC, Pan CS, Wang CY, Liu SW, Hsiao LD, Yang CM. Tumor necrosis factoralpha induces VCAM-1- mediated inflammation via c-Src-dependent transactivation of EGF receptors in human cardiac fibroblasts. *J Biomed Sci* 22: 1–15, 2015.
- 44. **Lubamba B, Dhooghe B, Noel S, Leal T**. Cystic fibrosis: insight into CFTR pathophysiology and pharmacotherapy. *Clin Biochem* 45: 1132–1144, 2012.
- 45. Luciani A, Villella VR, Esposito S, Brunetti-Pierri N, Medina D, Settembre C, Gavina M, Pulze L, Giardino I, Pettoello-Mantovani M, D'Apolito M, Guido S, Masliah E, Spencer B, Quaratino S, Raia V, Ballabio A, Maiuri L. Defective

CFTR induces aggresome formation and lung inflammation in cystic fibrosis through ROS-mediated autophagy inhibition. *Nat Cell Biol* 12: 863–875, 2010.

- Maiuri L, Luciani A, Giardino I, Raia V, Villella VR, D'Apolito M, Pettoello-Mantovani M, Guido S, Ciacci C, Cimmino M, Cexus ON, Londei M, Quaratino
 S. Tissue Transglutaminase Activation Modulates Inflammation in Cystic Fibrosis via PPAR Down-Regulation. J Immunol 180: 7697–7705, 2008.
- 47. **Martel G, Roussel L, Rousseau S**. The protein kinases TPL2 and EGFR contribute to ERK1/ERK2 hyper- activation in CFTRDF508-expressing airway epithelial cells exposed to Pseudomonas aeruginosa. *Biochem Biophys Res Commun* 441: 689–692, 2013.
- 48. Martin C, Coolen N, Wu Y, Thevenot G, Touqui L, Pruliere-Escabasse V, Papon JF, Coste A, Escudier E, Dusser DJ, Fajac I, Burgel P-R. CFTR dysfunction induces vascular endothelial growth factor synthesis in airway epithelium. *Eur Respir J* 42: 1553–1562, 2013.
- 49. **Morgan B, Sobotta MC, Dick TP**. Measuring E_{GSH} and H₂O₂ with roGFP2-based redox probes. *Free Radic Biol Med* 51: 1943–1951, 2011.
- Okiyoneda T, Barrière H, Bagdány M, Rabeh WM, K Du, Höhfeld J, Young JC,
 Lukacs GL. Peripheral protein quality control removes unfolded CFTR from the plasma membrane. *Science* 329: 805–810, 2010.
- 51. Ovrevik J, Refsnes M, Totlandsdal AI, Holme JA, Schwarze PE, Lag M. TACE/TGF-α/EGFR regulates CXCL8 in bronchial epithelial cells exposed to particulate matter components. *Eur Respir J* 38: 1189–1199, 2011.
- 52. **Park HS, Kim SR, Lee YC.** Impact of oxidative stress on lung diseases. *Respirology* 14: 27–38, 2009.
- 53. Pasqualon T, Pruessmeyer J, Weidenfeld S, Babendreyer A, Groth E, Schumacher J, Schwarz N, Denecke B, Jahr H, Zimmermann P, Dreymueller D, Ludwig A. A transmembrane C-terminal fragment of syndecan-1 is generated by the metalloproteinase ADAM17 and promotes lung epithelial tumor cell migration and lung metastasis formation. *Cell Mol Life Sci* 72: 3783–3801, 2015.
- 54. **Paulsen CE, Truong TH, Garcia FJ, Homann A, Gupta V, Leonard SE, Carroll KS.** Peroxide-dependent sulfenylation of the EGFR catalytic site enhances kinase activity. *Nat Chem Biol* 8: 57–64, 2011.
- 55. **Puddicombe SM**, **Polosa R**, **Richter A**, **Krishna MT**, **Howarth PH**, **Holgate ST**, **Davies DE**. Involvement of the epidermal growth factor receptor in epithelial repair in asthma. *FASEB J* 14: 1362–1374, 2000.

- 56. **Richter A, O'Donnell RA, Powell RM**. Autocrine ligands for the epidermal growth factor receptor mediate interleukin-8 release from bronchial epithelial cells in response to cigarette smoke. *Am J Respir Crit Care Med* 27: 85–90, 2002.
- 57. Roepstorff K, Grandal MV, Henriksen L, Knudsen SLJ, Lerdrup M, Grøvdal L,
 Willumsen BM, van Deurs B. Differential Effects of EGFR Ligands on Endocytic
 Sorting of the Receptor. *Traffic* 10: 1115–1127, 2009.
- 58. **Roum JH, Buhl R, McElvaney NG, Borok Z, Crystal RG**. Systemic deficiency of glutathione in cystic fibrosis. *J Appl Physiol* 75: 2419–2424, 1993.
- 59. **Rushworth GF, Megson IL**. Existing and potential therapeutic uses for Nacetylcysteine: the need for conversion to intracellular glutathione for antioxidant benefits. *Pharmacol Ther* 141: 150–159, 2014.
- 60. Sahin U, Weskamp G, Kelly K, Zhou H-M, Higashiyama S, Peschon J, Hartmann D, Saftig P, Blobel CP. Distinct roles for ADAM10 and ADAM17 in ectodomain shedding of six EGFR ligands. *J Cell Biol* 164: 769–779, 2004.
- 61. Scheller J, Chalaris A, Garbers C, Rose-John S. ADAM17: a molecular switch to control inflammation and tissue regeneration. *Trends Immunol* 32: 380–387, 2011.
- 62. Schnúr A, Hegyi P, Rousseau S, Lukacs GL, Veit G. Epithelial Anion Transport as Modulator of Chemokine Signaling. *Mediators Inflamm* 2016: 7596531, 2016.
- 63. Sham D, Wesley UV, Hristova M, van der Vliet A. ATP-Mediated Transactivation of the Epidermal Growth Factor Receptor in Airway Epithelial Cells Involves DUOX1-Dependent Oxidation of Src and ADAM17. *PLoS ONE* 8: e54391, 2013.
- 64. **Singh B, Coffey RJ**. Trafficking of Epidermal Growth Factor Receptor Ligands in Polarized Epithelial Cells. *Annu Rev Physiol* 76: 275–300, 2014.
- 65. Singhal SS, Wickramarachchi D, Yadav S, Singhal J, Leake K, Vatsyayan R, Chaudhary P, Lelsani P, Suzuki S, Yang S, Awasthi YC, Awasthi S. Glutathione-Conjugate Transport by RLIP76 Is Required for Clathrin-Dependent Endocytosis and Chemical Carcinogenesis. *Mol Cancer Ther* 10: 16–28, 2011.
- 66. **Sisto M, Lisi S, D'Amore M, Lofrumento DD**. The metalloproteinase ADAM17 and the epidermal growth factor receptor (EGFR) signaling drive the inflammatory epithelial response in Sjo[°]gren's syndrome. *Clin Exp Med* 15: 215–225, 2015.
- 67. Sloane PA, Shastry S, Wilhelm A, Courville C, Tang LP, Backer K, Levin E, Raju

SV, Li Y, Mazur M, Byan-Parker S, Grizzle W, Sorscher EJ, Dransfield MT, Rowe SM. A Pharmacologic Approach to Acquired Cystic Fibrosis Transmembrane Conductance Regulator Dysfunction in Smoking Related Lung Disease. *PLoS ONE* 7: e39809, 2012.

- 68. Sosnay PR, Siklosi KR, Van Goor F, Kaniecki K, Yu H, Sharma N, Ramalho AS, Amaral MD, Dorfman R, Zielenski J, Masica DL, Karchin R, Millen L, Thomas PJ, Patrinos GP, Corey M, Lewis MH, Rommens JM, Castellani C, Penland CM, Cutting GR. Defining the disease liability of variants in the cystic fibrosis transmembrane conductance regulator gene. Nat Genet 45: 1160–1167, 2013.
- 69. Stolarczyk M, Amatngalim GD, Yu X, Veltman M, Hiemstra PS, Scholte BJ. ADAM17 and EGFR regulate IL-6 receptor and amphiregulin mRNA expression and release in cigarette smoke-exposed primary bronchial epithelial cells from patients with chronic obstructive pulmonary disease (COPD). *Physiol Rep* 4: e12878, 2016.
- Taipale M, Krykbaeva I, Koeva M, Kayatekin C, Westover KD, Karras GI,
 Lindquist S. Quantitative Analysis of Hsp90-Client Interactions Reveals
 Principles of Substrate Recognition. *Cell* 150: 987–1001, 2012.
- 71. **Takezawa K, Ogawa T, Shimizu S, Shimizu T**. Epidermal growth factor receptor inhibitor AG1478 inhibits mucus hypersecretion in airway epithelium. *Am J Rhinol Allergy* 30: 1–6, 2016.
- 72. **Tan X, Lambert PF, Rapraeger AC, Anderson RA**. Stress-Induced EGFR Trafficking: Mechanisms, Functions, and Therapeutic Implications. *Trends Cell Biol* 28: 352–366, 2016.
- 73. **Tomas A, Futter CE, Eden ER.** EGF receptor trafficking: consequencesfor signaling and cancer. *Trends Cell Biol* 24: 26–34, 2014.
- 74. Tosco A, De Gregorio F, Esposito S, De Stefano D, Sana I, Ferrari E, Sepe A, Salvadori L, Buonpensiero P, Di Pasqua A, Grassia R, Leone CA, Guido S, De Rosa G, Lusa S, Bona G, Stoll G, Maiuri MC, Mehta A, Kroemer G, Maiuri L, Raia V. A novel treatment of cystic fibrosis acting on-target: cysteamine plus epigallocatechin gallate for the autophagy-dependent rescue of class IImutated CFTR. *Cell Death Differ* 23: 1380–1393, 2016.
- 75. **Val S, Belade E, George I, Boczkowski J, Baeza-Squiban A**. Fine PM induce airway MUC5AC expression through the autocrine effect of amphiregulin. *Arch Toxicol* 86: 1851–1859, 2012.
- 76. Veit G, Avramescu RG, Chiang AN, Houck SA, Cai Z, Peters KW, Hong JS, Pollard HB, Guggino WB, Balch WE, Skach WR, Cutting GR, Frizzell RA, Sheppard DN, Cyr DM, Sorscher EJ, Brodsky JL, Lukacs GL. From CFTR biology toward combinatorial pharmacotherapy: expanded classification of cystic

fibrosis mutations. Mol Biol Cell 27: 424-433, 2016.

- Veit G, Bossard F, Goepp J, Verkman AS, Galietta LJV, Hanrahan JW, Lukacs
 GL. Proinflammatory cytokine secretion is suppressed by TMEM16A or CFTR
 channel activity in human cystic fibrosis bronchial epithelia. *Mol Biol Cell* 23: 4188–4202, 2012.
- 78. Velsor LW, van Heeckeren A, Day BJ. Antioxidant imbalance in the lungs of cystic fibrosis transmembrane conductance regulator protein mutant mice. *Am J Physiol Lung Cell Mol Physiol* 281: L31–8, 2001.
- 79. **Wang X-J, Feng C-W, Li M**. ADAM17 mediates hypoxia-induced drug resistance in hepatocellular carcinoma cells through activation of EGFR/PI3K/Akt pathway. *Mol Cell Biochem* 380: 57–66, 2013.
- 80. Whitesell L, Mimnaugh EG, De Costa B, Myers CE, Neckers LM. Inhibition of heat shock protein HSP90-pp60v-src heteroprotein complex formation by benzoquinone ansamycins: essential role for stress proteins in oncogenic transformation. *Proc Natl Acad Sci U S A* 91: 8324–8328, 1994.
- Willems SH, Tape CJ, Stanley PL, Taylor NA, Mills IG, Neal DE, McCafferty J, Murphy G. Thiol isomerases negatively regulate the cellular shedding activity of ADAM17. *Biochem J* 428: 439–450, 2010.
- 82. Xu Y, Liu G, Yang D, Fu L, Zhang M, Zhang Z, Xia H, Yao S, Zhang S. Classically Activated Macrophages Protect against Lipopolysaccharide-induced Acute Lung Injury by Expressing Amphiregulin in Mice. *Anesthesiology* 124: 1086– 1099, 2016.
- 83. Zaiss DMW, Gause WC, Osborne LC, Artis D. Emerging Functions of Amphiregulin in Orchestrating Immunity, Inflammation, and Tissue Repair. *Immunity* 42: 216–226, 2015.
- 84. **Zhang H, Davies KJA, Forman HJ**. TGFβ1 rapidly activates Src through a noncanonical redox signaling mechanism. *Arch Biochem Biophys* 568: 1–7, 2015.
- 85. Zhang Y, Hegen M, Xu J, Keith JC, Jin G, Du X. Characterization of (2R, 3S)-2-({[4-(2-butynyloxy) phenyl] sulfonyl} amino)-N, 3-dihydroxybutanamide, a potent and selective inhibitor of TNF-α converting enzyme. *Int Immunol* 4: 1845–1857, 2004.
- Zhang Y, Xu J, Levin J, Hegen M, Li G. Identification and Characterization of 4-[[4-(2-Butynyloxy)phenyl]sulfonyl]-*N*-hydroxy-2,2-dimethyl-(3*S*)thiomorpholinecarboxamide (TMI-1), a Novel Dual Tumor Necrosis Factor- -Converting Enzyme/Matrix Metalloprotease Inhibitor for the Treatment of Rheumatoid Arthritis. *J Pharmacol Exp Ther* 309: 348–355, 2004.
- 87. Zhang Y, Zhu M, Yang Z, Pan X, Jiang Y, Sun W, Wang Q, Xiao W. The human Cathelicidin LL-37 induces MUC5AC mucin production by airway epithelial cells via TACE-TGF-α-EGFR pathway. *Exp Lung Res* 40: 333–342, 2014.
- Zhou Y, Lee JY, Lee CM, Cho WK, Kang MJ, Koff JL, Yoon PO, Chae J, Park HO, Elias JA, Lee CG. Amphiregulin, an epidermal growth factor receptor ligand, plays an essential role in the pathogenesis of transforming growth factor-β-induced pulmonary fibrosis. J Biol Chem 287: 41991–42000, 2012.
- Ziady AG, Hansen J. Redox balance in cystic fibrosis. Int J Biochem Cell Biol 52: 113–123, 2014.
- 90. **Zwang Y, Yarden Y**. p38 MAP kinase mediates stress-induced internalization of EGFR: implications for cancer chemotherapy. *EMBO J* 25: 4195–4206, 2006.

human disease CF lung COPD airway IL68 Inflammation AREG remodeling FGER shot fibrosis PBEC human disease CF lung COPD airway IL68 Infla**CHAPTER** 4 ADAM17 fibrosis PBEC human disease CF lung COPD airway IL5R inflammation AREG remodeling SAGER shedding A G remodeling Recom AREG rem op & inflammation OPD airwa PBEC hu 8 📲 🖓 🖓 🖓 🖓 lung COPD airway IL6R Inflammation AREG remodeling EGFR shedding ADAM17 fibrosis PBEC ase CF lung CORD airway JL68 inflammat fibrosis PBEC human dis Correction of lung inflammation in a F508del CFTR murine cystic fibrosis model by the sphingosine-1-phosphate lyase inhibitor LX2931. BEC human disease CF lung COPD airway IL6R inflammation shedding ADAM17 fibrosis PBEC human disease CF lung COPD airwa (published – Am J Physiol Lung Cell Mol Physiol) Veltman M, Stolarczyk M, Radzioch D, Wojewodka G, De Sanctis JB, Dik WA, ADAM17 fibrosis PBEC human di Dzyubachyk O, Oravecz T, de Kleer I, Scholte BJ della shedding ADAM17 fibrosis PBEC human disease CF lung COPD airway IL6R inflammation AREG rem

CALL FOR PAPERS | Biomarkers in Lung Diseases: from Pathogenesis to Prediction to New Therapies

Correction of lung inflammation in a F508del CFTR murine cystic fibrosis model by the sphingosine-1-phosphate lyase inhibitor LX2931

Mieke Veltman,¹ Marta Stolarczyk,¹ Danuta Radzioch,² Gabriella Wojewodka,² Juan B. De Sanctis,³ Willem A. Dik,⁴ Oleh Dzyubachyk,⁵ Tamas Oravecz,⁶ Ismé de Kleer,^{7,8} and Bob J. Scholte¹

¹Cell Biology, Erasmus MC, Rotterdam, The Netherlands; ²Departments of Medicine and Human Genetics, McGill University, Montreal, Canada; ³Faculty of Medicine. Universidad Central de Venezuela, Institute of Immunology, Caracas, Venezuela; ⁴Immunology, Erasmus MC, Rotterdam, The Netherlands; ⁵Department of Radiology, Division of Image Processing, Leiden University Medical Center, Leiden, The Netherlands; ⁶Lexicon Pharmaceuticals, Inc., The Woodlands, Texas; ⁷Department of Pediatrics, Division of Respiratory Medicine, Erasmus MC, Rotterdam, The Netherlands; and ⁸Laboratory of Pulmonary Medicine, Erasmus MC, Rotterdam, The Netherlands

Submitted 14 July 2016; accepted in final form 19 September 2016

Veltman M, Stolarczyk M, Radzioch D, Wojewodka G, De Sanctis JB, Dik WA, Dzyubachyk O, Oravecz T, de Kleer I, Scholte BJ. Correction of lung inflammation in a F508del CFTR murine cystic fibrosis model by the sphingosine-1-phosphate lyase inhibitor LX2931. Am J Physiol Lung Cell Mol Physiol 311: L1000-L1014, 2016. First published September 23, 2016; doi:10.1152/ajplung.00298.2016.-Progressive lung disease with early onset is the main cause of mortality and morbidity in cystic fibrosis patients. Here we report a reduction of sphingosine-1-phosphate (S1P) in the lung of unchallenged Cftr^{ImIEUR} F508del CFTR mutant mice. This correlates with enhanced infiltration by inducible nitric oxide synthase (iNOS)expressing granulocytes, B cells, and T cells. Furthermore, the ratio of macrophage-derived dendritic cells (MoDC) to conventional dendritic cells (cDC) is higher in mutant mouse lung, consistent with unprovoked inflammation. Oral application of a S1P lyase inhibitor (LX2931) increases S1P levels in mutant mouse tissues. This normalizes the lung MoDC/cDC ratio and reduces B and T cell counts. Lung granulocytes are enhanced, but iNOS expression is reduced in this population. Although lung LyC6+ monocytes are enhanced by LX2931, they apparently do not differentiate to MoDC and macrophages. After challenge with bacterial toxins (LPS-fMLP) we observe enhanced levels of proinflammatory cytokines TNF- α , KC, IFN γ , and IL-12 and the inducible mucin MUC5AC in mutant mouse lung, evidence of deficient resolution of inflammation. LX2931 does not prevent transient inflammation or goblet cell hyperplasia after challenge, but it reduces MUC5AC and proinflammatory cytokine levels toward normal values. We conclude that lung pathology in homozygous mice expressing murine F508del CFTR, which represents the most frequent mutation in CF patients, is characterized by abnormal behavior of infiltrating myeloid cells and delayed resolution of induced inflammation. This phenotype can be partially corrected by a S1P lyase inhibitor, providing a rationale for therapeutic targeting of the S1P signaling pathway in CF patients.

cystic fibrosis; mouse model; dendritic cells; macrophages; granulocytes

Address for reprint requests and other correspondence: B. J. Scholte, Erasmus MC, Rotterdam, Cell Biology Ee1034, PO Box 2040, 3000CA Rotterdam, NL (e-mail: b.scholte@erasmusmc.nl).

CYSTIC FIBROSIS (CF) is the most common recessive congenital disease in Caucasians, affecting ~75,000 patients worldwide. It is caused by a deficiency of the CF transmembrane conductance regulator (CFTR) involved in transcellular ion and fluid transport and characterized by severe progressive pancreatic, intestinal, and lung pathology. Progressive distal airway disease, causing severe morbidity, is already observed in a majority of CF infants younger than 5 years (63, 68). This is accompanied by airway remodeling, chronic inflammation and bacterial infections, frequent exacerbations, and eventually fatal loss of lung function. The molecular mechanisms involved in the development of CF lung disease are not completely understood despite intensive research in the past two decades. Experiments in cellular and animal models have suggested that CFTR dysfunction affects a network of interrelated regulatory signaling molecules, including growth factor shedding and cytokine secretion, which regulates cell fate decisions, inflammation, and tissue remodeling in the affected tissues. An effective therapy for a majority of CF patients is not available, despite recent progress with CFTR-targeted pharmaceutics. Therefore additional or alternative intervention with anti-inflammatory drugs should be considered (76).

Abnormal bioactive lipid metabolism in CF lung disease. Previous studies in CF humans and CF mutant mice have highlighted the involvement of various receptor-modulating bioactive lipids in the development of CF lung pathology. In particular the sphingosine pathway (77, 84) and phospholipase A-dependent signaling pathways (18) are affected and are likely involved in CF pathology through regulation of inflammation, airway plugging, and tissue remodeling. Several pharmaceutical interventions in these pathways were effective in animal models, and clinical trials are underway, highlighting the potential of anti-inflammatory treatment in addition to CFTR targeted therapeutics (7, 18, 22, 24, 76, 86).

The role of S1P in CF lung disease. In the present study we focus on sphingosine-1-phosphate (S1P), a receptor-activating lipid that is involved in several cellular processes that are crucial for the development of lung disease. S1P is produced by intracellular kinases (S1PK1 and S1PK2) from sphin-

1040-0605/16 Copyright © 2016 the American Physiological Society

gosines. S1P lyase (SPL) removes intracellular S1P from the sphingolipid pool and is considered an important drug target for cancer, inflammatory, and autoimmune disease (31, 32). The food colorant 2-acetyl-4-tetrahydroxy butylimidazole induces lymphopenia and effectively reduces SPL activity in mice (65). The orally available chemical derivative SPL inhibitor LX2931 (LX3305) reduced pathology in a mouse model of rheumatoid arthritis (5), was well tolerated, and showed clinical benefit in a phase II clinical trial (37).

S1P is transported to the extracellular compartment by ABC transporters and the major superfamily member Spinster 2 (19), which allows autocrine and paracrine signaling to a family of differentially expressed extracellular receptors $(S1P_{1-5})$, followed by activation of intracellular signaling pathways. S1P also acts on intracellular targets (1, 49, 60) and is involved in NFkB activation (4, 49). In cross talk with other signaling pathways, notably ceramides (77), S1P has important effects on cell migration, growth, and differentiation and plays a role in tissue repair. The importance of S1P metabolism in inflammatory lung disease and fibrosis has been documented in several publications in murine (50), rat (40), and canine models of acute lung injury (71). In a recent report Xu et al. (85) reported reduced S1P levels in bronchoalveolar lavage fluid (BALF) of CFTR knockout (KO) mice, correlating with abnormal dendritic cell infiltration.

In the present study we have confirmed and substantially extended these results using a unique mouse model (*Cftr^{im1EUR}*) for the most common mutation in CF patients (F508del CFTR), generated in our laboratory (21). These mice display electrophysiological abnormalities and intestinal disease typical of CF (82). In the lung, enhanced lung mucin secretion (82), abnormal lung inflammation and fibrotic response to bleomycin (27, 45), excessive response to LPS and *Pseudomonas* infection (48), and naphthalene injury (11) were observed. Furthermore, lung epithelial cells in *Cftr^{im1EUR}* mice show deficient autophagocytosis and accumulation of aggresomes, reportedly mediated by excessive production of reactive oxygen (42), which could affect the activity of key enzymes in S1P metabolism or intracellular S1P receptor systems.

Our data show that abnormal myeloid infiltration and delayed resolution of induced inflammation in *Cftr^{tm1EUR}* mice correlate with reduced lung tissue S1P levels, which can be partially corrected by chronic treatment with the SPL inhibitor LX2931.

METHODS

Animal studies. Breeding, maintenance, and genotyping of *Cftr^{im1EUR}* F508del CFTR mice (F12 C57Bl/6 backcross) was performed as described previously (21, 81, 82). Mice were kept in sterile ventilated cages, with sterile normal chow and water ad libitum. Adult males and females were used for intervention studies. All experiments were performed with approval (DEC 138-11-09) of the Independent Committee on Ethical Use of Experimental Animals, Rotterdam, according to national and European Union guidelines. Oral LX2931 treatment was performed by adding 500 mg/l LX2931, obtained from Lexicon Pharmaceuticals, to the drinking water for the period indicated.

Salivary gland activity measurement (8) was performed as described in Ref. 82. Briefly, in mice under ketamine/xylasine/diazepam anesthesia, basal cholinergic secretion activity is inhibited by subcutaneous injection of atropine to the cheek (50 μ l, 1 mM), and saliva

is collected at 3-min intervals in 2 \times 20-mm filter paper, measured by weighing the filter before and after collection. Adrenergic (CFTR-dependent) secretion is induced by subcutaneous injection of 50 μ l, 1 mM atropine 50 μ l, 100 μ M isoproterenol and followed for at least four further 3-min intervals.

Gated microCT lung scanning was performed on a Quantum FX CT scanner (Perkin Elmer), under isoflurane anesthesia. CT scans were viewed and analyzed with Caliper Analyze software. Scoring of CT scans was performed in a five-grade binning mode by measuring the relative surface of the lung volume occupied by high-density areas, indicating focal inflammation as illustrated in Fig. 8.

Histology and quantitative analysis. Animals were terminated with a lethal dose of ketamine/xylazine/diazepam, blood was removed from the circulation by ventricle perfusion with PBS. Lungs lobes and other tissues were removed immediately and either fixed in 4% paraformaldehyde/PBS and processed for standard paraffin histology or snap frozen in liquid nitrogen for analysis. Hematoxylin-eosin, periodic acid-Schiff (PAS)/Alcian blue (AB), and immunohistology staining were performed using standard procedures. Quantification of alveolar space sizes and septum thickness was performed using a modification of the method described in Ramnath et al. (58). After illumination correction, the final air space/septum masks are obtained from the binarized channel images, and air spaces smaller than 300 pixels $(31 \ \mu m^2)$ are removed. For measuring the septum thickness, the smallest distance between pairs of neighboring air spaces was used. The mean value of the aggregated vector of the pairwise distances defines a single septum thickness measure for the entire image. For defining the neighborhood relation between the air spaces, skeletonization of the identified septum region is performed. Consequently, the entire image domain is subdivided into connected regions separated by the obtained skeleton. Since each of the obtained regions contains only one air space, both the air space and the region that contains it are assigned the same unique label. Two air spaces are considered as neighboring if the regions containing them are separated by a single skeleton edge.

Cytokine analysis by sequential ELISA was performed essentially as described by Van den Berg et al. (79). We analyzed cytokines in total lung lysates made from the unperfused upper right lung lobe. The total lung lysates were obtained by homogenizing the lung lobe, two times in the TissueLyser II (Qiagen) for 90 s at 30 Hz, in ice-cold PBS with protease inhibitor cocktail (Complete Mini, Roche), and stored at -80°C for further analysis. Cytokines were analyzed in total lung lysates by Sequential Enzyme-Linked Immuno Sorbent Assay (ELISA) (R&D Systems Europe, Abingdon, UK), and 96-well microtiter plates (Nunc Immunoplate, Neptune, NJ) were coated with anti-cytokine antibody (Duoset, R&D Systems, Minneapolis, MN) diluted 1,000× in PBS (50 $\mu l/well)$ overnight rotating at 300 rpm at 4°C. Plates were washed with PBS 0.05% Tween 20, incubated for 1 h with blocking solutions: Blocker Blotto in TBS (eotaxin, MCP-1), Blocker Casein in PBS (KC, IL-6, IL-10, TNF-SR2, TNF-SR1, IL-1ra, IL-6R, IL-P70, MIP-2, IFNy), and Super Blocking Buffer in PBS (TNF-a, IL-1β) (150 µl/well, Pierce, Rockford, IL) at room temperature. After washing, the lung lysate samples diluted in PBS with 0.05% Tween 20 and standards (Duoset, R&D Systems) were incubated for 2 h at room temperature on a plate shaker at 300 rpm. Then the samples were removed and stored in a 96-deep-well polypropylene microtiter plate for further sequential use. After transfer, the plates were washed and the polyclonal biotinylated detection antibody in dilution buffer was added (Duoset, R&D Systems) (50 µl/well, 300 rpm). The plates were washed and incubated for 30 min with streptavidin-conjugated horseradish peroxidase (Bio-Rad), diluted 1:10,000 in PBS with 0.1% BSA and 0.05% Tween 20 (50 µl/well, 300 rpm), followed by 10-40 min incubation with the TMB peroxidase substrate system (KPL, Gaithersburg, MD) (1:1, 100 µl/well). The reaction was stopped with 1.0 M sulfuric acid (100 µl/well), and absorbance was measured using a plate reader (Bio-kinetics, BioTek) at 450 nm. Data are expressed as picograms

per milligram protein. Protein concentration in lung tissue samples were determined by BCA kit (Pierce).

SIP and sphingosine measurements. Lipids extracted from the unperfused middle lung lobe of *Cftr^{tm1EUR}* mutant animals and littermate controls aged 33-35 wk kept on normal chow were purified similarly as described previously (9). Briefly, tissues are spiked with internal standards, homogenized in organic solvent [chloroform:methanol 2:1 (vol/ vol) and butylated hydroxyl anisole]. One volume of cold water is added to the mixture and samples are mixed for 90 min at 4°C. The organic phase is collected, to which diethyl ether is added to remove protein contamination. The organic phase is dried and resuspended in 100 µl of chloroform, dried and resuspended in petroleum ether, purified by thin-layer chromatography (TLC), and detected by iodine. After scraping from the TLC plate, the amount of recovered lipids per sample was normalized according to the protein content in the initial samples measured via the bicinchoninic assay and then by normalization to same amount of inorganic phosphate in each of the analyzed samples. Final amounts were expressed in picomoles per nanomole of total phospholipid phosphate present in each analyzed sample. Contents of S1P and sphingosine in the samples were analyzed on a Thermo-Finnigan Trace DSQ fast-scanning single-quadrupole mass spectrometer using electron impact ionization. The instrument was tuned and calibrated for mass resolution and mass accuracy on a daily basis. Accurate mass determination and MS/MS fragmentation (LC/MS and LC/MS/MS): the LC/MS portion of the platform was based on a Waters ACQUITY UPLC and a Thermo-Finnigan LTQ-FT mass spectrometer, which had a linear ion-trap front end and a Fourier transform ion cyclotron resonance mass spectrometer back end. For ions with counts greater than 2 million, an accurate mass measurement could be performed. Accurate mass measurements could be made on the parent ion as well as fragments. The typical mass error was less than 5 ppm. Ions with less than 2 million counts require a greater amount of effort to characterize. Fragmentation spectra (MS/MS) were typically generated in a data-dependent manner. S1P measurements in unperfused lung tissues from LX2931-treated mutant mice were performed by LC-MS at Lexicon Pharmaceuticals laboratories, using procedures described previously (5).

Flow cytometry. For flow cytometry we obtained anti-mouse Ly6C-FITC (clone AL-21), SiglecF-PE (clone E50-2440), Ly6G-PE (clone 1A8), CD103-PE (clone E50-2440), CD19-APC_ Cy7 (clone ID3), and CD64-AF647 (clone X54-5/7.1) from Becton Dickinson (BD). Anti-mouse CD11b- Horizon450 (clone M1/70), CD11c- PE_Cy7 (clone N418), CD3-APC_Cy7 (clone I45-2c11) and MHCII-APC_Cy7 (clone I3/2.3), and inducible nitric oxide synthase (NOS2/iNOS) PE-cyanine7 (clone CXNFT) were obtained from eBioscience. Anti-mouse CD45-PE_TxRed (clone M5-114.5) was obtained from BioLegend, and arginase-1-FITC (polyclonal sheep IgG) from R&D Systems. The life-dead marker Aqua-AmCyan was obtained from Invitrogen. Lungs were digested using RPMI containing Liberase TM (Roche) and 10 U DNAse (Roche). Cell suspensions were lysed for 4 min at room temperature in 1 ml osmotic lysis buffer. All staining reactions were performed at 4°C for 30 to 45 min. Cells were incubated with 2.4G2 Fc receptor antibody to reduce nonspecific binding, together with monoclonal antibodies. To facilitate intracellular staining of arginase-1 and iNOS, cells were fixed and permeabilized using the Fixation/ Permeabilization Solution Kit of BD Biosciences. Cell data acquisition took place on a LSRII flow cytometer (BD). Gating strategy of CD45+ cells was as follows: granulocytes (CD11b+Ly6G+ Ly6C-), macrophages (CD11c+CD11b-MHCII- SiglecF+); Ly6C^{high} monocytes (CD11c-MHCII-CD11b+LyC6high+); B cells (CD19+ MHCII+ CD11c-); T cells (CD3+ MHCII-). Dendritic cells (DC, CD11c+MHCII+); conventional DC (cDC: CD11c+ MHCII+ CD11b+ CD64-); monocyte-derived DC (MoDC: CD11c+ MHCII+ CD11b+CD64+); CD103+ DC (CD11c+ MHCII+ CD11b-).

RESULTS

Reduced sphingosine-1-P in $Cftr^{tm1EUR}$ mouse lung. We analyzed lipid profiles by gas chromatography-mass spectrometry (GC-MS) in whole lung tissue from adult male mutant C57Bl/6 $Cftr^{tm1EUR}$ F508del CFTR mice. Lipids extracted from the unperfused middle lung lobe of mutant animals (n = 10) and littermate controls (n = 8) aged 33–35 wk, kept on normal chow, were analyzed in parallel as described in METHODS. We observed a reduction of total sphingosine species and in particular a threefold reduction of S1P in the lungs of unchallenged adult $Cftr^{tm1EUR}$ mutant mice compared with wild-type (Fig. 1).

Abnormal infiltration of myeloid cells and DC imbalance in F508del CFTR mouse lung. The reduction of lung S1P homozygous mutants in this strain correlates with enhanced mucus production, low-level basal lung inflammation, and enhanced response to bleomycin, lung injury, and Pseudomonas aeruginosa infection (27, 42, 45, 48, 82), suggesting a causal relationship between abnormal S1P tissue levels and CF lung pathology. Therefore, we investigated the effect of the SPL inhibitor LX2931 on basal and induced inflammation in this model. First, we established the basal myeloid cell profile in total lung by FACS analysis. The total number of CD45⁺ cells in F508del mutant mice is significantly higher than in wild-type littermates (Fig. 2A). This can be attributed mainly to a significant increase in total granulocytes (Fig. 2B) and to a lesser extent alveolar macrophages (Fig. 2C) and Ly6C^{high} monocytes (Fig. 2D). The total number of B cells and T cells is increased in mutant mice compared with wild-type, adding substantially to the total number of CD45 + cells (Fig. 2, E and F). The total number of DCs is higher on average but not significantly different from normal (Fig. 3A). Importantly, however, we show here for the first time that the percentage of CD11b+ cDC is reduced in F508del CFTR mutant mouse lung compared with normal (Fig. 3B), whereas we observe a significant increase in the percentage CD64+ MoDC, also known



Fig. 1. Sphingosine and sphingosine-1-phosphate (Sphingosine 1P) in F508del CFTR mouse lung. Lipids were extracted from total unperfused middle lung lobes from homozygous mutant mice (dd, n = 10) and wild-type littermates (WT, n = 8) aged 33–35 wk, kept on normal chow, and analyzed by LC-MS as described in METHODS. Data are shown as pmol/nmol total lipid phosphate. Average \pm SD; **P < 0.01, unpaired *t*-test.



Fig. 2. Abnormal myeloid infiltration in F508del CFTR mutant mouse lung. Total lung from wild-type (WT; n = 12), mutant (dd; n = 15), and mutant mice treated with SP-1 lyase inhibitor LX2931 in drinking water (dd LX; n = 6) were analyzed by FACS to quantitate myeloid subpopulations as described in METHODS. Total cell counts per lung (average \pm SE) are shown for myeloid cells (CD45+; A), granulocytes (CD11b+Ly6G+ Ly6C-; B), macrophages (CD11c+CD11b-MHCII-SiglecF+; C) LyC6^{high} monocytes (CD11c-MHCII-CD11b+LyC6+; D), B cells (CD19+MHCII+CD11c-; E), and T cells (CD3+MHCII-; F). **P < 0.01, *P < 0.05; 1-way ANOVA.

as inflammatory DC (Fig. 3*C*). In contrast to Xu et al. (85) in CFTR KO mice, we did not observe reduced CD103+ DC in our F508del CFTR mutant mice, either total numbers (not shown) or as a percentage of DC (Fig. 3*D*).

We conclude that unchallenged adult mutant *Cftr^{im1EUR}* mice under our conditions show enhanced myeloid cell infiltration and an increased MoDC/cDC ratio, compared with wild-type littermates, consistent with sterile lung inflammation, which correlates with reduced S1P levels in the lung.

An S1P lyase inhibitor corrects abnormal MoDC/cDC ratio in F508del CFTR mutant. To establish the role of S1P in this hyperinflammatory phenotype of mutant mice, and the therapeutic potential of pharmaceutical intervention in this pathway, we treated adult F508del CFTR mutant mice with the SPL inhibitor LX2931 in drinking water for 2 wk. As expected (5), this treatment caused a near-hundredfold increase in tissue S1P levels in total lung tissue measured by LC-MS [0.3 vs. 55 ng/mg, homozygous mutant (dd) vs. dd treated with LX2931,



Fig. 3. Abnormal dendritic cell population in F508del CFTR mutant mouse lung. Lung homogenates from wild-type (WT; n = 12), mutant (dd; n = 15), and mutant mice treated with SP-1 lyase inhibitor LX2931 in drinking water (dd LX; n = 6) as described in Fig. 2 were analyzed for markers of dendritic cells (DC). A: total DC per lung (CD11c+MHCII+). B: percentage monocyte-derived DC of total DC (MoDC, CD11c+ MHCII+ CD11b+ CD64+). C: percentage conventional DC (cDC, CD11c+ MHCII+ CD11b+ CD64-). D: percentage CD103+ DC (CD103+ CD11C+ MHCII+ CD11b-). Average \pm SE, **P < 0.01; *P < 0.05, 1-way ANOVA.

n = 10 in each group, P < 0.001]. Total white blood cell counts were reduced by LX2931 treatment in mutant mice $(4.36 \pm 0.45 \text{ vs. } 2.63 \pm 0.3 \times 10^9/\text{l}$, dd vs. dd treated with LX2931, n = 10 in each group, P < 0.01), no morbidity or mortality was observed in any of the LX2931-treated mutant animals. These data are comparable to those previously reported in wild-type mice with this compound (5). White and red blood cells in untreated mutant mice were not significantly different compared with normal littermates $(3.9 \pm 0.5, n =$ 10). Red blood cells in LX2931-treated mutant mice were comparable with normal and untreated mutant.

FACS analysis of myeloid cells from the lungs of LX2931treated mutant animals in parallel with untreated animals revealed a complex response. First, we observe no effect on total CD45+ (Fig. 2A) or total DC cell counts (Fig. 3A) in the lung. However, we see an increase in cDC and a concomitant reduction of MoDC. Thus LX2931 corrects the abnormal MoDC/cDC composition in mutant mice to wild-type levels (Fig. 3, *B* and *C*). Furthermore, LX treatment reduced the percentage of CD103+ DC (Fig. 3D), which are considered important regulators of the airway immune response (16, 34).

Remarkably, cells gated as granulocytes and Ly6C+ monocytes are significantly increased by LX2931 (Fig. 2, B and D). Consistent with this increase, the myeloid attractant MCP-1 was moderately enhanced in lung tissue of mutant mice by LX2931 treatment, but not MIP-2, eotaxin, IFN γ , and IL12p70 (Fig. 4).

The LX2931-mediated influx of granulocytes and Ly6C+ monocytes may represent myeloid-derived suppressor cells (MDSC), defined as CD11b+ Gr1/Ly6G+ myeloid cells. These cells are considered anti-inflammatory by suppressing NK cell activity and T cell proliferation (59, 66). In a recent study, Liu et al. (39) have shown that MDSC recruitment to the concanavalin A-induced inflamed liver in mice is enhanced by the S1P receptor modulator FTY720, reducing hepatic injury. Therefore, we stained in FACS analysis CD45+ cells from total lungs of F508del CFTR mutant mice, untreated and treated with LX2931 (n = 6 in each group) for the MDSC markers arginase (ARG1) and iNOS/NOS2. However, we did not observe staining for ARG1 in any of the myeloid subpopulations, including Ly6C+ monocytes, ruling out significant recruitment of MDSC in response to LX2931 (data not shown).

Interestingly, we observed a distinct iNOS+ granulocyte population in unchallenged mutant mice (Fig. 5), which was not detected in normal mice (n = 3, not shown). This confirms basal lung inflammation in mutant mice. This iNOS+ granulocyte population in mutant mice was significantly reduced by LX2931 treatment (Fig. 5), suggesting a reduction of lung inflammation.



Fig. 4. Total lung cytokines in unchallenged F508del mutant and LX2931-treated mice. Total lung extracts of untreated adult mutant mice (dd) or treated with LX2931 (dd LX), compared with wild-type littermates (WT), were analyzed in parallel by sequential ELISA as described in METHODS. Data are expressed as pg/mg protein. Average \pm SE, **P < 0.01; *P < 0.05, 1-way ANOVA.

Furthermore, LX2931 treatment caused a strong reduction of the enhanced B cells and T cells counts in mutant lung, even below values observed in normal mice (Fig. 2, E and F). The relevance of these results for CF pathology will be discussed below.

S1P lyase inhibitor does not recover F508del CFTR epithelial fluid transport activity. It has been postulated that inflammation and associated cellular stress reduces the efficiency of F508del CFTR trafficking or conversely that inefficient folding of mutant CFTR causes oxidative stress and defective membrane trafficking of murine and human F508del CFTR (14, 43). S1P has also been linked to inflammation, autophagy, and oxidative stress (26, 55). Therefore, we considered that modulation of S1P signaling could have an effect on F508del CFTR activity. We have investigated this in a separate experiment using several independent assays.

Consistent with previous data, adult (80–115 days) untreated male and female mutant mice were on average 10% underweight (Table 1) compared with age- and sex-matched wild-type littermates, primarily due to enterocyte CFTR deficiency (82). Two weeks of LX2931 treatment does not significantly affect the increase in body weight compared with untreated mutant mice. This confirms that LX2931 treatment does not negatively affect the health of the mice, and also that no major rescue of intestinal CFTR activity occurs under our conditions. Isoproterenol-induced salivary gland fluid secretion activity is severely affected in F508del CFTR mutant mice (82). LX2931-treated mutant mice on average did not show a



Fig. 5. SP-1 lyase inhibitor (LX2931) reduces iNOS expression in granulocytes from F508del CFTR mouse lung. In a separate experiment, lungs from unchallenged adult F508del CFTR mutant mice (n = 6) were analyzed by FACS, as in Figs. 2 and 3. A: percentage of granulocytes gated as iNOS+ were threefold higher in untreated mutant mice compared with mutant mice treated with LX2931 (**P < 0.01, unpaired *t*-test). B: representative scatter graphs of untreated mutant (dd) and LX2931-treated mutant (dd LX) show the distinct iNOS+ granulocyte subpopulation, which is absent in normal littermates (n = 3) and isotype control (not shown).

significant increase in isoproterenol-induced fluid secretion (Fig. 6) Furthermore, we report here for the first time frequent plugging of ducts in the parotic salivary gland with mucous material in F508del CFTR homozygous mice, which is virtually absent in wild-type mice. This is not observed in the submandibular and sublingual glands. Treatment with LX2931 for 3 wk does not significantly affect parotic duct plugging in mutant mice (Fig. 7).

We conclude that increasing S1P tissue levels by LX2931 treatment does not significantly affect F508del CFTR channel activity and associated fluid transport in secretory epithelia in this model. Changes affecting the behavior of myeloid cells cannot be excluded, however (10, 35), and are the subject of current studies.

S1P lyase inhibitor does not interfere with bacterial toxininduced myeloid infiltration in F508del CFTR mice. To establish whether LX2931 has an effect on bacterial toxin-induced inflammation in mutant mice, we treated mutant and wild-type mice intranasally with bacterial liposaccharides (LPS) in combination with the modified peptide fMLP, that is required for full polymorphonuclear leukocytes activation and neutrophil elastase secretion (13).

Mutant mice, both untreated and LX2931 treated, respond to LPS/fMLP with a substantial weight loss (10-20%) 48 h after

challenge, whereas wild-type mice suffered only a minor change (Table 1). Two of 13 untreated mutant animals did not survive the first week after treatment; no mortality was observed in the LX2931-treated group. One week after challenge, recovery of body weight is not complete in untreated and LX2931-treated mutant animals, in contrast to normal (Table 1). We conclude that F508del CFTR mutant mice are more susceptible to LPS/fMLP induced weight loss than wild-type, in agreement with previous findings using LPS only and *Pseudomonas aeruginosa* challenge (48). Furthermore, LX2931 treatment does not prevent this aspect of acute LPS-induced pathology.

Transient lung inflammation was monitored by microCT scanning before, 48 h after and 1 wk after challenge (Fig. 8). In all LPS/fMLP-treated animals we observed high-density areas covering 0–40% of the left and right lung lobes, indicating large areas of intense focal inflammation. In unchallenged mutant and wild-type animals this was not observed (not shown). Using a five-grade scoring scheme, microCT scores were high in all groups at 48 h (Fig. 8). One week after challenge, no remaining foci were observed in the wild-type group, whereas in a minority of the of the untreated and LX2931-treated mutant animals low but detectable focal microCT scores were find the statement of the untreated and LX2931-treated mutant animals low but detectable focal microCT scores were still present. Thus by this method we find

Table 1. Effect of LPS/fMLP and LX2931 on mouse body weight

55	5 5	<i>y y y</i>				
	Ν	Age, days (SD)	Initial weight, g (SD)	% Weight change 2 wk	% Weight change LPS 48 h	% Weight change LPS 7 days
dd female	7	81.4 (23.5)	18.3* (2.3)	4.2 (6.6)	-8.0* (8.9)	-0.6 (3.2)
dd male	6	112.5 (46.3)	22.3* (1.2)	9.4 (11.0)	-13.2*(8.2)	-8.2*(2.1)
dd female LX	5	90.2 (25.8)	17.8* (3.7)	15.9 (7.6)	-19.1*(2.7)	-12.9*(8.2)
dd male LX	7	90.2 (25.8)	22.8* (1.9)	9.8 (4.4)	-11.2*(5.3)	-12.9*(8.2)
WT female	12	95.0 (19.5)	20.9 (0.9)	4.3 (4.7)	0.5 (6.4)	0.1 (8.7)
WT male	8	115.6 (47.4)	25.8 (2.0)	5.0 (4.8)	-0.6 (6.3)	0.7 (4.5)

Initial body weight of age- and sex-matched F508del mutant (dd) and normal littermates (WT) were monitored in parallel. Data are presented are presented as average (\pm SD). A subset of the dd mice were treated with LX2931 in the drinking water, as described in METHODS. After 2 wk the percentage change in body weight was recorded (% weight change 2 wk). Subsequently all mice were treated intranasally with LPS/fMLP, and the change in body weight was determined after 48 h and 1 wk. Untreated male and female mutant mice are ~10% underweight compared with age- and sex-matched littermates, consistent with previous data, primarily due to intestinal disease. Two-week LX2831 treatment does not significantly affect average body weight of dm mice. LPS/fMLP challenge causes a reduction of body weight in mutant but not in normal animals after 48 h, but no mortality (*P < 0.01 compared with sex-matched WT, ANOVA-Tukey). One week after challenge, we observe an incomplete recovery of body weight in mutant mice. Treatment with LX2931 does not significantly affect this response. This suggests that LX2831 does not substantially correct mutant CFTR activity, nor does it have adverse effects during basal of challenge conditions.



Fig. 6. Salivary gland activity in LX2931-treated F508del mutant mice is not affected. LX2931-treated mutant mice (dd LX) did not show a significant increase in cAMP-induced salivary gland activity (mg·min⁻¹·kg⁻¹) compared with untreated mutant (dd), which is severely affected compared with wild-type littermates (WT) in F508del CFTR mice (1-way ANOVA, Tukey's test) (82). Isoproterenol-induced salivary gland fluid secretion was measured in parallel in adult mutant mice that were untreated (dd, n = 6) or treated with LX2931 (dd LX n = 7), compared with wild-type littermates (WT, n = 4). Data are expressed as mg·min⁻¹·kg⁻¹ body wt (average ± SE).

no evidence for a reduction of focal myeloid infiltration by LX2931 in mutant animals either in acute phase or during resolution.

Whereas focal infiltration was transient, an increase in thickness and apparent density of the septa of the alveolar region is observed in the microCT scans of all experimental groups 1 wk after challenge (not shown). This is confirmed by quantitative histology of all mice 2 wk after challenge. Inflammatory scores for septal infiltration, focal inflammation, and luminal monocytes were high compared with unchallenged in all experimental groups. In quantitative analysis of paraffin sections, septum thickness is also increased by LPS/fMLP challenge in mutant and wild-type mice compared with untreated (Fig. 9). Remarkably, this parameter is significantly higher in LX2931-treated mutant mice compared with untreated mutant, suggesting an increase in myeloid infiltration compared with untreated mutant 2 wk after LPS challenge, consistent with the microCT data (Fig. 8).

LPS/fMLP induced Muc5AC expression is reduced by LX2931. Unchallenged untreated or LX2931-treated mutant and wild-type mice do not show goblet cells in the bronchi by PAS/AB staining of paraffin sections. In contrast, distinct bronchial goblet cell hyperplasia is observed 2 wk after LPS/ fMLP challenge in all experimental groups to a comparable extent (Fig. 10, A and B). LX2931 treatment did not significantly reduce bronchial goblet cell numbers. Importantly, we observed a higher expression of the inflammation-responsive mucin mMUC5AC in mutant mice compared with wild-type in total lung extracts from LPS/fMLP-challenged mice (Fig. 10C). In LX2931-treated mutant mice a significant reduction toward wild-type values was observed. This suggests that LX2931 does not interfere with inflammation-induced goblet cell hyperplasia in this model but does reduce the expression of inducible gel-forming mucins, which is of considerable importance for CF pathology.

High LPS/fMLP induced cytokine levels in mutant mice restored by LX2391. We measured a panel of cytokines known to be involved in the resolution of inflammation by sequential ELISA in total lung homogenates of mutant and wild-type mice, 2 wk after challenge (Fig. 11). In F508del CFTR mutant mice we report here for the first time a twofold IFN γ and fourfold interleukin 12 (IL-12p70) level compared with wildtype 2 wk after LPS/fMLP challenge (Fig. 11, A and B). Also,





Fig. 7. LX2931 does not rescue parotic duct plugging in F508del CFTR. Paraffin sections of parotic glands were stained for mucus by PAS stain, from adult mutant mice that were untreated (dd, n = 7) or treated with LX2931 as described (dd, LX; n = 9), compared with wild-type littermates (WT, n = 5). In normal mice major ducts (arrows) were free of mucous material (A), whereas in F508del CFTR mice variable but frequent plugging with PAS positive material was observed (arrows) (B). This plugging was not observed in the submandibular and sublingual glands of mutant mice (not shown). C: multiple (3-6) nonsequential low sections of each mouse were analyzed and the percentage of plugged vs. open ducts was calculated. The graph shows the average per experimental group \pm SE. Plugging is significantly higher in mutant mice compared with WT (Kruskal-Wallis/Dunn's test). Treatment with LX2931 has no significant effect on parotic duct plugging.

AJP-Lung Cell Mol Physiol • doi:10.1152/ajplung.00298.2016 • www.ajplung.org

Fig. 8. Focal inflammation induced by bacterial toxins analyzed by microCT. Gated lung microcomputed tomography (microCT) scans on a Ouantum FX CT scanner (Perkin Elmer) under isoflurane anesthesia was performed on all animals treated with LPS/ fMLP as described in METHODS. CT scans were viewed and analyzed using Caliper Analyze software. Scoring of CT scans was performed in a 5-grade binning mode by measuring the relative surface of the lung volume occupied by high-density areas (arrows), as in the representative figures obtained with a F508del CFTR mutant mouse before (A), 48 h after (B), and 1 wk after (C)intranasal LPS/fMLP challenge of each animal. D: quantitative scoring of focal inflammation. Wild-type (WT, n = 17), homozygous F508del CFTR mutant (dd, n = 14), and homozygous mutant orally treated with LX2931 (dd LX, n = 11) for 3 wk prior to challenge and 2 wk after challenge (average score \pm SE). Before-challenge scores were zero in all animals (not shown). At 14 days lungs were collected for quantitative histology and ELISA (Figs. 9-11).



TNF α , IL-10, and KC showed increased levels in mutant mice after challenge (Fig. 11, *C*–*E*).

Importantly, the high IFN γ , IL12P70, KC, and IL-10 levels 2 wk after challenge are corrected toward normal values by LX2391 treatment (Fig. 11). TNF α shows a similar trend, but this does not reach statistical significance in this experiment. A notable exception to this pattern is the increase of the monocyte attractant MCP-1 (CCL2) induced by LX2931 (Fig. 11*F*), which is increased by LX2931 treatment. This is consistent with the enhanced myeloid infiltration after LPS/fMLP challenge (Figs. 8 and 9).

We conclude that resolution of LPS/fMLP-induced inflammation is delayed in F508del mutant mice compared with normal. Furthermore, LX2391 treatment corrects the enhanced expression of several, but not all proinflammatory cytokines under these conditions.

DISCUSSION

Abnormal dendritic cell ratio and myeloid infiltration in CF mouse lung. Unchallenged $Cftr^{Im1EUR}$ F508del CFTR mice kept under pathogen-free conditions display low-level lung inflammation (36, 43, 82). In this model we show here an enhanced level of myeloid cell (CD45+) infiltration, mainly granulocytes and B and T cells (Fig. 2), and low levels of cDC, whereas MoDC are increased (Fig. 3, B and C). MoDC arise from Ly6cHi monocytes at inflammatory sites (56) and play a proinflammatory role via the production of chemokines with chemotactic activity for effector T cells, monocytes and granulocytes (53). Therefore, the presence of high MoDC numbers in the lung may perpetuate inflammation in these mutant mice.

In addition, we observed a distinct iNOS-expressing granulocyte population that was not present in wild-type mice (Fig. 5). iNOS expression is associated with inflammation and tissue damage. This suggests that the iNOS+ granulocyte population in mutant mice is an aspect of basal inflammation caused by CFTR deficiency. A third subset, CD103+ DC, involved in antiviral responses (17), is not significantly different from normal in F508del CFTR mice (Fig. 3, A and D). Xu et al. (85) reported an increase in the percentage of CD11c+ CD11b+ DC in CFTR KO mice, which does not discriminate between CD64- cDC and CD64+ MoDC as in our approach. In contrast to our results (Fig. 3D), they reported a reduced percentage of CD103+ DC (CD11C+ MHCII+ CD11b-). The apparent discrepancy could be related to strain or husbandry differences. However, a direct comparison of these data sets is difficult, since Xu et al. do not report absolute counts and the use of a marker to exclude dead cells from analysis.

In CF infants with early-stage lung disease, CT scans show bronchial thickening and distal air-trapping even in the absence of apparent bacterial infection (41, 68), suggesting unprovoked inflammation in CF lung. In newborn CF piglets, however, a CF-like lung pathology develops within months without enhanced neutrophil infiltration prior to infection, arguing against significant sterile inflammation in newborn CF pig lung (2, 6, 69). However, no detailed quantitative analysis of tissue DC and other myeloid populations by FACS was reported in early stages of disease in either CF pig or human. Therefore, abnormalities similar to what we report here in adult mutant mice could be involved in the development of chronic CF lung disease in human.





Fig. 9. Quantitative analysis of myeloid infiltration after LPS/fMLP challenge. Representative hematoxylin-eosin stain of normal mouse lungs before (A) and 2 wk after (B) LPS/fMLP challenge show sustained septum thickening, associated with appearance of ARG1-positive interstitial and luminal myeloid cells in all experimental groups (not shown). C: quantitative analysis of septum thickness, as described in METHODS, of LPS/fMLP treated and untreated mice as in Fig. 5. A significant increase in septum thickness was observed 2 wk after LPS/ fMLP challenge in normal (WT), mutant (dd), and most strongly in mutant mice treated with LX2931 (dd LX) (n = 9 mice per experimental group, 3–4 nonadjacent sections per mouse were analyzed). **P < 0.01, ***P < 0.001, 1-way ANOVA.

Sustained response to LPS/fMLP challenge in F508del mutant mice. Enhanced levels of proinflammatory cytokines, including TNF- α and KC 2 wk after LPS/fMLP exposure, show defective resolution of inflammation in F508del CFTR mutant mice (Fig. 11). This is consistent with earlier reports of enhanced and sustained lung pathology after naphthaleneinjury (11), bleomycin (27), LPS, and *Pseudomonas aeruginosa* challenge (48) in this model.

Our present data show enhanced levels of both IFN γ and the active form of IL-12 (IL-12p70) in the lung of LPS/fMLP challenged mutant mice, 2 wk after exposure. IFN γ overexpression causes enhanced macrophage and neutrophil infiltration in mice (30). Th1-dependent IFN γ production is stimulated by IL-12, which in turn is secreted by triggered mature DC (30). Thus it seems likely that the high numbers of lung MoDCs in F508del CFTR mutant mice contribute to the delayed resolution of induced inflammation through activation of the IL12/IFN γ axis. Consistent with our finding, IFN γ is increased in BALF of young CF patients compared with non-CF (74).

IL-10 levels were high in mutant mice 2 wk after LPS/fMLP challenge compared with wild-type, which is also testimony to an abnormal resolution of inflammation. IL-10 is often considered an anti-inflammatory signal and therefore benign. How-

ever, this concept has been challenged in the context of chronic lung inflammation. Chronic overexpression of IL-10 causes a strong fibrotic response, likely through upregulation of MCP-1 (CCL2), enhanced myeloid infiltration, and a bias toward M2 macrophage differentiation (70). Proinflammatory macrophage polarization and accumulation were also observed in *Cftr*^{tm1EUR} (FVB) mice (45). Furthermore, a stronger fibrotic response to bleomycin-induced inflammation was reported in this strain, with enhanced myeloid infiltration and MCP-1 signaling (27). Both in the *Cftr*^{tm1EUR} and the IL-10 overexpression models isolated myofibrocytes displayed a profibrotic phenotype compared with control, suggesting that chronic lung inflammation in CF lung changes the epigenetic program of subepithelial connective tissue, involving the IL-10/MCP-1 axis.

Inflammation in Cftr^{imIEUR} mice correlates with low lung SIP levels. The effect of CFTR deficiency on myeloid infiltration and DC imbalance (Figs. 2 and 3) and resolution of inflammation (Fig. 10 and 11) is not readily explained at the molecular level. The myeloid attractants MCP-1, MIP-2, and eotaxin were not significantly increased in unchallenged mutant mice under our basal conditions (Fig. 4). However, CFTR deficiency affects other processes that control myeloid cell behavior in the lung, in particular the production of bioactive

L1010

Fig. 10. Enhanced MUC5AC production in F508del CFTR mutant mouse lung after LPS/fMLP challenge is reduced by SP-1 lyase inhibitor (LX2931). Representative micrographs (hematoxylin-eosin/ PAS staining) of unchallenged F508del CFTR mutant mouse (A) 2 wk after LPS/fMLP challenge (B) (bar: 100 μ m). Mutant (n = 20) and normal mouse lung (n = 20) show very few goblet cells in bronchi. Goblet cell hyperplasia is observed 2 wk after LPS/ fMLP challenge in bronchi of all normal (n = 17) and mutant (n = 14) animals, including animals treated with LX2931 (n = 9). C: total lung extracts of LPS/fMLP-treated mutant mice (dd) and wildtype littermates (WT) were analyzed in parallel by ELISA for murine MUC5AC 2 wk after challenge. Mutant mice produced more mMUC5AC antigen than normal, which was reduced to normal levels by LX2931 treatment (dd LPS LX) as in Fig. 5 (n =10). **P < 0.01; *P < 0.05, 1-way ANOVA.



lipids, as exemplified in our study. CFTR deficiency in epithelial cells reportedly causes destabilization of membrane protein scaffolds involving lysolipid receptors (25), ion transport systems (12, 20, 67) and tight junctions (64), activation of cellular phospholipase A (18), ceramidases (22), and enhanced oxidative stress associated with abnormal autophagy (43). Furthermore, CF lung DCs, macrophages, and neutrophils show phenotypic abnormalities (29, 54, 61, 78, 86). Any of these processes may affect inflammatory signaling in CF lungs. Here we report low sphingosines and S1P in total lung tissue of unchallenged Cftr^{tmIEUR} F508del CFTR mice that are kept on normal chow (Fig. 1). This confirms and extends the data of Xu et al. (85), who reported low S1P in BALF of a C57/Bl6 CFTR KO (Cftr^{1m1UNC}) strain. Low sphingosine levels have been reported in another CF mouse model related to enhanced susceptibility to bacterial infections (51, 72), which we also observe in our model (48).

Abnormalities of sphingolipid metabolism in CF mutant mice and patients have been extensively studied, focusing on the role of acid sphingomyelinase (SMase) in ceramide synthesis (23, 73, 83). In the *Cftr^{tm1EUR}* model, acid and alkaline sphingomyelinase (SMase), and neutral ceramidase activities are normally expressed in mutant mouse lung (47). Enzymes directly involved in S1P metabolism, in particular sphingosine kinase SPHK (33) and SPL (31), do not emerge from differential lung transcriptome analysis in our mouse model (Scholte BJ, unpublished observations). However, further studies are required to establish the relationship between CFTR deficiency and the activity and tissue distribution of key enzymes and transporters that determine the intracellular and extracellular S1P gradients that orchestrate S1P receptor activity.

Treatment with a SLP inhibitor corrects inflammation in unchallenged Cftr^{im1eur} mice. Oral treatment of adult F508del homozygous mice with the SPL inhibitor LX2931 strongly enhanced S1P lung tissue levels, as expected (5). No serious adverse effects were noticed during the 3 wk of the experiment. The relative abundance of the proinflammatory MoDC and reduced levels of cDC in mutant mouse lung are corrected by LX2931 treatment (Fig. 3). This suggests that the abnormal S1P level is a causative factor in CF lung inflammation in this model. We also observe a significant reduction of CD103+ DC in LX2931-treated mutant mice compared with untreated. Together, these data suggest that LX2931 shifts the mutant mouse lung DC population toward a lower state of inflammation.

LX2931 increases lung granulocyte and LyC6+ monocyte infiltration in unchallenged mutant mice (Fig. 2), consistent with an increased level of MCP-1 in LX2931-treated mutant mice (Fig. 4). These LX2931 induced LyC6+ monocytes do not have a suppressor cell (MDSC) phenotype, lacking the typical ARG1⁺ iNOS⁺ FACS stain. Importantly, however, they do not differentiate to proinflammatory MoDCs and macrophages, as they would under inflammatory conditions, since their total numbers are not affected (Fig. 2). Furthermore, the infiltrating granulocytes show reduced iNOS expression in LX2931-treated mutant mice (Fig. 5). Together, this suggests that LX2931 allows monocyte and granulocyte infiltration to the lung but induces a block in differentiation toward proinflammatory phenotypes.

Since our primary purpose was to establish the role of S1P in CF lung inflammation we did not test the effect of LX2931 on wild-type cells from our colony. In an SPL KO mouse model, alveolar myeloid infiltration is reported, comparable to our observations in mutant mice treated with LX2931 (3, 80) (Fig. 2, *B* and *D*). However, this was not observed in wild-type mice treated with LX2931 (T. Oravecz, unpublished data) or the parent compound THI (5), or in a strain expressing reduced levels (17%) of SPL activity (80). As our data indicate, CF mouse lungs may offer a different signaling environment compared with wild-type mice, resulting in enhanced infiltration of myeloid cells during LX2931 treatment, that do not differentiate to proinflammatory phenotypes.

Total B cell and T cell counts, which are increased in mutant mouse lungs, are strongly reduced by LX2931 treatment below wild-type levels (Fig. 2, *E* and *F*). The role of T and B cells in CF lung pathology is not well understood. Infiltrating B cells form lymphoid aggregates in CF lungs during end stage disease



Fig. 11. Abnormal cytokine levels in F508del CFTR mouse lung after LPS/fMLP challenge partially corrected by SP-1 lyase inhibitor (LX2931). Total lung extracts of wild-type (WT), F508del CFTR (dd), and mutant treated with LX2931 (dd LX) were analyzed by sequential ELISA 2 wk after LPS/fMLP challenge for IFN γ (*A*), IL12p70 (*B*), TNF- α (*C*), IL-10 (*D*), KC (*E*), and MCP-1 (*F*). Results from individual mice are shown as pg/mg total protein. ***P < 0.001; **P < 0.01; *P < 0.05, 1-way ANOVA.

(28). B cell differentiation factor (BAFF) is high in BALF from young CF patients compared with healthy controls, suggesting involvement in early CF lung disease (46). B cell depletion is considered a potential treatment in chronic inflammatory disorders (38) and may be beneficial to CF patients as well. T cell subpopulation (Th17/Th2) imbalance was reported in CF lung (74). However, subtype-specific quantitative data on T cell infiltration in early human CF or CF pig lung have not been reported to our knowledge. Clearly, further studies are required to establish the role of T and B cell subtypes in CF pathology and the potential use of pharmaceutical intervention.

LX2931 reduces LPS/fMLP induced inflammatory response in mutant mice. LX2931 treatment does not prevent the LPS/ fMLP induced acute myeloid infiltration (Fig. 8) or subsequent hyperplasia of goblet cells in mutant mice but reduces the enhanced production of mMUC5AC to normal levels (Fig. 10). This is highly relevant to CF since CFTR deficiency in humans causes abnormal mucus viscosity and clearance, which contributes to the progression of CF lung disease (52, 57, 63, 68). Whether LX2931 can prevent or reduce distal airway plugging in CF infants by reducing inflammation and associated mucin secretion remains to be established.

MCP1 levels are enhanced in LX2931-treated mutant animals after challenge (Fig. 11*F*), which is likely related to the increased myeloid infiltration under these conditions (Fig. 9). However, since LX2931 prevents proinflammatory differentiation of infiltrating cells this does not lead to more severe pathology (Table 1, Fig. 10). In contrast, the proinflammatory

IL8 orthologue KC, IL-10, as well as IFN γ and IL-12 are significantly reduced by LX2931 treatment after LPS/fMLP challenge (Fig. 11, *A* and *B*). This suggests that LX2931 treatment attenuates the IL12/IFN inflammation cascade, possibly through an effect on the resident DC population (Fig. 3).

The molecular mechanism of action of LX2931. Our data do not support a correction of mutant CFTR activity by LX2931, since it does not correct the low body weight (Table 1), CFTR-dependent salivary gland fluid transport activity (Fig. 6), or parotic duct plugging in CF mutant animals (Fig. 7). Chronic treatment with the SPL inhibitor THI, a parent compound to LX2931, strongly increased S1P tissue levels in medullary structural cells and also severely reduced S1P1 receptor expression on infiltrating thymocytes, probably by ligand-induced receptor inactivation (44). By analogy, it is highly likely that chronic LX2931 treatment inactivates S1P1 receptors by substantially increasing S1P levels in susceptible tissues. This may subsequently affect trafficking and differentiation of lymphoid cells.

Summary and perspective. Our results reveal for the first time by FACS analysis and sequential ELISA that unchallenged F508del CFTR mutant mouse lung displays abnormal myeloid infiltration and a delayed resolution of LPS/fMLP induced inflammation. This allows us to extend previous studies with experimental therapeutics in this model, targeting F508del mutant CFTR (15, 62, 75, 81) or using anti-inflammatory drugs (18, 43, 45). Our data and recent reports in the literature cited above imply that CF lung pathology should be considered a disease of myeloid cells as well as epithelial cells and can be at least partially corrected using drugs targeting affected signaling pathways (76).

Treatment with the SPL inhibitor LX2931 successfully reverses the low lung tissue S1P levels of F508del CFTR mutant mice and partially corrects the excessive basal and induced inflammation responses in these mice. This would support the hypothesis that pharmaceutic intervention in this pathway could be advantageous to the patients. Since the compound was shown to be safe in animals (Ref. 5; this study) and in humans (37), an intervention in CF patients could be considered.

ACKNOWLEDGMENTS

We gratefully thank J.C.P.A. (Conny) Van Holten-Neelen, Erasmus MC Immunology Dept., for invaluable assistance with the sequential ELISA assay. Present affiliation of T. Oravecz: Celgene, Cambridge, MA.

GRANTS

Part of this research was funded by grants from the Dutch CF Foundation (NCFS HIT-CF C6/7) and the Dutch Lung Fund (3.3.10.027). Part of this study was funded by a research grant from Lexicon Pharmaceuticals Inc.

DISCLOSURES

T. Oravecz was employed by Lexicon Pharmaceuticals Inc. at the time of the studies.

AUTHOR CONTRIBUTIONS

M.V., M.S., D.R., T.O., and B.J.S. conceived and designed research; M.V., G.W., J.B.D.S., W.A.D., T.O., I.D.K., and B.J.S. performed experiments; M.V., M.S., D.R., G.W., J.B.D.S., W.A.D., O.D., T.O., I.D.K., and B.J.S. analyzed data; M.V., M.S., D.R., G.W., J.B.D.S., W.A.D., O.D., T.O., I.D.K., and B.J.S. interpreted results of experiments; M.V., O.D., I.D.K., and B.J.S. prepared figures; M.V., T.O., I.D.K., and B.J.S. drafted manuscript; M.V., M.S., D.R., G.W., J.B.D.S., W.A.D., O.D., T.O., I.D.K., edited and revised manuscript; M.V., M.S., D.R., G.W., J.B.D.S., W.A.D., O.D., T.O., I.D.K., and B.J.S. approved final version of manuscript.

REFERENCES

- Adada MM, Canals D, Jeong N, Kelkar AD, Hernandez-Corbacho M, Pulkoski-Gross MJ, Donaldson JC, Hannun YA, Obeid LM. Intracellular sphingosine kinase 2-derived sphingosine-1-phosphate mediates epidermal growth factor-induced ezrin-radixin-moesin phosphorylation and cancer cell invasion. *FASEB J* 29: 4654–4669, 2015.
- Adam RJ, Michalski AS, Bauer C, Abou Alaiwa MH, Gross TJ, Awadalla MS, Bouzek DC, Gansemer ND, Taft PJ, Hoegger MJ, Diwakar A, Ochs M, Reinhardt JM, Hoffman EA, Beichel RR, Meyerholz DK, Stoltz DA. Air trapping and airflow obstruction in newborn cystic fibrosis piglets. *Am J Respir Crit Care Med* 188: 1434– 1441, 2013.
- Allende ML, Bektas M, Lee BG, Bonifacino E, Kang J, Tuymetova G, Chen W, Saba JD, Proia RL. Sphingosine-1-phosphate lyase deficiency produces a pro-inflammatory response while impairing neutrophil trafficking. J Biol Chem 286: 7348–7358, 2011.
- Alvarez SE, Harikumar KB, Hait NC, Allegood J, Strub GM, Kim EY, Maceyka M, Jiang H, Luo C, Kordula T, Milstien S, Spiegel S. Sphingosine-1-phosphate is a missing cofactor for the E3 ubiquitin ligase TRAF2. *Nature* 465: 1084–1088, 2010.
- 5. Bagdanoff JT, Donoviel MS, Nouraldeen A, Carlsen M, Jessop TC, Tarver J, Aleem S, Dong L, Zhang H, Boteju L, Hazelwood J, Yan J, Bednarz M, Layek S, Owusu IB, Gopinathan S, Moran L, Lai Z, Kramer J, Kimball SD, Yalamanchili P, Heydorn WE, Frazier KS, Brooks B, Brown P, Wilson A, Sonnenburg WK, Main A, Carson KG, Oravecz T, Augeri DJ. Inhibition of sphingosine 1-phosphate lyase for the treatment of rheumatoid arthritis: discovery of (E)-1-(4-((1R,2S,3R)-1,2,3,4-tetrahydroxybutyl)-1H-imidazol-2-yl)ethanone oxime (LX2931) and (1R,2S,3R)-1-(2-(isoxazol-3-yl)-1H-imidazol-4-yl)butane-1,2,3,4-tetraol (LX2932). J Med Chem 53: 8650–8662, 2009.
- Bartlett JA, Ramachandran S, Wohlford-Lenane CL, Barker CK, Pezzulo AA, Zabner J, Welsh MJ, Meyerholz DK, Stoltz DA, McCray PB Jr. Newborn cystic fibrosis pigs have a blunted early response to an inflammatory stimulus. Am J Respir Crit Care Med 194: 845–854, 2016.
- Becker KA, Riethmuller J, Luth A, Doring G, Kleuser B, Gulbins E. Acid sphingomyelinase inhibitors normalize pulmonary ceramide and inflammation in cystic fibrosis. *Am J Respir Cell Mol Biol* 42: 716–724, 2010.
- Best JA, Quinton PM. Salivary secretion assay for drug efficacy for cystic fibrosis in mice. *Exp Physiol* 90: 189–193, 2005.
- Bielawski J, Pierce JS, Snider J, Rembiesa B, Szulc ZM, Bielawska A. Comprehensive quantitative analysis of bioactive sphingolipids by highperformance liquid chromatography-tandem mass spectrometry. *Methods Mol Biol* 579: 443–467, 2009.
- Bryan AM, Del Poeta M, Luberto C. Sphingolipids as regulators of the phagocytic response to fungal infections. *Mediators Inflamm* 2015: 640540, 2015.
- Carvalho-Oliveira IM, Charro N, Aarbiou J, Buijs-Offerman RM, Wilke M, Schettgen T, Kraus T, Titulaer MK, Burgers P, Luider TM, Penque D, Scholte BJ. Proteomic analysis of naphthalene-induced airway epithelial injury and repair in a cystic fibrosis mouse model. *J Proteome Res* 8: 3606–3616, 2009.
- Collawn JF, Lazrak A, Bebok Z, Matalon S. The CFTR and ENaC debate — how important is ENaC in CF lung disease? *Am J Physiol Lung Cell Mol Physiol* 302: L1141–L1146, 2012.
- Corteling R, Wyss D, Trifilieff A. In vivo models of lung neutrophil activation. Comparison of mice and hamsters. *BMC Pharmacol* 2: 1, 2002.
- 14. De Stefano D, Villella VR, Esposito S, Tosco A, Sepe A, De Gregorio F, Salvadori L, Grassia R, Leone CA, De Rosa G, Maiuri MC, Pettoello-Mantovani M, Guido S, Bossi A, Zolin A, Venerando A, Pinna LA, Mehta A, Bona G, Kroemer G, Maiuri L, Raia V. Restoration of CFTR function in patients with cystic fibrosis carrying the F508del-CFTR mutation. *Autophagy* 10: 2053–2074, 2014.
- Dekkers JF, Wiegerinck CL, de Jonge HR, Bronsveld I, Janssens HM, de Winter-de Groot KM, Brandsma AM, de Jong NW, Bijvelds MJ, Scholte BJ, Nieuwenhuis EE, van den Brink S, Clevers H, van der Ent CK, Middendorp S, Beekman JM. A functional CFTR assay using primary cystic fibrosis intestinal organoids. *Nat Med* 19: 939–945, 2013.
- del Rio ML, Bernhardt G, Rodriguez-Barbosa JI, Forster R. Development and functional specialization of CD103+ dendritic cells. *Immunol Rev* 234: 268–281, 2010.

- 17. del Rio ML, Rodriguez-Barbosa JI, Kremmer E, Forster R. CD103– and CD103+ bronchial lymph node dendritic cells are specialized in presenting and cross-presenting innocuous antigen to CD4+ and CD8+ T cells. *J Immunol* 178: 6861–6866, 2007.
- Dif F, Wu YZ, Burgel PR, Ollero M, Leduc D, Aarbiou J, Borot F, Garcia-Verdugo I, Martin C, Chignard M, Israel-Biet D, Kita Y, Scholte BJ, Touqui L. Critical role of cytosolic phospholipase A2α in bronchial mucus hypersecretion in CFTR-deficient mice. *Eur Respir J* 36: 1120–1130, 2010.
- Donoviel MS, Hait NC, Ramachandran S, Maceyka M, Takabe K, Milstien S, Oravecz T, Spiegel S. Spinster 2, a sphingosine-1-phosphate transporter, plays a critical role in inflammatory and autoimmune diseases. *FASEB J* 29: 5018–5028, 2015.
- 20. Favia M, Guerra L, Fanelli T, Cardone RA, Monterisi S, Di Sole F, Castellani S, Chen M, Seidler U, Reshkin SJ, Conese M, Casavola V. Na⁺/H⁺ exchanger regulatory factor 1 overexpression-dependent increase of cytoskeleton organization is fundamental in the rescue of F508del cystic fibrosis transmembrane conductance regulator in human airway CFBE410- cells. *Mol Biol Cell* 21: 73–86, 2010.
- French PJ, van Doorninck JH, Peters RH, Verbeek E, Ameen NA, Marino CR, de Jonge HR, Bijman J, Scholte BJ. A delta F508 mutation in mouse cystic fibrosis transmembrane conductance regulator results in a temperature-sensitive processing defect in vivo. J Clin Invest 98: 1304– 1312, 1996.
- Grassme H, Carpinteiro A, Edwards MJ, Gulbins E, Becker KA. Regulation of the inflammasome by ceramide in cystic fibrosis lungs. *Cell Physiol Biochem* 34: 45–55, 2014.
- Grassme H, Riethmuller J, Gulbins E. Ceramide in cystic fibrosis. Hand Exp Pharmacol 216: 265–274, 2013.
- Guilbault C, Wojewodka G, Saeed Z, Hajduch M, Matouk E, De Sanctis JB, Radzioch D. Cystic fibrosis fatty acid imbalance is linked to ceramide deficiency and corrected by fenretinide. *Am J Respir Cell Mol Biol* 41: 100–106, 2009.
- Holcomb J, Jiang Y, Lu G, Trescott L, Brunzelle J, Sirinupong N, Li C, Naren AP, Yang Z. Structural insights into PDZ-mediated interaction of NHERF2 and LPA2, a cellular event implicated in CFTR channel regulation. *Biochem Biophys Res Commun* 446: 399–403, 2014.
- Huang LS, Berdyshev EV, Tran JT, Xie L, Chen J, Ebenezer DL, Mathew B, Gorshkova I, Zhang W, Reddy SP, Harijith A, Wang G, Feghali-Bostwick C, Noth I, Ma SF, Zhou T, Ma W, Garcia JG, Natarajan V. Sphingosine-1-phosphate lyase is an endogenous suppressor of pulmonary fibrosis: role of S1P signalling and autophagy. *Thorax* 70: 1138–1148, 2015.
- Huaux F, Noel S, Dhooghe B, Panin N, Lo Re S, Lison D, Wallemacq P, Marbaix E, Scholte BJ, Lebecque P, Leal T. Dysregulated proinflammatory and fibrogenic phenotype of fibroblasts in cystic fibrosis. *PLoS One* 8: e64341, 2013.
- Hubeau C, Lorenzato M, Couetil JP, Hubert D, Dusser D, Puchelle E, Gaillard D. Quantitative analysis of inflammatory cells infiltrating the cystic fibrosis airway mucosa. *Clin Exp Immunol* 124: 69–76, 2001.
- Ingersoll SA, Laval J, Forrest OA, Preininger M, Brown MR, Arafat D, Gibson G, Tangpricha V, Tirouvanziam R. Mature cystic fibrosis airway neutrophils suppress T cell function: evidence for a role of arginase 1 but not programmed death-ligand 1. *J Immunol* 194: 5520–5528, 2015.
- 30. Kim YS, Choi SJ, Choi JP, Jeon SG, Oh S, Lee BJ, Gho YS, Lee CG, Zhu Z, Elias JA, Kim YK. IL-12-STAT4-IFN-gamma axis is a key downstream pathway in the development of IL-13-mediated asthma phenotypes in a Th2 type asthma model. *Exp Mol Med* 42: 533–546, 2010.
- 31. Kumar A, Saba JD. Lyase to live by: sphingosine phosphate lyase as a therapeutic target. *Expert Opin Ther Targets* 13: 1013–1025, 2009.
- Kunkel GT, Maceyka M, Milstien S, Spiegel S. Targeting the sphingosine-1-phosphate axis in cancer, inflammation and beyond. *Nat Rev Drug Discov* 12: 688–702, 2013.
- Lai WQ, Wong WS, Leung BP. Sphingosine kinase and sphingosine 1-phosphate in asthma. *Biosci Rep* 31: 145–150, 2011.
- Lambrecht BN, Hammad H. Dendritic cell and epithelial cell interactions at the origin of murine asthma. Ann Am Thorac Soc 11, Suppl 5: S236–S243, 2014.
- Laval J, Touhami J, Herzenberg LA, Conrad C, Taylor N, Battini JL, Sitbon M, Tirouvanziam R. Metabolic adaptation of neutrophils in cystic fibrosis airways involves distinct shifts in nutrient transporter expression. *J Immunol* 190: 6043–6050, 2013.
- 36. Legssyer R, Huaux F, Lebacq J, Delos M, Marbaix E, Lebecque P, Lison D, Scholte BJ, Wallemacq P, Leal T. Azithromycin reduces

spontaneous and induced inflammation in DeltaF508 cystic fibrosis mice. *Respir Res* 7: 134, 2006.

- Lexicon Pharmaceuticals. Lexicon pharmaceuticals reports preliminary results from two phase 1 studies. http://www.lexpharma.com/media-center/ news/543-lexicon-pharmaceuticals-reports-preliminary-results-from-two-phase-1studies [2012; accessed 30 October 2016].
- Li R, Rezk A, Healy LM, Muirhead G, Prat A, Gommerman JL, Bar-Or A; MSSRF Canadian B Cells in MS Team. Cytokine-defined B cell responses as therapeutic targets in multiple sclerosis. *Front Immunol* 6: 626, 2015.
- 39. Liu G, Bi Y, Wang R, Yang H, Zhang Y, Wang X, Liu H, Lu Y, Zhang Z, Chen W, Chu Y, Yang R. Targeting S1P1 receptor protects against murine immunological hepatic injury through myeloid-derived suppressor cells. *J Immunol* 192: 3068–3079, 2014.
- Liu HB, Cui NQ, Wang Q, Li DH, Xue XP. Sphingosine-1-phosphate and its analogue FTY720 diminish acute pulmonary injury in rats with acute necrotizing pancreatitis. *Pancreas* 36: e10–e15, 2008.
- Loeve M, Gerbrands K, Hop WC, Rosenfeld M, Hartmann IC, Tiddens HA. Bronchiectasis and pulmonary exacerbations in children and young adults with cystic fibrosis. *Chest* 140: 178–185, 2011.
- 42. Luciani A, Villella VR, Esposito S, Brunetti-Pierri N, Medina D, Settembre C, Gavina M, Pulze L, Giardino I, Pettoello-Mantovani M, D'Apolito M, Guido S, Masliah E, Spencer B, Quaratino S, Raia V, Ballabio A, Maiuri L. Defective CFTR induces aggresome formation and lung inflammation in cystic fibrosis through ROS-mediated autophagy inhibition. *Nat Cell Biol* 12: 863–867, 2010.
- 43. Luciani A, Villella VR, Esposito S, Gavina M, Russo I, Silano M, Guido S, Pettoello-Mantovani M, Carnuccio R, Scholte B, De Matteis A, Maiuri MC, Raia V, Luini A, Kroemer G, Maiuri L. Targeting autophagy as a novel strategy for facilitating the therapeutic action of potentiators on DeltaF508 cystic fibrosis transmembrane conductance regulator. *Autophagy* 8: 1657–1672, 2012.
- 44. Maeda Y, Yagi H, Takemoto K, Utsumi H, Fukunari A, Sugahara K, Masuko T, Chiba K. S1P lyase in thymic perivascular spaces promotes egress of mature thymocytes via up-regulation of S1P receptor 1. *Int Immunol* 26: 245–255, 2014.
- 45. Meyer M, Huaux F, Gavilanes X, van den Brule S, Lebecque P, Lo Re S, Lison D, Scholte B, Wallemacq P, Leal T. Azithromycin reduces exaggerated cytokine production by M1 alveolar macrophages in cystic fibrosis. *Am J Respir Cell Mol Biol* 41: 590–602, 2009.
- 46. Neill DR, Saint GL, Bricio-Moreno L, Fothergill JL, Southern KW, Winstanley C, Christmas SE, Slupsky JR, McNamara PS, Kadioglu A, Flanagan BF. The B lymphocyte differentiation factor (BAFF) is expressed in the airways of children with CF and in lungs of mice infected with Pseudomonas aeruginosa. *PLoS One* 9: e95892, 2014.
- 47. Ohlsson L, Hjelte L, Huhn M, Scholte BJ, Wilke M, Flodstrom-Tullberg M, Nilsson A. Expression of intestinal and lung alkaline sphingomyelinase and neutral ceramidase in cystic fibrosis f508del transgenic mice. J Pediatr Gastroenterol Nutr 47: 547–554, 2008.
- Palomo J, Marchiol T, Piotet J, Fauconnier L, Robinet M, Reverchon F, Le Bert M, Togbe D, Buijs-Offerman R, Stolarczyk M, Quesniaux VF, Scholte BJ, Ryffel B. Role of IL-1beta in experimental cystic fibrosis upon *P. aeruginosa* infection. *PLoS One* 9: e114884, 2014.
- 49. Park ES, Choi S, Shin B, Yu J, Yu J, Hwang JM, Yun H, Chung YH, Choi JS, Choi Y, Rho J. Tumor necrosis factor (TNF) receptor-associated factor (TRAF)-interacting protein (TRIP) negatively regulates the TRAF2 ubiquitin-dependent pathway by suppressing the TRAF2-sphingosine 1-phosphate (S1P) interaction. J Biol Chem 290: 9660–9673, 2015.
- 50. Peng X, Hassoun PM, Sammani S, McVerry BJ, Burne MJ, Rabb H, Pearse D, Tuder RM, Garcia JG. Protective effects of sphingosine 1-phosphate in murine endotoxin-induced inflammatory lung injury. *Am J Respir Crit Care Med* 169: 1245–1251, 2004.
- 51. Pewzner-Jung Y, Tavakoli Tabazavareh S, Grassme H, Becker KA, Japtok L, Steinmann J, Joseph T, Lang S, Tuemmler B, Schuchman EH, Lentsch AB, Kleuser B, Edwards MJ, Futerman AH, Gulbins E. Sphingoid long chain bases prevent lung infection by Pseudomonas aeruginosa. *EMBO Mol Med* 6: 1205–1214, 2014.
- 52. Pillarisetti N, Williamson E, Linnane B, Skoric B, Robertson CF, Robinson P, Massie J, Hall GL, Sly P, Stick S, Ranganathan S. Infection, inflammation and lung function decline in infants with cystic fibrosis. Am J Respir Crit Care Med 184: 75–81, 2011.
- 53. Plantinga M, Guilliams M, Vanheerswynghels M, Deswarte K, Branco-Madeira F, Toussaint W, Vanhoutte L, Neyt K, Killeen N, Malissen B, Hammad H, Lambrecht BN. Conventional and monocyte-

L1014

derived CD11b(+) dendritic cells initiate and maintain T helper 2 cellmediated immunity to house dust mite allergen. *Immunity* 38: 322–335, 2013.

- 54. Pohl K, Hayes E, Keenan J, Henry M, Meleady P, Molloy K, Jundi B, Bergin DA, McCarthy C, McElvaney OJ, White MM, Clynes M, Reeves EP, McElvaney NG. A neutrophil intrinsic impairment affecting Rab27a and degranulation in cystic fibrosis is corrected by CFTR potentiator therapy. *Blood* 124: 999–1009, 2014.
- 55. Pyne NJ, McNaughton M, Boomkamp S, MacRitchie N, Evangelisti C, Martelli AM, Jiang HR, Ubhi S, Pyne S. Role of sphingosine 1-phosphate receptors, sphingosine kinases and sphingosine in cancer and inflammation. Adv Biol Regul 60: 151–159, 2016.
- Qu C, Brinck-Jensen NS, Zang M, Chen K. Monocyte-derived dendritic cells: targets as potent antigen-presenting cells for the design of vaccines against infectious diseases. *Int J Infect Dis* 19: 1–5, 2014.
- Quinton PM. Role of epithelial HCO₃⁻ transport in mucin secretion: lessons from cystic fibrosis. Am J Physiol Cell Physiol 299: C1222– C1233, 2010.
- 58. Ramnath NW, van de Luijtgaarden KM, van der Pluijm I, van Nimwegen M, van Heijningen PM, Swagemakers SM, van Thiel BS, Ridwan RY, van Vliet N, Vermeij M, Hawinkels LJ, de Munck A, Dzyubachyk O, Meijering E, van der Spek P, Rottier R, Yanagisawa H, Hendriks RW, Kanaar R, Rouwet EV, Kleinjan A, Essers J. Extracellular matrix defects in aneurysmal Fibulin-4 mice predispose to lung emphysema. *PLoS One* 9: e106054, 2014.
- Ray A, Chakraborty K, Ray P. Immunosuppressive MDSCs induced by TLR signaling during infection and role in resolution of inflammation. *Front Cell Infect Microbiol* 3: 52, 2013.
- 60. Ren S, Babelova A, Moreth K, Xin C, Eberhardt W, Doller A, Pavenstadt H, Schaefer L, Pfeilschifter J, Huwiler A. Transforming growth factor-beta2 upregulates sphingosine kinase-1 activity, which in turn attenuates the fibrotic response to TGF-beta2 by impeding CTGF expression. *Kidney Int* 76: 857–867, 2009.
- 61. Rieber N, Hector A, Carevic M, Hartl D. Current concepts of immune dysregulation in cystic fibrosis. *Int J Biochem Cell Biol* 52: 108–112, 2014.
- Robert R, Carlile GW, Liao J, Balghi H, Lesimple P, Liu N, Kus B, Rotin D, Wilke M, de Jonge HR, Scholte BJ, Thomas DY, Hanrahan JW. Correction of the Delta phe508 cystic fibrosis transmembrane conductance regulator trafficking defect by the bioavailable compound glafenine. *Mol Pharmacol* 77: 922–930, 2010.
- 63. Rosenow T, Oudraad MC, Murray CP, Turkovic L, Kuo W, de Bruijne M, Ranganathan SC, Tiddens HA, Stick SM; Australian Respiratory Early Surveillance Team for Cystic Fibrosis (AREST CF). PRAGMA-CF. A quantitative structural lung disease computed tomography outcome in young children with cystic fibrosis. Am J Respir Crit Care Med 191: 1158–1165, 2015.
- 64. Ruan YC, Wang Y, Da Silva N, Kim B, Diao RY, Hill E, Brown D, Chan HC, Breton S. CFTR interacts with ZO-1 to regulate tight junction assembly and epithelial differentiation through the ZONAB pathway. J Cell Sci 127: 4396–4408, 2014.
- Schwab SR, Pereira JP, Matloubian M, Xu Y, Huang Y, Cyster JG. Lymphocyte sequestration through S1P lyase inhibition and disruption of S1P gradients. *Science* 309: 1735–1739, 2005.
- 66. Scrimini S, Pons J, Sauleda J. Myeloid-derived suppressor cells: possible link between chronic obstructive pulmonary disease and lung cancer. *Arch Bronconeumol* 52: 29–35, 2015.
- 67. Seidler U, Singh AK, Cinar A, Chen M, Hillesheim J, Hogema B, Riederer B. The role of the NHERF family of PDZ scaffolding proteins in the regulation of salt and water transport. *Ann NY Acad Sci* 1165: 249–260, 2009.
- Stick S, Tiddens H, Aurora P, Gustafsson P, Ranganathan S, Robinson P, Rosenfeld M, Sly P, Ratjen F. Early intervention studies in infants and preschool children with cystic fibrosis: are we ready? *Eur Respir J* 42: 527–538, 2013.
- 69. Stoltz DA, Meyerholz DK, Pezzulo AA, Ramachandran S, Rogan MP, Davis GJ, Hanfland RA, Wohlford-Lenane C, Dohrn CL, Bartlett JA, Nelson GAt Chang EH, Taft PJ, Ludwig PS, Estin M, Hornick EE, Launspach JL, Samuel M, Rokhlina T, Karp PH, Ostedgaard LS, Uc A, Starner TD, Horswill AR, Brogden KA, Prather RS, Richter SS,

Shilyansky J, McCray PB Jr, Zabner J, Welsh MJ. Cystic fibrosis pigs develop lung disease and exhibit defective bacterial eradication at birth. *Sci Transl Med* 2: 29ra31, 2010.

- Sun L, Louie MC, Vannella KM, Wilke CA, LeVine AM, Moore BB, Shanley TP. New concepts of IL-10-induced lung fibrosis: fibrocyte recruitment and M2 activation in a CCL2/CCR2 axis. *Am J Physiol Lung Cell Mol Physiol* 300: L341–L353, 2011.
- Szczepaniak WS, Zhang Y, Hagerty S, Crow MT, Kesari P, Garcia JG, Choi AM, Simon BA, McVerry BJ. Sphingosine 1-phosphate rescues canine LPS-induced acute lung injury and alters systemic inflammatory cytokine production in vivo. *Transl Res* 152: 213–224, 2008.
- 72. Tavakoli Tabazavareh S, Seitz A, Jernigan P, Sehl C, Keitsch S, Lang S, Kahl BC, Edwards M, Grassme H, Gulbins E, Becker KA. Lack of sphingosine causes susceptibility to pulmonary *Staphylococcus aureus* infections in cystic fibrosis. *Cell Physiol Biochem* 38: 2094–2102, 2016.
- 73. Teichgraber V, Ulrich M, Endlich N, Riethmuller J, Wilker B, De Oliveira-Munding CC, van Heeckeren AM, Barr ML, von Kurthy G, Schmid KW, Weller M, Tummler B, Lang F, Grassme H, Doring G, Gulbins E. Ceramide accumulation mediates inflammation, cell death and infection susceptibility in cystic fibrosis. *Nat Med* 14: 382–391, 2008.
- 74. Tiringer K, Treis A, Fucik P, Gona M, Gruber S, Renner S, Dehlink E, Nachbaur E, Horak F, Jaksch P, Doring G, Crameri R, Jung A, Rochat MK, Hormann M, Spittler A, Klepetko W, Akdis CA, Szepfalusi Z, Frischer T, Eiwegger T. A Th17- and Th2-skewed cytokine profile in cystic fibrosis lungs represents a potential risk factor for Pseudomonas aeruginosa infection. Am J Respir Crit Care Med 187: 621–629, 2013.
- 75. Tomati V, Sondo E, Armirotti A, Caci E, Pesce E, Marini M, Gianotti A, Jeon YJ, Cilli M, Pistorio A, Mastracci L, Ravazzolo R, Scholte B, Ronai Z, Galietta LJ, Pedemonte N. Genetic inhibition of the ubiquitin ligase Rnf5 attenuates phenotypes associated to F508del cystic fibrosis mutation. *Sci Rep* 5: 12138, 2015.
- 76. Torphy TJ, Allen J, Cantin AM, Konstan MW, Accurso FJ, Joseloff E, Ratjen FA, Chmiel JF; Antiinflammatory Therapy Working Group. Considerations for the conduct of clinical trials with antiinflammatory agents in cystic fibrosis. A Cystic Fibrosis Foundation Workshop Report. Ann Am Thorac Soc 12: 1398–1406, 2015.
- Uhig S, Gulbins E. Sphingolipids in the lungs. Am J Respir Crit Care Med 178: 1100–1114, 2008.
- Uriarte SM. Novel insights related to CF neutrophils. Blood 124: 985– 986, 2014.
- van den Berg JW, Dik WA, van der Zee M, Bonthuis F, van Holten-Neelen C, Dingjan GM, Benner R, Ijzermans JN, Khan NA, de Bruin RW. The beta-human chorionic gonadotropin-related peptide LQGV reduces mortality and inflammation in a murine polymicrobial sepsis model. *Crit Care Med* 39: 126–134, 2011.
- Vogel P, Donoviel MS, Read R, Hansen GM, Hazlewood J, Anderson SJ, Sun W, Swaffield J, Oravecz T. Incomplete inhibition of sphingosine 1-phosphate lyase modulates immune system function yet prevents early lethality and non-lymphoid lesions. *PLoS One* 4: e4112, 2009.
- Wilke M, Bot A, Jorna H, Scholte BJ, de Jonge HR. Rescue of murine F508del CFTR activity in native intestine by low temperature and proteasome inhibitors. *PLoS One* 7: e52070, 2012.
- 82. Wilke M, Buijs-Offerman RM, Aarbiou J, Colledge WH, Sheppard DN, Touqui L, Bot A, Jorna H, de Jonge HR, Scholte BJ. Mouse models of cystic fibrosis: phenotypic analysis and research applications. J Cyst Fibros 10, Suppl 2: S152–S171, 2011.
- Wojewodka G, De Sanctis JB, Radzioch D. Ceramide in cystic fibrosis: a potential new target for therapeutic intervention. *J Lipids* 2011: 674968, 2011.
- Worgall TS. Lipid metabolism in cystic fibrosis. Curr Opin Clin Nutr Metab Care 12: 105–109, 2009.
- Xu Y, Krause A, Limberis M, Worgall TS, Worgall S. Low sphingosine-1-phosphate impairs lung dendritic cells in cystic fibrosis. Am J Respir Cell Mol Biol 48: 250–257, 2013.
- Zhang PX, Cheng J, Zou S, D'Souza AD, Koff JL, Lu J, Lee PJ, Krause DS, Egan ME, Bruscia EM. Pharmacological modulation of the AKT/microRNA-199a-5p/CAV1 pathway ameliorates cystic fibrosis lung hyper-inflammation. *Nat Commun* 6: 6221, 2015.

human disease CF lung COPD airway IL68 Inflammation AREC remodeling EGER s fibrosis PBEC human disease CF lung COPD airway IL68 InflaCHAPTER EGFR shedding ADAM17 fibrosis PBEC human disease CF lung COPD airway IL6R inflammation **General discussion** nflammation AREG remodeling EGFR shedding human disease CF lung COPD airway IL68 inflammation AREG G remodeling A REG remodeling a Sinflammation EGFR shedding ADAM17 fibrosis PBEC huses se CF lung 📲 uman disea airway IL6R Inflammation AREG remodeling EGFR sheddin 117 fibros ung COPD airway IL6R inflammation AREG remodeling EGFR shedding ADAM17 Thorosis PBEC / fibrosis PBEC human disease, CF lung COPD airway Is abnormal activation of EGFR-ADAM17 axis in COPD and CF shedding ADA a relevant target for therapeutic intervention? R inflammation AREG remo ainway IL6R Inflammation AREG remodeling EGFR sheddin Stolarczyk M, Scholte BJ an disea

GENERAL DISCUSSION AND FUTURE PERSPECTIVES

General introduction

Airway epithelium, apart from providing a structural barrier against microbes and inhaled particles (mucociliary clearance), also plays an active role in a first line of inflammatory responses [1], and thus emerged as a therapeutic target in chronic lung disease [2]. Airway epithelial cells respond dynamically to bacterial and viral infections [3] [4] [5] or inhaled noxious particles by transducing inflammatory signals [6] [7]. They produce cytokines, growth factors and other inflammatory mediators to recruit a range of inflammatory cells including neutrophils and macrophages [1] [8]. Importantly, airway epithelial cells also cross-talk with the underlying fibroblasts/myoblasts [9] [10]. This trans-signaling, when exaggerated by stress factors and bacterial infections during the course of disease development, leads to structural changes of the tissue structure termed lung tissue remodeling, a feature of both CF and COPD pathology [11] [12] [13].

A disintegrin and metalloprotease 17 (ADAM17), a sheddase, formerly known as TACE, that releases a broad spectrum of soluble biologically active substrates from airway epithelial cells is recognized as an important transducer of airway epithelial cis- and trans- signaling [14]. The growth factor AREG [15] and cytokine receptor IL6R [16], both shed by ADAM17, activate EGFR and gp130 respectively, and have been linked to lung pathology [17] [18] [19], but were never directly investigated in the context of COPD and CF lung malfunction. Therefore, the main aim of this thesis was to elucidate whether ADAM17 responds differently to external stimuli like cigarette smoke (CS) in COPD airway epithelial cells and whether CFTR deficiency activates ADAM17 as a cell autonomous defect. During the course of our studies it emerged that EGFR is an equally potent regulator of IL6R and AREG release and mRNA expression as ADAM17, indicating the role of the EGFR-ADAM17 axis in COPD and CF pathology (Chapter 2 and 3). It is known that both EGFR and ADAM17 play an essential role in lung tissue inflammation and regeneration [20] [21]. However, so far the role of ADAM17 and EGFR has only been broadly studied in carcinogenesis and little is known about the organization and function of this signaling network in differentiated CF and COPD airway epithelial cells.

In view of the comparable pathological events in CF and COPD [22] [23] [24] (Chapter 1) and the recently observed downregulation of CFTR by CS exposure [25] [26], we proposed that the molecular mechanism underlying the development of these diseases is similar and may involve abnormal activation of the EGFR-ADAM17 axis. Our studies and the literature cited show that ADAM17-mediated release of bioactive substrates is a tightly regulated, multistep, spatiotemporal process that likely requires activation by upstream kinase signaling (Chapter 2 and 3) [27], trafficking of ADAM17 towards the substrate in the intracellular compartment (Chapter 2) [28], thiol-isomerase and redox regulated disulfide bridge rearrangements (Chapter 3) [29], and dissociation of the inhibitory protein TIMP-3 [30]. EGFR/ADAM17 signaling cross-talks with many regulatory pathways like TLR5, G-coupled receptors, p38 MAPK, ERK1/2 (Chapter 2 and 3) [31] [32] [33] [34] [35]. In this thesis we addressed some of the issues of ADAM17 activation and the role of EGFR in this

signaling cascade in order to define the druggable targets in the ADAM17/EGFR axis in CF and COPD pathology (Chapter 1, Figure 7).

A model of lung pathology

Although mouse models can contribute considerably to our understanding of CF (Chapter 4, Appendix) and COPD lung pathology [36], we have focused on differentiated at air-liquid interface human airway epithelial cells in culture. Firstly, because it has been reported that ADAM17 activity towards its substrates is species specific, as exemplified by IL-6R [37] [38] [10]. Additionally, in humans and pigs, the loss of CFTR impairs respiratory host defenses due to acidification of ASL causing airway infection, but in mice this acidification is not observed [39]. CF mice do not develop CF type obstructive distal lung disease [40], however, they do show evidence of chronic lung disease characterized by inflammation and MUC5AC hypersecretion with abnormal myeloid infiltration and exaggerated responses to bacterial challenge (Chapter 4, Appendix). To provide clinically relevant data focusing on the contribution of airway epithelial cells to the signaling cascade, we investigated our hypothesis in human bronchial epithelial cells cultured at the air-liquid interface.

We examined the effect of cigarette smoke (CS) on the ADAM17/EGFR axis in welldifferentiated bronchial epithelial cells from COPD patients under air-liquid interface (ALI) (ALI-COPD HBEC) (Chapter 2). This gave us the opportunity to compare airway epithelial cells from COPD with non-COPD subjects. Apart from that we exposed the cells to whole cigarette smoke, instead of cigarette smoke extract (CSE) or noxious particles, that cannot be considered as the substitute of whole cigarette smoke [41] (Chapter 2). Although we were able to obtain significant differences, we noticed that primary human bronchial cells from different donors respond with high variability to external stimuli (Chapter 2). This can be considered as a disadvantage of the primary culture model in the event of studying the effect of CFTR deficiency on downstream pathways. Apart from that, cell material from the CF patient population allows us only limited access to end-stage disease after lung transplant. An alternative would be epithelial cell culture from nasal brushes, however, this does not allow expansion of large amounts of cells for culture, and this system differs from the distal airways and is also characterized by a large donor and subculture variation [42]. Thus, in further studies to define the effect of CFTR deficiency on ADAM17/EGFR axis activity, we used a well-characterized airway epithelial cell line CFBE410- differentiated under air-liquid interface with inducible CFTR expression (ALI-CFBE41o-). This model allowed us not only to overcome donor and culture variation (Chapter 3), but also the use of CFTR inhibitors or siRNA to obtain CFTR deficient cells is not preferred due to reported off target effects [43] [44]. In future studies in our laboratory induced pluripotent stem cells (iPS) from CF patients and parallel lines corrected by gene editing will be used. This will allow us to study CFTR deficiency in a single genetic background in differentiated epithelial lineages.

ADAM17 is the main sheddase of IL6R and AREG in human bronchial epithelial cells

Previous studies have shown that external stress factors (like cigarette smoke extract [7] [45] [46] [47] [48] [49], bacterial or viral infections [3] [4], cytokine or growth factor treatment [50] [51]) abnormally activate ADAM17 and EGFR in different airway epithelial

cell models, mainly in submerged cell lines. As an extension of previous finding, the studies described in this thesis show the mechanism of ADAM17 activation in CF and COPD lung disease in an advanced model of lung airway epithelial cells. By using the selective ADAM17 inhibitor TMI-2 (Wyeth), we showed that ADAM17 is the main sheddase responsible for AREG and IL6R release from differentiated primary airway epithelial cells (Chapter 2) and AREG from ALI-CFBE cell line (soluble IL6R has not been detected in supernatants from ALI-CFBE410- in our assay) (Chapter 3).

ADAM17 establishes a signaling axis with EGFR, regulating protein and mRNA levels of ADAM17 substrates.

To further investigate the signaling cascade leading to release of ADAM17 substrates, we tested several kinases that have been potential regulators of ADAM17 signaling cascade, like SRC [52] or ERK1/2/MAPK [35] [33], but EGFR emerged as the most potent regulator of AREG and IL6R release, with a strength equal to ADAM17 (Chapter 2). These data provided the proof of existence of a EGFR-ADAM17 positive feedback loop [53] [52] [54], also in differentiated airway epithelial cells (Chapter 1, Figure 4). Importantly, for the first time we present evidence that both EGFR and ADAM17 regulate not only protein release, but also mRNA expression of IL6R and AREG. The involvement of EGFR in mRNA regulation of other signaling molecules such as IL-8 has been already shown [7] [49]. Here we observed that also a selective ADAM17 inhibitor blocks IL8 mRNA production in ALI-HBEC through the feedback loop action of the system (Chapter 2). Thus, our results highlight that the EGFR-ADAM17 axis is a potent regulator of both ADAM17 substrate shedding and EGFR targets at the protein and mRNA level in airway epithelial cells, likely playing a crucial role in inflammation and tissue repair.

Airway epithelial cells from COPD patients respond more strongly to cigarette smoke than normal

In COPD ALI-HBEC we showed that under basal conditions release of ADAM17 substrates does not depend on COPD disease status, but upon CS exposure both AREG and IL-6R release is stronger in COPD cells. The difference between COPD and non-COPD in CSinduced ADAM17 activity is not caused by changes in the relative mRNA levels of AREG or IL-6R in COPD ALI-HBEC. Thus, we concluded that COPD lung disease makes airway epithelial cells more sensitive to external stress factors. Since our studies show induction of ADAM17 activity upon CS exposure, and previous studies of Amatngalim et al showed that CS also induces phosphorylation of EGFR [7], it is likely that the difference between COPD and non-COPD is due to post-transcriptional regulation of ADAM17 and EGFR activity (Chapter 2). This may include the level of ADAM17 and EGFR phosphorylation (Chapter 2 and 3) or structural changes (disulfide bridges) (Chapter 3). Epigenetic changes, involving DNA methylation and histone modifications, causing various levels of metaplasia, may also play a role. DNA methylation is affected in COPD lung tissue and mice exposed to CS [55], and is likely also a factor in CF lung disease development, though this has not been studied in detail. It has been proposed that both CF and COPD patients have altered microRNA (miRNA) expression [56] [57] [58] [59], which may affect ADAM17 [60] [61] [62] and EGFR [63] [64] activity. Of interest to our purpose is the fact that both miRNA and epigenetics offer new possibilities of therapeutic intervention.

CFTR dysfunction hyperactivates the ADAM17/EGFR axis and is a possible link between pathological events observed in COPD and CF

To further establish the importance of the EGFR-ADAM17 axis in lung pathology, we showed that CFTR deficient airway epithelial cells have induced ADAM17 activity and EGFR phosphorylation (Y1173), compared to CFTR expressing cells (Chapter 3). These data suggest that CFTR deficient cells have an intrinsic defect in the ADAM17/EGFR signaling pathway. Kim et al reported that CFTR inhibitor treatment increases EGFR and ADAM17 related inflammatory responses in submerged immortalized cells [65]. However, we used differentiated cells under ALI conditions with inducible CFTR expression, overcoming the off-target effects of CFTR inhibitors [43] [44] and donor variation observed in ALI-HBEC (Chapter 2). Importantly, enhanced ADAM17 shedding activity and EGFR phosphorylation in CFTR deficient cells is not caused by a difference in induced mRNA levels of ADAM17 or EGFR, suggesting that dysfunctional CFTR affects ADAM17/EGFR signaling posttranscriptionally. Moreover, the difference was only observed during ALI culture, and not in submerged conditions, suggesting that CFTR deficient cells in comparison to cells with induced CFTR activity adapt abnormally to the ALI environment, affecting the ADAM17/EGFR axis activity. This also highlights that 3D ALI culture, but not submerged cultures, resembles the in vivo conditions more precisely.

Interestingly, CS recently has been shown to reduce CFTR function, observed as CFTR internalization [25], reduced CFTR activity measured by nasal potential differences (NPD) in patients and ASL dehydration in airway epithelial cells [26], mimicking this aspect of CF pathology [23] [24] [66]. Recent studies show that treatment with cigarette smoke extract (CSE) transiently also reduced CFTR gating activity, due to formation of acrolein adducts, and that short term (1 h) application of VX-770 partially restores CFTR function [67]. Thus, the CS induced ADAM17/EGFR axis activity (Chapter 2) and the CFTR-dependent AREG release in CFBE cells (Chapter 3) suggest that CFTR dysfunction is a common cause of enhanced activity of the EGFR/ADAM17 axis observed in CF and COPD lung disease.

We measured CFTR related short-circuit current (Isc) 24h after CS exposure, when we observed induced shedding activity, but at this time point we did not observe abnormal CFTR function (data not shown). However, CFTR internalizes quickly after CS exposure and complete recovery takes approximately 24h [25] [68]. Therefore, further experiments with shorter time points are required to establish the relationship between CFTR activity and shedding in our CS exposure system.

Solone et al showed that the CFTR potentiator Ivacaftor (VX-770) activated CFTR-dependent chloride transport in non-CF patients, increased ASL height and mucociliary transport, suggesting this may be of therapeutic importance for COPD patients [26]. However, in our culture model we showed that long-term treatment (24h) of ALI-HBEC with VX-770 enhances CS triggered IL6R and AREG secretion significantly, in comparison to CS exposure alone (M Stolarczyk, 3 independent experiments in triplicate P<0.05; data not shown). In the light of conflicting data showing that chronic treatment with VX-770 either reduces wild-type CFTR function [69], improve CFTR function [26] or has no effect on CFTR activity [70], it

is difficult to establish whether the increase in shedding is related to CFTR inactivation or to an off target effect of VX-770 on the ADAM/EGFR axis. It is clear, however, our present data do not support the option to treat COPD or smoking patients with VX-770 chronically, as a therapy. In the future new generation potentiators, which are currently developed by several teams, could be further investigated in our system.

One should consider here the long-term and short-term effects of CS exposure. COPD patients still have COPD pathology long after they stop smoking and their CFTR levels are restored. This is probably related to long-term epigenetic changes and metaplasia in COPD caused by frequent CS exposure, as also suggested by our experiments (chapter 2), and not with the acute effects of CS. Long-term suppression of CFTR activity in CF patients and associated pathology would likely have similar effects. This may mean that fully developed CF lung disease cannot be treated with CFTR targeted agents only, but requires additional treatment to counter these effects.

EGFR/ADAM17 axis is regulated by CFTR related oxidative stress

The extracellular membrane proximal domain of ADAM17 [29] [71] and the intracellular domain of EGFR [72] are both activated in an oxidized environment, in particular by rearrangements of intramolecular disulfide bridges by thiol-isomerases (PDI) that are not only involved in ER protein maturation, but also act on the extracellular membrane [29] [71]. ADAM17 can also be activated by upstream signaling kinases sensitive to ROS, for instance SRC [73]. In line with these observations, we showed in Chapter 3 that AREG shedding mediated by ADAM17 is sensitive to ROS scavengers (NAC, DPI) and it is most strongly inhibited by addition of an impermeant natural thiol antioxidant glutathione (GSH) [74]. These data suggested that the ADAM17/EGFR axis is sensitive to the extracellular redox potential. By expression of roGFP-Grx-1 fused to GPI [75] [76], which enables ratiometric measurements of the apical extracellular redox potential, we detected enhanced oxidation in the apical lining fluid of CFTR deficient CFBE41o- cells. In contrast, we reported unchanged cytoplasmic redox potential in CFTR deficient CFBE410- cells expressing roGFP-Grx1. Thus, we propose that induced ADAM17 dependent AREG shedding in CFTR deficient cells (Chapter 3) is caused by an abnormal ratio of GSH/GSSG in the extracellular environment. However, at this stage it is still not clear whether this hyperoxidation occurs only in ASL or also in other compartments, such as the lateral and intra-vesicular space.

CFTR was previously reported to regulate the extracellular redox potential through GSH transport in the airways [77] [78], so it is likely that the abnormal transport of GSH causes ASL oxidation. In line with our speculations, in lung lavage fluids of CF patients the levels of GSH were reduced in comparison to healthy controls [79]. Furthermore, we observe that several lipid markers of oxidative stress in lung lavages (BALF) from juvenile CF patients, in particular isoprostanes which are peroxidation products of unsaturated fatty acids, correlate with the severity of CF lung disease (Scholte, Stick et al, in preparation). Current longitudinal studies by our group of bronchial lavage fluids from CF infants will focus on the relationship between oxidative stress signaling and the development of CF lung disease. Also in our F508del CFTR mouse model enhanced inflammation (Chapter 4, Appendix) correlates with lipid markers of oxidative stress (Scholte, Radzioch, in preparation). Importantly, GSH administration in CF patients reduced inflammatory markers in BALF [80].

The impaired GSH transport in CF may enhance downstream signaling [77] [80] and affect activity of ADAM17, which is sensitive to redox potential through its extracellular domain [29]. Since the extracellular domain of ADAM17 is not primarily exposed to the extracellular apical membrane, it is also plausible that the GSH/GSSG ratio, which is an indicator of oxidative stress, may be altered in the ER compartment [81], where it may affect protein disulfide bridge formation [82] in the ADAM17 molecule. However, redox potentials at the cell surface and in the ER were not different in CF nasal epithelial cells in comparison to CFTR corrected cells, as measured with roGFP1 [83]. But as indicated previously, nasal epithelial cells and bronchial epithelial cells may differ in their properties.

Interestingly, it has been proposed that GSH/GSSG balance may affect the optimal redox potential for PDI to catalyze disulfide bond formation not only at the external surface of the plasma membrane, but also in the ER [82] [83] [84]. PDI has been shown to be affected by GSH/GSSG balance, so it is likely that CFTR related ER oxidative stress [85] may interfere with ER GSH/GSSG balance and thus directly activate PDI/ADAM17 signaling cascade inducing EGFR/ADAM17 axis activity in CF bronchial epithelial cells. Importantly, PDI may also be a therapeutic target, as shown in the context of carcinogenesis [86], so further studies of membrane trafficking are of interest to establish the exact localization and mechanism how PDI and abnormal GSH levels affect ADAM17 activity in CFTR deficient bronchial epithelial cells.

Does ADAM17 shed its substrates at the cell membrane or in the intracellular vesicular compartment?

Most of the reports assume that ADAM17 cleaves it substrates at the extracellular membrane, but this has not been confirmed yet. Lorenzen et al showed that an ADAM17 construct containing the unmodified intracellular part was expressed at the cell surface to a lesser extent than the GPI anchored variant [87]. In accordance with this, ADAMs and its substrates have been localized in various subcellular compartments like lysosomes [88], endosomes [89] [90] and exosomes [91], thus it is difficult to identify an exact site of ADAM17 mediated shedding [28]. We show for the first time by proximity ligation assay (PLA) that upon CS exposure ADAM17 and ADAM17-P appears in close proximity with its substrates AREG and IL-6R in the intracellular compartment in ALI-HBEC, whereas under basal condition ADAM17 or ADAM17-P-substrate complexes were infrequent (Chapter 2). However, our data do not establish whether ADAM17 sheds its substrate in the intracellular vesicles followed by secretion, or whether the ADAM17/substrate complexes in these vesicles need to be transported to the membrane first in order to deliver ADAM17 and its substrate for cleavage. In both cases a vesicle trafficking and membrane fusion event is involved, which may be subject to regulation.

In line with our observation, Soond et al showed that expression of constitutively active MEK-1 and ERK in HeLa cells relocates ADAM17 fused to GFP (ADAM17-GFP) from the ER to intracellular vesicles that are distinct from the ER [35]. Also the phosphorylated form of ADAM17 at position T735 fused to GFP (ADAM17-T735-GFP) co-localized with the ER-Golgi transport marker Sec- 23/COPII and trans-Golgi network protein TGN-46 [35]. Furthermore, Lorenzen et al based on the co-immunoprecipitation of both immature and mature ADAM17 with its substrate suggested that ADAM17 and its substrate meet in the ER/Golgi

pathway, and that the pro-domain does not interfere with the enzyme-substrate interaction [87]. All these data may explain the CS triggered ADAM17-substrate interaction in the intracellular vesicular compartment observed by us (Chapter 2). However, they do not suffice to clarify the molecular regulation of ADAM17-mediated shedding.

According to some reports PKC activation by PMA induces ADAM17 dependent shedding, without increasing ADAM17 at the extracellular membrane [71] [92], whereas other studies did show that PMA stimulation increases expression of ADAM17 at the cell surface and decreases the presence of L-selectin at the cell surface, indicating that the cleavage of ADAM17 substrates occurs at the cell membrane [93]. Further studies by Xu et al. suggested that ADAM17 is localized at the cell surface as a dimer and associates with tissue inhibitor of metalloproteinase-3 (TIMP-3) [34]. Upon activation with PMA or anisomycin, the amount of ADAM17 at the cell surface increases, followed by dissociation of ADAM17 from TIMP-3, and ADAM17 dimers convert to monomers. According to the authors, these monomers correspond to the active form of ADAM17 and are predominantly found in cytoplasm [30], suggesting that the monomers are internalized and shedding may occur intracellularly. Also recent reports suggest that ADAM-mediated cleavage occurs from an intracellular vesicular pool in several cell types [88] [94]. In line with these studies, Doedens et al proposed that endocytosis from the cell surface is a pre-requisite for ADAM17 catalytic activity [95]. As an extension of these studies, [90], two proteins that regulate ADAM17-dependent shedding of EGFR ligands, annexins [96] and a phospho-furin acidic cluster sorting protein 2 (PACS-2) [90] have also been shown by proximity ligation assay (PLA) to be in close proximity to ADAM17 in the intracellular vesicular compartment, again suggesting that the shedding process may occur from intracellular compartment. However, all these data do not provide the complete picture of ADAM17 activation and localization. Thus, in light of other and our findings it is important to address where PDI leads to conversion of ADAM17 conformation, does it affect the interaction of ADAM17 with TIMP3 and what is the role of GSH in these events. These questions should be addressed in general terms as well as in CFTR experimental systems.

Taken together, our data showed that upon activation, ADAM17 or its phosphorylated form interact with the substrate in an intracellular vesicular compartment (Chapter 2). However, further studies focusing on dynamic trafficking events and protein-protein interactions during the activation of the ADAM17 system are required to reveal whether induced ADAM17 proteolytic activity occurs in the intracellular vesicles or requires membrane trafficking. These questions need to be addressed to allow the design of a next generation of drugs targeting ADAM17 activity, that avoids the disadvantages of current systemic ADAM inhibitors [97].

Is dynamic trafficking of EGFR and ADAM17 an additional regulatory mechanism of EGFR/ADAM17 axis activation?

• ADAM17-substrate complexes appear in the intracellular compartment, whereas EGFR internalizes after CS exposure

We showed that not only the localization of ADAM17 and it substrates changes upon activation by CS, but also EGFR undergoes internalization from the basolateral membrane into an intracellular compartment in CS treated ALI-HBEC (Chapter 2, Figure S2), associated with enhanced EGFR phosphorylation [7]. However, we have not established yet whether EGFR-P co-localizes with ADAM17-P-substrate complexes in the intracellular compartment, where they potentially may activate each other. The co-localization of EGFR and ADAM17-substrate complexes and their role in shedding in human airway epithelial cells upon activation is an important research question that should be further investigated. Additionally, the multifunctional sorting protein PACS-2 involved in retrograde trafficking of membrane proteins from endosomes and Golgi not only regulates ADAM17 trafficking toward an active pool and thus shedding of EGFR substrates, but also affects EGFR phosphorylation [90], giving another indication for the importance of co-trafficking as a potential regulatory mechanism of the ADAM17/EGFR axis.

Up till now, EGFR signaling has been believed to occur at the plasma membrane, however, upon activation by ligand binding, EGFR is rapidly internalized into endosomes [98]. Also EGFR ligands undergo intracellular trafficking [9]. Different stress factors lead to accumulation of EGFR in non-degradative endosomes [99] [100] [101] [102], and this endocytic trafficking has been correlated with active signal transduction [103] [104] [105] [91] [106] [107] [108]. We observed that CS exposure causes EGFR internalization (Chapter 2), which may prolong EGFR signaling from an intracellular vesicular pool. Interestingly, inhibition of EGFR with tyrosine kinase inhibitors (AG1478 and BIBW2948BS) prevents EGFinduced EGFR internalization [109]. These data have been confirmed in ex-vivo EGFstimulated airway biopsies form patients who inhaled EGFR inhibitor (BIBW2948 BS) prior to the experiment, which caused diminished EGFR internalization [109]. Thus, our and published data highlight the importance of investigating further the role of EGFR trafficking upon exposure to external stimuli, like CS or in CFTR deficiency. Apart from that, the secretion of EGFR and its ligands into exosomes has been reported [9] [110]. Exosomes containing EGFR or its ligands can be taken up by epithelial, endothelial, and fibroblast cells [91] [110], thus it would be interesting to investigating their role in the trans-signaling.

• EGFR appears in the apical compartment in CFTR deficient cells

The protein trafficking pathways together with tight junctions establish the apicalbasolateral polarity in epithelial cells. EGFR and its ligands are preferentially delivered to the basolateral cell surface and they all have a conserved basolateral sorting motif [9] [111]. Overexpressed EGFR in polarized epithelial kidney cells additionally appears at the apical cell membrane and as a consequence of this mistrafficking, EGFR is less sensitive to downregulation and undergoes less efficient endocytosis, which may result in prolonged EGFR activity [112]. We observed that upon induction of CFTR in CFBE410-ALI cells EGFR is expressed mainly at the basolateral membrane, but in CFTR deficient cells it appears more frequently at the apical side (Chapter 3). Thus, we speculate that this apically expressed receptor may be less sensitive to downregulation [112] and more susceptible to external stimuli. There are several possible scenarios that may explain abnormal EGFR trafficking in CFTR deficiency. Firstly, EGFR trafficking is highly pH sensitive. In acidic endosomes TGF- α dissociates from EGFR which results in recycling of unbound EGFR to the plasma membrane [113] and its exposure to other ligands [113] [111], whereas in an acidic environment binding of betacellulin to EGFR is highly resistant [113]. This suggests that ligand binding and EGFR behavior depends on acidification of its cellular environment. In CF human and CF pig the airway surface liquid (ASL) is acidified [114] [39]. The pH in the lateral compartment, where most EGFR appears to be localized, has not been investigated yet. Also, it is still contested whether endosomal and lysosomal pH differs in primary human airway epithelial cells from CF versus non-CF patients [115] [116] [117]. Therefore, further investigations are required to establish whether vesicular or apical pH affects EGFR trafficking in CFTR deficiency.

Another possible link between CFTR deficiency and abnormal EGFR activity is its effect on common factors regulating EGFR and CFTR trafficking. For instance, Wiskott-Aldrich Syndrome protein N-WASP when inhibited accelerates CFTR internalization in HT29 intestinal adenocarcinoma cells [118], whereas in murine and HeLa cells it impairs EGFmediated endocytosis of EGFR [119] [120]. Some studies also suggest a role of SRC family kinases in EGFR trafficking [102] [121], which also phosphorylate CFTR [122] [123] [124] [125]. Targeting these molecules in CFTR deficient tissues may normalize EGFR trafficking and thus downregulate EGFR signaling. Our observation that EGFR is mislocalized in CF cells (Chapter 3) may be related to the finding that CFTR depletion by siRNA delays EGFR degradation in 16HBE14o- cells. This delayed EGFR degradation can be abrogated by application of the ROS scavenger cystamine or BECN-1 expression that encodes an autophagy activator protein, Beclin-1. [126]. These data are part of a series of studies by Maiuri et al, which implicate CFTR deficiency in ROS related abnormalities in aggresome formation, defective autophagy and inflammatory signaling [127] [128] [129] [130] [131]. Thus, abnormal trafficking of EGFR and other elements of the EGFR/ADAM17 axis observed in CFTR deficiency is potentially linked to ROS induced defective autophagy [128].

Crosstalk between different pro-inflammatory pathways in CF lung inflammation.

The data presented in Chapter 4 and the Appendix, show that not only the EGFR/ADAM17 axis, but multiple pro-inflammatory and tissue remodeling signaling pathways are involved in CF lung disease, in accordance with abundant evidence presented by other authors. The molecular link between CFTR deficiency, the EGFR/ADAM17 axis and these parallel signaling pathways is not immediately obvious and requires further studies.

Multiple bioactive lipids are affected in CF, in particular the sphingosine pathway, probably signaling through the complex S1P receptor system (Chapter 4). Extracellular S1P induces ADAM17 activation via p38MAPK in an immortalized cell model [132]. Enhanced inflammatory responses mediated by interleukin-1-beta (IL-1 β) in F508del CFTR mutant mouse lung provides another node in the signaling web relating CFTR deficiency with inflammation (Appendix). The IL-1 β signaling pathways cross-talks with the EGFR/ADAM17 signaling cascade, in airway epithelial cell lines IL-1R activates ADAM17/EGFR/IL8 signaling [65]. Interestingly, in human chondrocytes extracellular S1P signaling down regulates IL-1 β -

induced phosphorylation of p38MAPK [133] suggesting cross-talk between S1P, IL-1 β and the ADAM17/EGFR pathway. However, how these signaling pathways interact in CF lungs in situ remains to be established.

Current studies in our laboratory also show a change in ceramide species ratio in bronchial lavage fluid (BALF) from preschool CF children, in F508del mutant mouse lung, and in primary bronchial cells from CFTR deficient piglets in ALI culture (Scholte et al in preparation). A CFTR dependent change in the ceramide membrane pool likely plays an important role in membrane protein trafficking and receptor signaling, in particular of EGFR during oxidative stress [134].

Moreover, we observe enhanced lysolecithin (LPC) and lysophosphatidic acid (LPA) concentrations in CF infant BALF, positively correlating with CF lung disease (Scholte et al, in preparation). This is probably related to enhanced activity of phospholipase A activity observed previously in CFTR deficient epithelia [150]. Preliminary studies in our laboratory also show that LPA receptor agonists strongly activate the EGFR/ADAM17 activity and secretion of pro-inflammatory cytokines in human bronchial cells, evidence of strong crosstalk between these pathways, likely contributing to chronic inflammation in the CF lung.

Taken together, these data provide a strong rationale for cross-talk between CFTR related oxidative stress, lysolipid and sphingosine metabolism, IL-1 β and the ADAM17/EGFR pathway in CFTR deficiency. At this time we cannot establish whether there is a simple linear cascade of events, or a more diffuse imbalance of multiple related signaling pathways caused by the CFTR primary defect leading to pro-inflammatory and tissue remodeling signaling. Therefore, this should be further investigated in ALI-HBEC or even better in 3D organotypic culture (lung on a chip),to mimic the conditions *in vivo*, in order to better understand the activation of EGFR/ADAM signaling pathway in CF and COPD lung disease.

Clinical relevance

Our studies show that lung disease in CF and COPD is characterized by exaggerated ADAM17/EGFR signaling, which is a proven druggable target. In a clinically relevant model, air-liquid interface culture of primary epithelial cells, we show that cells from COPD patients respond more pronouncedly to external triggers like CS exposure, and that these responses involve the ADAM17/EGFR axis (Chapter 2). In CFBE410- cells differentiated under ALI, we showed that the ADAM17/EGFR axis is activated as a cell autonomous defect caused by CFTR deficiency (Chapter 3). CFBE cells give a representation of the in vivo situation, however, the data should be verified in differentiated CF primary human airway epithelial cells and more advanced systems.

Considering the involvement of ADAM17 and EGFR in many pathological events observed in CF and COPD [21] [20], like inflammatory responses [135] [136], regeneration [137] [138] and mucus hypersecretion [139] [140], intervention in this signaling pathway could be beneficial to alleviate these symptoms. Importantly, both ADAM17 and EGFR not only mediate inflammatory processes, but also are linked with protective responses to injury

(wound healing), so complete blocking of EGFR or ADAM17 may have serious effects on the anti-inflammatory and pro-regenerative responses. Finding the balance between removing excessive inflammation while keeping the protective responses to infection and injury would be the main goal in CF and COPD therapy. Several inhibitors or antibodies against EGFR [109] [141] and ADAM17 [142] are currently in clinical trials showing positive results mainly in cancer treatment. Initial experiments in our laboratory testing systemic application of the ADAM inhibitors TMI1 and TMI2 (Wyeth) in our mouse model of CF showed only a partial reduction of pro-inflammatory cytokine signaling after a challenge with either naphthalene, causing airway injury, or intranasal LPS/fMLP causing acute neutrophilemia (Scholte et al, unpublished). Further studies should try to find the optimal dose range of available compounds, but also define alternative druggable targets in the ADAM17/EGFR pathway. Furthermore localized delivery methods, avoiding the disadvantages of systemic inhibitors that were initially developed for cancer treatment should be considered. Novel developments include the use of single chain humanized antibodies, and technology that allows targeted delivery of nano-carriers to inflamed tissue.

Our studies also highlight the involvement of oxidative stress (ROS/GSH) as one of the important regulators of EGFR/ADAM axis activity, which offers opportunities for treatment of both CF and COPD. Thus so far such attempts have met with limited success, due to the limitations of the available experimental treatments. Further development of such interventions would be warranted.

Conclusions and future direction

The results provided in this thesis help to explain the activation of the ADAM17/EGFR axis in CF and COPD lung disease. However, to define the druggable targets in this signaling cascade future studies should identify the molecular events leading to activation of ADAM17/EGFR axis in more detail.

Here we showed for the first time that EGFR is a potent regulator of ADAM17 activity and that they both regulate IL-6R and AREG protein release and mRNA synthesis (Chapter 2). As an extension of our results (Chapter 2), studies by Amatngalim et al show the importance of EGFR/ERK1/2/MAPK pathway in CS induced regulation of IL-8 mRNA levels [7]. This supports the hypothesis that EGFR establishes a positive feedback loop with ADAM17 (EGFR/ADAM17 axis). Furthermore, our studies point towards SRC as the druggable candidate in the ADAM17/EXFR axis. It not only affects ADAM17 activity (Chapter 3) [52], but also CFTR can be phosphorylated by tyrosine kinases from the SRC family (Fyn, Lyn, Fgr, Src) in vitro [122] [123] [124] [125]. However, we have not observed an induced SRC phosphorylation level in CFTR deficient compared to CFTR expressing CFBE410- cells (data not shown). Current studies are aimed at SRC regulation in primary differentiated cells in response to oxidative stress.

Since ROS activates ADAM17, EGFR and SRC directly or indirectly [29] [52] [72], and is thought to be induced in CFTR deficiency [128], we questioned whether it is a link between CFTR deficiency and ADAM17 activity. Our data show that airway epithelial cells are oxidized extracellularly, but not in the cytosol. Thus, we propose that extracellular

GSH/GSSG imbalance induces CFTR-mediated EGFR/ADAM17 signaling, contributing to inflammation and tissue remodeling. However, in the light of findings cited above, also a GSH/GSSG imbalance in the ER lumen may affect ADAM17 activity, by PDI-mediated conformational changes of ADAM17 [29] [84]. In line with these speculations, upon CS exposure ADAM17 appears in close proximity with its substrates in the intracellular compartment in ALI-HBEC (Chapter 2). Thus CFTR dependent GSH/GSSH imbalance in ER, may directly affect PDI-mediated ADAM17 activation, bypassing the ADAM17/EGFR axis. Our data show that extracellular GSH completely inhibits ADAM17 dependent AREG shedding, but not EGFR phosphorylation in CFBE cells (Chapter 3). However, further studies should reveal if CFTR deficiency and extracellular GSH affect PDI dependent ADAM17 mediated shedding in the ER lumen or in a related subcellular compartment.,

Based on our data and recent reports, we also speculate that trafficking of EGFR and ADAM17 may be one of the regulatory mechanisms activating EGFR/ADAM17 axis in CF and COPD pathology. Upon CS exposure ADAM17-substrate PLA signal appears in the intracellular compartment in ALI-HBEC (Chapter 2), where also EGFR internalizes. In line with our findings, the presence of EGFR in endosomes and exosomes has been described, similar to ADAM17 and its substrates, however, the mechanism of the EGFR/ADAM17 signaling via a vesicular compartment is still poorly understood. Also apical mislocalization of EGFR induced by CFTR deficiency in CFBE41o- cells (Chapter 3) raises the question how CFTR deficiency affects the EGFR and ADAM17 localization and thus whether it has consequence for activation of the EGFR/ADAM17 axis in CF pathology. Thus, validation of the CFTR-dependent EGFR trafficking to the apical membrane in primary CF bronchial epithelial cells would be important. So far several potentially common regulators of EGFR and CFTR endocytic trafficking have been identified, like an autophagy activator Becline-1 [128] [143] [144], [145], Transglutaminase 2 (TG2) [127] [128] [146] [147] [148], p62 [128] [126] and Rab5 [126] [149], providing potential druggable targets for mistrafficked EGFR in CFTR deficiency, which should be tested in primary CF airway epithelial cells.

Taken together, we highlight the need to further investigate the spatiotemporal regulation of the EGFR/ADAM17 axis, localize the place of shedding of the ADAM17 substrates, the role of redox potential (measured as GSH/GSSG ratio) and the druggable intermediates in the ADAM17/EGFR pathway (membrane/intracellular) in CFTR deficiency and upon exposure to stress factors like CS. This is also crucial in order to define the methods of potential drug delivery to the airways (permeable/impermeable agents).

LITERATURE

- 1. Hiemstra PS, McCray PB, Bals R. The innate immune function of airway epithelial cells in inflammatorylung disease. Eur Respir J 2015; 45: 1150–1162.
- 2. Gras D, Chanez P, Vachier I, Petit A, Bourdin A. Bronchial epithelium as a target for innovative treatments in asthma. Pharmacol Ther 2013; 140: 290–305.
- 3. Zhu L, Lee P, Lee W, Zhao Y, Yu D, Chen Y. Rhinovirus-induced major airway mucin production involves a novel TLR3-EGFR–dependent pathway. Am J Respir Cell Mol Biol 2009; 40: 610–619.
- Gomez MI, Sokol SH, Muir AB, Soong G, Bastien J, Prince AS. Bacterial Induction of TNF-α Converting Enzyme Expression and IL-6 Receptor Shedding Regulates Airway Inflammatory Signaling. J Immunol 2005; 175: 1930–1936.
- 5. Hudy MH, Proud D. Cigarette smoke enhances human rhinovirus-induced CXCL8 production via HuR-mediated mRNA stabilization in human airway epithelial cells. Respir Res 2013; 14: 88.
- 6. Parnia S, Hamilton LM, Puddicombe SM, Holgate ST, Frew AJ, Davies DE. Autocrine ligands of the epithelial growth factor receptor mediate inflammatory responsesto diesel exhaust particles. Respir Res 2014; 15: 1–10.
- 7. Amatngalim GD, van Wijck Y, de Mooij-Eijk Y, Verhoosel RM, Harder J, Lekkerkerker AN, Janssen RAJ, Hiemstra PS. Basal Cells Contribute to Innate Immunity of the Airway Epithelium through Production of the Antimicrobial Protein RNase 7. J Immunol 2015; 194: 3340–3350.
- 8. Robb CT, Regan KH, Dorward DA, Rossi AG. Key mechanisms governing resolution of lung inflammation. Semin Immunopathol 2016; 38: 425–448.
- 9. Singh B, Coffey RJ. Trafficking of Epidermal Growth Factor Receptor Ligands in Polarized Epithelial Cells. Annu Rev Physiol 2014; 76: 275–300.
- Lacroix M, Rousseau F, Guilhot F, Malinge P, Magistrelli G, Herren S, Jones SA, Jones GW, Scheller J, Lissilaa R, Kosco-Vilbois M, Johnson Z, Buatois V, Ferlin W. Novel Insights into Interleukin 6 (IL-6) Cis- and Trans-signaling Pathways by Differentially Manipulating the Assembly of the IL-6 Signaling Complex. J Biol Chem 2015; 290: 26943–26953.
- 11. Adam D, Roux-Delrieu J, Luczka E, Bonnomet A, Lesage J, Mérol JC, Polette M, Abély M, C C. Cystic fibrosis airway epithelium remodeling: involvement of inflammation. J Pathol 2015; 235: 408–419.
- 12. Pini L, Tiberio L, Tantucci C, Modina D, Pinelli V, Bezzi M. Central airways remodeling in COPD patients. Int J Chron Obstruct Pulmon Dis 2014; 9: 927–932.
- Boon M, Verleden SE, Bosch B, Lammertyn EJ, McDonough JE, Mai C, Verschakelen J, de Corput MK-V, Harm A W Tiddens, Proesmans M, Vermeulen FL, Verbeken EK, Cooper J, Van Raemdonck DE, Decramer M, Verleden GM, Hogg JC, Dupont LJ, Vanaudenaerde BM, De Boeck K. Morphometric analysis of explant lungs in cystic fibrosis. Am J Respir Crit Care Med 2016; 193: 516–526.

- 14. Scheller J, Chalaris A, Garbers C, Rose-John S. ADAM17: a molecular switch to control inflammation and tissue regeneration. Trends Immunol 2011; 32: 380–387.
- 15. Kasina S, Scherle PA, Hall CL, Macoska JA. ADAM-mediated amphiregulin shedding and EGFR transactivation. Cell Proliferat 2009; 42: 799–812.
- 16. Rose-John S. IL-6 Trans-Signaling via the Soluble IL-6 Receptor: Importance for the Pro-Inflammatory Activities of IL-6. Int J Biol Sci 2012; 8: 1237–1247.
- 17. Wang Y, Hu H, Wu J, Zhao X, Zhen Y, Wang S, Li W, Liang M, Bin Wu, Ma G. The IL6R gene polymorphisms are associated with sIL-6R, IgE and lung function in Chinese patients with asthma. Gene 2016; 585: 51–57.
- Pérez-Rubio G, Silva-Zolezzi I, Fernández-López JC, Camarena Á, Velázquez-Uncal M, Morales-Mandujano F, Hernández-Zenteno RDJ, Flores-Trujillo F, Sánchez-Romero C, Velázquez-Montero A, Espinosa de los Monteros C, Sansores RH, Ramírez-Venegas A, Falfán-Valencia R. Genetic Variants in IL6Rand ADAM19are Associated with COPD Severity in a Mexican Mestizo Population. COPD 2016; 13: 610–615.
- 19. Adib-Conquy M, Thierry Pedron, Petit-Bertron A-F, Tabary O, Harriet Corvol, Jacquot J, Annick Clément, Cavaillon J-M. Neutrophils in cystic fibrosis display a distinct gene expression pattern. Mol. Med. 2008; 14: 36–44.
- 20. Dreymueller D, Uhlig S, Ludwig A. ADAM-family metalloproteinases in lung inflammation: potential therapeutic targets. Am J Physiol Lung Cell Mol Physiol 2015; 308: L325–L343.
- 21. Vallath S, Hynds RE, Succony L, Janes SM, Giangreco A. Targeting EGFR signaling in chronic lung disease: therapeutic challenges and opportunities. Eur Respir J 2014; 44: 513–522.
- Johannesson B, Hirtz S, Schatterny J, Schultz C, Mall MA. CFTR Regulates Early Pathogenesis of Chronic Obstructive Lung Disease in βENaC-Overexpressing Mice. PLoS ONE 2012; 7: e44059.
- 23. Rab A, Rowe SM, Raju SV, Bebok Z, Matalon S, Collawn JF. Cigarette smoke and CFTR: implications in the pathogenesis of COPD. Am J Physiol Lung Cell Mol Physiol 2013; 305: L530–L541.
- 24. Mall MA, Hartl D. CFTR: cystic fibrosis and beyond. Eur Respir J 2014; 44: 1042– 1054.
- 25. Clunes LA, Davies CM, Coakley RD, Aleksandrov AA, Henderson AG, Zeman KL, Worthington EN, Gentzsch M, Kreda SM, Cholon D, Bennett WD, Riordan JR, Boucher RC, Tarran R. Cigarette smoke exposure induces CFTR internalization and insolubility, leading to airway surface liquid dehydration. FASEB J 2012; 26: 533–545.
- Sloane PA, Shastry S, Wilhelm A, Courville C, Tang LP, Backer K, Levin E, Raju SV, Li Y, Mazur M, Byan-Parker S, Grizzle W, Sorscher EJ, Dransfield MT, Rowe SM. A Pharmacologic Approach to Acquired Cystic Fibrosis Transmembrane Conductance Regulator Dysfunction in Smoking Related Lung Disease. PLoS ONE 2012; 7:
e39809.

- 27. Gooz M. ADAM-17: the enzyme that does it all. Crit Rev Biochem Mol Biol 2010; 45: 146–169.
- 28. Murphy G. Regulation of the proteolytic disintegrin metalloproteinases, the "Sheddases." Semin Cell Dev Biol 2009; 20: 138–145.
- 29. Düsterhöft S, Jung S, Hung C-W, Tholey A, Sönnichsen FD, Grötzinger J, Lorenzen I. Membrane-Proximal Domain of a Disintegrin and Metalloprotease-17 Represents the Putative Molecular Switch of Its Shedding Activity Operated by Proteindisulfide Isomerase. J Am Chem Soc 2013; 135: 5776–5781.
- 30. Xu P, Liu J, Sakaki-Yumoto M, Derynck R. TACE Activation by MAPK-Mediated Regulation of Cell Surface Dimerization and TIMP3 Association. Sci Signal 2012; 5: ra34.
- 31. Yu H, Zhou X, Wen S, Xiao Q. Flagellin/TLR5 responses induce mucus hypersecretion by activating EGFR via an epithelial cell signaling cascades. Exp Cell Res 2012; 318: 723–731.
- 32. Gschwind A, Hart S, Fischer OM, Ullrich A. TACE cleavage of proamphiregulin regulates GPCR-induced proliferation and motility of cancer cells. EMBO J 2003; 22: 2411–2421.
- 33. Yin J, Yu FSX. ERK1/2 Mediate Wounding- and G-protein-Coupled Receptor Ligands-Induced EGFR Activation via Regulating ADAM17 and HB-EGF Shedding. Invest Opthalmol Vis Sci 2009; 50: 132–139.
- Xu P, Derynck R. Direct Activation of TACE-Mediated Ectodomain Shedding by p38 MAP Kinase Regulates EGF Receptor-Dependent Cell Proliferation. Mol Cell 2010; 37: 551–566.
- 35. Soond SM, Everson B, Riches DWH, Murphy G. ERK-mediated phosphorylation of Thr735 in TNFα-converting enzyme and its potential role in TACE protein trafficking. J Cell Sci 2005; 118: 2371–2380.
- 36. Fricker M, Deane A, Hansbro PM. Animal models of chronic obstructive pulmonary disease. Expert Opin Drug Discov 2014; 9: 629–645.
- 37. Garbers C, Jänner N, Chalaris A, Moss ML, Floss DM, Meyer D, Koch-Nolte F, Rose-John S, Scheller J. Species Specificity of ADAM10 and ADAM17 Proteins in Interleukin-6 (IL-6) Trans-signaling and Novel Role of ADAM10 in Inducible IL-6 Receptor Shedding. J Biol Chem 2011; 286: 14804–14811.
- 38. Schumacher N, Meyer D, Mauermann A, Heyde von der J, Wolf J, Schwarz J, Knittler K, Murphy G, Michalek M, Garbers C, Bartsch JW, Guo S, Schacher B, Eickholz P, Chalaris A, Rose-John S, Rabe B. Shedding of Endogenous Interleukin-6 Receptor (IL-6R) Is Governed by A Disintegrin and Metalloproteinase (ADAM) Proteases while a Full-length IL-6R Isoform Localizes to Circulating Microvesicles. J Biol Chem 2015; 290: 26059–26071.
- 39. Shah VS, Meyerholz DK, Tang XX, Reznikov L, Abou Alaiwa M, Ernst SE, Karp PH,

Wohlford-Lenane CL, Heilmann KP, Leidinger MR, Allen PD, Zabner J, McCray PB, Ostedgaard LS, Stoltz DA, Randak CO, Welsh MJ. Airway acidification initiates host defense abnormalities in cystic fibrosis mice. Science 2016; 351: 503–507.

- 40. Wilke M, Buijs-Offerman RM, Aarbiou J, Colledge WH, Sheppard DN, Touqui L, Bot A, Jorna H, de Jonge HR, Scholte BJ. Mouse models of cystic fibrosis: Phenotypic analysis and research applications. J Cyst Fibros 2011; 10: S152–S171.
- 41. Thorne D, Adamson J. A review of in vitro cigarette smoke exposure systems. Exper Toxicol Pathol 2013; 65: 1183–1193.
- 42. Tosoni K, Cassidy D, Kerr B, Land SC, Mehta A. Using Drugs to Probe the Variability of Trans-Epithelial Airway Resistance. PLoS ONE 2016; 11: e0149550.
- 43. Zertal-Zidani S, Busiah K, Edelman A, Polak M, Scharfmann R. Small-molecule inhibitors of the cystic fibrosis transmembrane conductance regulator increase pancreatic endocrine cell development in rat and mouse. Diabetologia 2012; 56: 330–339.
- Kelly M, Trudel S, Brouillard F, Bouillaud F, Colas J, Nguyen-Khoa T, Ollero M, Edelman A, Fritsch J. Cystic Fibrosis Transmembrane Regulator Inhibitors CFTRinh-172 and GlyH-101 Target Mitochondrial Functions, Independently of Chloride Channel Inhibition. J Pharmacol Exp Ther 2010; 333: 60–69.
- 45. Heijink IH, Brandenburg SM, Noordhoek JA, Slebos D-J, Postma DS, van Oosterhout AJ. Role of aberrant metalloproteinase activity in thepro-inflammatory phenotype of bronchialepithelium in COPD. Respir Res BioMed 2011; 12: 110.
- 46. Shao MXG, Nakanaga T, Nadel JA. Cigarette smoke induces MUC5AC mucin overproduction via tumor necrosis factor-α-converting enzyme in human airway epithelial (NCI-H292) cells. Am J Physiol Lung Cell Mol Physiol 2004; 287: L420– L427.
- 47. Lemjabbar-Alaoui H, Sidhu SS, Mengistab A, Gallup M, Basbaum C. TACE/ADAM-17 Phosphorylation by PKC-Epsilon Mediates Premalignant Changes in Tobacco Smoke-Exposed Lung Cells. PLoS ONE 2011; 6: e17489.
- Lemjabbar H, Li D, Gallup M, Sidhu S, Drori E, Basbaum C. Tobacco Smoke-induced Lung Cell Proliferation Mediated by Tumor Necrosis Factor α-converting Enzyme and Amphiregulin. J Biol Chem 2003; 278: 26202–26207.
- 49. Richter A, O'Donnell RA, Powell RM. Autocrine ligands for the epidermal growth factor receptor mediate interleukin-8 release from bronchial epithelial cells in response to cigarette smoke. Am J Respir Crit Care Med 2002; 27: 85–90.
- 50. Petecchia L, Sabatini F, Usai C, Caci E, Varesio L, Rossi GA. Cytokines induce tight junction disassemblyin airway cells via an EGFR-dependent MAPK/ERK1/2-pathway. Lab Invest 2012; 92: 1140–1148.
- 51. Shaykhiev R, Zuo W-L, Chao I, Fukui T, Witover B, Brekman A, Crystal RG. EGF shifts human airway basal cell fate toward a smoking-associated airway epithelial phenotype. PNAS 2013; 110: 12102–12107.

- 52. Sham D, Wesley UV, Hristova M, van der Vliet A. ATP-Mediated Transactivation of the Epidermal Growth Factor Receptor in Airway Epithelial Cells Involves DUOX1-Dependent Oxidation of Src and ADAM17. PLoS ONE 2013; 8: e54391.
- 53. Breshears LM, Schlievert PM, Peterson ML. A Disintegrin and Metalloproteinase 17 (ADAM17) and Epidermal Growth Factor Receptor (EGFR) Signaling Drive the Epithelial Response to Staphylococcus aureus Toxic Shock Syndrome Toxin-1 (TSST-1). J Biol Chem 2012; 287: 32578–32587.
- 54. Baumgart A, Seidl S, Vlachou P, Michel L, Mitova N, Schatz N, Specht K, Koch I, Schuster T, Grundler R, Kremer M, Fend F, Siveke JT, Peschel C, Duyster J, Dechow T. ADAM17 Regulates Epidermal Growth Factor Receptor Expression through the Activation of Notch1 in Non-Small Cell Lung Cancer. Cancer Res 2010; 70: 5368– 5378.
- 55. Sundar IK, Rahman I. Gene expression profiling of epigenetic chromatin modification enzymes and histone marks by cigarette smoke: implications for COPD and lung cancer. Am J Physiol Lung Cell Mol Physiol 2016; 311: L1245–L1258.
- 56. McKiernan PJ, Greene CM. MicroRNA Dysregulation in Cystic Fibrosis. Mediators Inflam 2015; 2015: 1–7.
- 57. KARA M, KIRKIL G, KALEMCI S. Differential Expression of MicroRNAs in Chronic Obstructive Pulmonary Disease. Adv Clin Exp Med 2016; 25: 21–26.
- 58. Pietrusińska M, Pająk A, Górski P, Kuna P, Szemraj J, Goździńska-Nielepkowicz A, Pietras T. Preliminary studies: differences in microRNA expression in asthma and chronic obstructive pulmonary disease. Postepy Dermatol Alergol 2016; 4: 276–280.
- 59. Alipoor SD, Adcock IM, Garssen J, Mortaz E, Varahram M, Mirsaeidi M, Velayati A. The roles of miRNAs as potential biomarkers in lung diseases. Eur J Pharma 2016; 791: 395–404.
- 60. Doberstein K, Steinmeyer N, Hartmetz A-K, Eberhardt W, Mittelbronn M, Harter PN, Juengel E, Blaheta R, Pfeilschifter J, Gutwein P. MicroRNA-145 Targets the Metalloprotease ADAM17 and Is Suppressed in Renal Cell Carcinoma Patients. Neoplasia 2013; 15: 218–IN231.
- 61. Jiang-Tao C, Kun-Hou Y, Long H, Li-Ping Z, Chen-Yu W, Jun-Jie Z. miR-338-3p inhibits the proliferation and migration of gastric cancer cells by targeting ADAM17. Int J Clin Exp Pathol 2015; 8: 10922–10928.
- 62. Cai M, Wang Z, Zhang J, Zhou H, Jin L, Bai R, Weng Y. Adam17, a Target of Mir-326, Promotes Emt-Induced Cells Invasion in Lung Adenocarcinoma. Cell Physiol Biochem 2015; 36: 1175–1185.
- 63. Meng F, Wang F, Wang L, Wong SCC, Cho WCS, Chan LWC. MiR-30a-5p Overexpression May Overcome EGFR-Inhibitor Resistance through Regulating PI3K/AKT Signaling Pathway in Non-small Cell Lung Cancer Cell Lines. Front Genet 2016; 7: e05005.
- 64. Wang F, Meng F, Wang L, Wong SCC, Cho WCS, Chan LWC. Associations of

mRNA:microRNA for the Shared Downstream Molecules of EGFR and Alternative Tyrosine Kinase Receptors in Non-small Cell Lung Cancer. Front Genet 2016; 7: e05005.

- 65. Kim S, Beyer BA, Lewis C, Nadel JA. Normal CFTR Inhibits Epidermal Growth Factor Receptor-Dependent Pro-Inflammatory Chemokine Production in Human Airway Epithelial Cells. PLoS ONE 2013; 8: e72981.
- 66. Raju SV, Solomon GM, Dransfield MT, Rowe SM. Acquired Cystic Fibrosis Transmembrane Conductance Regulator Dysfunction in Chronic Bronchitis and Other Diseases of Mucus Clearance. Clin Chest Med. 2016; 37: 147–158.
- Raju SV, Lin VY, Liu L, McNicholas CM, Karki S, Sloane PA, Tang L, Jackson PL, Wang W, L W, Macon KJ, Mazur M, Kappes J, DeLucas LJ, Barnes S, Kirk K, Tearney GT, Rowe SM. The CFTR Potentiator Ivacaftor Augments Mucociliary Clearance Abrogating CFTR Inhibition by Cigarette Smoke. Am J Respir Cell Mol Biol 2016.
- 68. Rogan MP, Stoltz DA, Hornick DB. Cystic Fibrosis Transmembrane Conductance Regulator Intracellular Processing, Trafficking, and Opportunities for Mutation-Specific Treatment. Chest 2011; 139: 1480–1490.
- 69. Cholon DM, Quinney NL, Fulcher ML, Esther CR, Das J, Dokholyan NV, Randell SH, Boucher RC, Gentzsch M. Potentiator ivacaftor abrogates pharmacological correction of F508 CFTR in cystic fibrosis. Sci Transl Med 2014; 6: 246ra96.
- 70. Veit G, Avramescu RG, Perdomo D, Phuan PW, Bagdány M, Apaja PM, Borot F, Szollosi D, Wu YS, Finkbeiner WE, Hegedus T, Verkman AS, Lukacs GL. Some gating potentiators, including VX-770, diminish ΔF508-CFTR functional expression. Sci Transl Med 2014; 6(246):246ra97
- Willems SH, Tape CJ, Stanley PL, Taylor NA, Mills IG, Neal DE, McCafferty J, Murphy G. Thiol isomerases negatively regulate the cellular shedding activity of ADAM17. Biochem 2010; 428: 439–450.
- 72. Paulsen CE, Truong TH, Garcia FJ, Homann A, Gupta V, Leonard SE, Carroll KS. Peroxide-dependent sulfenylation of the EGFR catalytic site enhances kinase activity. Nat Chem Biol 2011; 8: 57–64.
- 73. Zhang H, Davies KJA, Forman HJ. TGFβ1 rapidly activates Src through a noncanonical redox signaling mechanism. Arch Biochem Biophys 2015; 568: 1–7.
- 74. Ghezzi P. Role of glutathione in immunity and inflammation in the lung. IJGM 2011;4: 105–113.
- 75. Gutscher M, Pauleau A-L, Marty L, Brach T, Wabnitz GH, Samstag Y, Meyer AJ, Dick TP. Real-time imaging of the intracellular glutathione redox potential. Nat Meth 2008; 5: 553–559.
- 76. Morgan B, Sobotta MC, Dick TP. Measuring EGSH and H2O2 with roGFP2-based redox probes. Free Radic Biol Med 2011; 51: 1943–1951.
- 77. Ziady AG, Hansen J. Redox balance in cystic fibrosis. Int J Biochem Cell Biol 2014; 52: 113–123.

- 78. Gould NS, Min E, Martin RJ, Day BJ. CFTR is the primary known apical glutathione transporter involved in cigarette smoke-induced adaptive responses in the lung Free Radic Biol Med 2012; 52: 1201–1206.
- 79. Roum JH, Buhl R, McElvaney NG, Borok Z, Crystal RG. Systemic deficiency of glutathione in cystic fibrosis. J Appl Physiol 1993; 75: 2419–2424.
- 80. Hartl D, Starosta V, Maier K, Beck-Speier I, Rebhan C, Becker BF, Latzin P, Fischer R, Ratjen F, Huber RM, Rietschel E, Krauss-Etschmann S, Griese M. Inhaled glutathione decreases PGE 2 and increases lymphocytes in cystic fibrosis lungs. Free Radic Biol Med 2005; 39: 463–472.
- 81. Birk J, Meyer M, Aller I, Hansen HG, Odermatt A, Dick TP, Meyer AJ, Appenzeller-Herzog C. Endoplasmic reticulum: reduced and oxidized glutathione revisited. J Cell Sci 2013; 126: 1604–1617.
- Chakravarthi S, Jessop CE, Bulleid NJ. The role of glutathione in disulphide bond formation and endoplasmic-reticulum-generated oxidative stress. EMBO Rep 2006; 7: 271–275.
- 83. Schwarzer C, Illek B, Suh JH, Remington SJ, Fischer H, Machen TE. Organelle redox of CF and CFTR-corrected airway epithelia. Free Radic Biol Med 2007; 43: 300–316.
- 84. Appenzeller-Herzog C. Glutathione- and non-glutathione-based oxidant control in the endoplasmic reticulum. J Cell Sci 2011; 124: 847–855.
- 85. Tang AC, Aabida Saferali, He G, Sandford AJ, Strug LJ, Turvey SE. Endoplasmic reticulum stress regulates chemokine production in cystic fibrosis airway cells through STAT3 modulation. J Infect Dis 2016; [Epub ahead of print]: 1–31.
- 86. Grek C, Townsend DM. Protein Disulfide Isomerase Superfamily in Disease and the Regulation of Apoptosis. ERSD 2014; 1: 4–17.
- Lorenzen I, Lokau J, Düsterhöft S, Trad A, Garbers C, Scheller J, Rose-John S, Grötzinger J. The membraneproximal domain of A Disintegrin and Metalloprotease 17 (ADAM17) is responsible for recognition of the interleukin6 receptor and interleukin1 receptor II. FEBS Lett 2012; 586: 1093–1100.
- Ebsen H, Lettau M, Kabelitz D, Janssen O. Subcellular localization and activation of ADAM proteases in the context of FasL shedding in T lymphocytes. Mol Immunol 2015; 65: 416–428.
- 89. Gephart JD, Singh B, Higginbotham JN, Franklin JL, Gonzalez A, Fölsch H, Coffey RJ. Identification of a Novel Mono-Leucine Basolateral Sorting Motif Within the Cytoplasmic Domain of Amphiregulin. Traffic 2011; 12: 1793–1804.
- 90. Dombernowsky SL, e-Petersen JSO, Petersen CH, Instrell R, Hedegaard A-MB, Thomas L, Atkins KM, Auclair S, Albrechtsen R, Mygind KJ, hlich CFO, Howell M, Parker P, Thomas G, Kveiborg M. The sorting protein PACS-2 promotes ErbB signaling by regulating recycling of the metalloproteinase ADAM17. Nat Comms 2015; 6: 7518.
- 91. Higginbotham JN, Beckler MD, Gephart JD, Franklin JL, Bogatcheva G, Kremers G-J,

Piston DW, Ayers GD, McConnell RE, Tyska MJ, Coffey RJ. Amphiregulin Exosomes Increase Cancer Cell Invasion. Curr Biol 2011; 21: 779–786.

- 92. Horiuchi K, Le Gall S, Schulte M, Yamaguchi T, Reiß K, Murphy G, Toyama Y, Hartmann D, Saftig P, Blobel CP. Substrate Selectivity of Epidermal Growth Factor-Receptor Ligand Sheddases and their Regulation by Phorbol Esters and Calcium Influx. Mol Biol Cell 2007; 18: 176–188.
- 93. Killock DJ, Ivetić A. The cytoplasmic domains of TNFα-converting enzyme (TACE/ADAM17) and L-selectin are regulated differently by p38 MAPK and PKC to promote ectodomain shedding. Biochem J 2010; 428: 293–304.
- 94. Gutwein P, Mechtersheimer S, Riedle S, Stoeck A, Gast D, Joumaa S, Zentgraf H, Fogel M, Altevogt P. ADAM10-mediated cleavage of L1 adhesion molecule at the cell surface and in released membrane vesicles. FASEB J 2002; 17: 292–294.
- 95. Doedens JR, Black RA. Stimulation-induced down-regulation of tumor necrosis factor-α converting enzyme. J Biol Chem 2000; 275: 14598–14607.
- 96. Tsukamoto H, Tanida S, Ozeki K, Ebi M, Mizoshita T, Shimura T, Mori Y, Kataoka H, Kamiya T, Fukuda S, Higashiyama S, Joh T. Annexin A2 Regulates A Disintegrin and Metalloproteinase 17–mediated Ectodomain Shedding of Pro–Tumor Necrosis Factor-α in Monocytes and Colon Epithelial Cells. Inflamm Bowel Dis 2013; 19: 1365–1373.
- 97. Arribas J, Esselens C. ADAM17 as a Therapeutic Target in Multiple Diseases. CPD 2009; 15: 2319–2335.
- 98. Tomas A, Futter CE, Eden ER. EGF receptor trafficking: consequences for signaling and cancer. Trends Cell Biol 2014; 24: 26–34.
- 99. Tan X, Lambert PF, Rapraeger AC, Anderson RA. Stress-Induced EGFR Trafficking: Mechanisms, Functions, and Therapeutic Implications. Trends Cell Biol 2016; 28: 352–366.
- 100. Zwang Y, Yarden Y. p38 MAP kinase mediates stress-induced internalization of EGFR: implications for cancer chemotherapy. EMBO J EMBO Press; 2006; 25: 4195–4206.
- 101. Walsh AM, Lazzara MJ. Differential parsing of EGFR endocytic flux among parallel internalization pathways in lung cancer cells with EGFR-activating mutations. Integr Biol 2014; 6: 312–323.
- 102. Dittmann K, Mayer C, Kehlbach R, Rodemann HP. Radiation-induced caveolin-1 associated EGFR internalization is linked with nuclear EGFR transport and activation of DNA-PK. Mol Cancer 2008; 7: 69.
- 103. Huang F, Khvorova A, Marshall W, Sorkin A. Analysis of Clathrin-mediated Endocytosis of Epidermal Growth Factor Receptor by RNA Interference. J Biol Chem 2004; 279: 16657–16661.
- 104. Sigismund S, Argenzio E, Tosoni D, Cavallaro E, Polo S, Di Fiore PP. Clathrin-Mediated Internalization Is Essential for Sustained EGFR Signaling but Dispensable

for Degradation. Dev Cell 2008; 15: 209-219.

- 105. Rappoport JZ, Simon SM. Endocytic trafficking of activated EGFR is AP-2 dependent and occurs through preformed clathrin spots. J Cell Sci 2009; 122: 1301–1305.
- 106. Goh LK, Huang F, Kim W, Gygi S, Sorkin A. Multiple mechanisms collectively regulate clathrin-mediated endocytosis of the epidermal growth factor receptor. J Cell Biol 2010; 189: 871–883.
- 107. Henriksen L, Grandal MV, Knudsen SLJ, van Deurs B, Grøvdal LM. Internalization Mechanisms of the Epidermal Growth Factor Receptor after Activation with Different Ligands. PLoS ONE 2013; 8: e58148.
- 108. Jones S, Rappoport JZ. Interdependent epidermal growth factor receptor signaling and trafficking. Int J Biochem Cell Biol 2014; 51: 23–28.
- 109. Wolff M, Tetzlaff K, Nivens MC, Schneider F-J, Jung B, Hohlfeld J, Heilker R. In vivo inhibition of epidermal growth factor receptor autophosphorylation prevents receptor internalization. Exp Cell Res 2011; 317: 42–50.
- 110. Lee H-H, Wang YN, Hung MC. Non-canonical signaling mode of the epidermal growth factor receptor family. Am J Cancer Res 2015. p. 2944–2958.
- 111. Singh B, Bogatcheva G, Washington MK, Coffey RJ. Transformation of polarized epithelial cells by apical mistrafficking of epiregulin. PNAS 2013; 110: 8960–8965.
- 112. Kuwada SK, Lund KA, Li XF, Cliften P, Amsler K, Opresko LK, Wiley HS. Differential signaling and regulation of apical vs. basolateral EGFR in polarized epithelial cells. Am J Physiol Cell Physiol 1998; 275: C1419–C1428.
- 113. Roepstorff K, Grandal MV, Henriksen L, Knudsen SLJ, Lerdrup M, Grøvdal L, Willumsen BM, van Deurs B. Differential Effects of EGFR Ligands on Endocytic Sorting of the Receptor. Traffic 2009; 10: 1115–1127.
- 114. Luckie DB, Van Alst AJ, Massey MK, Flood RD, Shah AA, Malhotra V, Kozel BJ. Chemical rescue of DF508-CFTR in C127 epithelial cells reverses aberrant extracellular pH acidification to wild-type alkalization as monitored by microphysiometry. Biochem Biophys Res Commun 2014; 451: 535–540.
- 115. Haggie PM, Verkman AS. Unimpaired Lysosomal Acidification in Respiratory Epithelial Cells in Cystic Fibrosis. J Biol Chem 2009; 284: 7681–7686.
- 116. Haggie PM, Verkman AS. Defective organellar acidification as a cause of cystic fibrosis lung disease: reexamination of a recurring hypothesis. Am J Physiol Lung Cell Mol Physiol 2009; 296: L859–L867.
- 117. Di A, Brown ME, Deriy LV, Li C, Szeto FL, Chen Y, Huang P, Tong J, Naren AP, Bindokas V, Palfrey HC, Nelson DJ. CFTR regulates phagosome acidification in macrophages and alters bactericidal activity. Nat Cell Biol 2006; 8: 933–944.
- 118. Ganeshan R, Nowotarski K, Di A, Nelson DJ, Kirk KL. CFTR surface expression and chloride currents are decreased by inhibitors of N-WASP and actin polymerization. Biochem Biophys Acta 2007; 1773: 192–200.

- 119. Galovic M, Xu D, Areces LB, van der Kammen R, Innocenti M. Interplay between N-WASP and CK2 optimizes clathrin-mediated endocytosis of EGFR. J Cell Sci 2011; 124: 2001–2012.
- 120. Benesch S. N-WASP deficiency impairs EGF internalization and actin assembly at clathrin-coated pits. J Cell Sci 2005; 118: 3103–3115.
- 121. Lo H-W, Hsu S-C, Ali-Seyed M, Gunduz M, Xia W, Wei Y, Bartholomeusz G, Shih J-Y, Hung M-C. Nuclear interaction of EGFR and STAT3 in the activation of the iNOS/NO pathway. Cell Cancer 1999; 6: 575–589.
- 122. Cesaro L, Marin O, Venerando A, Donella-Deana A, Pinna LA. Phosphorylation of cystic fibrosis transmembrane conductance regulator (CFTR) serine-511 by the combined action of tyrosine kinases and CK2: the implication of tyrosine-512 and phenylalanine-508. Amino Acids 2013; 45: 1423–1429.
- 123. Venerando A, Cesaro L, Marin O, Donella-Deana A, Pinna LA. A "SYDE" effect of hierarchical phosphorylation: possible relevance to the cystic fibrosis basic defect. Cell Mol Life Sci 2014; 71: 2193–2196.
- 124. Billet A, Luo Y, Balghi H, Hanrahan JW. Role of Tyrosine Phosphorylation in the Muscarinic Activation of the Cystic Fibrosis Transmembrane Conductance Regulator (CFTR). J Biol Chem 2013; 288: 21815–21823.
- 125. Billet A, Jia Y, Jensen T, Riordan JR, Hanrahan JW. Regulation of the cystic fibrosis transmembrane conductance regulator anion channel by tyrosine phosphorylation. FASEB J 2015; 29: 3945–3953.
- 126. Villella VR, Esposito S, Bruscia EM, Vicinanza M, Cenci S, Guido S, Pettoello-Mantovani M, Carnuccio R, De Matteis MA, Luini A, Maiuri MC, Raia V, Kroemer G, Maiuri L. Disease-relevant proteostasis regulation of cystic fibrosis transmembrane conductance regulator. Cell Death Differ 2013; 20: 1101–1115.
- 127. Maiuri L, Luciani A, Giardino I, Raia V, Villella VR, D'Apolito M, Pettoello-Mantovani M, Guido S, Ciacci C, Cimmino M, Cexus ON, Londei M, Quaratino S. Tissue Transglutaminase Activation Modulates Inflammation in Cystic Fibrosis via PPAR Down-Regulation. J Immunol 2008; 180: 7697–7705.
- 128. Luciani A, Villella VR, Esposito S, Brunetti-Pierri N, Medina D, Settembre C, Gavina M, Pulze L, Giardino I, Pettoello-Mantovani M, D'Apolito M, Guido S, Masliah E, Spencer B, Quaratino S, Raia V, Ballabio A, Maiuri L. Defective CFTR induces aggresome formation and lung inflammation in cystic fibrosis through ROS-mediated autophagy inhibition. Nat Cell Biol 2010; 12: 863–875.
- 129. Luciani A, Villella VR, Esposito S, Brunetti-Pierri N, Medina DL, Settembre C, Gavina M, Raia V, Ballabio A, Maiuri L. Cystic fibrosis: A disorder with defective autophagy. Autophagy 2011; 7: 104–106.
- 130. Luciani A, Villella VR, Vasaturo A, Giardino I, Raia V, Pettoello-Mantovani M, D'Apolito M, Guido S, Leal T, Quaratino S, Maiuri L. Correction: SUMOylation of Tissue Transglutaminase as Link between Oxidative Stress and Inflammation. J Immunol 2014; 193: 5347–5349.

- 131. Tosco A, De Gregorio F, Esposito S, De Stefano D, Sana I, Ferrari E, Sepe A, Salvadori L, Buonpensiero P, Di Pasqua A, Grassia R, Leone CA, Guido S, De Rosa G, Lusa S, Bona G, Stoll G, Maiuri MC, Mehta A, Kroemer G, Maiuri L, Raia V. A novel treatment of cystic fibrosis acting on-target: cysteamine plus epigallocatechin gallate for the autophagy-dependent rescue of class II-mutated CFTR. Cell Death Differ 2016; 23: 1380–1393.
- 132. Hirata N, Yamada S, Shoda T, Kurihara M, Sekino Y, Kanda Y. Sphingosine-1phosphate promotes expansion of cancer stem cells via S1PR3 by a ligandindependent Notch activation. Nat Comms 2014; 25: 4806.
- 133. Stradner MH, Gruber G, Angerer H, Huber V, Setznagl D, Kremser M-L, Moazedi-Fürst FC, Windhager R, Graninger WB. Sphingosine 1-Phosphate Counteracts the Effects of Interleukin-1β in Human Chondrocytes. Arthritis Rheum 2013; 65: 2113– 2122.
- 134. Goldkorn T, Filosto S, Chung S. Lung Injury and Lung Cancer Caused by Cigarette Smoke-Induced Oxidative Stress: Molecular Mechanisms and Therapeutic Opportunities Involving the Ceramide-Generating Machinery and Epidermal Growth Factor Receptor. Antioxidants Redox Signal 2014; 21: 2149–2174.
- 135. Pruessmeyer J, Martin C, Hess FM, Schwarz N, Schmidt S, Kogel T, Hoettecke N, Schmidt B, Sechi A, Uhlig S, Ludwig A. A Disintegrin and Metalloproteinase 17 (ADAM17) Mediates Inflammation-induced Shedding of Syndecan-1 and -4 by Lung Epithelial Cells. J Biol Chem 2009; 285: 555–564.
- 136. Bergin DA, Greene CM, Sterchi EE, Kenna C, Geraghty P, Belaaouaj A, Taggart CC, O'Neill SJ, McElvaney NG. Activation of the Epidermal Growth Factor Receptor (EGFR) by a Novel Metalloprotease Pathway. J Biol Chem 2008; 283: 31736–31744.
- 137. Pasqualon T, Pruessmeyer J, Weidenfeld S, Babendreyer A, Groth E, Schumacher J, Schwarz N, Denecke B, Jahr H, Zimmermann P, Dreymueller D, Ludwig A. A transmembrane C-terminal fragment of syndecan-1 is generated by the metalloproteinase ADAM17 and promotes lung epithelial tumor cell migration and lung metastasis formation. Cell Mol Life Sci 2015; 72: 3783–3801.
- 138. Broekman W, Amatngalim GD, de Mooij-Eijk Y, Oostendorp J, Roelofs H, Taube C, Stolk J, Hiemstra PS. TNF- α and IL-1 β -activated humanmesenchymal stromal cells increase airwayepithelial wound healing in vitro viaactivation of the epidermal growth factorreceptor. Respir Res 2016; 17: 1–12.
- 139. Deshmukh HS, Case LM, Wesselkamper SC, Borchers MT, Martin LD, Shertzer HG, Nadel JA, Leikauf GD. Metalloproteinases Mediate Mucin 5AC Expression by Epidermal Growth Factor Receptor Activation. Am J Respir Crit Care Med 2005; 171: 305–314.
- 140. Kim S, Nadel JA. Fibrinogen binding to ICAM-1 promotes EGFR-dependent mucin production in human airway epithelial cells. Am J Physiol Lung Cell Mol Physiol 2009; 297: L174–L183.
- 141. https://www.mycancergenome.org/clinicaltrials/?disease=NON-SMALL+CELL+LUNG+CANCER&gene=EGFR.

- 142. https://clinicaltrials.gov/ct2/results?term=TACE&Search=Search. 760: 1–8.
- 143. Wei Y, Zou Z, Becker N, Anderson M, Sumpter R, Xiao G, Kinch L, Koduru P, Christudass CS, Veltri RW, Grishin NV, Peyton M, Minna J, Bhagat G, Levine B. EGFR-Mediated Beclin 1 Phosphorylation in Autophagy Suppression, Tumor Progression, and Tumor Chemoresistance. Cell 2013; 154: 1269–1284.
- 144. Li B, Antonyak MA, Druso JE, Cheng L, Nikitin AY, Cerione RA. EGF potentiated oncogenesis requires a tissue transglutaminase-dependent signaling pathway leading to Src activation. PNAS 2010; 107: 1408–1413.
- 145. Li Y-Y, Lam S-K, Mak JC-W, Zheng C-Y, Ho JC-M. Erlotinib-induced autophagy in epidermal growth factor receptor mutated non-small cell lung cancer. Lung Cancer 2013; 81: 354–361.
- 146. Luciani A, Villella VR, Vasaturo A, Giardino I, Raia V, Pettoello-Mantovani M, D'Apolito M, Guido S, Leal T, Quaratino S, Maiuri L. SUMOylation of Tissue Transglutaminase as Link between Oxidative Stress and Inflammation. J Immunol 2009; 183: 2775–2784.
- 147. Li Z, Xu X, Bai L, Chen W, Lin Y. Epidermal Growth Factor Receptor-mediated Tissue Transglutaminase Overexpression Couples Acquired Tumor Necrosis Factor-related Apoptosis-inducing Ligand Resistance and Migration through c-FLIP and MMP-9 Proteins in Lung Cancer Cells. J Biol Chem 2011; 286: 21164–21172.
- 148. Wang R, Ahmed J, Wang G, Hassan I, Strulovici-Barel Y, Salit J, Mezey JG, Crystal RG. Airway Epithelial Expression of TLR5 Is Downregulated in Healthy Smokers and Smokers with Chronic Obstructive Pulmonary Disease. J Immunol 2012; 189: 2217–2225.
- 149. Chen PI, Kong C, Su X, Stahl PD. Rab5 Isoforms Differentially Regulate the Trafficking and Degradation of Epidermal Growth Factor Receptors. J Biol Chem 2009; 284: 30328–30338.
- 150. Dif F, Wu YZ, Burgel PR, Ollero M, Leduc D, Aarbiou J, Borot F, Garcia-Verdugo I, Martin C, Chignard M, Israel-Biet D, Kita Y, Scholte BJ, Touqui L, Critical role of cytosolic phospholipase A2{alpha} in bronchial mucus hypersecretion in CFTRdeficient mice. Eur Respir J. 2010;36(5):1120-30.

human disease CF lung COPD airway IL68 inflammation AREG remodeling EGER sh fibrosis PBEC human disease CF lung COPD airway IL68 infla**APPENDIX** steling G remodeling a Inflammation PBEC hus ??? airway IL6R Inflammation AREG remodeling EGFR shedding 17 fibros a uman disea lung COPD airway IL6R Inflammation AREG remodeling EGFR shedding ADAM17 fibrosis PBEC Role of IL-1β in experimental cystic fibrosis upon shedding ADAM17 fibrosis PBEC human disease CF lung COPD P.aeruginosa infection EG reme (published – PLoS One) ADAM17 fibrosis PBEC human disease CF lung COPD airway Palomo J, Marchiol T, Piotet J, Fauconnier L, Robinet M, Reverchon F, Le Bert M, Togbe D, Buijs-Offerman R, Stolarczyk M, Quesniaux VF, Scholte BJ, Ryffel B



GOPEN ACCESS

Citation: Palomo J, Marchiol T, Piotet J, Fauconnier L, Robinet M, et al. (2014) Role of IL- 1β in Experimental Cystic Fibrosis upon *P. aeruginosa* Infection. PLoS ONE 9(12): e114884. doi:10.1371/journal.pone.0114884

Editor: Min Wu, University of North Dakota, United States of America

Received: October 3, 2014

Accepted: November 14, 2014

Published: December 12, 2014

Copyright: © 2014 Palomo et al. This is an openaccess article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability: The authors confirm that all data underlying the findings are fully available without restriction. All relevant data are within the paper.

Funding: This study was part of the European project n°10IM1016 financed by the French association "Vaincre La Mucoviscidose", supported by the ERASMUS Medical Center of Rotterdam (The Netherlands), and by Région Centre (Respig project) and CNRS of Orléans (France). RBO was supported by the Dutch CF foundation NCFS; MS was supported by the Dutch Lung foundation (Longfonds 3.3.10.027). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors confirm that co-author Bernhard Ryffel is a PLOS ONE Editorial Board member. And the authors confirm that they have declared that no competing interests exist. T. Marchiol, L. Fauconnier, M. Robinet and D. Togbe are employed by the commercial company "Artimmune", but this does not alter the authors' adherence to PLOS ONE policies on sharing data and materials. RESEARCH ARTICLE

Role of IL-1 β in Experimental Cystic Fibrosis upon *P. aeruginosa* Infection

Jennifer Palomo^{1,2}, Tiffany Marchiol³, Julie Piotet^{1,2}, Louis Fauconnier³, Marieke Robinet³, Flora Reverchon^{1,2}, Marc Le Bert¹, Dieudonnée Togbe³, Ruvalic Buijs-Offerman⁴, Marta Stolarczyk⁴, Valérie F. J. Quesniaux^{1,2}, Bob J. Scholte^{4*}, Bernhard Ryffel^{1,2,5*}

1. CNRS, UMR7355, Orleans, France, 2. Experimental and Molecular Immunology and Neurogenetics, University of Orléans, Orléans, France, 3. Artimmune, Orléans, France, 4. Erasmus MC, Cell Biology department, Rotterdam, The Netherlands, 5. Institute of Infectious Disease and Molecular Medicine, IDM, Cape Town, South Africa

*b.scholte@erasmusmc.nl (BS); bryffel@cnrs-orleans.fr (BR)

Abstract

Cystic fibrosis is associated with increased inflammatory responses to pathogen challenge. Here we revisited the role of IL-1 β in lung pathology using the experimental F508del-CFTR murine model on C57BL/6 genetic background (Cftr^{tm1eur} or d/d), on double deficient for d/d and type 1 interleukin-1 receptor (d/d X IL-1R1 $^{-/-}$), and antibody neutralization. At steady state, young adult d/d mice did not show any signs of spontaneous lung inflammation. However, IL-1R1 deficiency conferred partial protection to repeated P. aeruginosa endotoxins/LPS lung instillation in d/d mice, as 50% of d/d mice succumbed to inflammation, whereas all $d/d \times IL-1R1^{-/-}$ double mutants survived with lower initial weight loss and less pulmonary collagen and mucus production, suggesting that the absence of IL-1R1 signaling is protective in d/d mice in LPS-induced lung damage. Using P. aeruginosa acute lung infection we found heightened neutrophil recruitment in d/d mice with higher epithelial damage, increased bacterial load in BALF, and augmented IL-1 β and TNF- α in parenchyma as compared to WT mice. Thus, F508del-CFTR mice show enhanced IL-1 β signaling in response to *P. aeruginosa*. IL-1 β antibody neutralization had no effect on lung homeostasis in either d/d or WT mice, however P. aeruginosa induced lung inflammation and bacterial load were diminished by IL-1 β antibody neutralization. In conclusion, enhanced susceptibility to P. aeruginosa in d/d mice correlates with an excessive inflammation and with increased IL-1 β production and reduced bacterial clearance. Further, we show that neutralization of IL-1 β in d/d mice through the double mutation d/d x IL-1R1^{-/-} and in WT via antibody neutralization attenuates inflammation. This supports the notion that intervention in the IL-1R1/IL-1 β pathway may be detrimental in CF patients.

Introduction

Cystic fibrosis (CF) is an autosomal recessive genetic disease that affects 1 newborn out of 3,500 in the USA (WHO), and 1 per 2,000–3,000 in Europe, with an average 40-year life expectancy. This pathology is caused by mutations within the gene encoding the CFTR (Cystic Fibrosis Transmembrane Conductance *Regulator*) chloride channel [1]. In humans, the most common mutation (F508del) found in patients is a deletion of phenylalanine 508 (F508del) in the CFTR chloride channel [2], the modified protein is not efficiently folded and is rapidly degraded [3]. CF affects secretory epithelia from different organs, leading to gastro-intestinal tract pathology, with a reduced pancreatic digestive enzyme production [4] and chronic intestinal malabsorption [5]. CF mortality and morbidity are mainly due to respiratory disease, characterized by the production of abnormally viscous mucus, plugging of distal airways, and increased susceptibility to chronic infections with opportunistic bacteria, excessive lung inflammation, bronchiectasis and fibrosis leading to progressive loss of lung function [6]. The exact molecular link between CFTR mutations and hypersensitivity to infections remains unclear and controversial. In CF patients, S. aureus predominates in the lung of children and teenagers, while P. aeruginosa prevails in adults [7]. Exopolysaccharide-enriched biofilms produced by P. aeruginosa increase the mucus viscosity, resistance to antibiotics and host immune effectors. Chronic bacterial infections are common in CF patients and facilitates lung inflammation, mucous obstruction and tissue remodeling, resulting in fatal loss of function [1]. CF lungs display excessive inflammatory response, especially with increased neutrophil recruitment [8], the mechanism of this phenomenon is not adequately explained. However, intervention in this process likely will benefit CF patients.

The role of the pro-inflammatory signaling cytokine Interleukin 1 β (IL-1 β) in CF lung disease has been reported before. *P. aeruginosa* induces IL-1 β or IL-18 production through NLRC4 inflammasome activation [9, 10]. *P. aeruginosa* flagellin and highly acylated LPS is recognized by TLR5 [11] and TLR4 [12] respectively. Human polymorphisms observed in the *IL1B* gene were associated with CF disease [13]. CFTR deficient mice were found to be more susceptible to acute [14, 15] and chronic [16] *P. aeruginosa* infection and display an exacerbated inflammatory response to LPS and *P. aeruginosa* activated alveolar macrophages from F508del mutant mice have enhanced expression of IL-1 β [17, 18]. Huaux et al recently showed a deregulated inflammatory and fibrotic response in F508del mutant mice to bleomycin, which is IL-1R1 signaling dependent [19, 20].

Here we revisited the role of IL-1 β in the resolution of *P. aeruginosa* infection, in a murine model based on mice carrying the most common CF mutation F508del CFTR [21–23]. In this study, we show that excessive activation of IL-1 β correlates with increased bacterial load, inflammation and lung damage in F508del CFTR mice. Further, we show that IL-1 β antibody neutralization attenuates the inflammatory response to *P. aeruginosa* infection.

Materials and Methods

Mice

Mice were on C57BL/6(J) background, wild type (WT), or homozygotes for F508del CFTR mutation in the murine *Cftr* gene (*Cftr*^{tm1eur} or d/d) [<u>21–23</u>], and deficient for type 1 interleukin 1 receptor (d/d X IL-1R1^{-/-}) [<u>24</u>]. Mice, obtained from Erasmus MC Rotterdam [<u>21</u>] were bred at TAAM – UPS 44, Orléans (Institut Transgenose) and manipulated in an SPF zone. This study was carried out in strict accordance with the recommendations of CNRS, and in the Guide "Animaleries de laboratoire". The protocol was approved by the Committee on the Ethics of CNRS of Orléans (Permit Number: CLE CCO 2012-042). All animals are under daily inspection by the animal facility staff and the experimenters. All efforts were made to minimize suffering.

IL-1 β antibody administration

Mice were treated intraperitoneally, once a week for 8 weeks, with anti-IL-1 β antibody (Dr. Hermann Gram, Novartis Pharma, Basel) [25], 10 mg/kg. Control mice received 150 μ L PBS. Mice were euthanatized 1 week after the last treatment, by carbon dioxide inhalation (80–90% in a dedicated inhalation chamber).

Repeated LPS induced lung inflammation (4 weeks)

Mice were treated intranasally with 80 μ g of *P. aeruginosa* endotoxins/LPS [Sigma Chemical Co., St. Louis, MO] in 40 μ L PBS, under isoflurane anesthesia, once a week for 4 weeks. Control mice were untreated. Mice were euthanatized 24 h after the last challenge, by carbon dioxide inhalation (80–90% in a dedicated inhalation chamber). All animals are under daily inspection by the experimenters. Humane endpoints were used during this survival study: in case that mice unexpectedly present obvious health problems (over 20% weight loss, signs of suffering, difficulties in their movements and uptake of water and food, apathy) were sacrificed, by carbon dioxide inhalation (80–90% in a dedicated inhalation chamber).

Pseudomonas aeruginosa infection

Mice were infected with freshly prepared inoculum of *P. aeruginosa* strain 2310.55 of serotype IATS O11. An overnight culture in 10 mL BHI medium was prepared, starting from the frozen stock at 37 °C and shaken at 150 rpm. Of this culture, 2.5 mL was taken to start a fresh 10 mL BHI culture. The culture was stopped when an OD of about 0.4 was reached (corresponding to a bacterial titer of about 2×10^8 bacteria/ml). Mice were anaesthetized with a low dose of intravenous ketamine/xylazine (1.25 mg/ml/0.5 mg/mL) and 40 µl of the bacterial solution or the corresponding vehicle solution (isotonic saline) was applied intranasally using an ultrafine pipette tip. Mice were monitored until they woke up and evaluated 6 h and 24 h after infection.

IL-1 β antibody administration to infected mice

Mice were treated intraperitoneally, with anti-IL-1 β antibody (Novartis Pharma, Basel) [25], 200 µg/mice, 15 h and 1 h before *P. aeruginosa* infection. Control mice received 150 µL PBS. Mice were euthanatized 20 h after infection, by carbon dioxide inhalation (80–90% in a dedicated inhalation chamber).

Bacterial load in lung

Lung total weights were recorded after sacrifice and expressed as a percentage of the body weight. Lung homogenates were prepared in 1.5 ml of isotonic saline solution using a Dispomix tissue homogenizer (Medic Tools). Tenfold serial dilutions of homogenate were plated on BHI agar (BHI 37g/l, Agar 15g/l) plates (Biovalley). Plates were incubated at 37 °C and 5% CO₂ and the numbers of CFU were enumerated after 24 h.

Bronchoalveolar lavage (BAL)

After CO₂ inhalation deep euthanasia, bronchoalveolar lavage fluid (BALF) was collected by cannulating the trachea and washing the lung with 1 mL saline at room temperature. The lavage fluid was centrifuged at 2,000 rpm for 10 min at 4°C and the supernatant was stored at -80°C for analysis. The cell pellet was resuspended in PBS, counted in a haemocytometer chamber and cytospin preparations were made using a Shandon cytocentrifuge (1000 rpm for 10 min). The cells were stained with Diff-Quick (Dade Behring, Marburg, Germany) and counted for neutrophils, macrophages, lymphocytes and eosinophils.

Total RNA extraction and RT-q-PCR from lung

Total mRNA was isolated from homogenized lung using TRI-Reagent (Sigma), purified by RNeasy Mini Kit (Qiagen, Valencia, CA), and quantified by NanoDrop (Nd-1000). Reverse transcription was performed in with SuperScriptIII Kit according manufacturer's instructions (Invitrogen). cDNA was subjected to quantitative real-time PCR using primers for *Il1b, Il6* or *Ccl2* (Qiagen) and GoTaq qPCR-Master Mix (Promega). *GAPDH* and *18S* expression was used for normalization. Raw data were analyzed using the Relative Expression Software Tool (REST, <u>http://www.rest.de.com/</u>).

Cytokine and chemokine measurement

TNF- α , IL-6, IL-1 β and keratinocyte-derived chemokine (KC or CXCL-1) concentrations in BALF and lung homogenates were measured by ELISA (Duoset Kit; R&D Systems) according to the manufacturer's instructions.

MPO activity in lung

Lungs were homogenized in Dispomix [MedicTools. Zug, CH] with 1.5 mL NaCl 0.9%, and centrifuged. The supernatant was discarded. The pellet was homogenized in 1 mL PBS - HTAB 0.5% - EDTA 1 mM. After centrifugation, 20 μ L supernatant were added with: 1 mL HBSS, 200 μ L of PBS - HTAB 0.5% - EDTA 1 mM buffer, 100 μ L o-Dianisidine (1.25 mg/mL) and 100 μ L H₂O₂ 0.048%. The reaction was stopped with 100 μ L NaN₃ 1%. MPO (myeloperoxidase) activity was determined as absorbance at 460 nm against medium.

Protein assay

Total protein concentration was analyzed in BALF, using Bradford assay with BIORAD DC Protein assay kit (Biorad), according to the manufacturer's instructions.

Histology

Lungs were fixed in 4% buffered formalin, dehydrated in ethanol and embedded in paraffin. Sections (3 μ m) were stained with haematoxylin and eosin (H&E), CAB (collagen staining) or PAS (mucus staining) and all lung sections were evaluated by two independent observers for pathological changes, collagen deposition, mucus production and cellular recruitment.

Statistical analysis

Statistical evaluation of differences between the experimental groups was determined by using Mann-Whitney non-parametric test, or Log Rank test for survival. All tests were performed with GraphPad Prism [GraphPad Software Inc., San Diego, CA, USA; <u>www.graphpad.com</u>]. A *P*-value <0.05 was considered significant and symbolized with *; ** for p<0.01 and *** for p<0.001. A *P*-value>0.05 was considered not significant and symbolized with ns.

Results

1. IL-1 β signaling enhances lung inflammation in F508del CFTR mutants in response to LPS

Since LPS, *P. aeruginosa* [14–16] and bleomycin induced injury and inflammation [26] are increased in F508del CFTR (d/d mice), and since the IL-1R1 pathway is involved [20, 27] and required for the development of bleomycin-induced fibrosis [20], we studied the role of the IL-1 β pathway in the response to inflammation and lung damage in F508del CFTR mice. In order to induce chronic lung damage, repeated *P. aeruginosa* LPS challenges were performed in d/d and in double mutant d/d x IL-1R1^{-/-} mice, and compared to untreated WT littermate controls. Lung analysis was performed at day 29, 24 h after the last LPS challenge. Survival upon repeated LPS challenge was reduced to 50% in the d/d mice, while

PLOS ONE





Fig. 1. IL-1 β participates in pathologic inflammation in F508del CFTR mutants, in response to *P. aeruginosa* LPS. d/d and double KO d/d x IL-1R1^{-/-} mice were treated intranasally with 80 µg of *P. aeruginosa* endotoxins/LPS in 40 µL PBS, once a week for 5 weeks. Untreated WT mice (UT) were used as control. Survival is presented in (A) and body weight variation 24 h after the last LPS challenge in (B). Absolute numbers of cells, (macrophages, lymphocytes and neutrophils) were measured in BALF 24 h after the last LPS challenge (C). CCL2 (D), IL-6 (E) and IL-1 β (F) were measured in lung homogenate. Cell infiltration was observed on H&E staining (G), mucus production on PAS staining (H) and collagen deposition on CAB staining (I). Histopathological scores are shown in (J). n=6–7 initially, n=3 for d/d+LPS mice 24 h after last challenge instillation. Values are in mean +/– SEM; * for p<0.05 and ns for non-significant.

doi:10.1371/journal.pone.0114884.g001

PLOS ONE

100% of double KO d/d x IL- $1R1^{-/-}$ mice survived (Fig. 1.A). The results suggest that in F508del CFTR mutant, the active IL-1 β pathway is deleterious in response to repeated inflammatory injury. Reduced survival of d/d mice was apparently associated with higher loss of body weight at day 7, as compared to d/d x IL-1R1⁻ mice (Fig. 1.B). Nevertheless, all surviving mice recovered their initial weight after 4 weeks, indicating tolerance to repeated LPS challenges. LPS-induced inflammation was confirmed by increased total cell number in BALF (macrophages and lymphocytes) 24 h after the last challenge but no significant difference was observed between d/d and d/d X IL- $1R1^{-/-}$ mice at this time point (Fig. 1.C). Similarly, P. aeruginosa LPS also induced a recruitment of neutrophils in BALF of both d/d and d/d X IL- $1R1^{-/-}$ mice (Fig. 1.C). No increase compared to untreated WT mice of the CCL2 chemokine and IL-6 could be detected in lung homogenate after repeated LPS challenges (Fig. 1.D and E). These results also suggest tolerance since LPS is known to induce these cytokines after single challenge [28]. Single acute LPS challenge is also known to induce IL-1 β [29]. Indeed, the pulmonary IL-1 β production was still augmented after repeated LPS challenges in d/d mice, but to a lesser extend in d/d x IL- $1R1^{-/-}$ mice (Fig. 1.F). Semi-quantitative analysis of the lung histology indicated a distinct increase of cell infiltration, mucus production and collagen deposition in d/d mice upon repeated LPS challenges as compared to untreated control mice. Numerous inflammation parameters were reduced in d/d x $IL-1R1^{-/-}$ mice compared to d/d, but did not reach statistical significance likely due to the low number of surviving d/d mice (Fig. 1.A and F-J). In conclusion, the present data support the notion that in response to LPS challenge, IL-1 β participates in inflammation and fibrosis in F508del CFTR mutants, and that genetic ablation of IL-1R1 signaling has a protective effect.

2. Increased inflammatory response after *P. aeruginosa* injection in cystic fibrosis lung

To further support this hypothesis in another model of CF pathology we used *P. aeruginosa* infection, a common lung infection occurring in CF patients. We established the model in our facility previously and showed that antibody neutralization of LPS inhibited acute and fatal *P. aeruginosa* induced pneumonia [30]. First, a *P. aeruginosa* (PA011) infection was performed in WT and d/d mice, WT mice administered saline were used as control. Twenty hours after *P. aeruginosa* infection, all mice developed severe pathology, as indicated by weight loss (Fig. 2.A). Lung inflammation, analyzed by the myeloperoxidase (MPO) activity, total cell and neutrophil counts in BALF was significantly increased in all



PLOS ONE | DOI:10.1371/journal.pone.0114884 December 12, 2014



Fig. 2. *P. aeruginosa* infection induces an increased inflammation in d/d mice. WT and d/d mice received 2×10^5 CFU of *P. aeruginosa* PA011 intranasally, and WT mice treated with NaCl 0.9% were used as control. Body weight variation 20 h after infection is shown in (A). Myeloperoxidase activity was quantified (B) and absolute numbers of cells; (macrophages and neutrophils) were measured in BALF (C). Bacterial load (total CFU) was determined in BALF and in lung homogenate (D). Total protein concentration was evaluated in BALF (E) and KC (F), IL-1 β (G) and TNF- α (H) were measured in lung homogenate. Cell infiltration was observed on H&E stained slide (I) and scored (J). (n=5–6) Values are in mean +/– SEM; * for p<0.05; ** for p<0.01 and *** for p<0.001. ns for non-significant.

doi:10.1371/journal.pone.0114884.g002

groups compared with uninfected mice (Fig. 2.B and C). The bacterial load in the lung at 20 h did not differ between the groups (Fig. 2.D). However, bacterial load (CFU) detected in BALF of d/d mice was two logs higher in d/d mice than in WT mice (Fig. 2.D). Increased CFU in BALF of d/d mice was associated with heightened neutrophil recruitment as compared with WT mice (Fig. 2.C). This correlated with increased epithelial injury in d/d mice, compared to WT mice (Fig. 2.E) indicating that pulmonary epithelial damage and protein release were enhanced in d/d mice compared to WT mice (Fig. 2.G and H). KC chemokine (CXCL1), known to attract neutrophils is increased in all groups after *P. aeruginosa* infection, as compared with uninfected mice (Fig. 2.F). In d/d mice, IL-1 β and TNF- α levels were substantially higher in the lung than in WT mice. Histological analysis showed augmented cell infiltration in d/d mice as compared with WT mice after *P. aeruginosa* infection (Fig. 2.I and J).

3. IL-1 β antibody neutralization attenuates *P. aeruginosa*-induced inflammation

First, to study IL-1 β function at steady state in the lung, we investigated the effect of a long-lasting anti-IL-1 β antibody treatment in wild type mice or in mice with F508del CFTR mutation (d/d). Following 8-week antibody treatment, the young adult mice (14-week-old females and males) were investigated. The treatment was well tolerated; no mortality or morbidity was observed in any of the experimental groups. Histological analysis did not reveal any significant differences of cell recruitment, mucus production and collagen deposition in the lung (Fig. 3.A-C) between WT and d/d mice treated with control PBS or anti-IL-1 antibody. Lung mRNA expression of *Il1b*, *Il6* and *Ccl2* was assessed by qPCR and no significant difference was found among the control and the anti-IL-1 β antibody administered groups (Fig. 3.D and E). Therefore, this investigation suggests that under basal conditions, in the absence of stimulation, no significant difference is detected between d/d and WT control mice. The F508del CFTR murine model (on C57BL/ 6 background), displayed no significant spontaneous lung inflammation under these conditions. Furthermore, long-term IL-1β antibody treatment had neither a significant effect on pulmonary morphology, nor on basal expression of Il1b, Il6 and Ccl2, suggesting that IL-1β plays no significant role in steady state conditions.

Subsequently, the role of IL-1 β upon *P. aeruginosa* infection was studied using neutralizing antibodies in C57BL/6 mice. Mice were infected by intranasal instillation with 10⁶ CFU of *P. aeruginosa* (strain PA011) in order to induce an acute inflammation analyzed at 20 h. To test the role of IL-1 β , mice received two



Fig. 3. Anti-IL-1 β antibody treatment has no significant effect on lung ultrastructure, nor on IL-1 β , IL-6 and CCL2 mRNA production in lungs. Wild type (WT) and F508del CFTR mutation homozygote (d/d) mice (14-week-old females and males) were treated intra-peritoneally, once a week for 8 weeks, an anti-IL-1 β antibody (10 mg/kg), or PBS. Mice were euthanatized 1 week after the last treatment and lung structure was observed on H&E stained slides observed at \times 20 magnification (A). Cellular infiltration (B) was quantified on these H&E slides and mucus production (C) was analyzed on CAB stained lung slides. Lung injury score was recorded in anti-IL-1 β antibody treated mice compared with PBS treated mice. (D and E) mRNA production of IL-1 β , IL-6 and CCL2 was measured, and normalized with 2 housekeeping gene expression (*Hprt1* and *Gapdh*). The effect of anti-IL-1 β antibody treatment compared to PBS control in WT and d/d animals is shown in (D), and the effect of F508del CFTR mutation compared to WT is shown in (E). (n=6-7)

doi:10.1371/journal.pone.0114884.g003



Fig. 4. IL-1 β participates in *P. aeruginosa*-induced inflammation at 20 h in WT mice. WT mice received 10⁶ CFU of *P. aeruginosa* PA011 intranasally, and WT mice treated with NaCl 0.9% were used as control. A group of WT mice was also treated intraperitoneally with anti-IL-1 β antibody, 200 µg/mice, 15 h and 1 h before infection. Body weight variation 20 h after infection is shown in (A). Bacterial load was determined in BALF and in lung homogenate (B), myeloperoxidase activity was quantified (C) and absolute numbers of cells, (macrophages and neutrophils) were measured in BALF (D). Protein concentration was evaluated in BALF (E) and IL-1 β (F), KC (G), TNF- α (H) and IL-6 (I), were measured in BALF and in lung homogenate. Cell infiltration was observed on H&E stained slide (J) and scored (K). (n=5–6) Values are in mean +/– SEM; * for p<0.05; ** for p<0.01 and *** for p<0.001. ns for non-significant.

doi:10.1371/journal.pone.0114884.g004

peritoneal injections of 200 µg neutralizing antibodies 15 h and 1 h before infection. As reported before (Figure 2 and in [30]), infection of C57BL/6 mice with *P. aeruginosa* caused a rapid weight loss (Fig. 4.A), with a bacterial load of 10⁴ CFU in BALF and total lung (Fig. 4.B). P. aeruginosa infection provoked neutrophil recruitment (revealed by myeloperoxidase activity measurement and cell counts in BALF), and epithelial damage suggested by an increase of proteins in BALF in all groups post infection (Fig. 4.C-E). However, IL-1 β neutralization had a distinct effect on infection and lung inflammation. Indeed, IL-1β blockade significantly reduced bacterial load in BALF, and to a lesser extent in lung (Fig. 4.B), suggesting a substantial difference in bacterial clearance between the two groups. This was correlated with a decrease in neutrophil recruitment (Fig. 4.C and D) and an inhibition of protein in BALF as measure of epithelial damage (Fig. 4.E). Moreover KC and TNF- α production in BALF were diminished (Fig. 4.G and H), whereas IL-6 was not modified (Fig. 4.I). In agreement with this, histological analysis showed reduced cell infiltration in anti-IL-1 β treated mice as compared with WT untreated mice after *P. aeruginosa* infection (Fig. 4.J and K). These data suggest that IL-1 β antibody neutralization may attenuate P. aeruginosa-induced acute lung inflammation.

Discussion

CF patients display increased susceptibility to chronic infections with opportunistic bacteria, excessive lung inflammation and fibrosis leading to fatal loss of function eventually [6]. *P. aeruginosa* has been shown to induce IL-1 β through NLRC4 inflammasome activation [9, 10]. CFTR deficient mice were found to be more susceptible to *P. aeruginosa* infection (acute [14, 15] and chronic [16]) and have an exacerbated inflammatory response to LPS and *P. aeruginosa*. Furthermore, F508del CFTR mutant alveolar macrophages display an enhanced IL-1 β production in response to LPS stimulation [18]. Thus, we were interested in clarifying the role of IL-1 β in the resolution of *P. aeruginosa* infection, in a murine model with the most common CF mutation F508del CFTR [21–23]. Here, we show that, after *P. aeruginosa* infection, excessive activation of IL-1 β in F508del CFTR (d/d) mice compared to WT was accompanied by increased CFU in BALF, inflammation and lung damage. Further we show that a therapeutic antibody administration attenuates inflammatory response in an acute model of infection using WT mice.

In CF patients bacterial infections persist and their chronicity facilitates lung inflammation, and subsequent tissue remodeling with severe loss of function [1]. CF lungs display excessive immune response, with increased neutrophil recruitment [8]. We first investigated the role of IL-1 β under pathologic conditions, induced by repeated *P. aeruginosa*-LPS challenges. We observed LPS-induced mortality in d/d mice, that was absent in d/d x IL-1R1^{-/-} double mutants, and a higher inflammation in d/d mice compared to the double mutant, suggesting a detrimental role of the IL-1R1 pathway in LPS-challenged d/d mice

(Fig. 1). The deficiency in IL-1R1 apparently protected the mice against LPSinduced mortality in d/d mice. *P.aeruginosa*-LPS challenges were responsible for leucocyte recruitment, including neutrophils. Despite increased cell infiltration in the lung of d/d mice, collagen deposition and mucus production were only slightly increased after multiple LPS challenges. Repeated LPS challenges did not cause epithelial damage and chronic inflammation, which we explain by tolerance induction [<u>31</u>], that is the reason to switch to *P. aeruginosa* infection.

No spontaneous inflammation was found in young adult F508del CFTR mutant on C57BL/6 genetic background under our conditions. Whereas signs of inflammation and tissue remodeling develop early in most CF patients [32], results on basal inflammation in murine models are variable [23]. These data are consistent with published data on the absence of spontaneous pathology such as mucus plugging, neutrophil accumulation or bronchiectasis, in young mice deficient or mutated for the *Cftr* gene [15, 33–35], but neutrophilic inflammation has been reported in absence of infection [36, 37]. Spontaneous lung pathology development seems to be dependent on the age of mice [37], genetic background and animal facility health status [23]. However, in CF mutant mice challenged with pro-inflammatory agents (*P. aeruginosa*, LPS) enhanced injury and inflammation compared to wild type was reported in several independent studies [14, 15, 26, 38].

Therefore, to get closer to a model of lung infections that occur in CF patients, we used an acute P. aeruginosa infection model, which is the most current pathogen found in CF patient [7], in F508del CFTR mice. Twenty hours after P. aeruginosa infection, d/d mice displayed increased neutrophil recruitment in BALF as compared with WT mice (Fig. 2). They also showed an increase in bacterial load and cytokine production, such as IL-1 β and TNF- α , in BALF. These results suggest a higher sensitivity to infection in d/d mice, consistent with previous studies in CFTR mutant mice, showing an augmentation of bacterial load and production of pro-inflammatory cytokines after P. aeruginosa infection [14, 15, 38]. The enhanced production of IL-1 β compared to WT after infection was associated with increased epithelial damage, cell infiltration and CFU (Fig. 2). Then, we analyzed the effect of IL-1 β neutralization at steady state in the lung. We found that in d/d as well as in WT mice, IL-1 β antibody had no significant effect on survival. Furthermore d/d and WT mice displayed no difference in basal Il1b mRNA expression, and repeated anti-IL-1ß antibody administration had no significant effect on basal inflammatory parameters and cytokine production. However, our data suggest that IL-1 β may cause an excessive and pathologic inflammation in challenged CF mutant lungs. IL-1ß over-expression in d/d mice infected with P. aeruginosa was accompanied by a higher bacterial load. Indeed, excessive production of inflammatory cytokines has been associated with bacterial persistence in other studies as well. High concentrations of cytokine, such as IL-1 β , IL-6 or TNF- α , enhance intracellular and extracellular bacterial growth of *P*. *aeruginosa*, or *S. aureus* in the presence of monocytes [39, 40]. Furthermore, IL-1 β release was linked to mortality in a murine model of *P. aeruginosa* infection after thermal injury [41].

Therefore, we tested the effect of a therapeutic antibody against murine IL-1 β in our acute P. aeruginosa infection model in WT mice. IL-1ß neutralization has been successfully tested previously with this antibody in a model of intestinal inflammation [25]. Twenty hours after *P. aeruginosa* infection, IL-1β antibody administration caused a decrease in the inflammatory parameters and in bacterial load, as compared with NaCl-treated mice. This is consistent with studies by Schultz et al. who showed that inhibition of the IL-1R1 pathway by either the IL-1R1 mutation or application of the IL-1 β antagonist IL-1RA improved antibacterial host defense and reduced pro-inflammatory cytokine production [42]. Reduced neutrophil recruitment by IL-1 β neutralization could be due to reduced endothelial activation, as proinflammatory cytokines activate endothelial cells. For instance, TNF- α and IL-17 synergize to induce in cultured endothelial cells the expression of P- and E-selectin, as well as neutrophil chemokines, increasing neutrophil transmigration [43]. IL-1 β produced by activated monocytes was shown to augment the expression of adhesion molecules (ICAM-1, Eselectin) in vitro [44]. Thus, the diminished leucocyte recruitment we observe with IL-1 β neutralization could be related to a reduced endothelial activation. Abnormal recruitment and metabolic adaptation of neutrophils in human CF airways has been demonstrated [45], the molecular mechanism remains to be established but likely involves IL-1 β signaling based on the arguments mentioned above. We cannot exclude a role of lung fibroblasts and myoblasts. Indeed, IL-1 β stimulated human cardiac fibroblasts overexpress adhesion molecules and neutrophil chemoattractant [46].

In conclusion we propose that the IL-1 β pathway is critical to drive excessive and detrimental inflammation in F508del mouse model of CF. We show here that antibody neutralization of IL-1 β is well tolerated in mice, has no effect on the unchallenged lung in WT or F508del CFTR mice, whereas it can reduce pathology induced by acute bacterial lung infection.

Author Contributions

Conceived and designed the experiments: J. Palomo BJS BR. Performed the experiments: J. Palomo TM J. Piotet LF MR FR. Analyzed the data: J. Palomo MR. Contributed reagents/materials/analysis tools: J. Palomo MLB DT RBO MS VFJQ. Wrote the paper: J. Palomo BJS BR.

References

- Gaspar MC, Couet W, Olivier JC, Pais AA, Sousa JJ (2013) Pseudomonas aeruginosa infection in cystic fibrosis lung disease and new perspectives of treatment: a review. Eur J Clin Microbiol Infect Dis.
- 2. O'Sullivan BP, Freedman SD (2009) Cystic fibrosis. Lancet 373: 1891–1904.
- Cheng SH, Gregory RJ, Marshall J, Paul S, Souza DW, et al. (1990) Defective intracellular transport and processing of CFTR is the molecular basis of most cystic fibrosis. Cell 63: 827–834.
- Steward MC, Ishiguro H, Case RM (2005) Mechanisms of bicarbonate secretion in the pancreatic duct. Annu Rev Physiol 67: 377–409.

- Winfield RD, Beierle EA (2006) Pediatric surgical issues in meconium disease and cystic fibrosis. Surg Clin North Am 86: 317–327, viii-ix.
- Lubamba B, Dhooghe B, Noel S, Leal T (2012) Cystic fibrosis: insight into CFTR pathophysiology and pharmacotherapy. Clin Biochem 45: 1132–1144.
- CFF (2012) Patient registry Annual data report 2011. Cystic Fibrosis Foundation <u>http://www.cff.org/</u> UploadedFiles/research/ClinicalResearch/2011-Patients-Registry.pdf accessed online at 22 July 2013.
- Zhou Y, Song K, Painter RG, Aiken M, Reiser J, et al. (2013) Cystic fibrosis transmembrane conductance regulator recruitment to phagosomes in neutrophils. J Innate Immun 5: 219–230.
- Sutterwala FS, Mijares LA, Li L, Ogura Y, Kazmierczak BI, et al. (2007) Immune recognition of Pseudomonas aeruginosa mediated by the IPAF/NLRC4 inflammasome. J Exp Med 204: 3235–3245.
- Faure E, Mear JB, Faure K, Normand S, Couturier-Maillard A, et al. (2014) Pseudomonas aeruginosa Type-3 secretion system dampens host defense by exploiting the NLRC4-coupled inflammasome. Am J Respir Crit Care Med.
- Ben Mohamed F, Garcia-Verdugo I, Medina M, Balloy V, Chignard M, et al. (2012) A crucial role of Flagellin in the induction of airway mucus production by Pseudomonas aeruginosa. PLoS ONE 7: e39888.
- Ernst RK, Hajjar AM, Tsai JH, Moskowitz SM, Wilson CB, et al. (2003) Pseudomonas aeruginosa lipid A diversity and its recognition by Toll-like receptor 4. J Endotoxin Res 9: 395–400.
- Levy H, Murphy A, Zou F, Gerard C, Klanderman B, et al. (2009) IL1B polymorphisms modulate cystic fibrosis lung disease. Pediatr Pulmonol 44: 580–593.
- Saadane A, Soltys J, Berger M (2006) Acute Pseudomonas challenge in cystic fibrosis mice causes prolonged nuclear factor-kappa B activation, cytokine secretion, and persistent lung inflammation. J Allergy Clin Immunol 117: 1163–1169.
- van Heeckeren AM, Schluchter MD, Xue W, Davis PB (2006) Response to acute lung infection with mucoid Pseudomonas aeruginosa in cystic fibrosis mice. Am J Respir Crit Care Med 173: 288–296.
- Paroni M, Moalli F, Nebuloni M, Pasqualini F, Bonfield T, et al. (2013) Response of CFTR-deficient mice to long-term chronic Pseudomonas aeruginosa infection and PTX3 therapy. J Infect Dis 208: 130– 138.
- Legssyer R, Huaux F, Lebacq J, Delos M, Marbaix E, et al. (2006) Azithromycin reduces spontaneous and induced inflammation in DeltaF508 cystic fibrosis mice. Respir Res 7: 134.
- Meyer M, Huaux F, Gavilanes X, van den Brule S, Lebecque P, et al. (2009) Azithromycin reduces exaggerated cytokine production by M1 alveolar macrophages in cystic fibrosis. Am J Respir Cell Mol Biol 41: 590–602.
- **19.** Huaux F, Noel S, Dhooghe B, Panin N, Lo Re S, et al. (2013) Dysregulated proinflammatory and fibrogenic phenotype of fibroblasts in cystic fibrosis. PLoS One 8: e64341.
- Gasse P, Mary C, Guenon I, Noulin N, Charron S, et al. (2007) IL-1R1/MyD88 signaling and the inflammasome are essential in pulmonary inflammation and fibrosis in mice. J Clin Invest 117: 3786– 3799.
- 21. van Doorninck JH, French PJ, Verbeek E, Peters RH, Morreau H, et al. (1995) A mouse model for the cystic fibrosis delta F508 mutation. EMBO J 14: 4403–4411.
- 22. French PJ, van Doorninck JH, Peters RH, Verbeek E, Ameen NA, et al. (1996) A delta F508 mutation in mouse cystic fibrosis transmembrane conductance regulator results in a temperature-sensitive processing defect in vivo. J Clin Invest 98: 1304–1312.
- 23. Wilke M, Buijs-Offerman RM, Aarbiou J, Colledge WH, Sheppard DN, et al. (2011) Mouse models of cystic fibrosis: phenotypic analysis and research applications. J Cyst Fibros 10 Suppl 2: S152–171.
- 24. Glaccum MB, Stocking KL, Charrier K, Smith JL, Willis CR, et al. (1997) Phenotypic and functional characterization of mice that lack the type I receptor for IL-1. J Immunol 159: 3364–3371.
- Villeret B, Brault L, Couturier-Maillard A, Robinet P, Vasseur V, et al. (2013) Blockade of IL-1R signaling diminishes Paneth cell depletion and Toxoplasma gondii induced ileitis in mice. Am J Clin Exp Immunol 2: 107–116.

- Lubamba B, Huaux F, Lebacq J, Marbaix E, Dhooghe B, et al. (2012) Immunomodulatory activity of vardenafil on induced lung inflammation in cystic fibrosis mice. J Cyst Fibros 11: 266–273.
- Hudock KM, Liu Y, Mei J, Marino RC, Hale JE, et al. (2012) Delayed resolution of lung inflammation in II-1rn-/- mice reflects elevated IL-17A/granulocyte colony-stimulating factor expression. Am J Respir Cell Mol Biol 47: 436–444.
- 28. Nojiri T, Hosoda H, Tokudome T, Miura K, Ishikane S, et al. (2014) Atrial natriuretic peptide inhibits lipopolysaccharide-induced acute lung injury. Pulm Pharmacol Ther.
- 29. Moreno-Vinasco L, Quijada H, Sammani S, Siegler J, Letsiou E, et al. (2014) Nicotinamide phosphoribosyltransferase inhibitor is a novel therapeutic candidate in murine models of inflammatory lung injury. Am J Respir Cell Mol Biol.
- Secher T, Fauconnier L, Szade A, Rutschi O, Fas SC, et al. (2011) Anti-Pseudomonas aeruginosa serotype O11 LPS immunoglobulin M monoclonal antibody panobacumab (KBPA101) confers protection in a murine model of acute lung infection. J Antimicrob Chemother 66: 1100–1109.
- **31.** del Campo R, Martinez E, del Fresno C, Alenda R, Gomez-Pina V, et al. (2011) Translocated LPS might cause endotoxin tolerance in circulating monocytes of cystic fibrosis patients. PLoS ONE 6: e29577.
- Stick S, Tiddens H, Aurora P, Gustafsson P, Ranganathan S, et al. (2013) Early intervention studies in infants and preschool children with cystic fibrosis: are we ready? Eur Respir J 42: 527–538.
- van Heeckeren AM, Schluchter M, Xue L, Alvarez J, Freedman S, et al. (2004) Nutritional effects on host response to lung infections with mucoid Pseudomonas aeruginosa in mice. Infect Immun 72: 1479– 1486.
- Fisher JT, Zhang Y, Engelhardt JF (2011) Comparative biology of cystic fibrosis animal models. Methods Mol Biol 742: 311–334.
- Bragonzi A (2010) Murine models of acute and chronic lung infection with cystic fibrosis pathogens. Int J Med Microbiol 300: 584–593.
- Durie PR, Kent G, Phillips MJ, Ackerley CA (2004) Characteristic multiorgan pathology of cystic fibrosis in a long-living cystic fibrosis transmembrane regulator knockout murine model. Am J Pathol 164: 1481–1493.
- **37.** Tirkos S, Newbigging S, Nguyen V, Keet M, Ackerley C, et al. (2006) Expression of S100A8 correlates with inflammatory lung disease in congenic mice deficient of the cystic fibrosis transmembrane conductance regulator. Respir Res 7: 51.
- Schroeder TH, Reiniger N, Meluleni G, Grout M, Coleman FT, et al. (2001) Transgenic cystic fibrosis mice exhibit reduced early clearance of Pseudomonas aeruginosa from the respiratory tract. J Immunol 166: 7410–7418.
- Kanangat S, Meduri GU, Tolley EA, Patterson DR, Meduri CU, et al. (1999) Effects of cytokines and endotoxin on the intracellular growth of bacteria. Infect Immun 67: 2834–2840.
- 40. Meduri GU (2002) Clinical review: a paradigm shift: the bidirectional effect of inflammation on bacterial growth. Clinical implications for patients with acute respiratory distress syndrome. Crit Care 6: 24–29.
- Tsay TB, Yang MC, Chen PH, Lai KH, Huang HT, et al. (2013) Blocking TNF-alpha enhances Pseudomonas aeruginosa-induced mortality in burn mice through induction of IL-1beta. Cytokine 63: 58– 66.
- 42. Schultz MJ, Rijneveld AW, Florquin S, Edwards CK, Dinarello CA, et al. (2002) Role of interleukin-1 in the pulmonary immune response during Pseudomonas aeruginosa pneumonia. Am J Physiol Lung Cell Mol Physiol 282: L285–290.
- **43.** Griffin GK, Newton G, Tarrio ML, Bu DX, Maganto-Garcia E, et al. (2012) IL-17 and TNF-alpha sustain neutrophil recruitment during inflammation through synergistic effects on endothelial activation. J Immunol 188: 6287–6299.
- 44. Shikama Y, Aki N, Hata A, Nishimura M, Oyadomari S, et al. (2014) Palmitate-stimulated monocytes induce adhesion molecule expression in endothelial cells via IL-1 signaling pathway. J Cell Physiol.
- Laval J, Touhami J, Herzenberg LA, Conrad C, Taylor N, et al. (2013) Metabolic adaptation of neutrophils in cystic fibrosis airways involves distinct shifts in nutrient transporter expression. J Immunol 190: 6043–6050.



46. Turner NA, Das A, O'Regan DJ, Ball SG, Porter KE (2011) Human cardiac fibroblasts express ICAM-1, E-selectin and CXC chemokines in response to proinflammatory cytokine stimulation. Int J Biochem Cell Biol 43: 1450–1458. numan disease CF lung COPD airway IL6R inflammation APCC removeling COPD airway IL6R infla**SUMMARY**EGER s ADAM17 fibrosis PBEC human disease CF lung COPD airway IL6R inflammation AREG remodelin ng ADAM17 fibro SGFR shedding // deling EGFR s AREG ren a sinflammatio EGFR shedding ADAM17 fibrosis PBEC human disease CF EGFR shedding ADAM17 fibrosis PBEC huses inflammation AREG remodeling EGFR shedding ADAM17 ise iuman, dis airway IL6R Inflammation AREG remodeling EGFR sheddin to DAV17 fibrosis lung COPD airway ILER Inflammation AREG remodeling EGFR shedding ADAM17 fibrosis PBEC

SUMMARY

Cystic Fibrosis (CF) lung disease and Chronic Obstructive Pulmonory Disease (COPD) are progressive and eventually fatal lung diseases characterised by airway mucus hypersecretion, airway surface liquid dehydratation, chronic inflammation and lung tissue remodelling. Though the etiology of the two diseases is obviously different (CF results form CFTR deficiency, whereas COPD from chronic CS exposure), however, the molecular mechanisms involved in the progression of lung malfunction are likely overlapped, but not completely elucidated. By studying pathological processes in the CF and COPD airway model systems, we aim to establish novel therapeutic targets.

In our studies we focused on ADAM17 and EGFR which are both linked to lung pathology, by regulating inflammation, tissue regeneration, and mucus production. ADAM17 is a membrane bound protease that releases receptor-activating proteins from the cell surface by cleaving extracellular domains from a large number of target proteins. EGFR is a receptor for extracellular growth factors, which activates intracellular protein kinase cascades. They are both activated by external signals, including pollutants like cigarette smoke, viral and bacterial infections. Importantly, EGFR and ADAM17 function and activity are closely linked, so that they establish an EGFR/ADAM17 axis. ADAM17 sheds most of the EGFR ligands that activate the receptor in an autocrine and paracrine manner. When triggered by external signals, ADAM17 releases IL-6R and AREG from airway epithelial cells that transactivate IL-6st/gp130 and EGFR receptors respectively, on the underlying fibroblasts, myofibroblasts and smooth muscle cells. The IL-6/IL-6R/gp-130 and AREG/EGFR pathways converge in STAT3 activation, a transcription factor and genetic modulator of CF lung disease, involved in lung tissue repair, inflammation and fibrosis.

We hypothesized that hyperactivity of airway epithelial ADAM17/EGFR axis may contribute to the excessive inflammation and lung tissue remodelling characteristic of CF and COPD lung disease. Thus the focal point of this thesis was to establish the role of ADAM17 and EGFR in pathology of CFTR-related lung diseases: CF and COPD.

The aim of this thesis was to determine whether and how the COPD status and CF conditions affects basal and induced ADAM17/EGFR activity in airway epithelium. We addressed our hypothesis in advanced airway epithelial cell models: COPD and non-COPD primary bronchial airway epithelial cells exposed to cigarette smoke, and immortalized CF mutant CFBE410- cells with induced CFTR expression, both cultured on permeable filters in air-liquid interface (ALI-HBEC, ALI-CFBE410-). In ALI-HBEC we compared the effect of cigarette smoke exposure on ADAM17/EGFR in COPD and non-COPD ALI-HBEC cells, and studied the mechanism of this activation. In ALI-CFBE410- cells we investigated whether CFTR deficiency hyperactivates ADAM17/EGFR as a cell autonomous defect, and we further defined the druggable targets upstream and downstream of this regulator to potentially apply them as an early intervention CF lung therapy.

Chapter 1 provides a general introduction to CF and COPD pathology and describes the importance of anti-inflammatory therapy in CF. In the second part, ADAM17 and EGFR are introduced and their role in lung inflammation and remodeling is underlined.

In **Chapter 2** we compared CS-induced ADAM17 mediated protein shedding and mRNA expression of sIL-6R and AREG in COPD and non-COPD (ex)smokers. We showed that CS significantly induces IL6R and AREG shedding into the basolateral medium from COPD cells, but not in non-COPD cells suggesting that COPD disease makes cells more sensitive to external triggers. Furthermore, by using selective inhibitors of ADAM17 and EGFR (TMI-2 and AG1478), we showed that they not only regulate shedding of IL-6R and AREG, but also their mRNA levels. Additionally, we showed that upon CS activation ADAM17 meets its substrate in an intracellular vesicular compartment. Also EGFR is internalized after CS exposure. This suggests that trafficking of ADAM17 and EGFR is one of the regulatory mechanisms of ADAM17/EGFR axis activation.

Chapter 3 shows that functional expression of CFTR in ALI-CFBE410- cells reduces AREG release, suggesting that CFTR deficient cells have intrinsically altered ADAM17 dependent responses. In line with this, we also observed that CFTR deficiency elevates levels of EGFR phosphorylation. As in ALI-HBEC, AREG release depends on EGFR and ADAM17 to the same extent. In CFBE410-cells we observed that CFTR deficiency induces increased apical localization of EGFR, which may lead to abnormal activation of ADAM17/EGFR axis, however, these data need to be confirmed in differentiated primary airway epithelial cells. Finally, we observed that ADAM17 dependent shedding is sensitive to ROS scavengers, in particular extracellular glutathione (GSH). Using the fluorescent redox probes Grx1-roGFP expressed in cytoplasm and Grx1-roGFP-GPI expressed at the apical membrane, we showed that extracellular, but not intracellular redox potential is changed to a more oxidised state by deficient CFTR. Thus, based on these data we propose that deficient GSH secretion in CFTR deficient cells, shown by other authors in analogous systems, causes oxidation of extracellular and possible intravesicular membranes, which in turn leads to enhanced activation of ADAM17 by a known redox dependent conformation change, and concomitant abnormal EGFR signalling and localization.

Chapter 4 Illustrates the role of abnormal sphingosine metabolism in CF lung disease, providing an alternative intervention pathway. We show that unchallenged F508del CFTR mutant mice have enhanced infiltration with pro-inflammatory cells and abnormal dendritic cell activity which correlates with reduced sphingosine-1-phosphate S1P in lungs. Oral administration of an inhibitor of S1P-lyase (LX2931, Lexicon Pharmaceutics), increases tissue S1P levels and reduces inflammation and dendritic cell imbalance. Upon challenge with bacterial toxins (LPS-fMLP) LX2931 reduces induced MUC5AC production and pro-inflammatory cytokines in mutant mice (TNF- α , KC, IFN- γ , IL-12). The S1P receptor network is known to interact with the EGFR/ADAM axis through intracellular kinases, providing a mechanistic link between these observations.

Chapter 5 Summarizes and critically discusses the studies presented in this thesis and provides possible future directions in investigation of EGFR-ADAM17 axis in the context of CF and COPD lung disease.

Appendix presents evidence that CFTR mutant mice show excessive inflammation, increased IL-1 β production and reduced bacterial clearance. CFTR deficient mice with the IL-1R1 knock-out (double mutant F508del x IL-1R1) have attenuated inflammation, providing a
rationale for targeting the IL-1 β /IL-1R1 pathway in CF patients. A therapeutic antibody reduces infection related pathology in this model. Importantly in the context of this thesis, activation of IL-1R also activates ADAM17/EGFR/IL8 signalling, in airway epithelial cell lines. Thus intervention in the IL-1 β /IL-R1 β pathway may be also beneficial in attenuation of EGFR/ADAM17 signalling in the CF or COPD disease state, which can be further evaluated in differentiated human airway epithelial cells.

Taken together, our data provide evidence for the involvement of the ADAM17/EGFR axis in CF and COPD lung disease. Although CF and COPD differ in primary cause, we showed that they share a molecular mechanism possibly leading to the development and progression of the disease. CS exposure in COPD and CFTR deficiency in differentiated airway epithelial cells induce ADAM17 and EGFR dependent protein shedding of IL-6R and AREG. ADAM17 dependent release of EGFR ligands may affect epithelial (autocrine) and subepithelial (transsignaling) receptors. This implicates the ADAM17/EGFR axis in inflammation, lung tissue repair and fibrosis. Importantly we identified ROS, which inactivates phosphatases, as a highly plausible link between CFTR deficiency and EGFR/ADAM17 axis. Since we identified several potential druggable targets for intervention in this signalling pathway (GSH, PDI, SRC), future studies evaluating their exact mechanism and relevance in CF and COPD models are of importance.

Future studies are directed towards defining the role of CFTR deficiency in the abnormal regulation of redox potentials, bioactive lipid synthesis, and membrane protein trafficking, which results in inappropriate inflammatory and repair responses in CF lung tissue, and applying these in novel therapeutic approaches.

numan disease CF lung COPD airway ILGR inflation SAMENVATTING FR st ADAM17 fibrosis PBEC human disease CF lung COPD airway IL6R inflammation AREG remodeling Sand CAGER shedding A deling EGFR sh G remodeling OPD airwa EGFR shedding ADAM17 fibrosis PBEC hu and ise CF lung airway IL6R Inflammation AREG remodeling EGFR sheddin 📲 🖓 IUmani, dise ung COPD airway IL6R Inflammation AREG remodeling EGFR shedding ADAM17 fibrosis PBEC

SAMENVATTING

Longziekte bij cystische fibrose (CF) en COPD (chronic obstructive pulmonary disease, rokerslong) neemt toe met de leeftijd, is niet goed te behandelen, en leidt tot ernstig verlies van longfunctie met vaak de dood tot gevolg. Beide ziektes worden gekenmerkt door de ophoping van slijm in de luchtwegen, uitdroging van de luchtwegen, chronische ontsteking, moeilijk te bestrijden bacteriële infecties, en onomkeerbare verandering van de structuur van de longen. De primaire oorzaak van de ziektes is duidelijk verschillend. CF wordt veroorzaakt door een erfelijke afwijking in een chloride kanaal betrokken bij vloeistof transport door de slijmcellaag (CFTR). COPD wordt vooral veroorzaakt door blootstelling aan sigarettenrook. Toch vertonen de moleculaire mechanismes die een rol spelen bij het ontwikkelen van deze chronische longziektes overeenkomsten. Het doel van ons onderzoek was deze mechanismes in het laboratorium nader te onderzoeken, zodat we nieuwe geneesmiddelen kunnen testen voor beide ziektes.

We hebben ons geconcentreerd op de rol van twee eiwitten, waarvan we weten dat ze betrokken zijn bij ontstekingsreacties, slijmproductie, en weefselherstel in de long. ADAM17 is een enzym dat meerdere eiwitten van de celmembraan losmaakt, die op hun beurt via specifieke bindingseiwitten (receptoren) processen in de cel regelen. EGFR (epidermal growth factor receptor) is een van de ontvangers van die signalen, die via het celoppervlak worden doorgegeven aan een complex netwerk van eiwit fosforylerende enzymen (kinases) in de cel. Deze activiteit wordt in de long versterkt door diverse uitwendige prikkels zoals rook, beschadiging, en infecties. Zoals ook uit ons onderzoek blijkt beïnvloeden ADAM17 en EGFR elkaar wederzijds zodat ze samen een regelsysteem vormen (EGFR/ADAM17). ADAM17 knipt na stimulatie niet alleen EGFR bindende eiwitten van een cel los, waaronder Amphireguline (AREG), maar ook de receptor van een ontstekingsfactor (interleukine 6 receptor, IL6R). Hierdoor worden ook nabij-liggende cellen die een EGFR receptor of een interleukine receptor hebben geactiveerd. Op deze manier kunnen prikkels die het oppervlak van de long beïnvloeden worden doorgegeven aan het onderliggende bindweefsel en witte bloed cellen (trans-stimulatie). De ADAM17/AREG/EGFR en IL6/IL6R/gp130 receptor systemen activeren samen de productie van eiwitten nodig voor normale ontstekingsreacties en weefselherstel, onder andere via STAT3, een eiwit dat in de celkern de productie van specifieke messenger RNA's regelt, en waarvan bekend is dat het mede de ernst van CF longziekte bepaalt.

De Hypothese. We veronderstelden bij het begin van ons onderzoek dat een verhoogde activiteit van het ADAM17/EGFR systeem bijdraagt aan de heftige ontstekingsreacties en structuurveranderingen van de long bij CF en COPD. Daarom waren onze experimenten er vooral op gericht om de rol van ADAM17 en EGFR bij het ontstaan van chronische longziekte te analyseren.

We onderzochten daarvoor het effect van COPD en CF op de activiteit van ADAM17/EGFR in long-slijmhuid cellen (epitheel). Hiervoor kweekten we cellen van COPD patiënten, en van personen zonder COPD. Deze cellen werden onder gecontroleerde omstandigheden blootgesteld aan sigarettenrook. Zo konden we de activering van ADAM17/EGFR en het effect van chronische COPD onderzoeken. Daarnaast gebruikten we epitheel cellen oorspronkelijk geïsoleerd uit de long van een CF patiënt, waarin we het ontbrekende eiwit CFTR konden activeren. Op die manier konden we vaststellen dat het ontbreken van CFTR de activiteit van EGFR/ADAM17 inderdaad verhoogt. Door onder andere remmers van ADAM17 en van EGFR te testen, hebben we laten zien dat dit proces door experimentele geneesmiddelen beïnvloed kan worden. **Hoofdstuk 1** voorziet in een algemene inleiding over CF en COPD longziekte, en beschrijft het belang van anti-ontstekingstherapie bij de bestrijding hiervan. In het tweede deel worden ADAM17 en EGFR in detail beschreven, en hun mogelijke relatie met longziekte.

Hoofdstuk 2 beschrijft een onderzoek met longcellen die zodanig op filters zijn gekweekt, dat ze zich gedragen als slijmhuidcellen in de luchtwegen, door ze bloot te stellen aan lucht (air-liquid interface culture). Deze experimenten laten zien dat cellen afkomstig van COPD patiënten na behandeling met sigarettenrook een sterkere ADAM17/EGFR activiteit vertonen dan cellen van niet-CF patiënten, waarbij de groeifactor AREG en de ontstekingsfactor IL6R aan de bloedzijde van de cellaag worden uitgescheiden. Dit suggereert dat de longen van COPD patiënten een sterkere reactie op rook vertonen dan normaal. Verder hebben we laten zien dat remmers van zowel EGFR als ADAM17 niet alleen de uitscheiding van AREG en IL6R verminderen, maar ook de productie van de bijbehorende messenger RNA's, wat duidt op de betrokkenheid van transcriptiefactoren in de kern. Een onverwacht resultaat is dat ADAM17 na activering door sigarettenrook binnen in de cel een interactie aangaat met AREG en IL6R, niet op het oppervlak waar de producten worden uitgescheiden. Ook EGFR wordt na activering vanaf het celoppervlak naar binnen getransporteerd. Het transport van deze membraaneiwitten in de cel na stimulatie is blijkbaar een onderdeel is van het regelmechanisme. Dit moet nader onderzocht worden, en zal van belang zijn bij de ontwikkeling van nieuwe geneesmiddelen.

Hoofdstuk 3 laat zien dat onze hypothese juist was. De productie van de groeifactor amphireguline (AREG) door CF luchtweg epitheelcellen neemt af wanneer we ze een actieve vorm van eiwit CFTR laten maken. Productie van AREG is ook in deze cellen volledig afhankelijk van ADAM17 en EGFR. De activering van EGFR, gemeten als fosforylering door kinases is hoger in cellen zonder CFTR. Verder zien we meer EGFR op het aan de lucht blootgestelde deel van de celmembraan. Het verband tussen deze verhoogde activiteit en de afwezigheid van CFTR hebben we nader onderzocht. Onze metingen laten zien dat het CFTR defect een verhoogde oxidatie van de celmembraan veroorzaakt. We constateerden ook dat stoffen die de oxidatie van de celmembraan verminderen, met name glutathion, de uitscheiding van AREG sterk verminderen. Oxidatie veroorzaakt een stressreactie in de cel, en die beschermt zich daartegen onder andere door de uitscheiding van een natuurlijk antioxidant: glutathion. Andere onderzoekers hadden al laten zien dat glutathion transport in CF cellen minder is dan normaal. We concluderen dan ook dat in CF luchtweg cellen de oxidatie-stress veroorzaakt door onvoldoende glutathion transport leidt tot hyperactiviteit van EGFR/ADAM17. Waarschijnlijk komt dit door een verandering in de structuur van ADAM17, gecombineerd met een verandering van het intracellulaire transport van EGFR/ADAM17. Samenvattend betekent dit waarschijnlijk dat ook CF epitheel cellen in de long van een patiënt spontaan een verhoogde activiteit van het EGFR/ADAM17 vertonen, ook als er geen bacteriële infectie is in die long, met versterkte inflammatie en fibrose tot gevolg.

Hoofdstuk 4 maakt gebruik van een CF muismodel dat we eerder gemaakt hebben, met dezelfde erfelijke afwijking als de meeste CF patiënten (F508del CFTR). In dit model stelden we eerst met een geavanceerde methode vast dat de muizen longontsteking vertonen, zonder dat ze worden blootgesteld aan infecties of andere stressfactoren. Blijkbaar is het CFTR defect op zichzelf voldoende om witte bloedcellen te activeren tot een ontstekingsreactie. Verder vonden we dat de productie van een bekende signaalstof betrokken bij longontsteking, sphingosine-fosfaat (S1P) sterk verlaagd is in de long van de CF muizen. Wanneer we de S1P verhogen met behulp van een experimenteel medicijn tegen reumatische ontstekingen (LX2931 Lexicon Pharmaceutics), zien we een afname van

de longontsteking in CF muizen. Ook wordt de ontstekingsreactie op bacteriële toxines in de long minder langdurig. S1P en zijn receptoren maken deel uit van een groter signaalnetwerk, waar ook EGFR/ADAM17 deel van uitmaakt, en dat beïnvloed wordt door CFTR.

Hoofdstuk 5 geeft een samenvatting en een kritische analyse van de studies, en geeft aan wat de volgende stappen zijn in het onderzoek naar de relatie tussen EGFR/ADAM17 en chronische longziekte.

Appendix beschrijft in een gerelateerde studie dat F508del CFTR muizen meer moeite hebben om een bacteriële infectie te overleven dan gezonde muizen, en dat ze daarbij een verhoogde productie van het signaaleiwit IL1b vertonen. Een mutatie in een receptor van IL1b, en de toediening van een antilichaam tegen IL1b verminderen de gevoeligheid van muizen voor bacteriële infectie. Daarbij is van belang voor onze studie dat IL1b EGFR/ADAM17 activeert, en dus ook deel uitmaakt van dit CFTR afhankelijke regelnetwerk. Dit biedt een alternatieve mogelijkheid om in te grijpen in het proces van excessieve long inflammatie in CF en mogelijk bij COPD.

Samenvattend geven onze resultaten aan dat het EGFR/ADAM17 systeem betrokken is bij zowel CF als COPD longziekte. Hoewel deze ziektes verschillen wat betreft hun primaire oorzaak, laten we zien dat ze een gemeenschappelijk moleculair mechanisme hebben dat mede verantwoordelijk kan zijn voor het ontstaan en voortschrijden ervan. Sigarettenrook en CFTR defecten in luchtwegslijmhuidcellen activeren EGFR/ADAM17 afhankelijke uitscheiding van signaalmoleculen (IL6R en AREG). Dit kan de slijmproductie, het bindweefsel, en de witte bloedcellen in het weefsel activeren, wat kan bijdragen tot de typische ziekteverschijnselen van CF en COPD. Verder wijzen onze studies er op dat vooral oxidatiestress de verbindende schakel is tussen het CFTR defect en EGFR/ADAM17 activering. Dit biedt meerdere mogelijkheden tot nader onderzoek naar toepassing van nieuwe geneesmiddelen die dit regelsysteem beïnvloeden.

Toekomstige studies worden gericht op de rol van CFTR bij het regelen van oxidatiestress, de productie van signaalmoleculen, en het transport van membraan eiwitten, wat in de CF long leidt dit tot heftige ontstekingsreacties en abnormaal weefselherstel. De resultaten kunnen ons helpen om nieuwe geneesmiddelen te ontwikkelen.

CV & PhD PORTFOLIO ing Af disease CF lung COPD airway IL6R inflammation AREG remodeling human disease CF lung COPD airway IL6R inflammation AREG Sea SER shedding At deling EGFR sh G remodeling OPD airway EGFR shedding ADAM17 fibrosis PBEC hu to the nflammation AREG remodeling EGFR shedding ADAM17 117 fibrosi 🐂 🔐 Iuman' disea airway IL6R Inflammation AREG remodeling EGFR sheddin ung COPD airway IL6R Inflammation AREG remodeling EGFR shedding ADAM17 fibrosis PBEC

CURRICULUM VITAE

PERSONAL DETAILS

First and last name:	Marta Stolarczyk
Education level:	MSc in Molecular Biology
	(PhD candidate at ErasmusMC:
	University Medical Center Rotterdam)
Phone number:	(+49) 162 153 184 6
E-mail:	stolarczyk.marta@gmail.com
Address:	Gustav-Kirchhoff Str 2b, 69120 Heidelberg, Germany
Date of Birth:	May 21 th ,1986, Iława, POLAND

EDUCATION	
2011 – 2015	PhD candidate, ERASMUS MC: UNIVERSITY MEDICAL CENTER ROTTERDAM, DEPARTMENT OF CELL BIOLOGY <u>Thesis title:</u> EGFR/ADAM17 axis in Lung Diseases: Cystic Fibrosis and COPD.
2008 - 2010	MSc in Molecular Biology, UNIVERSITY OF WARSAW, BIOTECHNOLOGY, DEPARTMENT OF METABOLIC REGULATION <u>Thesis title:</u> Binding test optimization and examination of nuclear receptor PPARD activity changes under various candidate drugs stimulation
2005 - 2008	BSc in Biotechnology, UNIVERSITY OF WARSAW, BIOTECHNOLOGY, DEPARTMENT OF METABOLIC REGULATION <u>Thesis title:</u> Agents stimulating insulin secretion
2003 - 2005	High School No.2, Olsztyn

PROFESSIONAL EXPERIENCE

2016 – present	PostDoc, UniversitatsKlinikum, Heidelberg, Hematology 3D culture models for precision medicine of blood cancers
2011 – 2015	PhD candidate, ERASMUS MC: UNIVERSITY MEDICAL CENTER ROTTERDAM, DEPARTMENT OF CELL BIOLOGY Investigation of the inflammatory responses mediated by ADAM17 in lung <i>ex vivo</i> models: primary bronchial human cells, CFBE cells, organoids.
2009 – 2010	Pharmaceutical Company " ADAMED " – a contract for a specific task Investigation of the role of PPARbeta/delta agonists in diabetes type II in primary and stable cell lines (Hepa-1,6, HepG2, C2C12, 3T3L1)
2009	 Biotechnology Company "BIOTON" – a student training transfection with bacterial DNA plasmid and plasmids isolation from bacterial culture, various electrophoresis techniques, PCR, electroporation, biochemical tests API (BioMerieux), SDS-PAGE electrophoresis, ELISA test for E.coli proteins, transfection with GFP
2009	 Fertility Clinic in Warsaw "NOVUM" – a student training <i>in vitro</i> fertilization (IVF), cytogenetic and bacteriology analysis, blood tests, urine samples analysis

CURRICULUM VITAE

(INTER)NATIONAL CONFERENCES

2014	North American Cystic Fibrosis Conference, NACFC, Atlanta Poster presentation
2013	Basic Science Meeting, Malta Oral and Poster presentation
2013	NRS Young Investigator Symposium, Animal models in pulmonary research, Utrecht
2012	Poster presentation The 7th European CF Young Investigator Meeting, Paris Oral presentation: A dom 17 a main playar in CF lung disease?
2011	FEBS, Sorrento Poster Presentation: Cell Biology and Pharmacology of Mendelian Disorders.
2011-2014	Long Dagen, Utrecht Poster presentations

FELLOWSHIPS

1	2	0	1	5
	-	v		-

Short-term fellowship EMBO award 4 month intership in Gergely L. Lukacs laboratory at McGill University, Montreal

WORKSHOPS AND MEETINGS

2011-2013	PhD Workshops, Maastricht (2011), Dusseldorf (2012), Luxemburg (2013)
2011-2015	Regular meetings of the Netherlands CF groups, Utrecht
2011-2015	Regular meetings with CF experts in Sophia Hospital, Rotterdam
2011-2014	Journal Club Presentations

SUPERVISING INTERNSHIPS

2012-2014

Supervision of 3 internships Duration: 9 months

COURSES AND TRAININGS

2016	KNIME
	Heidelberg Center for Human Bioinformatics (HD-HuB)
2014	Photoshop and Illustrator C6 Workshop,
	ErasmusMC, Rotterdam
2010	Western Blotting Course
	MERCK, Warsaw
2008	Training in Animal Handling
	University of Warsaw, Poland
2007	Radiation Safety and Isotopic Techniques Training
	University of Warsaw, Poland

PUBLICATIONS

In preparation (1 st author)		Stolarczyk et al, Lack of CFTR function in polarized bronchial epithelial cells	
		results in increased extracellular redox potential leading to augmented ADAM17	
		and EGFR activity and enhanced AREG release.	
2016	(1 st author)	Stolarczyk et al., ADAM17 and EGFR regulate IL-6 receptor and amphiregulin in	
		COPD bronchial epithelial cells, Physiol Rep	
2016		Veltman et al., Correction of Lung Inflammation in a F508del CFTR murine Cystic	
		Fibrosis model by the Sphingosine-1-Phosphate Lyase Inhibitor LX2931, Am J	
		Physiol Lung Cell Mol Physiol	
2014		Palomo, et al., Role of IL-1β in experimental cystic fibrosis upon P.aeruginosa	
		infection, PLoS One	
2011		Stolarczyk et al, Nuclear receptors PPAR as a drug target in metabolic disorders,	
		Postepy Biochem.	



PhD Portfolio

Summary of PhD training and teaching

Name PhD student: Marta Stolarczyk PhD p		PhD period: 2011-2015			
Erasmus MC Department: Cell Biology Promotor(s): Prof. D		. Frank Grosvelt			
Research School: MGC Postgraduate School Supervisor: Dr. Bob J			Scholte		
1. F	1. PhD training				
			Year	Workload	
Gei	neral courses				
-	Safety Laboratory Techniques		2011	0.3 ECTS	
-	Biophysics & Biochemistry		2011	3 ECTS	
-	Cell and Developmental Biology		2012	3 ECTS	
-	Genetic		2012	3 ECTS	
-	Laboratory Animal Science		2012	3 ECTS	
-	Literature discussion		2013	2 ECTS	
-	Statistic		2013	2 ECTS	
-	English for Academic Purpose		2014	3 ECTS	
-	Photoshop and Illustrator C6 Workshop		2014	0,3 ECTS	
Ser	ninars and workshops				
-	PhD Workshop, Maastricht		2011	32 hours	
-	NRS Young Investigator Symposium, A	Animal models in	2011	0,3 ECTS	
-	pulmonary research, Utrecht PhD Workshop, Dusseldorf (poster presentat	tion: The role of the	2012	32 hours	
	sheddasses (Adam17) in Cystic Fibrosis – Chronic lung disease)				
-	- NRS Young Investigator Symposium, Animal models in pulmonary		2012	0,3 ECTS	
	research, Utrecht				
-	- PhD Workshop, Luxemburg (oral presentation: COPD and Cystic				
	Fibrosis a tale of two diseases)		2013	32 hours	
Pre	sentations				
-	Oral presentation, Leiden (The role of ADAM?	17 and IL6 pathway	2011	1 ECTS	
	in Cystic Fibrosis in HT29 adenocarinoma cel	ls)			
-	Oral presentation, Utrecht (The role of ADAM17 and IL6 pathway in Cystic Eibrosis Lungs)		2011	1 ECTS	
-	One year evaluation presentation (The role of	of ADAM17 and IL6	2012	1 ECTS	
	pathway in Cystic Fibrosis and COPD)				
-	Oral presentation, Leiden (The role of ADAM	17 and IL6 pathway	2012	1 ECTS	
	in chronic lung disease: Cystic Fibrosis and C	OPD)			
-	Oral presentation Utrecht (Does CFTR reg	gulate shedding of	2012	1 ECTS	
	cytokines and growth factors in CF ling disease?)				
-	Monday morning meeting (CF)		2012	1 ECTS	
-	Monday morning meeting (COPD)		2013	1 ECTS	
-	Oral Presentation, Sophia Children Hospital (COPD and CF)		2013	1 ECTS	
-	Oral Presentation, Utrecht (The role of A	DAM17 in chronic	2013	1 ECTS	
	inflammatory lung diseases: CF and COPD)				
-	- Oral Presentation, ErasmusMC, CF meeting (Metalloproteases in		2013	1 ECTS	
	CF and COPD, primary bronchial epithelial	cells in culture as a			
	surrogate model for lung pathology)				

-	Monday morning meeting	2013	1 ECTS
	(The mode of action of ADAM17 in resolution of inflammation in		
	CF. The pulmonary models to understand the mechanisms of	2013	0.5 ECTS
	lung fibrosis)		
	Long Dagen, Utrecht (poster presentation)	2014	1 ECTS
-	NRS Young Investigator Symposium, Animal models in	2014	1 ECTS
	pulmonary research, Utrecht (poster presentation)		
-	Oral presentation, Utrecht CF meeting (ADAM17 and	2014	1 ECTS
	inflammatory responses in CFTR deficient cells and Lung		
	organoids as a tool for drug screening in CF pathology)		
-	Monday morning meeting (The influence of CFTR potentiators	2014	1 ECTS
	and correctors on inflammatory responses in CF lung disease)		
-	Monday Morning meeting (The strategies in CF therapy)	2014	1 ECTS
-	Long Dagen (poster presentation)		
-	Oral Presentation, Montreal (The mode of action of ADAM17 in	2015	1 ECTS
	resolution of inflammation in chronic lung diseases)		
(Int	er)national conferences		
-	FEBS, Italy (Poster Presentation)	2011	1.4 ECTS
-	The 7th European CF Young Investigator Meeting, Paris (Oral	2013	1.4 ECTS
	presentation)		
-	Basic Science Meeting Malta (Oral and Poster presentation)	2014	1.4 ECTS
-	North American Cystic Fibrosis Conference, NACFC, Atlanta		
	(Poster presentation)	2014	1.4 ECTS
Oth	ner		
-	Short-term fellowship EMBO award	2015	4 months
	(4 month internship in Professor Lukacs GL Laboratory at McGill		
	University (Montreal) under supervision of Veit G, Associate		
	Professor)		
2. 1	Feaching		
Su	pervising practicals and excursions, Tutoring		
-	Internship Technician (M. Veltman)	2011	9 months
-	Internship Technician (S. Nieuwenhuize)	2012	9 months
-	Internship Technician (A. Labasati)	2013	7 months
Oth	ner		
-	Journal Club (presentation: "Cigarette smoke exposure induces	2012	0.25 ECTS
	CFTR internalization and insolubility, leading to airway surface liquid		
	dehydration")		
-	Journal Club (presentation: An HNF4a-miRNA Inflammatory	2012	0.25 ECTS
	Feedback Circuit Regulates Hepatocellular Oncogenesis)		
-	Journal Club (presentation: Cell surface annexin regulate ADAM-	2013	0.25 ECTS
	mediated ectodomain shedding of proAREG)		
-	Journal Club (Functional engrafment of colon epithelium expanded	2014	0.25 ECTS
	in vitro from a single adult Lgr5+ stem cell)		
-	Journal Club (A special population of regulatory T cells potentiates	2015	0.25 ECTS
1	muscular repair)		
	muscular repair)		
	muscular repair)		

human disease CF lung COPD airway IL68 inflammation AREG remodeling ACKNOWLEDGEMENTS G remodeling e leling EGFR she Ren AREG remo a sinflammation EGFR shedding ADAM17 fibrosis PBEC hui OPD airway ise nflammation AREG remodeling EGFR shedding ADAM17 📫 🖓 Iumani disea airway IL6R Inflammation AREG remodeling EGFR sheddin ung COPD airway IL6R Inflammation AREG remodeling EGFR shedding ADAM17 fibrosis PBEC h

ACKNOWLEDGEMENTS

After an intensive period of PhD programme, finally came the moment of the finishing touch on my thesis: writing the acknowledgements. As Murray Gell-Mann, an American physicist said: "No individual is alone responsible or a single stepping stone along the path of progress (...)", also the progress of this thesis would not be possible without support, inspiration and motivation of many people who I met and encountered during my PhD path.

Firstly, I would like to express my sincere gratitude to my promoter **Prof. Frank Grosveld** for the opportunity to be enrolled in a PhD programme at the Cell Biology Department, for being the part of the committee and all advices given on a few occasional meetings. Dear **Prof. Dr. Huylebroeck**, you took a lead on my last years of PhD programme. I really appreciated your guidance, which helped me in all the time of the writing of this thesis.

My sincere thanks goes also to my co-promoter and daily advisor **Dr. Bob Scholte**. Bob thank you for providing me an opportunity to join your lab and the continuous support during my PhD research and also writing of this thesis (here goes also acknowledgements to the NCFS and Longfonds for sponsoring my project). I also appreciated your insightful comments and hard questions, which incentivized me to constantly deepen my knowledge. Finally, I thank you for allowing me to travel to various conferences and opening the opportunities for collaborations, which all had widened my perspective on the project. Finally, thank you for determination in allowing me to finish my PhD and all your help with the finalization of the manuscript, especially in the last year, when our lab encounter difficulties and I was not employed anymore. I took it for granted that in the morning I will get an e-mail from you with a question "Marta, how and where are you?". You were always available for my questions. It was tough for both of us to finish this thesis without financial support, however, "I was taught that the way of progress was neither swift nor easy" (Maria Skłodowska-Curie, the first women to win the Nobel Prize, a Polish physicist and chemist).

Besides, I would like to thank the rest of the small thesis committee members: **Prof. Dr. Pieter S Hiemstra, Associate Prof. Dr Robbert Rottier** and **Prof. Dr. Ruud W Hendriks** for reviewing the thesis and providing comments, suggestions and critics to the manuscript.

Dear Prof. Dr. Pieter S Hiemstra and **Dr. Gimano D. Amatngalim** thank you for our nice collaboration. It was a pleasure to work with you. Your guidance and immense knowledge helped me in all the time of research. You always inspired me and gave a lot of motivation to widen my research.

Dear Associate Prof. Guido Veit and **Prof. Dr. Gergely L. Lukacs** thank you for pursuing the part of my project together and supplying us with your unique CF research model. **Guido** there is no words to express my sincere thanks to you. Thank you for inviting me to your laboratory (here goes also acknowledgements to the EMBO Founding Awards for supporting financially my visit). You were one of my best mentors during my PhD period. You not only gave me a good example in the laboratory practice, but also showed how to improve the organization skills. It was really a great pleasure to have a chance to work with you and have your expertise. We went through the cold winter up to the warm spring in Montreal ;). It is also a good moment to thank **Andrea** for your support and kindness during my stay in the Lukacs lab. You immediately made me feel as in my home lab. **Brian**, it was nice to share with you the office at McGill. Many thanks to **Pierre-Aurelien**, **Radu**, **Miklos**, **Pirjo**, **Hector**, **Haijin**, you were all very kind and helpful during this 4-month stay.

I would like to thank **Dr Gert Jansen** and **Dr Niels Galjart** for meetings, discussions and support.

Dear **Robbert**, it was great to share the lab with your group and always have a good support and feedback from you. I thank my fellow lab-mates for the stimulating discussions and for all the fun we had in the last four years. **Mieke**, it was great to have you as my first intern and later as a technician. Thank you for all your help and willingness for crazy parties! Thanks, smiley ⁽ⁱ⁾! **Heelen**, you were not only a person to talk to, but also a person to dance and joke ⁽ⁱ⁾. Take care of your family! **Marjon**, I loved your positive attitude everyday, simply do not change ;)! A lot thanks to **Joshua**, you was the first person showing me around. **Yadi**, it was nice to have evening's chats with you. **Kim** and **Anna** thanks for having you around. I will miss you all a lot! **Xiao**, I really enjoyed Food-Festival with you and our collaboration ⁽ⁱ⁾! It was fun to do experiments together. Thanks to all the people on the 10th floor! **Alex, Nesrin, Luca, Jessica, Thomas, Andrea, Polynikis, Sreya, Liu, Dave**. It was great to have you around!

Marcel, thanks for enlightening me organoid cultures. **Hugo**, thank you for all your advices! **Paulina**, it was nice time in Atlanta! **Isabellus**, it was amazing trip to Antwerp. Good luck with finding nubs! I also would like to thank **Jeffrey**, **Johanna**, **Ludwik** for our regular CF meetings in UMC. I also thank all the people that I met during conferences, who I have not mentioned here by name, but they brought knowledge for fresh ideas

Marike, thank you for all your help. I will never forget your support with the final steps in order to submit my thesis. Thanks for all your actions, mails (sometimes late in the evening) and calming me down.

Dorota bralas udzial przy moim procesie rekrutacyjnym. Dziekuje za przetlumaczenie mojej polskiej pracy magisterskiej. Dzieki za pozytywny glos ⁽²⁾! **Madzia** dziekuje za nasze mile spotkania i dzielenie sie ze mna Twoimi doswiadczeniami, ktore pomogly mi kierowac moja sciezka zawodowa. **Karolina**, dziekuje za wspolne rozmowy przy kawusi i nasze krotkie wycieczki. **Kasia**, dziekuje za wspolnie spedzone swieta, a Lukaszowi za pyszniusie jedzonko! Powodzenia w NL, trzymam kciuki ⁽²⁾.

Finally I would like to thank to my current supervisor. Dear **Sascha**, thank you for trusting me and hiring me without PhD. I know that in Germany it is a challenge, so I am very thankful for your effort. It is a great pleasure to work with you. **Sophie, Caro, Lena**, girls, you are amazing! Thank you for unwind and relax moments, cinema evenings and eating-out "white pizza". No words to thank you for your support, especially in the last days before departure from Heidelberg to the Netherlands!

Least, but not last I would like to thank my close relatives for supporting me spiritually throughout writing this thesis and my life in general.

I will be never thankful enough to my dear parents. **Mama**, jestes niesamowita! Zawsze, mimo ciezkiej pracy, bylas kiedy Cie ptrzebowalam. Dziekuje za nasze cieple rozmowy i motywacje, ale tez konstruktywna krytyke. Nie ma slow ktore moga wyrazic moja wdziecznosc za to gdzie dzis jestem i kim jestem, a wszystko dzieki Tobie i Twoim staraniom! Dziekuje z calego serca **Pata**, dziekuje za wszelka pomoc i za to ze jestes zawsze, gdy tego potrzebuje.

My dear **Maciejasek**, we met at the end of my PhD contract. It was a period of my life that pushed me to the emotional and financial boarders (no job, no home, no money). There are no words to express my thanks for being with me in this hardest period of my PhD/life. We even did not know each other well and you decided to take care of me. Thank you for your kindness, warmness and love. I need to say here that without your financial support, this thesis would not be finished! Thanks for taking me for short trips and skis to have at least some fun in this hard situation. Dziekuje rowniez Twoim rodzicom za goscinnosc ⁽ⁱⁱ⁾. Thank you for all the tremendous, but also hard moments we shared. It's amazing to conquer the world together. Hope to see you soon in Heidelberg... ⁽ⁱⁱ⁾! I miss you a lot!

Piotrek, you encouraged me to apply for PhD in the Netherlands and shared with me all my ups and downs during this 4 year PhD period. Thank you for helping me out in our daily life and picking me up in the midnights from work when I had long-run experiments. Thanks that you were there for running, dancing, traveling, cooking, but also when I needed advises. Thank you for being my friend and partner. Our every year Asian journeys were amazing and inspiring. Thank you for all beautiful moments.

Mariusz, as typical older brother you were always demanding to your younger sister. But you also always believe in me and provide a lot of support and motivation. During your studies at the Michigan University, you helped me to snake into the genetic course. At that time I was in high school, but it was a moment when I decided to become a researcher. Thanks for your inspiration.

Dear **Raquel**, we met during one of The European CF Young Investigator Meeting in Paris. We approached each other and immediately realized that we found a soul mate. We were exploring Paris together and at that time we did not know that we would have a chance to meet each other again. When I got a short-term fellowship in Canada, you immediately made me feel like at home. Thank you for your amazing hospitality, our chats, delicious brunches, the Mexican B-day party, trips and tips. Thanks for encouraging your boyfriend to organize a trip to waterfalls ...! You are amazing! I wish you all the best and "congratz" with your thesis [©].

Martijn, I really enjoyed our tennis evenings and Saturday morning coffee. Thank you for helping me with the movement and keeping my Gazelle for half a year !

Thanks to the **'Lindy Hop Gang'**, you were all like family to me. Without you my stay in the Netherlands would not be such a great fun. You always increased my positive energy level. Special thanks go to **Priscilla**, **Erik**, **Flo**, **Meine**, **Jamila**, **Spyros**, **Kayleigh**, **Mathijs**, **Alexandre**, **Jill** and all the others that I have not mentioned here. Thanks for our every Thursday fun, Brux exchange trip, our eating-out and all dance parties! It was amazing time with you, guys!

This thesis was written in the Netherlands (Rotterdam), Canada (Montreal),

Poland (Wroclaw, Olsztyn, Pawlowice) and **Germany** (Heidelberg).

CF lung COPD airway IL6R inflammation AREG remodeling EGFR shedding ADAM17 fibrosis PBEC human disease CF lung COPD airway IL6R inflammation AREG remodeling EGFR shedding ADAM17 fibrosis PBEC human disease CF lung COPD airway IL6R inflammation AREG remodeling EGFR shedding ADAM17 fibrosis PBEC human disease CF lung COPD airway IL6R inflammation AREG remodeling EGFR shedding ADAM17 fibrosis PBEC human disease CF lung COPD airway IL6R inflammation AREG remodeling EGFR shedding ADAM17 fibrosis PBEC human disease CF lung COPD airway IL6R inflammation AREG remodeling EGFR shedding ADAM17 fibrosis PBEC human disease CF lung COPD airway IL6R inflammation AREG remodeling EGFR shedding ADAM17 fibrosis PBEC human disease CF lung COPD airway IL6R inflammation AREG remodeling EGFR shedding ADAM17 fibrosis PBEC human disease CF lung COPD airway IL6R inflammation AREG remodeling EGFR shedding ADAM17 fibrosis PBEC human disease CF lung COPD airway IL6R inflammation AREG remodeling EGFR shedding ADAM17 fibrosis PBEC human disease CF lung COPD airway IL6R inflammation AREG remodeling EGFR shedding ADAM17 fibrosis PBEC human disease CF lung COPD airway IL6R inflammation AREG remodeling EGFR shedding ADAM17 fibrosis PBEC human disease CF lung COPD airway IL6R inflammation AREG remodeling EGFR shedding ADAM17 fibrosis PBEC human disease CF lung COPD airway IL6R inflammation AREG remodeling EGFR shedding ADAM17 fibrosis PBEC human disease CF lung COPD airway IL6R inflammation AREG remodeling EGFR shedding ADAM17 fibrosis PBEC human disease CF lung COPD airway IL6R inflammation AREG remodeling EGFR shedding ADAM17 fibrosis PBEC human disease CF lung COPD airway IL6R inflammation AREG remodeling EGFR shedding ADAM17 fibrosis PBEC human disease CF lung COPD airway IL6R inflammation AREG remodeling EGFR shedding ADAM17 fibrosis PBEC human disease CF lung COPD airway IL6R inflammation AREG remodeling EGFR shedding ADAM17 fibrosis PBEC human disease CF lung COPD airway IL6R inflammation AREG remodeling EGFR shedding ADAM17 fibrosis PBEC human disease CF lung COPD airway IL6R inflammation AREG remodeling EGFR shedding ADAM17 fibrosis PBEC human disease CF lung COPD airway IL6R inflammation AREG remodeling EGFR shedding ADAM17 fibrosis PBEC human disease CF lung COPD airway IL6R inflammation AREG remodeling EGFR shedding ADAM17 fibrosis PBEC human disease CF lung COPD airway IL6R inflammation AREG remodeling EGFR shedding ADAM17 fibrosis PBEC human disease CF lung COPD airway IL6R inflammation AREG remodeling EGFR shedding ADAM17 fibrosis PBEC human disease CF lung COPD airway IL6R inflammation AREG remodeling EGFR shedding ADAM17 fibrosis PBEC human disease CF lung COPD airway IL6R inflammation AREG remodeling EGFR shedding ADAM17 fibrosis PBEC human disease CF lung COPD airway IL6R inflammation AREG remodeling EGFR shedding ADAM17 fibrosis PBEC human disease CF lung COPD airway IL6R inflammation AREG remodeling EGFR shedding ADAM17 fibrosis PBEC human disease CF lung COPD airway IL6R inflammation AREG remodeling EGFR shedding ADAM17 fibrosis PBEC human disease CF lung COPD airway IL6R inflammation AREG remodeling EGFR shedding ADAM17 fibrosis ...

...to be continued.